

LSHTM Research Online

Mwangi, N; (2020) Diabetic Retinopathy in Kenya: assessment of services and interventions to improve access. PhD (research paper style) thesis, London School of Hygiene & Tropical Medicine. DOI: https://doi.org/10.17037/PUBS.04658189

Downloaded from: https://researchonline.lshtm.ac.uk/id/eprint/4658189/

DOI: https://doi.org/10.17037/PUBS.04658189

Usage Guidelines:

 $Please \ refer \ to \ usage \ guidelines \ at \ https://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$

Available under license: http://creativecommons.org/licenses/by-nc-nd/2.5/



Diabetic Retinopathy in Kenya: assessment of services and interventions to improve access

Nyawira Mwangi

Thesis submitted in accordance with the requirements for the degree of **Doctor of Philosophy** of the **University of London**

2020

Clinical Research Department Faculty of Infectious and Tropical Diseases London School of Hygiene & Tropical Medicine

Funded by the **Queen Elizabeth Diamond Jubilee Trust** – through the **Commonwealth Eye Health Consortium**

Research group affiliation: Disability and Eye Health Group

Declaration

I, Nyawira Mwangi, confirm that the work presented in this thesis is my own. Where the information has been derived from other sources, I confirm that this has been indicated in the thesis.



Supervisors

Prof Allen Foster Dr Covadonga Bascaran Mr Lawrence Muthami

Advisory Committee

Prof Allen Foster Dr Covadonga Bascaran Mr Lawrence Muthami Dr Consuela Moorman Dr Stephen Gichuhi Mr David Macleod Ms Min Kim

Collaborating Institutions

London School of Hygiene and Tropical Medicine, London, UK Kenya Medical Research Institute, Nairobi, Kenya Kenya Medical Training College, Nairobi, Kenya Ministry of Health, Nairobi, Kenya College of Ophthalmology of Eastern, Central and Southern Africa

PhD Thesis Examiners

Prof Colin Cook Dr Edward Fottrell

Acknowledgements

I thank God who has made it possible for me to do this work—to Him be the glory.

It has been a privilege to learn from my supervisors, and I thank them for being highly invested and responsive to facilitate a flexible learning experience throughout the PhD work:

Allen Foster, for his wonderful nurturing approach in supervision, connecting me to the right people and resources, helping me see both the big picture and the small picture, and allowing me to forge my own path in this work. The supervisory meetings were particularly inspirational and I will draw on the learning way beyond the PhD.

Covadonga Bascaran, who encouraged me to register on this PhD, thank you for seeing my potential, pastoral support, robust discussions and link to opportunities. Glad to have you as a supervisor both on the MSc and then on this PhD.

Lawrence Muthami for your valuable presence in field activities related to this project, offering valuable ideas and impressing upon me the value of a statistical plan



With my supervisors, Prof Allen Foster(right) and Dr Covadonga Bascaran(left)

I would like to offer great thanks to my advisory team:

Stephen Gichuhi and Consuela Moorman, for your enthusiasm to discuss the project design, implementation and the results in every step, and giving constructive critique.

Min Kim and David Macleod– for offering sage guidance on statistical approach and for dedicated support during data analysis.

I am grateful to Hannah Kuper and Karl Blanchet who examined me at upgrade and provided very helpful feedback and suggestions that I have applied in this work.

This research would not have been possible without the constant support of the diligent and committed implementation partners in Kenya. In particular:

Michael Gichangi, head of Ophthalmic Services, Ministry of Health

The Kenya Diabetic Retinopathy working group

Kirinyaga county department of health and Kerugoya County hospital

Diabetes support groups in Kirinyaga, who welcomed me into their groups, and whose contribution is central to the evidence around how to improve diabetic retinopathy services

The field teams, research assistants and peer-supporters that facilitated project implementation

Thanks to Jacqueline Ramke from whom I have learnt a lot on writing and reviewing. Lindsay Hampton thanks for reading my drafts and helping with illustrations. Jenny Evans thank you for reading through the thesis and providing feedback to improve it.

Sally Parsley, Deon Minnies, and Daksha Patel – thanks for the opportunity to work with you in open education, and mentorship to develop an open education course.

I would also like to thank all the members of the Disability and Eye Health group for offering very helpful guidance, feedback and suggestions along the way.

I also thank the following institutions:

Kenya Medical Training College for the study leave to take the PhD

The London School of Hygiene and Tropical Medicine for a very memorable learning experience and giving me opportunity to represent the students on the School Athena Swan committee – this helped me to get more involved with life at the school.

Kenya Medical Research Institute for technical support

The Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium for funding this research

I am indebted to the PhD student community within the Disability and Eye Health Group as well as other LSHTM research groups during the time of my PhD. I enjoyed the vibrant discussions we had at the school, the fun moments we shared and the warm atmosphere you created.

To my examiners, I realize that examining a thesis is immense work. Thank you for the countless hours spent reading my thesis cover to cover, for the brilliant suggestions and for making the viva an enjoyable experience. I look forward to collaborating with you in the future. Finally, I owe massive thanks to my wonderful family and friends for your remarkable solidarity and for the moral support throughout the PhD, even though we were far apart much of the time. This is for you.

The post-viva thesis corrections have been made during a very significant time for my family because of three reasons: dad's illness, the global airspace closure (due to COVID-19) and our subsequent extended stay in the United States for many months. Amidst this crisis, dad and each of my family members, both at home and abroad, have frequently enquired about my progress. My supervisors and examiners have similarly supported and encouraged me during this unexpected turn of events. I am absolutely honored for this support.

I thank each of you for this.



Left to right: Prof Allen Foster, Prof Colin Cook, Dr Nyawira Mwangi, Dr Edward Fottrell, Dr Covadonga Bascaran at the London School of Hygiene and Tropical Medicine, just after the PhD examination

Dedication

This thesis is dedicated to my maternal grandmother Mrs Phoebe Wangechi Kimburi, and to the memory of my late paternal grandmother Mrs Mary Wothaya Ngumo (I did not get a chance to meet my grandfathers).

You didn't have a chance to attend school, due to the barriers to entry to education for people in your circumstances. Nevertheless, you passionately encouraged me to p ursue education diligently.

Like you, many persons living with diabetes do not have access to the services that they need.

We have shown that we can break the barriers to get them to attend screening for diabetic retinopathy as a way of preventing vision loss and blindness.

This thesis is also dedicated to my parents Mr and Mrs Mwangi Ngumo in recognition that you gave my siblings and I everything you could to enable us obtain education. You gave us opportunities you did not have yourselves.

This is for each of you.

Contributors to the thesis

Contributor	Role	Institution
Allen Foster	Supervisor	London School of Hygiene and Tropical Medicine, UK
Covadonga Bascaran	Supervisor	London School of Hygiene and Tropical Medicine, UK
Lawrence Muthami	Supervisor	Kenya Medical Research Institute, Kenya
Consuela Moorman	Advisory committee	Oxford University NHS Trust, UK
David Macleod	Advisory committee	London School of Hygiene and Tropical Medicine, UK
Min Kim	Advisory committee	London School of Hygiene and Tropical Medicine, UK
Stephen Gichuhi	Advisory committee	University of Nairobi, Kenya
Edward Fottrell	PhD Examiner	University College London Centre for Global Non-communicable Diseases, UK
Colin Cook	PhD Examiner	University of Cape Town, South
Atieno Jalango	Implementing partner	Kabarak University, Nakuru, Kenya
Esbon Gakuo	Implementing partner	Kirinyaga county department of health, Kenya
Githeko Kibata	Implementing partner	City Eye Hospital, Nairobi, Kenya
Mark Nganga	Implementing partner	Kerugoya County Referral Hospital, Kenya
Martin Musyoki	Implementing partner	Nakuru County Referral Hospital, Kenya
Michael Gichangi	Implementing partner	Ministry of Health Kenya
Peter Tum	Implementing partner	Kenya Medical Training College, Kenya
Reuben Magoko	Implementing partner	Kenya Defeat Diabetes Association, Kenya

Hannah Kuper	Upgrade examiner	London School of Hygiene and Tropical Medicine, UK
Karl Blanchet	Upgrade examiner	London School of Hygiene and Tropical Medicine, UK
Matthew Burton	Advisor	London School of Hygiene and Tropical Medicine, UK
Clare Gilbert	Advisor	London School of Hygiene and Tropical Medicine, UK
GV Murthy	Advisor	London School of Hygiene and Tropical Medicine, UK
Helen White	Research Degrees Administrator	London School of Hygiene and Tropical Medicine, UK
Sarah Polack	Departmental Research Degrees Coordinator	London School of Hygiene and Tropical Medicine, UK
Daksha Patel	Support with online DR course development	London School of Hygiene and Tropical Medicine, UK
Deon Minnies	Mentorship on online DR course development	Community Eye Health Institute, University of Cape Town, South Africa
Jacqueline Ramke	Support with developing research and scholarly skills	University of Auckland, New Zealand
Jennifer Evans	Support with developing research and scholarly skills	Cochrane Eye and Vision Group, London School of Hygiene and Tropical Medicine, UK
Marcia Zondervan	Link with resources and leadership training in the LINKS program	London School of Hygiene and Tropical Medicine, UK
Maria Zuurmond	Advice on qualitative research	London School of Hygiene and Tropical Medicine, UK
Sally Parsley	Mentorship on DR online course development	London School of Hygiene and Tropical Medicine, UK
Tunde Peto	Training on DR screening & grading	Moorfields Eye Hospital, London, UK

Project	London School of Hygiene and
administrator	Tropical Medicine, UK
Project	London School of Hygiene and
administrator	Tropical Medicine, UK
Project	London School of Hygiene and
administrator	Tropical Medicine, UK
Project Manager	College of Ophthalmology of Eastern, Central and Southern Africa
Project	London School of Hygiene and
administrator	Tropical Medicine, UK
	Project administrator Project administrator Project administrator Project Manager Project administrator

Abstract

Background

The majority of the estimated 425 million people living with diabetes (PLWD) reside in low and middle-income countries. An estimated 35% of PLWD have diabetic retinopathy (DR) and 10% have vision threatening retinopathy (VTDR) requiring urgent treatment. There is a high unmet need for DR services in many of these countries, which highlights the need to strengthen health systems. The purpose of this PhD is to provide evidence on the factors and interventions for promoting access and utilisation of services for DR. The study setting is Kenya, which has adopted Universal Health Coverage (UHC) as a target for 2030. The specific objectives were: i) To synthesise the literature on the magnitude, needs and priorities for diabetic retinopathy services; ii)To conduct an assessment of the health system for PLWD and diabetic retinopathy in Kenya; iii) To use the evidence from the literature and evidence from the health system assessment as a platform for health system strengthening; and iv) To develop and test an intervention through a randomized clinical trial to improve uptake of eye care services for PLWD in Kenya.

Methods

The first objective was achieved through a literature review. For the second objective, a health system assessment for diabetes and DR was conducted, based on the World Health Organization's framework for health systems building blocks and the tracer condition approach. For the third objective, as part of health system strengthening, national guidelines for DR and a short online course for health care workers were developed. For the fourth objective, using the evidence from the health system assessment, we designed and implemented a two arm (1:1) pragmatic cluster randomized controlled trial to test whether a peer-supporter-led package of interventions can increase uptake of DR screening among members of diabetes support groups who have never had screening. The primary outcome measure was attendance at DR screening in the 14 clusters and 734 participants followed up over six months.

Results

A review of the literature on the epidemiology and interventions for diabetes and DR, identified that the key priorities of health systems in reducing the incidence of DR-related blindness are: (1) control of diabetes (2) early detection of DR and (3) appropriate treatment of DR. However, most of the available evidence is from highincome countries. The health system assessment for diabetes and DR in Kenya identified a high unmet need for DR screening, and uptake of screening was the bottleneck to entry to DR services. The barriers were lack of referral and inadequate knowledge of diabetes eye health. The lack of clinical guidelines and inadequate opportunities for continuous professional development limit the availability of DR services. To address this barrier, national clinical guidelines were developed through a process of adaptation from generic guidelines. The scope of the guidelines is screening, diagnosis, treatment and follow-up for DR. In addition, an online training course on the control of DR has been developed using the theories of adult learning and instructional design. The results of the cluster-randomized trial indicate that the peer-led interventions are acceptable and reach the population most vulnerable to DR. One in two of the participants in the intervention arm compared to one in ten in the control arm attended DR screening. We also found that most of the effect occurs early—within the first month of the intervention.

12

Conclusions

This thesis provides evidence about how national DR guidelines and online training programmes for health workers can be developed to strengthen health systems and improve eye services for PLWD. The thesis also provides evidence that the uptake of DR screening by PLWD can be significantly increased through a community based intervention utilising diabetes support groups and peer-supporters. This can lead to earlier diagnosis of DR and the prevention of blindness.

Acronyms

\$	US dollar
ADDIE	Analysis, Design, Development, Implementation, Evaluation
	Framework
AGREE	Appraisal of Guidelines, Research and Evaluation
AI	Artificial Intelligence
AMREF	African Medical Research Foundation
Anti-VEGF	Anti-vascular Endothelial Growth Factor
BMC	Biomed Central
BMI	Body Mass Index
ССМ	Chronic Care Model
CD	Communicable Disease
CEA	Cost-Effectiveness Analysis
CHEW	Community Health Extension Worker
CHV	Community Health Volunteer
CI	Confidence Interval
CIFR	Consolidated Framework for Implementation Research
CONSORT	Consolidated Standards of Reporting Trials
CPD	Continuous Professional Development
cRCT	Cluster Randomized Controlled Trial
CSME	Clinically Significant Macula Oedema
DALY	Disability Adjusted Life Years
DCCT	Diabetes Control and Complications Trial
DED	Diabetic Eye Disease
DM	Diabetes Mellitus
DMO	Diabetic Macula Oedema
DR	Diabetic Retinopathy
DRS	Diabetic Retinopathy Screening
DSG(s)	Diabetes Support Group(s)
DTA	Diagnostic Test Accuracy
DURE	Uptake of Retinal Examination in Diabetes Study
EHSA	Eye Health System Assessment
EMR	Electronic Medical Records
EPICOT	Evidence, Population, Intervention, Comparator, Outcome, Timeline
ETDRS	Early Treatment Diabetic Retinopathy Study
FGD	Focus Group Discussion
GDP	Gross Domestic Product
HSA	Health System Assessment
Hb	Haemoglobin
HbA1c	Glycated Haemoglobin
HMIS	Health Management Information System
HSS	Health System Strengthening
ICD	International Classification of Diseases

IDF	International Diabetes Federation
INGO	International Non-Governmental Organization
Intl\$	International Dollar
KDDA	Kenya Defeat Diabetes Association
KDHS	Kenya Demographic and Health Survey
Ksh	Kenya shilling
LMIC	Low and Middle Income County
LSHTM	London School of Hygiene & Tropical Medicine
Mg/dl	Milligrams per Decilitre
Mmol/l	Millimole per Litre
МОН	Ministry of Health
mol	Mole
MRC	Medical Research Council
NCD	Non-Communicable Disease
NGO	Non-Governmental Organization
NHIF	National Hospital Insurance Fund
NHS	National Health Service
000	Ophthalmic Clinical Officer
ОСТ	Optical Coherence Tomography
OEU	Operation Eyesight Universal
OGTT	Oral Glucose Tolerance Test
ON	Ophthalmic Nurse
OOP	Out of Pocket
OR	Odds Ratio
PDR	Proliferative Diabetic Retinopathy
PhD	Doctor of Philosophy
PI	Principal Investigator
PICO	Population, Intervention, Comparator and Outcome
PLWD	People Living With Diabetes
PPP	Public Private Partnership
PS	Peer-supporter
PSED	Posterior Segment Eye Diseases
Q	Quarter
QALY	Quality Adjusted Life Year
RCT	Randomized Controlled Trial
SDG	Sustainable Development Goals
SE	Self-Efficacy
SMS	Short Message Service
STDR	Sight-Threatening Diabetic Retinopathy
STEPS	STEPwise Survey for Surveillance
UHC	Universal Health Coverage
UKPDS	United Kingdom Prospective Diabetic Retinopathy Study
UN	United Nations
UNESCO	United Nations Educational, Scientific and Cultural Organization
US	United States
VEGF	Vascular Endothelial Growth Factor

VI	Visual Impairment
VR	Vitreo-Retinal Service
VTDR	Vision-Threatening Diabetic Retinopathy
WESDR	Wisconsin Epidemiologic Study of Diabetic Retinopathy
WHO	World Health Organization

Table of Content

			Page
Declaration			2
Acknowledgement		4	
Dedication			7
Contributors t	o the	thesis	8
Abstract			11
Acronyms			14
SECTION A		GENERAL INTRODUCTION	
Chapter One		Introduction	25
	1.0	Overview	25
	1.1	Definition of terms	25
	1.2	Research domain	27
	1.3	Impetus of the research	28
	1.4	Purpose of the research	30
	1.5	Rationale	30
	1.6	Aim and objectives	31
	1.7	Thesis structure	32
	1.8	Roles of the candidate in the research	35
	1.9	Study timelines	36
			~ ~
Chapter Two		Literature review	39
Chapter Two	2.0	Literature review Overview	39 39
Chapter Two	2.0 2.1	Literature review Overview The epidemiology of diabetes	39 39 39
Chapter Two	2.0 2.1 2.2	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care	39 39 39 53
Chapter Two	2.0 2.1 2.2 2.3	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions	39 39 39 53 56
Chapter Two	2.0 2.1 2.2 2.3 2.4	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy	39 39 39 53 56 61
Chapter Two	2.0 2.1 2.2 2.3 2.4 2.5	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda	39 39 39 53 56 61 82
Chapter Two SECTION B	2.0 2.1 2.2 2.3 2.4 2.5	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT	39 39 39 53 56 61 82
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy	 39 39 39 53 56 61 82 101
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview	 39 39 39 39 53 56 61 82 101 101
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment	 39 39 39 39 53 56 61 82 101 101 101 101
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1 3.2	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment Conceptual framework of the health system	 39 39 39 39 53 56 61 82 101 101 101 101 102
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1 3.2 3.3	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment Conceptual framework of the health system Health system frameworks	 39 39 39 39 53 56 61 82 101 101 101 102 104
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1 3.2 3.3 3.4	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment Conceptual framework of the health system Health system frameworks Adaptation of the WHO health system framework for this study	 39 39 39 39 53 56 61 82 101 101 101 101 102 104 108
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1 3.2 3.3 3.4 3.5	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment Conceptual framework of the health system Health system frameworks Adaptation of the WHO health system framework for this study A summary of study methods	 39 39 39 39 53 56 61 82 101 101 101 101 102 104 108 109
Chapter Two SECTION B Chapter Three	2.0 2.1 2.2 2.3 2.4 2.5 3.0 3.1 3.2 3.3 3.4 3.5 3.6	Literature review Overview The epidemiology of diabetes Chronic care model for diabetes care Peer support interventions Diabetic retinopathy Control of diabetic retinopathy within an NCD, UHC and the health systems agenda HEALTH SYSTEM ASSESSMENT Health system assessment for diabetes and diabetic retinopathy Overview Purpose of the health system assessment Conceptual framework of the health system Health system frameworks Adaptation of the WHO health system framework for this study A summary of study methods Results	 39 39 39 39 53 56 61 82 101 101 101 101 102 104 108 109 112

	3.8	Priority setting – health system strengthening	132
	3.9	Priority setting- Intervention to improve access to	133
		DR screening	
Chapter Four		Additional findings from health system assessment	140
	4.0	Overview	140
	4.1	Preface-Research paper 1	140
	4.2	Preface - Research paper 2	141
		Research paper 1	143
		Research paper 2	159
SECTION C		HEALTH SYSTEM STRENGTHENING	
Chapter Five		Health system strengthening: Development of	183
chapter rive		clinical guidelines for diabetic retinopathy in Kenya	105
	5.0	Overview	184
	5.1	Preface - Research paper 3	185
	5.2	Preface - Research paper 4	186
		Research paper 3	189
		Research paper 4	199
Chapter Six		Health system strengthening: The development of	207
		an open online training course on control of DR	
	6.0	Overview	207
	6.1	Online learning for continuous professional	208
		development of health workers	
	6.2	Adult learning theory	210
	6.3	The ADDIE instructional design framework	210
	6.4	Mentorship for course development	211
	6.5	Developing the online course on control of DR using the ADDIE framework	212
	6.6	Potential challenges in running the online course	223
	6.7	Lessons learnt during course development	225
	6.8	Conclusion	226
SECTION D		DURF Cluster randomized controlled trial	231
Chapter Seven	1	Study protocol	231
	7.0	Overview	233
	7.1	Intervention development	233
	72	Preface - Research paper 5	235
	,.2	Research paper 5	238
Chapter Eight		Pilot trial	251
	8.0	Overview	251
	8.1	Preface - Research paper 6	252
		Research paper 6	255

Chapter Nine	Process evaluation	281
9.0	Overview	281
9.1	Preface - Research paper 7	282
	Research paper 7	285
Chapter Ten	Outcome evaluation	305
10.0	Overview	304
10.1	Preface - Research paper 8	305
	Research paper 8	308
SECTION E		
Chapter 11	Thesis Summary	340
11.0	Overview	340
11.1	The PhD narrative	340
11.2	Overview of DR services	341
11.3	DR services in Kenya	343
11.4	Health system strengthening	346
11.5	Development of an intervention to increase uptake	348
11.0	of screening	250
11.6		350
11.7	Implications	355
11.8	Dissemination	361
11.9	Conclusion	361
Appendices		364
Appendix 1	Training - courses undertaken	364
Appendix 2	PhD timeline	366
Appendix 3	Standard Operating Procedures for measuring anthropometric, clinical and metabolic parameters	367
Appendix 4	CONSORT checklist for pilot trial	370
Appendix 5	Development of training programme for peer- supporters	371
Appendix 6	CONSORT checklist cluster randomized trials	373
Appendix 7	CONSORT checklist pragmatic trials	379
Appendix 8	Intra-cluster correlation coefficients	383
Appendix 9	Statistical Analysis Plan	384
Appendix 10	Dissemination	391
Appendix 11	Ethics approvals	393
Appendix 12	Epilogue	401

List of Tables

		Page
Table 2-1	Prevalence of diabetic retinopathy in sub-Sahara Africa	65
Table 2-2	Planning for diabetic retinopathy services in Kenya	66
Table 2-3	Risk factors for onset and progression of diabetic retinopathy	69
Table 2-4	Wilson and Jungner classic screening criteria	72
Table 2-5	Emerging screening criteria proposed since 1968	73
Table 2-6:	Interventions for control of vision loss from diabetic retinopathy	82
Table 3-1	The suitability of diabetes and diabetic retinopathy as tracer conditions	107
Table 3-2	Health workforce for diabetes services - physicians and diabetes educators	120
Table 3-3	Distribution of eye care workers	121
Table 4-1	Diabetes and diabetic retinopathy as tracer conditions	163
Table 4-2	Integration as a theme in the different building blocks	166
Table 4-3	Examples of concepts of integration in a sample of health system plans and strategies	168
Table 4-4	Benefits of integration	172
Table 6-1	Tasks and outputs within the ADDIE framework in this course	213
Table 6-2	EPICOT description for the Control of Diabetic Retinopathy course	215
Table 6-3	Course units	218
Table 8-1	Demographic and health statistics- Kirinyaga county	260
Table 8-2	Eligibility criteria	262
Table 8-3	Trial interventions	264
Table 8-4	Baseline characteristics	267
Table 10-1	Eligibility criteria	315
Table 10-2	Baseline characteristics of study participants	320

Table 10-3	Time period of trial in which attendance at diabetic retinopathy screening occurred	325
Table 10-4	Predictors of attendance at diabetic retinopathy screening	335
Table 11-1	Table 11-1 Is the evidence sufficient?	359

List of Figures

		Daga
Figure 1 1	Cohoma of the theorie	Page
Figure 1-1	Scheina of diabatas in different regions	33
Figure 2-1		40
Figure 2-2	Character and the world map	47
Figure 2-3	Chronic care model	54
Figure 2-4	Retinal images showing classification of diabetic retinopathy	61
Figure 2-5	Association of prevalence of diabetic retinopathy with duration of diabetes	63
Figure 3-1	Conceptual Framework for health system assessment	103
Figure 3-2	Comparison of building block focused frameworks	106
Figure 3-3	Study counties and the diabetes belt	111
Figure 3-4	Access to HbA1c testing by county	115
Figure 3-5	Access to diabetic retinopathy screening by county	116
Figure 3-6	Diabetic retinopathy screening rate in different categories of diabetes clinics	117
Figure 4-1	Conceptual framework for integrated diabetes and diabetic retinopathy (DR) services	182
Figure 6-1	Conceptual model of the ADDIE framework as applied in the development if this course	213
Figure 7-1	Development, piloting, implementation and evaluation of the clinical trial	234
Figure 8-1	Flow chart for the pilot trial	279
Figure 8-2	Time to screening among PLWD in intervention and control arms	264
Figure 10-1	Location of the clusters and the eye clinic within Kirinyaga county	318
Figure 10-2	Flow diagram for the trial	333
Figure 103	Proportion of PLWD that attended screening in each cluster	322
Figure 10-4	Self-efficacy scores in intervention and control arms	323
Figure 10-5	Kaplan-Meier analysis of time from first group talk to attendance at DR screening	324



Section A

"You think you understand two because you understand one and one. But you must also understand 'and'."

— Mawlana Jalal-al-Din Rumi

13th century Persian poet (1207-1273)



Finalists for the Three Minute Thesis competition at the London School of Hygiene and Tropical Medicine, May 2017; Nyawira Mwangi is 2nd left.

Chapter 1- Introduction

1.0 Overview

This chapter presents an overview of the research described in this thesis through a brief description of: definition of key terms; research domain; purpose; impetus; rationale; aims and objectives; structure of the thesis and the role of the candidate in the research.

1.1 Definition of key terms

Diabetes

Diabetes mellitus (herein referred to as diabetes) is a group of metabolic diseases characterized by chronic hyperglycaemia resulting from defects in insulin secretion, insulin action, or both.¹ The World Health Organization (WHO) diagnostic criteria for diabetes is based on any of the following parameters: fasting plasma glucose of \geq 126 mg/dl (7.0 mmol/L); plasma glucose after 2-h oral glucose tolerance test (OGTT) \geq 200 mg/dl (11.1 mmol/L); glycated haemoglobin (HbA1c) \geq 6.5% (48 mmol/mol Hb) or a random plasma glucose of \geq 200 mg/dl (11.1 mmol/L) along with symptoms of hyperglycaemia.²

Diabetic retinopathy

Diabetic retinopathy (DR), a common ocular complication of diabetes, refers to progressive pathology of the retinal microvasculature (capillaries, arterioles and venules) with subsequent leakage or occlusion of the vessels, and retinal ischaemia.³ The Early Treatment Diabetic Retinopathy Study (ETDRS) described the vascular lesions in DR, of which the microaneurysm is the hallmark.⁴ DR is a key cause of visual impairment and blindness in the working age population.⁵

Screening for diabetic retinopathy

Screening refers to the application of tests for detection of DR in persons with diabetes who have no symptoms of DR. The methods for screening include dilated fundus examination and retinal photography.

Health system and health system strengthening

A health system comprises all the activities, institutions, organizations and resources whose primary purpose is to promote, restore or maintain health.^{6, 7} It includes broad categories such as formal health care services (public or private), informal care (home care, herbal and alternative medicine), screening programs, disease control campaigns, health policies, and organizations involved in health insurance, research, training and service delivery organizations.

Health system strengthening (HSS) refers to 'purposeful and significant efforts or actions to improve the performance of the health system'.⁷ HSS is often needed to facilitate progress towards desired targets, for example universal health coverage (UHC) and better population health.

Universal Health Coverage

The WHO defines UHC as "access to key promotive, preventive, curative and rehabilitative health interventions for all at an affordable cost, thereby achieving equity in access."⁸ UHC is a multi-dimensional concept, operationalized in terms of population coverage, financial protection, and access to quality health care according to need, pursued within the framework of legal right to health.⁹

Visual impairment

The International Classification of Diseases 11 (2018) defines visual impairment (VI) as reduction in visual acuity and classifies it as follows: (A) distance VI can be mild

(presenting visual acuity worse than 6/12), moderate (presenting visual acuity worse than 6/18), severe (presenting visual acuity worse than 6/60) and blindness (presenting visual acuity worse than 3/60). (B) Near VI is defined as presenting near visual acuity worse than N6 or M.08 with existing correction.¹⁰

1.2 Research domain

Epidemiologic evidence indicates that the prevalence of diabetes has increased internationally over the last four decades, with a more pronounced rise in low and middle-income countries (LMICs).¹¹⁻¹³ In 2017, the global prevalence was estimated at 8.8%, affecting 415 million people aged 20-79 years.^{12, 14} The majority of people living with diabetes (PLWD) live in LMICs and are predominantly below the age of 65 (88%).^{12, 14-16}

Diabetic retinopathy is the most common ocular complication of diabetes and a major cause of visual impairment in PLWD, especially when there are no screening programs.¹⁷⁻²⁰ Poor vision is associated with reduced quality of life, increased risk of fall-related injuries, cognitive decline, poor mental health, decreased independence, difficulties in managing comorbidities, and increased health expenditure.²¹

All persons living with any type of diabetes are at risk of DR. Poor control and increasing duration of diabetes are key risk factors for DR. Good management of diabetes is therefore important for the prevention of DR-related vision loss. Yau and colleagues in 2012 in a global review of the literature reported that 34.6% of PLWD at any given time have DR, while 10.2% have vision-threatening DR.²² Over the past few decades, major advances have been made in the management of DR to reduce associated vision loss. There is evidence that screening, early detection and prompt treatment of vision-threatening DR (VTDR) prevents or delays diabetes-related VI.²³⁻

geography or health system contexts is lacking.²⁶ Resource constraints,²⁷ differences in prevalence of DR,¹⁹ and health system barriers²⁸ contribute to the increasing importance of DR as a cause of avoidable VI.

In LMICs, there are more PLWD, often at a younger age, who are at risk of visual impairment. To address this public health challenge, health systems need to be ready to deliver an increasing volume and quality of services, commensurate to the predicted increase in need. Universal health coverage has become a central theme of global health efforts to provide health care for all people that is effective and sustainable. To achieve this, there is a need to strengthen health systems to address the public health needs of specific populations and specific disease conditions. How can LMICs ensure that all PLWD receive all the services that they need, e.g. eye care services for DR? As LMICs face many health system barriers, it is important to understand how relevant innovations to address DR can be effectively introduced. The work in this thesis focuses on assessing and improving services for DR in Kenya.

1.3 Impetus for the research

The PhD was inspired by observation of the Kenyan health system, where I work at the intersection of ophthalmology, public health and health systems. The research questions have been generated from looking at Kenya's health system through that lens. This research is bidirectional because the health system provided the research questions, and in turn the research has provided ways to strengthen the health system, some of which are already being implemented. During my practice, I have noted that PLWD often fall through the care net. The typical patient with DR presents to the eye clinic with advanced disease already causing VI or blindness, despite having been in contact with the health system often for many years for routine diabetes management. The results of treatment at this stage are less than optimal

28

and the cost of treatment is often high with most patients not having health insurance. The aspirations of universal eye health, (as part of UHC, which Kenya has adopted) cannot be achieved with this status quo. There is a need to find the right policies and interventions for PLWD to prevent visual loss from DR.

An important window of opportunity to implement change at scale in Kenya is presented by the upcoming development of a new national eye health plan. However, eye health plans need to be informed by evidence regarding interventions that can improve eye health services for PLWD.²⁹ Towards this, it is important to synthesize evidence from the literature, test hypotheses, and provide local evidence. Given that interventions require considerable investment, it is also essential to understand the context and consider the practical ways in which the health system might work differently. Drawing on the literature review in section A and the health system assessment in section B, I present the need for knowledge translation, which underpins the interventions in section C, in addition to the strategic use of existing resources and the empowerment of PLWD, which underpin the interventions described in section D.

Perhaps because of my medical background, or because the late presentation of PLWD with DR was the most visible gap, I originally thought the main need was to improve treatment services. Therefore, at entry to the PhD my intention was to develop interventions to improve treatment. However, after assessing the health system I found that the main bottleneck was the lack of access of PLWD to regular eye screening. A cluster randomized controlled trial has therefore tested an intervention to improve access to screening, and I present the evidence for its effectiveness in chapters 9 and 10.

29

Reflecting on what I have learnt through the PhD, at the end of chapter 11, I revisit the question of the priority challenges facing the health system. I conclude that the main challenge to preventing visual loss from DR in PLWD in Kenya (and any LMIC) is broader than either screening or treatment, but that access to screening supported by good treatment services is a starting point. Given that there are no simple solutions to address the diabetes epidemic and its effects,¹³ evidence -based health system strengthening (HSS) has a critical role in the pursuit of UHC for PLWD.²⁹⁻³² Such HSS requires the engagement of multiple actors and actions in the health system. The work described in this thesis has provided an opportunity to consider the unique contribution that researchers can bring to this agenda.

1.4 Purpose of the research

The purpose of the research is to contribute to improvement of DR services and prevention of vision loss in PLWD. In order to inform practice in the real world settings, a pragmatic approach is adopted.

1.5 Rationale

The number of people affected by VI and blindness from DR increased worldwide between 1990 and 2015.³³ This is avoidable vision loss because DR can be prevented and treated. Of the 415 million PLWD (20-79 years) in 2017, it is estimated that 145 million have DR and 45million have vision-threatening DR (VTDR).³⁴ Predictions suggest that by 2040, 642 million adults will be living with diabetes, 224 million will have some form of DR and 70 million will have VTDR.³⁴ There is evidence that appropriate treatment can delay the onset and slow the progression of DR.^{35, 36} Thus eye health services for PLWD are important in reducing the incidence of vision loss from DR. In sub-Saharan Africa (SSA) in particular, investments in health system strengthening may forestall the increase in vision loss from DR.^{30, 34, 37} This region is at the beginning of the diabetes and DR epidemic, and although data is limited, there is evidence that health systems can be strengthened to make control of diabetes and DR accessible, affordable and sustainable.³⁸ SSA can apply existing tools for planning DR services, and learn from DR screening programs in other countries.³⁹

Although PLWD have frequent contact with health services, eye care for PLWD is not routinely provided in many situations.⁴⁰ This contributes to the risk of vision loss from DR, and increases the cost of care, since patients with DR are often identified in advanced stages. Investing resources on effective eye health services for PLWD can reduce the extra morbidity of vision loss in PLWD. As a starting point, there is a need to identify the current gaps in DR services and provide evidence on well-defined interventions to address these gaps.

1.6 Aim and objectives

1.6.1 Aim

To provide evidence to improve eye care services for PLWD in Kenya in order to reduce vision loss in PLWD.

1.6.2 Specific objectives

- To synthesise the literature on the magnitude, needs and priorities for diabetic retinopathy services (Chapter 2)
- To conduct an assessment of the health system for PLWD and diabetic retinopathy in Kenya (Chapter 3 and 4)
- 3. To use the evidence from the literature and evidence from the health system assessment as a platform for health system strengthening (Chapter 5 and 6)

4. To develop and test an intervention through a randomized clinical trial to improve eye care services for PLWD in Kenya (Chapter 7-10)

1.7 Thesis structure

The thesis is organised in five sections (A, B, C, D and E), each of which has two or more chapters, Figure 1-1.

Section A is the general introduction and consists of two chapters. Chapter 1 gives the background and general direction of the thesis. It provides an overview of the research area, how the research ideas developed and the purpose of the research. Chapter 2 is a synthesis of the published evidence. As diabetes and DR are the subject of independent literature, the literature is analysed to identify the priorities and challenges in global health for eye care for PLWD, particularly in LMICs.

Section B is the health system assessment. In Chapter 3, I discuss the different frameworks for health system assessment and describe the approach for assessing the health system for PLWD and DR in Kenya. This assessment is of interest on its own, because it provided useful evidence on the strengths and weaknesses of the system. This evidence will be useful to policy-makers and implementers. In addition, it provided evidence for developing interventions to improve eye services for PLWD, which were tested in the randomized trial. Chapter 4 consists of two research papers discussing results from the health system assessment —the predictors of uptake of retinal screening and the rationale for integration of diabetes and DR services.



Figure 1-1: Schema of the thesis

Section C provides a discussion on health system strengthening – its importance in this study and how we applied it. The content includes technical and methodological contributions on developing clinical guidelines and a training course about DR, which might be useful to other LMICs wanting to undertake these activities. Chapter 5 consists of two papers on guidelines development process and outputs. Chapter 6 describes the development of an open online course on control of DR.

Section D describes the DURE (uptake of retinal examination in diabetes) intervention to improve access to DR services, developed in accordance to the four phases of the Medical Research Council's (MRC) framework for developing complex interventions. Chapter 7 describes the intervention development and includes the published trial protocol. Chapter 8 describes the pilot and feasibility trial, in a research paper. Chapter 9 describes the process evaluation of the full trial while chapter 10 describes the outcome evaluation of the trial. The four chapters provide evidence for the application of the MRC framework in DR research in a LMIC setting. Section E comprises Chapter 11, which is the thesis summary consisting of the discussion, conclusions and next steps. It ends with a personal reflection on my learning from the PhD.

Section	Paper	What the paper describes
Section B	Paper 1 (published)	Results of health system assessment (Predictors of uptake of retinal examination)
	Paper 2 (submitted)	Results of health system assessment (Rationale for integration of diabetes and DR services)
Section C	Paper 3 (published)	Process of adapting clinical guidelines for DR in Kenya (process and outputs)
	Paper 4	A summary of the recommendations in the clinical guidelines
	(published)	
Section D	Paper 5	Protocol for cluster randomized controlled trial
	(published)	
	Paper 6	Results: Pilot trial
	(submitted)	
	Paper 7	Results: Process evaluation of the trial
	(submitted)	
	Paper 8	Results: Outcome evaluation of the trial
	(submitted)	

The content of the research papers in each section are outlined below:

1.8 Roles of the candidate in the research

I have carried out the work described in all the sections of this thesis, guided by my supervisors and advisory committee. Some of the specific roles are highlighted below:

Before the research	Applying for a grant
	Conceptualizing the study
During the research	Taking the necessary courses (Listed in Appendix 1)
	Managing the grant
	Engaging stakeholders to determine priorities, plan and implement the study
Section A	Conducting the literature review
Section B	Writing the protocol
	Obtaining ethics approval
--------------------	--
	Developing research tools
	Training research assistants
	Conducting fieldwork
	Data analysis
	Dissemination of findings
Section C	Conducting literature review specific to guideline development
	Coordinating the guideline development team
	Writing the outputs
	Dissemination
	Developing the training program through mentorship
Section D	Obtaining ethics approval
	Developing protocol
	Developing peer training materials
	Developing questionnaires
	Training research team
	Coordinating field work
	Mixed methods process evaluation
	Data analysis
	Dissemination
Section E	Interpreting the evidence from the PhD
After the research	Synthesis, writing up, Identifying next steps

1.9 Study timelines

The timeframe for the study activities are shown in Appendix 2.

REFERENCES

1. Kharroubi AT, Darwish HM. Diabetes mellitus: The epidemic of the century. World Journal of Diabetes. 2015;6(6):850–67.

2. World Health Organization. WHO definition and diagnosis of diabetes mellitus and intermediate hyperglycaemia: report of a WHO/IDF consultation. Geneva: WHO; 2006.

3. Scanlon PH. Diabetic Retinopathy Medicine. 2019;47(2).

4. Early Treatment Diabetic Retinopathy Study Research Group. Grading diabetic retinopathy from stereoscopic color fundus photographs--an extension of the modified Airlie House classification. ETDRS report number 10. Ophthalmology. 1991;98:786-806.

5. Bourne RA, Stevens GA, White RA, Smith JL, Flaxman SR, Price H, et al. Causes of vision loss world wide 1990-2010: a systematic analysis. Lancet Global Health. 2013:e339-e49.

6. World Health Organization. The World Health Report 2000: improving performance. Geneva: WHO; 2000.

 Kutzin J, Sparkes SP. Health system strengthening, universal health coverage, health security and resilience. Bulletin of the World Health Organization.
 2016;92(2):537-9.

8. World Health Organization. Sustainable health financing, universal coverage and social health insurance. Geneva: World Health Organization; 2005.

9. Abiiro GA, De Allegri M. Universal health coverage from multiple perspectives: a synthesis of conceptual literature and global debates. BMC International Health and Human Rights. 2015;15(1):17.

10. World Health Organization. International Classification of Diseases, 11th Revision (ICD-11). Geneva: WHO; 2018.

11. NCD Risk Factor Collaboration (NCG-RisC). Worldwide trends in diabetes since 1980: a pooled analysis of 751 population-based studies with 4·4 million participants. Lancet. 2016;387:1513-30.

12. Cho NH, Shaw JE, Karuranga S, Huang Y, Fernandes JD, Ohlrogge AW, et al. IDF Diabetes Atlas: Global estimates of diabetes prevalence for 2017 and projections for 2045. Diabetes Research and Clinical Practice. 2018;138(2018):271-81.

13. World Health Organization. Global Report on Diabetes. Geneva: World Health Organization; 2016.

14. International Diabetes Federation. IDF Diabetes Atlas, Eighth edition. 2017.

15. Chan JCN, Gregg E, Sargent J, Horton R. Reducing global diabetes burden by implementing solutions and identifying gaps: a Lancet Commission. The Lancet. 2016;387(10027):1494-5.

16. International Diabetes Federation. IDF Diabetes Atlas, Seventh Edition. 2015.

17. Liew G, Michaelides M, Bunce C. A comparison of the causes of blindness certifications in England and Wales in working age adults (16–64 years), 1999–2000 with 2009–2010. BMJ Open. 2014;4(e004015).

18. Scanlon PH, Aldington SJ, Leal J, Luengo-Fernandez R, Oke J, Sivaprasad S, et al. Development of a cost-effectiveness model for optimisation of the screening interval in diabetic retinopathy screening. Health Technology Assessment. 2015;19(74).

19. Sivaprasad S, Gupta B, Crosby-Nwaobi R, Evans J. Prevalence of Diabetic Retinopathy in Various Ethnic Groups: A Worldwide Perspective. Survey of Ophthalmology. 2012;57(4):347-70.

20. Scanlon PH. Screening Intervals for Diabetic Retinopathy and Implications for Care. Current Diabetes Reports. 2017;17(10):96.

21. National Academies of Sciences Engineering and Medicine. Making Eye Health a Population Health Imperative: Vision for Tomorrow. Washington, DC: The National Academies Press; 2016.

22. Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556-64.

23. Ting DSW, Cheung GCM, Wong TY. Diabetic retinopathy: global prevalence, major risk factors, screening practices and public health challenges: a review. Clinical and Experimental Ophthalmology. 2015;44:260-77.

24. Fong DS, Aiello L, Gardner TW, King GL, Blankenship G, Cavallerano JD, et al. Retinopathy in Diabetes. Diabetes Care. 2004;27:S84-S7.

25. Marozas LM, Fort PE. Diabetic Retinopathy—Update on Prevention
Techniques, Present Therapies, and New Leads. US Ophthalmic Rev. 2014;7(1):54-8.
26. Andermann A, Pang T, Newton JN, Davis A, Panisset U. Evidence for Health I:
Producing evidence for improving health and reducing inequities. Health Research

Policy and Systems. 2016;14(18).

27. Cockburn N, Steven D, Lecuona K, Joubert F, Rogers G, Cook C, et al. Causes and Socio-Economic Determinants of Vision Loss in Cape Town, South Africa. PLoS One. 2012;7(2):e30718.

28. Piyasena MMPN, Murthy GVS, Yip JLY, Gilbert C, Zuurmond M, Peto T, et al. Systematic review on barriers and enablers for access to diabetic retinopathy screening services in different income settings. PLoS ONE.14(4):e0198979.

29. Ramke J, Zwi AB, Silva JC, Mwangi N, Rono H, Gichangi M, et al. Evidence for national universal eye health plans. Bull World Health Organ. 2018;96(10):695-704.

30. Blanchet K, Gilbert C, de Savigny D. Rethinking eye health systems to achieve universal coverage: the role of research. British Journal of Ophthalmology. 2014;98:1325-8.

31. Horton R, Das P. Universal health coverage: not why, what, or when—but how? The Lancet. 2015;385:1156-7.

32. United Nations General Assembly. Resolution adopted by the General Assembly 66/2. Political Declaration of the High-Level Meeting of the General Assembly on the Prevention and Control of Noncommunicable Diseases. New York: United Nations; 2011.

33. Flaxman SR, Bourne RA, Resnikoff S, Ackland P, Braitwaite T, Cicinelli MV, et al. Global causes of blindness and distance vision impairment: 1990-2015 and projections to 2020 – a systematic review and meta-analysis. The Lancet Global Health. 2017;5(12):e1221-34.

34. International Agency for the Prevention of Blindness (IAPB). IAPB Vision Atlas 2016 [Available from: <u>http://atlas.iapb.org/vision-trends/diabetic-retinopathy/</u>.

35. Diabetes Control and Complications Trial (DCCT) Research Group. The Effect of Intensive Treatment of Diabetes on the Development and Progression of Long-Term Complications in Insulin-Dependent Diabetes Mellitus. New England Journal of Medicine. 1993;329:977-86.

36. Heng LZ, Comyn O, Peto T, Tadros C, Ng E, Sivaprasad S, et al. Diabetic retinopathy: pathogenesis, clinical grading, management and future developments. Diabetic Medicine. 2013;30:640-50.

37. Blanchet K, Patel D. Applying principles of health system strengthening to eye care. Indian Journal of Ophthalmology. 2012;60 (5):470–4.

38. Chan JCN, Gregg EW, Sargent J, Horton R. Reducing global diabetes burden by implementing solutions and identifying gaps: a Lancet Commission. The Lancet. 2016;387:1494-5.

39. Poore S, Foster A, Zondervan M, Blanchet K. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. Int J Health Policy Manag. 2015;4(1):19-28.

40. Global Diabetic Retinopathy Advocacy Initiative. Integrated care for diabetes and eye health: A global compendium of good practice. Melbourne, Australia; 2018.

Chapter Two: Diabetes, Diabetic retinopathy and Visual Impairment – A synthesis of the literature

2.0. Overview

This chapter addresses the first objective of the thesis and follows a narrative structure. Section 2.1 describes the epidemiology of diabetes. Section 2.2 describes the care model for diabetes. Section 2.3 discusses peer support for diabetes. Section 2.4 discusses diabetic retinopathy (DR), visual impairment from DR, including the epidemiology, the impact, and the control measures. The concluding section (2.5) summarizes the priority needs in DR control within a non-communicable disease (NCD), Universal Health Coverage (UHC) and health systems agenda.

2.1 The epidemiology of diabetes

2.1.1 Diabetes globally

The prevalence of diabetes and the number of people living with diabetes (PLWD) has been rising steadily over the past four decades¹.The International Diabetes Federation (IDF) estimates that among adults aged 20–79 years in 2017 there was an estimated 415 million PLWD with a global prevalence of diabetes of 8.8%. This is about a fourfold increase from the number of PLWD in 1980, which the World Health Organization (WHO) estimated to be 108 million, with a global prevalence of 4.7%. ^{2, 3} There are regional disparities in the prevalence: in 2017, North America and the Caribbean region had the highest prevalence (11%) while Africa had the lowest prevalence (4.2%), as shown in Figure 2-1. We can expect the number of PLWD to continue to increase in the next two decades. There is also a high level of undiagnosed diabetes in all regions, at 37.6% in North America and the Caribbean and 69.2% in Africa in 2017.⁴



Figure 2-1: Prevalence of diabetes in different regions in 2017 and 2015 (data from the IDF Diabetes Atlas 8th edition)

There are three main types of diabetes (Type 1, 2 and gestational diabetes), all of which are treatable, with the goal of achieving normal levels of glucose. Type 2 diabetes is largely preventable but it accounts for 90% of the diabetes prevalence, and the majority of undiagnosed diabetes. ^{4, 5} People with impaired glucose tolerance are at high risk of developing Type 2 diabetes and are classified as pre-diabetes. For PLWD in the different regions, whether diagnosed or not, access to diabetes services is a priority. This includes health education, self-management, treatment and interventions for diabetes complications. In order to reduce the prevalence of diabetes it is also crucial to reduce the risk factors for diabetes in the general population.

Given the escalating number of PLWD, diabetes is not only a global health concern, but also an economic and social burden. Individual PLWD, their families, health systems and countries face a large, growing and long-term economic burden. This includes indirect costs due to disability, loss of productivity and premature mortality, besides the direct costs, particularly treatment costs. The loss in workplace productivity due to diabetes and its complications is considerable because 72% PLWD worldwide are within the working age bracket (20-64 years).⁴ This reduction in productivity has an adverse impact in all regions.⁶

The direct annual cost of diabetes in the world is estimated at Intl\$ 825 billion, and a significant proportion of these costs are paid out-of-pocket, especially in LMICs, where out of pocket payment accounts for 25-80 % of this health expenditure.^{2, 6, 7} The presence of diabetes complications such as visual impairment and blindness from DR significantly increases the direct costs. PLWD and governments struggle to meet these costs, and the protection offered by health insurance is often incomplete.⁴ As an example, the health care costs of people with diabetes in the United States (US) are nearly two and a half times of those without diabetes within the same sociodemographic bracket.⁸ Jaspers and colleagues in their systematic review on the global impact of non-communicable diseases (NCDs) on households found that financial catastrophe due to NCDs, including diabetes, was seen in all countries and at all income levels, but with a higher proportion of households affected in lower income areas.⁹ When the costs of diabetes treatment lead to financial hardship, they are drivers for impoverishment. In turn, poverty is a driver for poor compliance with treatment, poor control of diabetes and the development of complications of diabetes.

The rising magnitude and the high cost of diabetes point to the need for cost-effective control measures. The WHO has identified proven, priority cost-effective interventions 'best buys' for diabetes that can be implemented in all regions. These

are effective glycaemic control, preventive foot care for PLWD, DR screening for all PLWD and laser photocoagulation for prevention of blindness. ¹⁰ These 'best buys' may result in social, health and economic benefits by delaying complications and hence preserving health, quality of life and productivity. They are also potentially costsaving on the treatment costs for these complications. Failure to implement proven interventions will increase health care costs and is therefore counterproductive.

The recognition of the global impact of diabetes has led to several global commitments to catalyse action. In 2011, the Political Declaration on the Prevention and Control of NCDs at the 66th General Assembly of the United Nations (UN) identified diabetes as one of four priority NCDs that constitute the largest NCD burden and are largely preventable.¹¹ To generate global, regional and national momentum for action on these NCDs, the World Health Assembly in 2013 adopted the WHO Global Action Plan for the prevention and control of NCDs, which included targets for the control of diabetes and its risk factors.^{3, 12} Further, in 2015, all UN member states committed to the 2030 Sustainable Development Goals agenda which includes targets to reduce NCD-related premature mortality, achieve UHC and increase access to affordable treatment for these conditions.¹³ These commitments have been instrumental in raising awareness of the urgent need to place NCDs on the agenda of governments.⁷ Diabetes can be used as a tracer condition for examining the progress towards UHC ^{14, 15} and also for assessing health systems.¹⁶

What does UHC mean for PLWD globally? With UHC, all PLWD should obtain the services they need without suffering financial hardship. Many high-income countries such as those of Western Europe, and some middle-income countries such as Thailand, Brazil and Mexico have some form of UHC. For example, in Thailand, a country that has had UHC since 2002, PLWD can access a comprehensive package of

services with financial protection.¹⁷ We highlight some of the lessons the country has to offer to other countries below. One lesson is that not only does Thailand have UHC, but also importantly, the UHC package includes the diabetes services. In many countries, UHC packages exclude even basic diabetes care components, such as HbA1c testing or DR screening, because they are perceived as expensive. Thus having UHC does not necessarily guarantee universal access to diabetes services. Without UHC, PLWD often receive inadequate services. Many undiagnosed PLWD are uninsured, yet health insurance is significantly associated with the likelihood of timely diagnoses, improved diagnosis and control of diabetes.¹⁸ For these reasons, although health-financing schemes cannot cover all possible interventions, it is of utmost importance to ensure that UHC packages address the needs of PLWD.

Many PLWD, especially among poor, rural, less educated and other vulnerable or disadvantaged populations have been 'left behind', and do not access health services. Applying an equity lens to UHC can help to address this modifiable gap, by developing UHC packages that target these people without excluding the rest of the population.¹⁹ This approach has been useful in some Latin American countries, such as Costa Rica.²⁰ Reaching the disadvantaged populations may be facilitated through investment in primary care, and the use of community volunteers as in the case of Thailand another lesson from this country.¹⁷ Rwanda, an LMIC that has made significant progress to UHC, has also strengthened primary care through community health volunteers.²¹ In addition, Costa Rica and the National Health Service (NHS) in England have implemented UHC through strengthening primary care, where regular assessment of body mass index (BMI), HbA1c and foot examinations are provided to PLWD.^{20, 22} Health system strengthening (HSS), such as better trained health workforce and improved infrastructure as well as addressing the other health system building blocks is critical if PLWD are to access good health services.^{17, 22} Ong and colleagues in their systematic review identified several health system factors that influence the response to the diabetes epidemic.²³ They found removal of out of pocket payments and integrated diabetes models enabled the health system to address diabetes. Additional HSS interventions that have been found useful for diabetes services in the literature include: training multidisciplinary health professionals in primary care about diabetes and its complications; avoiding stock-out of supplies; providing essential technologies, such as blood glucose measurement and engaging pharmacists in diabetes, hypertension, and dyslipidaemia in Costa Rica ²⁰ and incentive schemes for primary care providers in the NHS²² are further examples.

2.1.2 Diabetes in LMICs

The prevalence of diabetes is increasing disproportionately faster in resource-poor regions.²⁶⁻²⁸ As in other parts of the world, the rise is mainly for Type 2 diabetes. The risk factors for diabetes, such as sedentary lifestyle, high BMI and impaired glucose tolerance often go unrecognized, leading to a huge unmet need for prevention.²⁹ The surge in incidence and prevalence of diabetes is driven by increasing population growth and life expectancy, industrialization, urbanisation, lifestyle changes such as the nutrition transition (shift in dietary patterns) and physical inactivity.^{1, 2, 4, 25, 30} An estimated 79% of PLWD live in LMICs, and these countries also have the highest proportion of undiagnosed diabetes in the world.^{4, 28} Notably, PLWD in LMICs are predominantly below the age of 65 years, experience a large unmet need for

treatment and have inadequate glycaemic control.^{24, 31} These are important risk factors for the development of complications such as retinopathy.

In sub-Saharan Africa (SSA), where many of the LMICS are located, public health strategies to manage the diabetes epidemic are known to be inadequate.^{32, 33} The priorities for health systems have hitherto been maternal and child health, and communicable diseases that require acute care. In contrast, PLWD require regular and lifelong engagement with the health system, which should mainly be at the primary care level. This is challenging for health systems that are oriented to acute care instead of overall health system development, and which are also fragile particularly at primary care.³⁴ As an example, large losses to care for PLWD occur at each stage of the diabetes care cascade (testing, diagnosis, treatment, glycaemic control).²⁴ Only an estimated 11% of PLWD (range 7-33%) are retained throughout the care cascade than the older PLWD.²⁴ As noted earlier, PLWD in LMICs are on average younger than PLWD in other regions. As they are 'left behind' in access to care, they are at greater risk of complications, with a high social and economic impact.

The ongoing epidemiological transition between communicable diseases and noncommunicable diseases (NCDs) presents an opportunity to orient health systems to NCD-relevant responses. This includes health system changes at community level, primary care and secondary care, including referral systems (such as for DR screening) and follow-up mechanisms. As described earlier, effective interventions such as the best buys for diabetes proposed by the WHO exist. Cost-effectiveness analysis (CEA) for these best buys has shown that the cost of implementation is approximately 1\$ per disability-adjusted life year (DALY) averted in LMICs.¹⁰ This suggests that with a modest increase in investment, the best buys are feasible for implementation even in

low-resource settings where resources are limited. Despite this evidence, there is very low coverage of these interventions in most LMICs. Given the scarcity of evidence about the best ways to improve coverage of these interventions within health systems, there is a need for robust implementation research that can inform strategies to improve coverage.

The majority of diabetes-related premature mortality, disability and loss of productivity occurs in LMICs, leading to very high indirect costs of diabetes ^{2, 31}. The direct costs of diabetes treatment are largely borne by PLWD and their families, rather than governments or insurance schemes.^{35, 36} This is likely to be a driver for disparities in access. In addition, those who fall out of care are likely to present with complications whose treatment will incur much higher costs. Although countries in the African region spend an average 6% of the national health budget on diabetes, the mean expenditure per PLWD per year is only Intl\$ 444 (compared to 3132 and 8396 Intl\$ in Europe and North America respectively), and is inadequate to provide the required interventions.^{4, 31} The only option available, therefore, is to prioritize a core set of evidence-based interventions, particularly those that can reduce the cost of treating complications. The implementation of the best buys described above should therefore be a priority.

2.1.3 Diabetes in Kenya

The population of Kenya (Fig 2-2) in 2019 is estimated at 52.6 million, with a 2.3% annual rate of population increase, life expectancy at birth of 66 years and 60% of the population being under 25 years.³⁷ This is a relatively young population, of whom 73% reside in rural areas. Responsibilities for health care in Kenya are shared between the national government and the 47 semi-autonomous county governments. The county level is responsible for essential health service delivery while the national level is

responsible for health policy and regulation, technical assistance to counties and management of national referral hospitals. Achoki and colleagues have recently discussed the gradual escalation in the relative contribution of NCDs to the disease burden at subnational level (epidemiological transition), and the challenge it poses to UHC.³⁸



Figure 2-2: Kenya on the world map

Both public and private sectors provide service delivery, with the private sector providing about 49% of services. ³⁹ Health insurance coverage for the population is low, at about 19%.⁴⁰ Within this, the national health insurance fund (NHIF), a public corporation, is the largest provider of health insurance, covering 16% of Kenyans. In addition, the NHIF's contracted facility network is only 40% of the health care facilities in the country; hence, services obtained outside this network are not covered. Though NHIF has a comprehensive benefit package, covering the costs of many medicines, laboratory and radiologic tests in the public sector, some of these services are not available in these facilities. ⁴⁰ Not all costs in the private sector are covered, therefore even patients who are covered still incur out of pocket expenses.⁴¹

The Kenya Health Policy 2014-2030 elaborates the commitment to achieve UHC by scaling up health services to populations in need.⁴² The policy defines four levels of care within public health services: community level, primary care, county level and national level. The primary care level is the first contact with the health system, which comprises dispensaries, health centres, maternity units and nursing homes. In practice, primary care level often concentrates on acute episodic care and defers chronic disease to county hospitals.

The county level consists of hospitals that complement the primary level and provide secondary and tertiary care. Diabetes clinics and general medical clinics at county level provide the diabetes services. The national level consists of highly specialized tertiary hospitals. Services for diabetes testing, diagnosis and treatment should be available at all the four levels, though not all the laboratory tests and medication are available at all the facilities.⁴³ Specialist clinics at county and national hospitals provide screening for complications. As there is no gate-keeping mechanism, PLWD are allowed frequent engagement with any level of the services.

At the community level, under the community health strategy, health services are organized around community units of about 5000 people.⁴⁴ Each unit has 50 community health volunteers (CHVs, also called community health workers), each responsible for 20 households.⁴⁵ The strategy specifies the roles of the CHVs, which include visiting households on a door-to-door basis to identify health problems, provide some interventions related to disease prevention, maternal and newborn care, home-based care, observe treatment and some curative tasks. They are also responsible for community mobilisation and linking the community to the primary

health facility. The CHVs receive an initial 10 days training with regular refreshers and have a kit of supplies. A community health extension worker (CHEW), who is a trained government health worker, provides support and supervision to the CHV. Two CHEWS are responsible for each community unit. A community health committee, consisting of community members, participates in the governance of this level. ^{44, 45}

The success of implementation of the community level of health care varies at subnational levels. Some of the challenges faced are high attrition of CHVs and conflict of workload for CHEWs.^{45, 46} Diabetes services are generally not available at this level, but health promotion around the NCD risk factors may be provided. There is potential to leverage CHVs and CHEWs to deliver NCD interventions with diabetes as an entry point, such as regular BMI measurement or blood sugar testing for adults on a door-to-door basis.

At the community level, there are diabetes support groups (DSGs) for PLWD in many of the counties.⁴⁷ The Kenya Defeat Diabetes Association, a civil organization, is the umbrella body for these groups. The groups have monthly meetings where they engage in peer-to-peer support, group education, monitoring BMI and metabolic control. They also engage in advocacy with the government and NGOs, and conduct community diabetes screening camps. These groups have links with the CVs and the primary health facilities. As this peer support resource is known to be useful in the management of diabetes,^{48, 49} its role is discussed in detail in subsequent sections.

Evidence from the Global Burden of Disease Study and country-level evidence shows an increase in NCDs over time,^{38, 50, 51} with disparities across counties.³⁸ Kenya has an NCD division within the Ministry of Health, and an NCD strategy to halt and reverse this rising burden of NCDs.⁴² However, the NCD program faces challenges due to workforce shortages, frequent stock-outs of medicines in public-sector pharmacies,

and problems guaranteeing the quality of medicines.^{52, 53} The Kenya Service Availability and Readiness Assessment Mapping (SARAM) in 2014 showed that only 5% of all facilities offered all NCD services defined in the Kenya Essential Package of Health (KEPH) and only 25% of health facilities had different tracer commodities for NCDs with huge regional variation.⁵⁴ The most available services related to health promotion, while the least available services were screening and rehabilitation.⁵⁴ Many people obtain the NCD diagnosis in the public sector but buy their drugs in the private sector, and these out-of-pocket payments can be high.^{53, 55}

The Lancet Diabetes and Endocrinology Commission on diabetes in sub-Saharan Africa underscored the need for locally derived epidemiologic evidence on diabetes at country level.³⁵ Such evidence is important because it allows for contextual analysis, identifies opportunities for improving outcomes at country level and can reliably inform subsequent policy, practice and research. A nationally representative STEPwise approach to surveillance (STEPS) survey for NCD risk factors in 2015 reported the prevalence of diabetes to be 2% among the age group 18-64 years.⁵⁶ The estimates from the IDF were very similar, at 2.2%.²⁸ This availability of data from a national survey is a significant starting point that allows consideration of some of the public health implications of the findings in the STEPS survey, which are discussed below.

The age-adjusted prevalence of pre-diabetes and diabetes was 3.1% and 2.4% respectively.⁵⁷ As the prevalence of pre-diabetes exceeded that of diabetes, the prevalence of diabetes is likely to increase in the medium-term rather than stabilise, through conversion from the large pool of individuals with pre-diabetes. As complications such as DR can occur even in pre-diabetes,⁵⁸ there is need for an integrated response that includes prevention, diagnosis, treatment, and regular screening for complications.

The age adjusted prevalence of diabetes was highest among urban (3.4%) compared to rural (1.9%) residents. Similarly, the prevalence was highest in the richest wealth quintiles (5.2%) compared to the poorest wealth quintile (1.6%). Previous population based studies had also identified the urban-rural variation.⁵⁹ That diabetes is still more prevalent in the urban and more affluent segments of the population suggests that the epidemic is still in the early stages. However, given that the majority of the population is rural, even a small increase in the prevalence in the rural areas (especially from conversion from the pre-diabetes pool) would have a significant impact. A cohort study of people age 50+ years in Nakuru Kenya found the six-year cumulative incidence of diabetes to be 61 per 1000, equating to about 10 new cases of diabetes per 1000 per year.⁶⁰ Assuming 10% of the 53 million is aged 50+, (5.3million) this would be 53,000 additional PLWD every year, excluding the new cases aged <50 years. A large scale up of service provision in the diabetes care cascade will be required.

The odds for diabetes increased with age and the prevalence was highest in 45–59 age category.⁵⁷ Given that diabetes affects working-age people, the consequences of diabetes extend beyond the treatment costs. Besides disability and death, other productivity effects have been described, including: absenteeism, presenteeism (the practice of being present at one's place of work for more hours than is required, especially among workers with reduced productivity due to health issues, as a manifestation of insecurity about one's job), unemployment and premature retirement, leading to reduced income and savings, low self-esteem and low quality of life.⁶¹⁻⁶³ This suggests the need for effective targeted intervention to help workers better manage their disease. Employers need to be engaged as pertinent stakeholders in diabetes control.

The proportion of undiagnosed diabetes in the STEPs survey was 52.8%, which although not surprising, has far-reaching implications for health-care providers in the country, as many currently undiagnosed PLWD carry the risk of advanced diabetic microvascular complications being present by the time they are diagnosed, resulting in kidney failure, retinopathy, neuropathies and diabetic foot. For patients diagnosed at a late stage, the complications are likely to be progressive.

Among the 47% of PLWD who were aware of their diagnosis, only 41% were on treatment, and among those on treatment only 33% had achieved glycaemic control.⁵⁷ This means that out of every 100 PLWD, 47 are aware of the diagnosis, 19 are on treatment and only six have glycaemic control; the other 93 do not achieve glycaemic control. In the poorest households, none of the PLWD were on treatment and as expected, none of them had achieved glycaemic control.⁵⁷ This combination points to low awareness, poor access, low treatment coverage, inadequate management of diabetes, and loss of PLWD along the diabetes care cascade. People with uncontrolled diabetes are likely to present with complications, especially the microvascular complications. This means that interventions are required at all the four levels of health care, rather than only primary care. Access to treatment should be a priority in order to prevent the development of complications. Interventions targeting the poor are required.

As expected, the direct cost of managing diabetes and its complications is much higher in private than public facilities. The largest costs are medication, clinical tests, inpatient costs and treatment of microvascular complications, which can result in catastrophic expenditure even in the public sector.^{7, 55, 64} The indirect cost is similarly high given that a high number of disability-adjusted life years (DALYS) lost are attributable to diabetes-related illnesses (364 per 100,000 person years of

observation).^{64, 65} Seeing that diabetes is a costly disease in Kenya,⁵⁵ and that the complications are the largest source of the overall costs, there is a need for programs that can control these complications. Possible interventions include removing out-of-pocket payments, which will require more effective engagement of insurers and other funders of health care.

Based on the epidemiological data from the STEPS survey, to reduce the burden of diabetes at the population level, there is need to ensure all people with diabetes get tested, diagnosed and enter the continuum of care and are retained without dropping out. Successes in access to and retention in the diabetes care cascade can lead to better diabetes outcomes, and reduction in the health, social and economic burden of diabetes. This is likely to be achieved through a chronic care model, which is discussed below.

2.2 Chronic care model for diabetes care

The *Chronic Care Model* (CCM), developed by Wagner and colleagues in the 1990s, ^{66,} ⁶⁷ has been proposed as a suitable model for managing diabetes, Fig 3-3. The CCM envisions linking actively engaged patients with proactive health care providers.⁶⁸ Studies have shown that CCM is beneficial to chronic care especially at primary care, even in SSA.⁶⁹⁻⁷¹ Some CCM-related interventions in Kenya have been reported, ^{72, 73} and some aspects are routinely implemented in the country, but not as a systematically defined package.

The CCM model enumerates six essential elements that encourage high-quality chronic disease care. These elements are *self-management support, delivery system design, decision support and clinical information systems, the health system, and community resources.* Cost-effective interventions within each of these components

can be implemented as a package.⁶⁸ Complex interventions that incorporate several of these elements have been shown to be more effective than single interventions, but none of them has been found to be the dominant driver of success in diabetes care.⁶⁸ The actual interventions within each component also show heterogeneity in most studies. The next section provides a description of how these components might be operationalized.





Diabetes Self-management support should focus on patient/family empowerment, and development of skills such as self-monitoring/home-based monitoring, administering insulin injections, action-planning and problem-solving.^{74, 75} A clinician or trained diabetes educator may provide diabetes education through different channels such as face-to-face individual education, group education, education materials, online modules or mobile applications for the promotion of healthy lifestyle.⁷⁵

Delivery system support is about strengthening the coordination of patient-centred services for PLWD, both during hospital visits and follow-up. This includes giving standing orders for annual screening tests, automated reminders, and developing a multidisciplinary team (nurses, clinicians, physicians, pharmacists, nutritionists, diabetes educators and other specialists). Team roles are specified (for example who takes care of foot examinations or DR screening), teams are trained, and they meet regularly to review progress.^{74, 75}

Decision support helps to ensure that health workers attending to PLWD have the expertise that they need. Typical activities include training of multidisciplinary health care providers on the use of evidence-based guidelines or protocols, and providing these guidelines.^{70, 74} This can promote diabetes care by strengthening the workforce to provide quality care, and ensuring timely referral of PLWD who need additional care.

Electronic records, a diabetes registry and reminder or recall systems are important aspects of the *clinical information systems* for diabetes.^{74, 75} This can help track referrals, clinical and public health outcomes, and strengthen engagement of PLWD along the care cascade to reduce the potential for dropout. A call and recall mechanism can also ensure that patients are reminded of their screening schedule.

The leadership provided by the *health system* determines how interventions for diabetes are implemented.^{70, 76} Providing sufficient resources (space, workforce, standards, information systems, funds); retraining health workers to give them a chronic disease orientation; removing barriers to care; integration of diabetes care

across different health care providers; increasing collaboration between specialists that attend to PLWD; implementing quality assurance measures and removing health system barriers can ensure that PLWD get all the services they require.^{70, 74}

Community resources may help patients acquire self-management skills, such as achieving lifestyle modifications that positively influence chronic disease control. For example, community health volunteers (CHVs) and peer supporters are a flexible and cost-effective strategy in the control of diabetes in multiple settings.⁷⁷⁻⁸⁷ Local gyms, exercise programs and weight loss programs are also useful for lifestyle goals. Diabetes programs and PLWD should be encouraged to engage with these resources to fill gaps in care. As diabetes support groups (DSGs) are an existing community resource external to the formal health system in Kenya, and since the peer supporters are PLWD themselves, the next section explores their unique contribution.

2.3 Peer support interventions

2.3.1 Peer support in health care

Peer support as a health intervention has been defined as 'the provision of emotional, appraisal, and informational assistance by a created social network member who possesses experiential knowledge of a specific behaviour'. ^{8889, 90} The peer supporter has similar characteristics as the target population and the relationship relies on the following factors: non-hierarchical, flexible and reciprocal relationships,^{79, 88, 89} sharing similar life experiences, mastery of self-care behaviours and reduced social distance between peers.^{78, 87}

There are diverse modes of peer support interaction (individual or group sessions, face-to-face, telephone or online interaction), which occur in diverse settings such as homes, health facilities, schools, prisons. Support can be structured as an informal or

a formal interaction (for example through training peer supporters). There are varied types of peer supporter training and peer group composition (homogenous or mixed, disease type) ^{78, 91}.

There are multiple applications of peer support in health care. It has been used as a health promotion approach to encourage behaviour change interventions such as for exclusive breastfeeding, prevention of Human Immunodeficiency Virus (HIV) and Sexually Transmitted Infections (STI), smoking cessation, mental health and cancer screening among other areas. In Kenya, peer support has been used to help patients with HIV transition between inpatient and outpatient care, for continuity of follow-up.⁹²

A systematic review of peer support for breastfeeding found the role of peer support to be most important during the early postnatal period.^{90, 93} Individual peer support for exclusive breastfeeding in Uganda was reported to be acceptable to women.⁹⁴ In Bangladesh, women's groups with high population coverage reduced neonatal mortality.⁹⁵ In Malawi, women's groups and health education by peer counsellors were similarly found to have positive effects on childhood mortality, maternal mortality and breastfeeding rates.^{93 88, 90, 96} In China, community-based peer support for patients with severe mental illness has been found to be acceptable.⁹⁷

2.3.2 Peer support in diabetes

There is evidence that peer support can promote the implementation of diabetes care.^{80, 82, 85, 98} Peer supporters can assist people with their daily diabetes selfmanagement activities, provide emotional and social support, assist and encourage clinical care and be available when needed. ^{99, 100} A needs assessment in the Kilimanjaro Diabetic Program in Tanzania found that living with diabetes is not associated with stigma, and PLWD have a desire to raise community awareness of diabetes and support others to live with the disease and secure social support to access hospital services.¹⁰¹

Peer approaches in other settings have been found to promote physical activity, healthy eating, glycaemic control and reduce hospitalisation of PLWD.^{80, 99, 102-104} A peer intervention led to significant reduction in the combined prevalence of type 2 diabetes and intermediate hyperglycaemia as well as a 2-year incidence of type 2 diabetes among an intermediate hyperglycaemia cohort of adults in Bangladesh.¹⁰⁵

Several systematic reviews on the effectiveness of peer support have been published. Qi and colleagues in 2015 found that peer support had a significant impact on improving HbA1c levels in patients with poor glycaemic control.⁷⁸ Dale et al in 2012 found that peer support improved clinical and behavioural outcomes in some adults with diabetes.¹⁰⁶ Fisher et al in 2017 found that peer support could reduce HbA1c levels. ⁴⁸ Gatlin et al in 2017 found that peer led self-management education can increase diabetes knowledge and reduce HbA1c.¹⁰⁷The studies included in these reviews were mostly from high-income settings and the models of peer support were heterogeneous. Having found no literature on the role of peer support specific to the long-term complications of diabetes, research in this area would help to expand knowledge and expand the application of the chronic care model in diabetes care.

Although peer support is a scalable and potentially low cost intervention that would be particularly useful in low resource settings, ^{79, 100, 108}some literature cautions against over-optimism in its potential to influence health outcomes. ^{81, 109} Local evidence, especially mixed-methods implementation research would be needed as peer support interventions are developed and evaluated, in order to identify what types of peer support work, where, how and why.

Peer support is not a monolithic 'one-size fits all' concept, and these studies have used different types of peer support. There are community based, home-based and facility-based approaches. Peer support can be peer-led or health worker-led. It can be individual or groups peer support, and can be implemented through face-to-face, telephone, internet or other strategies. Brownson and Heisler summarized six essential contributions of peer supporters to diabetes care in any of these models, which are listed below.⁸⁷ These points outline the utility of peer support.

- Improving access to regular, safe, high-quality clinical care: Peer supporters conduct outreach and case finding; make referrals; help PLWD navigate the healthcare system; serve as liaisons between PLWD and healthcare settings; coordinate care/services (case management); provide language translation; assist with applications and paperwork for insurance or other services/programs.
- Individualized assessment and tailored management: Assess needs of PLWD; assess readiness to change, level of literacy, and other life influences on their ability to self-manage; individualize education and support; and provide services in non-traditional settings (e.g. home visits).
- Collaborative behavioural goal setting and problem solving: Help patients set and reach specific behavioural goals; help in problem solving to overcome barriers.
- Education and skills for managing diabetes: Conduct outreach and recruitment for educational services, lead (or assist with) culturally appropriate and accessible self-management training and education; and teach/reinforce self-management skills.

- Ongoing follow-up and support: Provide non-judgmental follow-up, informal counselling, social support and encouragement; and provide instrumental support.
- 6. Linkage to community resources: Identify needed resources; develop relationships with community organizations; provide information and support to PLWD regarding available community resources; advocate for needed services; develop capacity within communities to support healthy behaviours

These points provide us with an understanding of the utility of peer support. Looking at these roles, we can construe that the goal of peer support is to empower the individual PLWD and to ensure regular access to diabetes services.

Turning to diabetic retinopathy, can peer support strengthen prevention, screening, diagnosis, treatment or rehabilitation for DR? We did not find any published evidence on peer support for DR. We suggest that this is an important area for investigation.

2.4 Diabetic retinopathy



Source: Community Eye Health Journal <u>https://creativecommons.org/licenses/by-nc/2.0/legalcode</u>

The long-term sequelae of diabetes, including diabetic eye disease (DED), make diabetes a complex comorbidity. DEDs include cataract, glaucoma, rubeosis iridis, optic neuropathy, ocular muscle dysfunction and retinopathy. The increase in prevalence of type 2 diabetes in LMICs will occasion proportionate increases in the rates of these complications.¹¹⁰ Cataract and glaucoma may occur in the absence of diabetes, but DR is a specific ocular manifestation of end-organ damage from diabetes (Fig 2-4). These conditions are potentially blinding and PLWD are 25 times more likely to go blind than the general population.¹¹¹

2.4.1 Epidemiology of diabetic retinopathy

Diabetic retinopathy is a progressive disease of retinal microvasculature and the most frequent microvascular complication of diabetes.¹¹² All PLWD are at risk of developing DR; the risk increases with the duration of diabetes as Fig 2-5 illustrates. After 20 years of living with diabetes, 80% of PLWD will have some form of DR.^{113, 114} At any point in time, 35% of PLWD have DR, and about 10% of PLWD (or about a third of those who have DR) have vision threatening DR (VTDR) with a high risk of severe visual impairment from DR.¹¹⁵ The early stages of DR are asymptomatic, but with progression over time, the main vision-threatening sequelae are diabetic macular oedema (DMO) and proliferative diabetic retinopathy (PDR), both of which are part of VTDR, Figure 4. These complications make DR a leading cause of acquired vision loss particularly in the working age population.^{112, 113, 116}



Figure 2-5: Association of prevalence of DR with duration of diabetes

The 2017 report of the Vision Loss Expert Group shows that the crude prevalence of blindness due to all causes except DR reduced between 1990 and 2015 globally.¹¹⁷ During these 25 years, the number of people with DR-related blindness and visual impairment (VI) increased, and the overall causal contribution of DR to blindness increased by 7.7%.¹¹⁷ This trend suggests that DR has potential to become one of the leading causes of VI globally in the future. At present, the prevalence of DR-related VI in Asia and SSA is much lower than in high-income countries.¹¹⁷ This also has potential to reverse, based on the evidence of the escalating prevalence of diabetes and the increasing life expectancy in these regions. An increased focus on the control of DR is thus appropriate. Although large population-based studies on DR in SSA are lacking, posterior-segment eye diseases (PSEDs), grouped together, are the second leading cause of blindness in the region, accounting for 13 to 37% of blindness.¹¹⁸

A recent systematic review has shown the annual incidence of DR globally ranges from 2.2% to 12.7% and progression from 3.4% to 12.3% among adult PLWD.¹¹⁹ In the Malawi Diabetic Retinopathy cohort study, progression from no retinopathy and

background retinopathy to VTDR over five years was approximately 5 and 3 times that reported in the Liverpool Diabetic Eye Study.¹²⁰ The 5-year incidence of any DR in Malawi was 48.4%. Poor glycaemic control and HIV infection were predictors of progression.^{121, 122} This was the first report of an association between HIV infection and progression of retinopathy. It is a pointer that infectious diseases and other variables in LMICs may affect the epidemiology of DR. The 6-year incidence of DR among PLWD age 50+ in the Nakuru cohort study in Kenya was 225 cases per 1000.⁵⁰ The other available incidence data for LMIC is from Bangladesh, where the cumulative incidence of DR over a 15-year period was 50.6%.¹²³ It is difficult to compare these findings because of different study designs, follow-up rates and insufficient stratification of data by age and other parameters. It is still uncertain whether DR occurs earlier or progresses faster in African populations as compared to PLWD of a similar age in other populations. However, the data is useful for conceptualisation of the need for services such as screening programmes for early detection of these cases.

The prevalence range for any DR among PLWD in SSA, as reported in individual studies (Table 2-1), is 7.0 to 62.4%.¹²⁴ This wide variation may result from: the variation in the mean duration of diabetes among the participants in the different studies; differences in study methods and changes in the prevalence of DR over time.

Table 2-1: Prevalence of diabetic retinopathy in sub-Sahara Africa

Country	Author	Year	Study setting	Sample size (n)	Prevalence (%)
Botswana	Mengesha et al ¹²⁵	2006	Clinic-based	401	9.2
Cameroon	Jingi et al ¹²⁶	2015	Clinic-based	407	17.4
Ethiopia	Seyoum and Mengistu ¹²⁷	2001	Clinic-based	302	37.8
Ethiopia	Teshome and Melaku ¹²⁸	2004	Clinic-based	1390	28.7
Ethiopia	Gill et al ¹²⁹	2008	Clinic-based	105	21
Kenya	Mwendwa <i>et al¹³⁰</i>	2005	Clinic-based	100	7.0
Kenya	Mwale <i>et al</i> ¹³¹	2007	Clinic-based	96	22.6
Malawi	Glover <i>et al¹³²</i>	2012	Clinic-based	249	32.5
Nigeria	Omolase <i>et al¹³³</i>	2010	Clinic-based	100	15
Nigeria	Onakpoya et al ¹³⁴	2010	Clinic-based	80	21.6
Southern Africa	Rotchford and Rotchford ¹³⁵	2002	Clinic-based	253	40.43
Southern Africa	Carmichael et al ¹³⁶	2005	Clinic-based	1517	26.5
Southern Africa	Mash <i>et al¹³⁷</i>	2007	Clinic based	400	62.4
Southern Africa	Reed and Cook ¹³⁸	2007	Clinic-based	248	32.3
Tanzania	Mumba et al ¹³⁹	2007	Clinic-based	86	20.9
Tanzania	Cleland et al ¹⁴⁰	2016	18 diabetes clinics	3187	27.9
Zambia	Lewis et al ¹⁴¹	2018	Clinic-based	2153	52
Malawi	Burgess et al ¹⁴²	2014	Cohort study	357	50.1
Kenya	Mathenge <i>et al¹⁴³</i>	2014	Population based survey	4414	35.9
Nigeria	Kyari <i>et al³²</i>	2014	Population- based national survey	1595	20.5

In Kenya, a Nakuru study reported PSED to account for 30.4% of blindness, of which DR accounted for 2%.^{144, 145} The Nakuru cohort study found the prevalence of any DR among PLWD to be 35.9%.^{143 115} This data can be used for planning services in Kenya, although more studies would help to consolidate the evidence. The Rapid Assessment of Avoidable Blindness (RAAB + DR) is a potential method to collect more local prevalence estimates.^{146, 147} Table 2-2 shows how the data on incidence and prevalence can be used for planning DR programs. We propose that interventions to increase access to screening among PLWD are vital in preventing vision loss.

Table 2-2: Planning for DR services in Kenya

NEED FOR DR SERVICES IN KENYA

Population at risk	
Total Population	53million
Population to be covered (consider age- above 20yrs = 45%)	24million
Diabetes Mellitus (PLWD)	
Prevalence of diabetes (number DM / 100 pop)	2%
Number of people with diabetes	480,000
Diabetic Retinopathy	
Proportion (%) of PLWD with DR	35%
Number of people with DR	168,000
Vision threatening diabetic retinopathy (VTDR)	
Proportion of PLWD with VTDR	10%
Number of people with Vision Threatening DR (VTDR) (need treatment)	48,000

2.4.2 Risk factors for onset and progression of DR

The major risk factors leading to development and progression of DR have been elucidated, Table 2-3. Key factors that are consistently associated with DR include longer duration of diabetes, prolonged hyperglycemia, and systemic hypertension.^{112,} ¹¹⁵ These factors apply to all types and stages of DR.¹¹⁵ Dyslipidaemia is associated with DMO, but its role in other types and stages of DR is the subject of debate.^{115, 148} The duration of diabetes is considered the strongest independent predictor for DR,¹¹⁴ but in Kenya where >50% of PLWD are undiagnosed, DR may be the first presentation of diabetes.⁵⁷ This raises the need to increase early detection of both diabetes and DR.

The Diabetes Control and Complications Trial (DCCT) and the United Kingdom Prospective Diabetic Retinopathy Study (UKPDS) showed the protective effect of good glycaemic control (HbA1c < 7%) in reducing the risk and progression of DR in type 1 and 2 diabetes respectively.¹¹⁴ This effect has been quantified, such that a 1% decrease in HbA1c corresponds to a reduced risk of retinopathy by 40%, progression of DR to VTDR by 25%, need for laser therapy by 25%, and blindness by 15%.¹¹³ On the other hand, tight glycaemic control carries the risk of hypoglycaemia, and requires a strong compliance from both PLWD and physicians. High levels of undiagnosed diabetes, low treatment coverage for diabetes, and non-adherence to medication, which have been already highlighted in Kenya, are likely to be a challenge for achieving good glycaemic control.^{51, 57}

The UKPDS showed the positive influence of tight blood pressure control. Every 10mmHg reduction in systolic blood pressure results in a decreased risk of retinopathy progression by 35%, need for laser therapy by 35%, and visual loss by 50%.¹¹³ Crucial to note, the STEPS survey in Kenya found that 56% of the

population 18-64 years had never checked their blood pressure and 22% of those who were already diagnosed with hypertension were not on treatment.^{50, 51} This is another challenge for DR control programs.

Given these risk factors, prevention of DR-related vision loss will require close collaboration between diabetes and eye care services.¹⁴⁹ At present, such collaboration is minimal.¹⁵⁰ The landmark trials mentioned above, along with the Early Treatment Diabetic Retinopathy Study (ETDRS) have shown that blindness from DR is almost entirely preventable with early detection of DR, timely treatment and effective diabetes control.¹⁵¹ To prevent vision loss, metabolic control and yearly retinal examinations are recommended for PLWD, beginning at the time of diagnosis of type 2 diabetes or 5 years after diagnosis of type 1 diabetes, with increasing frequency once DR is identified.^{152, 153} At present, although PLWD in LMICs have some awareness of DR as a possible diabetes complication, specific knowledge about DR is rare, and many PLWD do not have an eye examination before they experience vision loss.¹⁴⁷ One study reported that the mean time between diabetes diagnosis and DR screening for PLWD in different parts of LMICs ranged from 8.0 (±7.0) to 14.4 (±10.8) years.¹⁵⁴ Screening programs must therefore have a strong health education and advocacy component.

As can be seen from Table 2-3, three priority risk factors for NCDs (physical inactivity, tobacco use, unhealthy diet) have been associated with onset and progression of DR. There is therefore optimism that the proposed interventions for NCDs will be useful to prevent and control DR. The association of DR with alcohol intake, another priority risk factor for NCDs, is the subject of debate, and the evidence is inconclusive at present, an area for further research.^{155, 156} There is no published evidence on the association of DR with the fifth priority risk factor for NCD, air pollution. Crucially,

there is a need for more evidence on other risk factors that are of special interest to

Africa, such as co-pathology with HIV, malaria, tuberculosis, malnutrition,

micronutrient deficiencies and anaemia.33, 142

Table 2-3: Risk factors for onset and progression of Diabetic Retinopathy

Patient factors

Non-modifiable: increasing duration of diabetes, ^{157, 158} increasing age, ¹⁵⁹⁻¹⁶¹ African, South Asian and Latin American ethnicity, ^{159, 162} male gender^{159-161, 163} metabolic hormones (leptin and adiponectin)¹⁶⁴

Modifiable: hyperglycaemia ^{158, 165}, high blood pressure ^{158, 165}, dyslipidaemia^{158, 166}, physical inactivity¹⁵⁷, smoking,¹⁶⁷ obesity,¹⁶⁴ pregnancy,¹⁶⁴ lack of awareness,¹⁶⁸ lack of access to services¹⁶⁹ inflammation and oxidative stress¹⁶⁴

Health worker factors

Lack of knowledge, lack of education in diabetes care, poor communication with patients, failure to individually recommend patients for diabetes screening¹⁶⁸ lack of collaboration between providers of diabetes and eye services¹⁶⁹ ambivalence towards referral of patients, non-adherence to practice guidelines¹⁷⁰

Health system factors

Limited and irregular anti-diabetes drug supplies¹⁶⁸, high cost of treatment and tests¹⁶⁸, lack of glucometers, strips and syringes¹⁶⁸, lack of screening programs¹⁶⁸, lack of diabetes registries¹⁶⁸, lack of coordination of care¹⁶⁸, stigma associated with diabetes in the society,¹⁶⁸ lack of clinical guidelines,¹⁷⁰ lack of endocrinologists¹⁶⁸ and ophthalmologists¹⁶⁸

2.4.3 Impact of DR-related vision loss

The early stages of DR are asymptomatic but the development of severe nonproliferative DR, proliferative DR and diabetic macular oedema causes irreversible vision loss.^{113, 171, 172} In the early stages of DR, vision specific functions are maintained at a similar level with those with no DR. However, persons with VTDR are up to 6 times more likely to have a low overall vision-specific functioning (participation in vision-dependent activities such as reading, watching television or driving) compared with those with less severe stages of DR, independent of their presenting visual acuity.¹⁷³ Activities requiring fine visual acuity such as reading small print are particularly affected. Loss of contrast sensitivity also occurs, especially with maculopathy. Laser photocoagulation is destructive to the retina and causes loss of visual field. Further, PLWD who have DR are at an increased risk of falls.¹⁷⁴ As a result of vision loss, physical mobility is reduced, PLWD become socially isolated and often dependent on others.¹⁷⁵ Thus, it is important to bring in interventions that can preserve vision.

PLWD have difficulties in managing diabetes once they are visually impaired. For example, they have difficulty measuring their blood glucose or measuring the dose of insulin. This can lead to poor glycaemic control, deterioration in DR, frequent need for hospitalization and increased mortality. Furthermore, patients with DR are at greater risk of other systemic vascular complications, including stroke, coronary heart disease, and congestive heart failure.¹⁷³ Physicians and ophthalmologists should therefore ascertain that patients with DR are receiving appropriate assessment and treatment for these comorbidities.¹⁶⁴ Diabetes, DR and vision loss are independently associated with mental health problems such as emotional distress, depression and reduced psychosocial functioning.¹⁷⁶⁻¹⁷⁸ Fear and anxiety relating to potential visual loss as well as fear of laser photocoagulation treatment for DR has been reported.¹⁷⁵

There is clear evidence of the heavy economic burden caused by diabetes, DR and vision loss. ^{4, 64, 179} DR incurs substantial treatment costs, often necessitates increased health care utilization and significantly reduces health-related quality of life, once VI occurs. ⁵⁵ Further, diabetes and DR are associated with loss of productivity, in terms of loss of employment and lost workdays. In a US population, the presence of diabetes itself reduced employment of PLWD by 3.5%, and the presence of complications reduced employment by 12% as compared to similar population without diabetes.¹⁸⁰ Given the economic impact of diabetes and DR to the PLWD, their

families and society, and given that vision loss worsens this burden and the quality of life, prevention of visual impairment is crucial. Early detection and timely treatment are essential.

2.4.4 The role of health systems in the control of DR

To reduce the incidence of blindness due to DR, a health system should: (1) ensure appropriate metabolic control of diabetes, as previously described; (2) enable early detection of DR through regular retinal screening; and (3) provide appropriate treatment and follow-up for patients with DR.¹⁸¹ As both diabetes and DR are increasing, and DR awareness remains patchy and low in many populations, there is a need to increase the awareness in PLWD and in health workers regarding these services. ¹⁸²

Retinal Screening

The first screening tests for DR were conducted in 1980 in Iceland. ¹⁸³ Wilson and Jungner (1968) established the criteria for screening in medicine, which was adopted by the WHO (Table 2-4). ¹⁸⁴ Subsequent debate on the criteria has led to additional specifications of these criteria (Table 2-5). The St Vincent Declaration of 1989 was the first international commitment to apply screening to DR, targeting the reduction of DR-related blindness by a third over the next 5 years. Three decades since then, the application of these criteria continues. In 2005, the Liverpool Declaration sought to reduce DR-related blindness through systematic screening with a coverage of at least 80 % of PLWD by 2010, using trained personnel and access to the proper therapy. Since then many developed countries have initiated screening programs that use ophthalmoscopy, retinal photography and telemedicine.¹⁸⁵ Understanding how the screening criteria can be translated into practice in LMICs where there is no
established systematic screening programmes is crucial. The implications of these screening criteria in the context of LMICs need to be considered. Although there is an assumption that these criteria apply universally, successful DR screening (DRS) programs should be suited to the context, and context influences the translation of the criteria into practice.^{147, 186}

	Table 2-4:Wilson and Jungner classic screening criteria ¹⁸⁴
1.	The condition sought should be an important health problem
2.	The natural history should be adequately understood
3.	There should be a detectable early/latent/asymptomatic stage
4.	There should be a suitable test for the early stage
5.	The test should be acceptable
6.	Treatment at an early stage should be of more benefit that at a later stage
7.	There should be an agreed policy on whom to treat as patients
8.	There should be adequate health service provision for the extra clinical workload resulting from the screen
9.	The costs of diagnosis and treatment should be balanced against benefits
10.	Case finding should be a continuing process and not a "once and for all" project

Table 2-5: Emerging screening criteria proposed since 1968¹⁸⁷

- 1. The screening programme should respond to a recognized need
- 2. The objectives of screening should be defined at the outset
- 3. There should be a defined target population
- 4. There should be scientific evidence of screening programme effectiveness
- 5. The programme should integrate education, testing, clinical services and programme management
- 6. There should be quality assurance, with mechanisms to minimize potential risks of screening
- 7. The programme should ensure informed choice, confidentiality and respect for autonomy
- 8. The programme should promote equity and access to screening for the entire target population
- 9. Programme evaluation should be planned from the outset
- 10. The overall benefits of screening should outweigh the harm

In the previous sections the concept of DR as a chronic public health problem with a known natural history has been explored.^{152, 188} The previous sections have presented evidence that the increasing prevalence, and the adverse health, social and economic impact are significant factors for PLWD, heath care providers and policy-makers. The goal of DR screening is to identify PLWD with asymptomatic DR and provide timely treatment for those with VTDR.¹⁸⁹ The target population for screening is well defined, since all PLWD are at risk, and will develop some DR if they live long enough. There is consensus that screening is required. However, the evidence base so far is from high income settings. While it is known that DR is generally progressive, the rate of progression in different settings is uncertain. It is also unknown whether the probability of adverse visual outcomes for PLWD in LMIC corresponds to the evidence from developed countries. Yet this is a learning point for LMICs – local evidence collected from screening programs would help identify the optimal screening intervals and methods for LMIC with more certainty.

As DR has a latent asymptomatic stage and effective treatment is available, annual examination of the retina is recommended.^{152, 190} However, screening all patients with diabetes for retinopathy has remained an elusive target in most countries.¹⁹¹ Lack of appropriate screening is a risk factor for VTDR and vision loss. The determinants of uptake of an eye examination are context-specific and can be categorised as personal, provider and health system factors.^{171, 192, 193} Evidence on the barriers from different LMIC settings have been collated,¹⁹⁴ but screening programs will need to continually collect and use the evidence from their own settings.

Retinal examination can detect the latent, asymptomatic phase of DR.¹⁸⁹ International classifications of DR gives the specific parameters to be used for classifying DR and the subsequent clinical pathways to pursue.¹⁹⁵ All stages of DR (as specified in these classifications) are detectable with the same screening test. Recognition of early DR is helpful as it provides evidence of early disease, and appropriate metabolic control can prevent or delay further deterioration. There is considerable evidence that prompt treatment of VTDR can prevent vision loss.^{113, 196} The challenge in LMICs is to ensure that all PLWD found to have VTDR get appropriate referral for treatment and access the treatment on time. Therefore, it is important that treatment services are available before establishing a systematic screening programme.

Retinal examination is considered safe, non-invasive and relatively quick. One of the requirements that can be challenging is mydriasis (pupil dilatation). Mydriasis is considered safe in SSA, as acute angle closure is not common; in Asia angle-closure is common hence non-mydriatic modes of retinal examination may be preferable.¹⁴⁷ Temporary blurring of near vision after pupil dilatation can discourage PLWD from uptake of retinal examination, especially if adequate explanation is not provided.¹⁸⁶ Further challenges that can discourage uptake of screening include the cost of travel

and time away from work. False positives may raise alarm, while false negatives may raise false security. It is therefore important to consider the diagnostic test accuracy (DTA) of different screening methods in comparison to the gold standard, which for routine practice is clinical examination of the retina by an eye specialist.¹⁹⁷ The recommended threshold for tests used in systematic population-based screening is over 80% sensitivity, over 95% specificity and less than 5% technical failure rate for satisfactory images or examination. ^{189, 198, 199}

Dilated slit lamp bio-microscopy and ophthalmoscopy (direct or indirect) are the main modes of screening in LMICs.¹⁸⁶ The sensitivity of direct ophthalmoscopy for detection of VTDR is only 65%, which is below the 80% threshold. However it is still useful for opportunistic case-detection.¹⁸⁹ Indirect ophthalmoscopy has the limitation of requiring extensive training, but it's accuracy is high and it is considered an alternative to direct ophthalmoscopy.^{200, 201} Slit lamp bio-microscopy through a dilated pupil has a high sensitivity and specificity of 89% and 94% respectively, but is time consuming and requires a slit lamp.²⁰² The main disadvantages of these clinical examination methods is that the findings cannot be captured and stored for quality assurance. Fundus fluorescein angiography is expensive, time consuming, invasive, requires specific expertise, has a small risk of anaphylaxis and is not readily available in LMICs, hence it is not used as a screening tool. A combined colour camera and Optical Coherence Tomography (OCT) is useful for detection of diabetic maculopathy, however it not widely available or cost-effective for population-based screening.¹⁸⁹ Screening programs in LMICs are gradually adopting mydriatic or non-mydriatic retinal photography. The limitation is the need for investment, as a single camera may cost \$20,000, although low-cost fundus cameras are becoming more available.²⁰¹ A recent systematic review reports that single field views of both mydriatic and non-

mydriatic digital imaging methods generate a satisfactory level of sensitivity, i.e. 86% for the detection of any DR, once ungradable images are excluded from analysis.²⁰² The specificity is 91% (non-mydriatic camera) and 87% (mydriatic camera), which is below the 95% recommended threshold, but this improves to meet the threshold if two or more fields are taken. For the detection of VTDR, the sensitivity and specificity also increases with the number of fields of view, providing evidence that the number of fields taken is an important consideration in photography-based screening. In one study the use of mydriasis reduced the number of ungradable images from 27.1% to 8.3%, using a single-field strategy.¹⁸⁶ This is important because cataract and corneal opacities can be a source of ungradable images.¹⁸³ However, non-mydriatic imaging reduces the screening time and inconvenience of pupil dilatation, and may thus be more efficient and acceptable to both PLWD and health care providers.

Bragge and colleagues in their meta-analysis that included 20 studies found that screening involving non-medical photographers yielded lower specificity as compared to screening involving photographers with specialist medical or eye qualification, particularly with non-mydriatic imaging.²⁰³ Low specificity has potential to increase the false positives and unnecessary referrals. It may be attributable to imaging difficulties or poorer image quality in their photographs. However in a more recent analysis that included 26 studies, Piyasena and colleagues found that non-medical retinal image graders could achieve the threshold level of sensitivity and specificity in both mydriatic and non-mydriatic strategies.¹⁸⁶ As screening is not an efficient use of ophthalmologists' time, task-shifting to trained retinal graders is a useful option, as already exemplified by the English DR Screening program.²⁰⁴ While recommending this option, we emphasise the pre-requisite for training, certification and quality assurance mechanisms for the graders.

Screening at scale is cost-effective but can be expensive and labour intensive particularly for rural communities. One tool to address this is telemedicine. The diagnostic accuracy of telemedicine using digital imaging in DR is overall high (95% sensitivity and 86% specificity for detection of any DR). ^{116, 205} Considering the high sensitivity in the detection of any clinical level of DR, telemedicine can be used widely for DR screening. It has the potential to deliver cost-effective screening to rural, remote and hard-to-reach PLWD. On the downside, it still requires a fundus camera, which is expensive. Secondly, noncompliance with follow-up appointments has been reported in telemedicine programs, leading to low take-up of the screening results and hospital referrals.¹⁴⁷ One study in Kenya found that only 58% of PLWD return to a tele-ophthalmology site to receive screening results.²⁰⁶ This illustrates that the telemedicine programs must extend efforts beyond capturing the images, to keep PLWD engaged. Other essential components for the use of telemedicine are robust technology, clear referral pathways and quality assurance. The retinal imaging devices must be validated and meet the specifications for image resolution which some programs have recommended as 20 pixels per degree with a resolution of at least 10-15 mm.201

Another tool is automated image analysis. A systematic review found that automated screening has high sensitivity (87-95%) but low specificity (49.6-68.8%). In this review, false negatives were likely to be mild DR with low risk of progression within one year. However, several studies in this review also reported missed cases of DME, which is a concern.²⁰⁷ The use of cloud-based image analysis software is an additional emerging trend. There is need for more evidence on the diagnostic accuracy of the analytic software especially in the African population. There are only two studies that have tested the use of artificial intelligence (AI) in imaging African eyes, which are usually more pigmented than other ethnic groups, and the results have been positive.^{208, 209}

There is also need for research on the ethical considerations, given the data security concerns.

A third tool for which there is growing interest is smart-phone based retinal screening. Validation studies are ongoing for different prototypes that use the principles of direct or indirect ophthalmoscopy. Bilong et al have reported the findings for a study where retinal images were taken through a dilated pupil, using a smartphone attached to an adaptable camera. Indirect ophthalmoscopy was also performed. The retinal images were sent via internet to a retinal specialist for interpretation. Sensitivity and specificity for the detection of any DR, was 73.3% and 90.5%. For DME it was 78% and 95%. However, for severe NPDR and PDR the sensitivity increased above the standard threshold.²¹⁰ At present, there is insufficient evidence to make any recommendations regarding this modality in screening.

The cost-effectiveness of DR screening compares well with other screening programs.¹⁹⁹ Data from the Liverpool Diabetic Eye Study showed that photographybased systematic screening is also more cost-effective than opportunistic screening or examination by a human screener, even if the screener has a sensitivity of 95%.¹⁹¹ Systematic screening is also cost-effective in terms of sight years preserved compared with no screening.²¹¹ Early diagnosis and early treatment of diabetic retinopathy is cost-effective, and even cost-saving²¹² if the cost of VTDR is considered and if there is at least 80% compliance to screening.²¹³ Coverage is an important driver for costeffectiveness.

Achieving the desired coverage may be challenging especially in rural and remote areas. Outreach DR screening is one way of improving screening coverage and can reduce overall health care costs, although quality assurance should be maintained.²⁰³ The highest DTA (sensitivity 90% and specificity 95%) is observed when screening is

delivered at secondary/tertiary level clinics. ¹⁸⁶ However DTA is not the sole factor that needs consideration. Context, resources, costs and coverage vary widely and the success of screening programs depends on understanding the 'best fit' within the existing system.

The appropriate timing and frequency of screening is an important consideration. DRassociated vision loss has been found to increase as the screening interval is extended from one to five years.²¹⁴ Annual screening is widely recommended, although some programs are moving to examination every two years if the initial examination is normal.^{191, 195, 215} The evidence to-date is that screening every two years is costeffective, but annual screening is outside the cost-effectiveness range, defined as less than three times the gross domestic product (GDP), or <US\$3183 per quality-adjusted life year (QALY) gained.^{147, 201, 216} On the other hand, long screening intervals may be a disincentive for PLWD, due to the potential for loss of contact and the interpretation that vision loss is unlikely and therefore not a concern.²¹⁷ Therefore yearly intervals for those with no DR are generally preferred, unless individual risk of progression (glycaemic and blood pressure control) can be assessed with certainty.²¹⁷ For those with DR, screening intervals depend on the severity of the DR.

There is consensus that VTDR should be treated, and the goal of treatment is to halt progression and prevent vision loss; restoration of lost vision is uncommon.¹¹³ Because treatment is aimed at preventing vision loss and retinopathy can be asymptomatic, it is important to identify and treat patients early in the disease. Early laser treatment for diabetic retinopathy is safe, effective and universally agreed.^{190, 218} The DRS and ETDRS studies showed that timely laser decreases the risk of severe vision loss (15 letters) by 52%; early treatment of clinically significant macula oedema

(CSME) reduces the risk of severe vision loss by 50% at five years and reduces the need for treatment to only 8% at two years.^{196, 199, 219}

Anti-vascular endothelial growth factor (anti-VEGF) therapy is increasingly being used for DMO and has been shown to maintain visual acuity if used early and frequently in the course of the disease.¹⁹⁶ While both laser and anti VEGF injections are available treatment options for DMO, there is ongoing debate regarding which is the best treatment in LMICs, given the logistical and cost implications for each method.

The realities in LMICs provide important considerations about what policies should be prioritized. Based on the discussion above, it is clear that DR screening as applied to LMICs meets the screening criteria. However, in an environment where there are operational or implementation complexities, screening tends to be neglected. These complexities include infrastructural capacity (equipment, technology), trained human resources, program coordination and evaluation, logistics (reaching hard to reach groups), social considerations (e.g. compliance, informed consent, confidentiality and autonomy) and sparsity of local evidence. Given these uncertainties, innovative and context-specific solutions to improve screening services are required.

Photography-based systematic screening strategies are recommended, as they offer patient convenience, efficient use of skills, quality assurance and ability to detect other eye conditions. Retinal imaging devices must meet the minimum acceptable standard for screen resolution. The standards used in the English program is a vertical resolution of 1080 (1920 9 1080) with an achievable and recommended standard of a minimum of 1200 (1920 9 1200 or higher). It is recommended that a minimum of 60% of the image should be viewable on the grading screen to avoid too much scrolling to see the full image.²⁰⁴ Opportunistic screening programs can use the tools available.

A potential leverage spot for screening programs is program management. To make the most of resources, and cause ripple effect in the capacity, volume, quality and cost-effectiveness, it is necessary to strengthen capacity to manage screening programs. For example, a program manager may help to bring endocrinologists, physicians, ophthalmologists, epidemiologists, economists, funders and policy-makers together on one page, for a coordinated and concerted response. There is no published literature specifically investigating approaches to sustain and coordinate program management in DR programs. This is an important area for future research.

Table 2-6 summarizes the interventions for control of DR-related vision loss. In LMICs barriers to the access and utilization of these interventions abound, including lack of diagnostic equipment and treatment infrastructure (lasers, laser maintenance, anti-VEGF, vitrectomy machines).³³ International clinical practice guidelines (CPGs) on the use of these interventions have been developed,¹⁹⁰ but locally adapted CPGs are often lacking.^{33, 150}In SSA where the mean ophthalmologist density is 3.7 per million population,²²⁰ there is need to invest in developing sufficient workforce as well as task shifting.²²¹

Table 2-6: Interventions for control of vision loss from DR

1 Prevention of diabetes

Lifestyle measures, especially healthy diet, regular exercise and weight vigilance.^{58,}

170, 222

2 Prevention of DR in PLWD

Good blood glucose control, BP control, control of hyperlipidaemia; avoid

smoking^{113, 114, 222, 223}

3 Prevention of visual loss from DR

Annual retinal screening examination, timely identification, referral and treatment

of VTDR (laser for PDR, and laser or anti-VEGF or intravitreal steroids for DMO),

with regular follow-up.^{191, 212, 224-227}

Restoration of vision in severe PDR with vision loss from vitreous haemorrhage

Good tertiary eye care services, including vitreo-retinal surgery^{113, 195, 223, 228}

5 Improving functioning in PLWD with visual loss

Visual rehabilitation services^{164, 195, 228}

2.5 Control of DR within an NCD, UHC and the health systems agenda

The previous sections have presented evidence from the literature that the health, economic and social burden of both diabetes and DR is likely to increase, and that LMICs carry a disproportionately high burden. It is clear that diabetes is a global health priority along with other NCDs. The NCD agenda recognizes diabetes not as a single disease, but as a complex comorbidity.⁵ To this extent, NCD 'best buy' interventions that are relevant to DR have been recommended in the agenda. What is less clear is why relevant targets and indicators for DR, or any other long-term complication of diabetes, are not included in the global NCD targets and indicators.²²⁹ The NCD targets focus on preventing mortality from diabetes. As we have described, DR causes morbidity and other social and economic effects, but it is not in itself fatal. Including targets relevant to DR, such as the number of people going blind or the loss in productivity from DR, might increase its visibility and attract investment in DR control at country level.

Many PLWD in LMICs who would benefit from the interventions for diabetes and DR do not use them. For example many are undiagnosed and untreated, or fall off the diabetes care cascade.^{4, 5} Addressing DR at the individual PLWD level is therefore a priority need for PLWD. However, interventions that only have effect at the individual level are insufficient for control of DR, as there is need for system level impact. This chapter has made a case about the relevance of interventions based on the multifaceted chronic care model.

To reduce the burden with NCD interventions, there is also need to prioritize surveillance and research, strengthening services and equity. These are some of the strategies that have contributed to the success of UHC and DR programs, for example in the United Kingdom.^{22, 198} Although LMICs cannot directly copy strategies used in high-income countries, they can learn from them. The Diabetic Retinopathy Network (DR-NET) provides an important learning opportunity in which existing VISION 2020 LINKS between UK and overseas eye departments share learning on DR screening and treatment.^{171, 230} At present there are 22 DR LINKS in the network representing 18 countries in Africa, the Caribbean and the Pacific.²³⁰ The DR-NET helps LMICs to develop collaboration between diabetes and eye services, set up diabetes registries, acquire screening and treatment infrastructure, develop clinical guidelines and plan for DR services using the DR-NET toolkit.²²⁸ Although three in four of PLWD reside in LMICs, there is a dearth of data from these countries, which makes it challenging to define needs and priorities.²³¹ National governments are making effort to address this; for example, the Ministry of Health Kenya has carried out a STEPS survey on the risk factors of NCDs.⁵¹ The data has helped to understand the epidemiological profile of diabetes, as discussed in this chapter. Local evidence about DR is required for planning DR services. In Kenya, the DR prevalence data so far has come from three observational studies (Table 1) and one longitudinal cohort study that provided the incidence data.⁶⁰ While using this data to plan for services, we need solution-based research to identify feasible approaches to engage PLWD and prevent non-adherence along the care cascade. We also need surveillance data for both process and clinical outcomes, such as the number of PLWD screened for DR and the number that progress to vision-threatening DR. This evidence might then serve to attract investment in services.

Like other NCDs, the control of DR has strategic links with health systems and UHC. The services required for control of DR are the interventions for prevention, screening, diagnosis, treatment and rehabilitation. These services have to be provided within a health system that is responsive to the changing epidemiology.^{232, 233} There is need to build a health system that will help people live in a way that prevents diabetes, achieve good control of their diabetes, provide regular retinal screening to detect DR in its early stages, and provide accessible and affordable laser or anti-VEGF treatment for those with VTDR to prevent blindness.¹⁷¹ As countries aspire to attain UHC, these health systems need these services to be comprehensive, have sufficient coverage, high quality and be at an optimal cost.²² Heath systems therefore play a crucial role in the response to the burden of both diabetes and DR. The WHO's global report on diabetes in 2016 reported variation in the capacity of health systems to deliver diabetes services.²³³ For example some LMICs countries did not have national policies or guidelines, basic technology for diagnosis of diabetes, or sufficient supplies of insulin in primary care. Planning services for DR also requires an assessment of the health system capacity.¹⁴⁹ There is value in periodic health system assessment, specific to diabetes and DR, as it helps to identify the needs, and facilitate subsequent health system strengthening to address priority gaps.

As NCDs become the leading disease category in developing countries, the provision of NCD-related services will increasingly form the backbone of health systems. Given the pluralistic needs of PLWD, there is need for investment in systems that make DR control programmes accessible, affordable, and sustainable. Weak health systems are an obstacle to access services, as well as to equity.¹⁵ At present, in many African health systems, laser photocoagulation and anti-VEGF drugs for DR are either unavailable or unaffordable, while the health workforce is insufficient, especially in rural areas. ^{149, 234}. Without equitable coverage, for example with the 'best buy' interventions, the poor / rural PLWD will continue to be left behind and jeopardize the attainment of the 2030 SDG agenda. Populations of PLWD that are disadvantaged should be prioritized, not necessarily by interventions that target them exclusively, but those interventions that bring them most benefit. As health systems differ, the interventions to improve equity are likely to be context specific.

REFERENCES

1. Kharroubi AT, Darwish HM. Diabetes mellitus: The epidemic of the century. World Journal of Diabetes. 2015;6(6):850–67.

2. NCD Risk Factor Collaboration (NCG-RisC). Worldwide trends in diabetes since 1980: a pooled analysis of 751 population-based studies with 4·4 million participants. Lancet. 2016;387:1513-30.

3. World Health Organization. Global Report on Diabetes. Geneva: World Health Organization; 2016.

4. International Diabetes Federation. IDF Diabetes Atlas, Eighth edition. 2017.

5. Chan JCN, Gregg E, Sargent J, Horton R. Reducing global diabetes burden by implementing solutions and identifying gaps: a Lancet Commission. The Lancet. 2016;387(10027):1494-5.

6. Seuring T, Archangelidi O, Subreke M. The economic costs of Type 2 diabetes: a global systematic review. PharmacoEconomics. 2015;33:811-31.

7. Mutyambizi C, Pavlova M, Chola L, Hongoro C, Groot W. Cost of diabetes mellitus in Africa: a systematic review of existing literature. Globalization and Health. 2018;14(1):3.

8. American Diabetes Association. Economic costs of diabetes in the US in 2012. Diabetes Care. 2013;36:1033-46.

9. Jaspers L, Colpani V, Chaker L, van der Lee SJ, Muka T, Imo D, et al. The global impact of non-communicable diseases on households and impoverishment: a systematic review. European Journal of Epidemiology. 2015;30(3):163-88.

10. World Health Organization. 'Best Buys' and other recommended interventions for the prevention and control of non-communicable diseases: updated appendix 3 of the Global Action Plan for the prevention and control of non-communicable diseases. Geneva: WHO; 2017.

11. United Nations General Assembly. Resolution adopted by the General Assembly 66/2. Political Declaration of the High-Level Meeting of the General Assembly on the Prevention and Control of Noncommunicable Diseases. New York: United Nations; 2011.

12. World Health Organization. Global action plan for the prevention and control of non communicable diseases 2013-2020. Geneva: WHO; 2013.

13. United Nations General Assembly. Transforming our world: the 2030 Agenda for Sustainable Development. New York: United Nations; 2015.

14. Hogan DR, Stevens GA, Hosseinpoor AR, Boerma T. Monitoring universal health coverage within the Sustainable Development Goals: development and baseline data for an index of essential health services. Lancet Global Health. 2018;6:e152-68.

15. World Health Organization, International Bank for Reconstruction and Development, World Bank. Tracking universal health coverage: 2017 global monitoring report. Geneva: WHO; 2017.

16. Nolte E, Bain C, McKee M. Diabetes as a Tracer Condition in International Benchmarking of Health Systems. Diabetes Care. 2006;29(5):1007-11.

17. Patcharanarumol W, Panichkriangkrai W, Wangmo S, Thammatacharee J, Uechi M, Wanwong Y. Diabetes prevention and care in the universal health coverage context: The example of Thailand. WHO South East Asia Journal of Public Health. 2016;5(1):27-33.

18. Hogan DR, Danaei G, Ezzati M, Clarke PM, Jha AK, Salomon JA. Estimating The Potential Impact Of Insurance Expansion On Undiagnosed And Uncontrolled Chronic Conditions. Health Affairs. 2017;34(9):1554-62.

19. Rodin A. Accelerating action towards universal health coverage by applying a gender lens. Bulletin of the World Health Organization. 2013;91:710-1.

20. Atun R, De Andrade LOM, Almeida G, Cotlear D, Dmytraczenko T, Frenz P, et al. Health-system reform and universal health coverage in Latin America. Lancet. 2015;385(9974):1230-47.

Chemouni B. The political path to universal health coverage: Power, ideas and community-based health insurance in Rwanda. World Development. 2018;106:87-98.
 Friebel R, Molloy A, Leatherman S, Dixon J, Bauhoff S, Chalkidou K. Achieving high quality universal health coverage: a perspective from the National Health Service in England. BMJ Global Health. 2018;3.

23. Ong SE, Koh JJK, Toh S-AES, Chia KS, Balabanova D, McKee M, et al. Assessing the influence of health systems on Type 2 Diabetes Mellitus awareness, treatment, adherence, and control: A systematic review. PLoS ONE 13(3). 2017;13(3).

24. Manne-Goehler J, Geldsetzer P, Agoudavi, K A-BG, Aryal KK BB, et al. Health system performance for people with diabetes in 28 low- and middle-income countries: A cross-sectional study of nationally representative surveys. PLoS Med. 2019;16(3):e1002751.

25. Krug EG. Trends in diabetes: sounding the alarm. Lancet. 2016;387:1485-6.

26. Peer N, Kengne AP, Motala AA, Mbanya JC. Diabetes in the Africa Region: an update. Diabetes Res Clin Pract. 2014;103(2):197-205.

27. Guariguata L, Whiting DR, Hambleton I, Beagley J, Linnenkamp U, Shaw JE. Global estimates of diabetes prevalence for 2013 and projections for 2035. Diabetes Res Clin Pract. 2014;103(2):137-49.

International Diabetes Federation. IDF Diabetes Atlas, Seventh Edition. 2015.
 Fottrell E, Ahmed N, Sanjit Kumer Shaha SK, Jennings H, Kuddus A, Morrison J,

et al. Distribution of diabetes, hypertension and non-communicable disease risk factors among adults in rural Bangladesh: a cross-sectional survey. BMJ Glob Health. 2018;3(e000787).

30. Williams R, Airey M, Baxter H, Forrester J, Kennedy-Martin T, Girach A. Epidemiology of diabetic retinopathy and macular oedema: a systematic review. Eye (Lond). 2004;18(10):963-83.

31. Cho NH, Shaw JE, Karuranga S, Huang Y, Fernandes JD, Ohlrogge AW, et al. IDF Diabetes Atlas: Global estimates of diabetes prevalence for 2017 and projections for 2045. Diabetes Research and Clinical Practice. 2018;138(2018):271-81.

32. Kyari F, Tafida A, Sivasubramaniam S, GVS. M, Peto T, Gilbert CE, et al. Prevalence and risk factors for diabetes and diabetic retinopathy: results from the Nigeria national blindness and visual impairment survey. BMC Public Health. 2014;14(1299):1-12.

33. Burgess PI, Msukwa G, Beare NAV. Diabetic Retinopathy in sub-Saharan Africa: meeting the challenges of an emerging epidermic. BMC Medicine. 2013;11(157).

34. Mendis S, O'Brien E, Seedat YK, Yusuf S. Hypertension and Diabetes: Entry Points for Prevention and Control of the Global Cardiovascular Epidemic. International Journal of Hypertension. 2013;1:1-3.

35. Atun R, Davies JI, Gale EAM, Bärnighausen T, Beran D, Kengne AP, et al. Diabetes in sub-Saharan Africa: from clinical care to health policy. The Lancet Diabetes & Endocrinology. 2017;5(8):622-67.

36. Kankeu HT, Saksena P, Xu K, Evans DB. The financial burden from noncomunicable diseases in low and middle income countries: a literature review. Health Research, Policy and Systems. 2013;11(31). 37. United Nations. World Population Prospects 2019 2019 [Available from: https://population.un.org/wpp/.

38. Achoki T, Miller-Petrie MK, Glenn SD, Kalra N, Lesego A, Gathecha GK, et al. Health disparities across the counties of Kenya and implications for policy makers, 1990–2016: a systematic analysis for the Global Burden of Disease Study 2016. Lancet Global Health. 2019;7(1):e81-e95.

39. Chuma J, Maina T, Ataguba J. Does the distribution of health care benefits in Kenya meet the principles of universal coverage? BMC Public Health. 2012;12(20).
40. Barasa E, Rogo K, Mwaura N, Chuma J. Kenya National Hospital Insurance Fund Reforms: Implications and lessons for Universal Health Coverage. Health Systems and Reforms. 2018;4(4):346-61.

41. Kenya National Bureau of Statistics, Ministry of Health/Kenya, National AIDS Control Council/Kenya, Kenya Medical Research Institute, Population NCf, Development/Kenya. Kenya Demographic and Health Survey 2014. Rockville, MD, USA; 2015.

42. Ministry of Health Kenya. Kenya Health Policy Framework 2014-2030:
Towards attaining the highest standards of health Nairobi: Ministry of Health; 2014.
43. Evaluation IoHMa. Health Service Provision in Kenya: Assessing facility
capacity, costs of care and patient perspectives. Seattle, Washington: IHME; 2014.
44. Ministry of Health. Strategy for Community Health 2014-2019 Transforming
health: Accelerating the attainment of health goals. Nairobi: Ministry of Health; 2014.

45. McCollum R, Otiso L, Mireku M, Theobald S, de Koning S, Hussein S, et al. Exploring perceptions of community health policy in Kenya and identifying implications for change. Health Policy and Planning. 2016;31:10-20.

46. Ngugi AK, Nyaga LW, Lakhani A, Agoi F, Hanselman M, Lugogo G, et al. Prevalence, incidence and predictors of volunteer community health worker attrition in Kwale County, Kenya. BMJ Global Health. 2018;3(4):e000750.

47. Ministry of Health. Kenya National Diabetes Strategy 2010-2015. Nairobi: Ministry of Health; 2010.

48. Fisher EB, Boothroyd RI, Elstad EA, Hays L, Henes A, Maslow GR, et al. Peer support of complex health behaviors in prevention and disease management with special reference to diabetes: systematic reviews. Clinical Diabetes and Endocrinology. 2017;3(1):4.

49. Shannon GD, Haghparast-Bidgoli H, Chelagat W, Kibachio J, Skordis-Worrall J. Innovating to increase access to diabetes care in Kenya: an evaluation of Novo Nordisk's base of the pyramid project. Global health action. 2019;12(1):1605704-.

50. Wamai RG, Kengne AP, Levitt N. Non-communicable diseases surveillance: overview of magnitude and determinants in Kenya from STEPwise approach survey of 2015. BMC Public Health. 2018;18(1224).

51. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases; 2015.

52. Siddharthan T, Ramaiya K, Yonga G, Mutungi GN, Rabin TL, List JM, et al. Noncommunicable Diseases In East Africa: Assessing The Gaps In Care And Identifying Opportunities For Improvement. Health affairs (Project Hope). 2015;34(9):1506-13.

53. Onyango MA, Vian T, Hirsch I, Salvi DD, Laing R, Rockers PC, et al. Perceptions of Kenyan adults on access to medicines for non-communicable diseases: A qualitative study. PLOS ONE. 2018;13(8):e0201917.

54. Government of Kenya. Kenya Service Availability and Readiness Assesment Mapping (SARAM). Nairobi: Ministry of Health; 2014.

55. Subramanian S, Gakunga R, Kibachio J, Gathecha G, Edwards P, Ogola E, et al. Cost and affordability of non-communicable disease screening, diagnosis and treatment in Kenya: Patient payments in the private and public sectors. PLOS ONE. 2018;13(1):e0190113.

56. Ministry of Health Kenya. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. In: Diseases DoN-C, editor. Nairobi2015.

57. Mohammed SF, Mwangi M, Mutua MK, Kibachio J, Hussein A, Ndegwa Z, et al. Prevalence and factors associated with prediabetes and diabetes mellitus in Kenya: results from a national survey. BMC Public Health. 2018;18(1215).

58. Diabetes Prevention Program Research G. The prevalence of retinopathy in impaired glucose tolerance and recent-onset diabetes in the Diabetes Prevention Program. Diabetic medicine : a journal of the British Diabetic Association. 2007;24(2):137-44.

59. Christensen DL, Friis H, Mwaniki DL, Kilonzo B, Tetens I, Boit MK, et al. Prevalence of glucose intolerance and associated risk factors in rural and urban populations of different ethnic groups in Kenya. Diabetes Research and Clinical Practice. 2009;84(3):303-10.

60. Bastawrous Andrew, Mathenge Wanjiku, Wing Kevin, Bastawrous Madeleine, Rono Hillary, Weiss Helen A, et al. The incidence of diabetes mellitus and diabetic retinopathy in a population-based cohort study of people age 50 years and over in Nakuru , Kenya. 2017:1-14.

61. Breton M-C, Guénette L, Amiche MA, Kayibanda J-F, Grégoire J-P, Moisan J. Burden of diabetes on the ability to work: a systematic review. Diabetes care. 2013;36(3):740-9.

62. Tunceli K, Bradley CJ, Nerenz D, Williams LK, Pladevall M, Elston Lafata J. The Impact of Diabetes on Employment and Work Productivity. Diabetes Care. 2005;28(11):2662-7.

63. Magliano DJ, Martin VJ, Owen AJ, Zomer E, Liew D. The Productivity Burden of Diabetes at a Population Level. Diabetes Care. 2018;41(5):979-84.

64. Moucheraud C, Lenz C, Latkovic M, Wirtz VJ. The costs of diabetes treatment in low- and middle-income countries: a systematic review. BMJ Global Health. 2019;4(1):e001258.

65. Etyang AO, Munge K, Bunyasi EW, Matata L, Ndila C, Kapesa S, et al. Burden of disease in adults admitted to hospital in a rural region of coastal Kenya: an analysis of data from linked clinical and demographic surveillance systems. Lancet Glob Health. 2014;2(4):e216-e24.

66. MacColl Center for Health Care Innovation. Chronic Care Model. Improving Chronic Illness Care. 1997.

67. Wagner EH, Austin BT, Von Korff M. Organizing care for patients with chronic illness. Milbank Q. 1996;74(511).

68. Beaglehole R, Epping-Jordan J, Patel V, Chopra M, Ebrahim S, Kidd M, et al. Improving the prevention and management of chronic disease in low-income and middle-income countries: a priority for primary health care. The Lancet. 2008;372(9642):940-9.

69. Ku GM, Kegels G. Adapting chronic care models for diabetes care delivery in low-and-middle-income countries: a review. World J Diabetes. 2015;6:566-75.

70. Lall D EN, Devadasan N, Horstman K, Criel B. Models of care for chronic conditions in low/middle-income countries: a 'best ft' framework synthesis. BMJ Glob Health. 2018;3(e001077).

71. Kane J, Landes M, Carroll C, Nolen A, Sodhi S. A systematic review of primary care models for non-communicable disease interventions in SubSaharan Africa. BMC Fam Pract. 2017;18(46):1-12.

72. Pastakia SD, Ali SM, Kamano JH, Akwanalo CO, Ndege SK, Buckwalter VL, et al. Screening for diabetes and hypertension in a rural low income setting in western Kenya utilizing home-based and community-based strategies. Globalization and Health. 2013;9(1):21.

73. Bloomfield GS, Kimaiyo S, Carter EJ, Binanay C, Corey GR, Einterz RM, et al. Chronic noncommunicable cardiovascular and pulmonary disease in sub-Saharan Africa: an academic model for countering the epidemic. Am Heart J. 2011;161(5):842-7.

74. Stellefson M, Dipnarine K, Stopka C. The Chronic Care Model and Diabetes Management in US Primary Care Settings: A Systematic Review. Preventing Chronic Disease. 2013;10(120180).

75. Stuckey HL, Adelman AM, Gabbay RA. Improving care by delivering the Chronic Care Model for diabetes. Diabetes Management. 2011;1(1):37-52.

76. Samb B, Desai N, Nishtar S, Mendis S, Bekedam H, Wright A, et al. Prevention and management of chronic disease: a litmus test for health-systems strengthening in low-income and middle-income countries. The Lancet. 2010;376(9754):1785-97.

77. Shah M, Kaselitz E, Heisler M. The role of community health workers in diabetes: update on current literature. Curr Diab Rep. 2013;13(2):163-71.

78. Qi L, Liu Q, Qi X, Wu N, Tang W, Xiong H. Effectiveness of peer support for improving glycaemic control in patients with type 2 diabetes: a meta-analysis of randomized controlled trials. BMC Public Health. 2015;15:471.

79. Paul G, Smith SM, Whitford D, O'Kelly F, O'Dowd T. Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes. BMC Health Serv Res. 2007;7:136.

80. Haltiwanger EP, Brutus H. A culturally sensitive diabetes peer support for older Mexican-Americans. Occup Ther Int. 2012;19(2):67-75.

81. van der Ven N. Psychosocial group interventions in diabetes care.pdf. Diabetes Spectrum. 2003;16(2):88-95.

82. Paul GM, Smith SM, Whitford DL, O'Shea E, O'Kelly F, O'Dowd T. Peer support in type 2 diabetes: a randomised controlled trial in primary care with parallel economic and qualitative analyses: pilot study and protocol. BMC Fam Pract. 2007;8:45.

83. Baksi AK. Experiences in peer-to-peer training in diabetes mellitus: challenges and implications. Fam Pract. 2010;27 Suppl 1:i40-5.

84. Bui TD, Kadzakumanja O, Munthali C. Mobilizing for the Lilongwe Diabetes Peer Support Programme in Malawi. Malawi Medical Journal. 2014;26(4):124-5.

85. Tang TS, Funnell MM, Brown MB, Kurlander JE. Self-management support in "real-world" settings: an empowerment-based intervention. Patient Educ Couns. 2010;79(2):178-84.

86. World Health Organization. Peer Support Programs in DiabetesDiabetes. Geneva: World Health Organization; 2007.

87. Brownson CA, Heisler M. The role of peer support in diabetes care and selfmanagement. Patient. 2009;2(1):5-17.

88. Dennis C-L. Peer support within a health care context: a concept analysis. International Journal of Nursing Studies. 2003;40(3):321-32.

89. Riddell MA, Renwick C, Wolfe R, Colgan S, Dunbar J, Hagger V, et al. Cluster randomized controlled trial of a peer support program for people with diabetes: study

protocol for the Australasian peers for progress study. BMC Public Health. 2012;12(843).

90. Kaunonen M, Hannula L, Tarkka MT. A systematic review of peer support interventions for breastfeeding. J Clin Nurs. 2012;21(13-14):1943-54.

91. Embuldeniya G, Veinot P, Bell E, Bell M, Nyhof-Young J, Sale JE, et al. The experience and impact of chronic disease peer support interventions: a qualitative synthesis. Patient Educ Couns. 2013;92(1):3-12.

92. Karwa R, Maina M, Mercer T, Njuguna B, Wachira J, Ngetich C, et al. Leveraging peer-based support to facilitate HIV care in Kenya. PLOS Medicine. 2017;14(7):e1002355.

93. Lewycka S, Mwansambo C, Rosato M, Kazembe P, Phiri T, Mganga A, et al. Effect of women's groups and volunteer peer counselling on rates of mortality, morbidity, and health behaviours in mothers and children in rural Malawi (MaiMwana): a factorial, cluster-randomised controlled trial. The Lancet. 2013;381(9879):1721-35.

94. Nankunda J, Tumwine JK, Nankabirwa V, Tylleskar T, Group P-ES. "She would sit with me": mothers' experiences of individual peer support for exclusive breastfeeding in Uganda. Int Breastfeed J. 2010;5:16.

95. Fottrell E, Azad K, Kuddus A, Younes L, Shaha S, Nahar T, et al. The Effect of Increased Coverage of Participatory Women's Groups on Neonatal Mortality in Bangladesh A Cluster Randomized Trial. JAMA Paediatrics. 2013;167(9):816-25.

96. Jirojwong S, MacLennan R. Health beliefs, perceived self-efficacy, and breast self-examination among Thai migrants in Brisbane. Journal of Advanced Nursing. 2002;41(3):241-9.

97. Fan Y, Ma N, Ma L, Xu W, Steven Lamberti J, Caine ED. A community-based peer support service for persons with severe mental illness in China. BMC Psychiatry. 2018;18(1):170-.

98. Tang TS, Funnell MM, Gillard M, Nwankwo R, Heisler M. Training peers to provide ongoing diabetes self-management support (DSMS): results from a pilot study. Patient Educ Couns. 2011;85(2):160-8.

99. Whittle J. When does peer support improve glycemic control in persons with diabetes mellitus? JAMA Intern Med. 2014;174(6):982-3.

100. Ghorob A, Vivas MM, De Vore D, Ngo V, Bodenheimer T, Chen E, et al. The effectiveness of peer health coaching in improving glycemic control among low-income patients with diabetes: protocol for a randomized controlled trial. BMC Public Health. 2011;11(208).

101. Hall CE, Hall AB, Kok G, Mallya J, Courtright P. A needs assessment of people living with diabetes and diabetic retinopathy. BMC Res Notes. 2016;9:56.

102. Baumann LC, Frederick N, Betty N, Jospehine E, Agatha N. A demonstration of peer support for Ugandan adults with type 2 diabetes. Int J Behav Med. 2015;22(3):374-83.

103. Fisher EB, Earp JA, Maman S, Zolotor A. Cross-cultural and international adaptation of peer support for diabetes management. Fam Pract. 2010;27 Suppl 1:i6-16.

104. Chau YC, Shiu ATY, Ma SF, Au TY. A nurse-led walking exercise program for Hong Kong Chinese diabetic patients: implications for facilitating self-efficacy beliefs. Journal of Clinical Nursing. 2005;14:1247-59.

105. Fottrell E, Ahmed N, Morrison J, Kuddus A, Shaha SK, King C, et al. Community groups or mobile phone messaging to prevent and control type 2 diabetes and intermediate hyperglycaemia in Bangladesh (DMagic): a cluster-randomised controlled trial. Lancet Diabetes Endocrinology. 2019;7:200-12.

106. Dale JR, Williams SM, Bowyer V. What is the effect of peer support on diabetes outcomes in adults? A systematic review. Diabetic Medicine. 2012;29(11):1361-77.

107. Gatlin TK, Serafika R, Johnson M. Systematic review of peer education intervention programmes among individuals with type 2 diabetes. Journal of Clinical Nursing. 2017;26(23-24):4212-22.

108. Mash R, Kroukamp R, Gaziano T, Levitt N. Cost-effectiveness of a diabetes group education program delivered by health promoters with a guiding style in underserved communities in Cape Town, South Africa. Patient Educ Couns. 2015;98(5):622-6.

109. Campbell C. Creating environments that support peer education: experiences from HIV/AIDS prevention in South Africa. Health Education 2004;104(4):197-200.

International Diabetes Federation. IDF Diabetes Atlas 7th Edition. 2015.
 Williams R, Airey M, Baxter H, Forrester J, Kennedy-Martin T, Girach A.
 Epidemiology of Diabetic Retinopathy and macula oedema: a systematic review. Eye.
 2004:18:963-83.

112. Klein BEK. Overview of epidemiologic studies of diabetic retinopathy. Ophthalmic Epidemiology. 2007;14(4):179-83.

113. Cheung Ning, Mitchell Paul, Wong Tien Yin. Diabetic retinopathy. The Lancet. 2010;376:124-36.

114. Fong DS, Aiello L, Gardner TW, King GL, Blankenship G, Cavallerano JD, et al. Retinopathy in Diabetes. Diabetes Care. 2004;27:S84-S7.

115. Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556-64.

116. Sabanayagam C, Wanfen Y, Ting D, Tan G, Wong TY. Ten emerging trends in the epidemiology of diabetic retinopathy. Ophthalmic Epidemiol. 2016;23(4).

117. Flaxman S, Bourne RRA, Resnikoff S, Ackland P, Braithwaite T, Cicinelli MV, et al. Global causes of blindness and distance vision impairment 1990 -2020: a systematic review and meta-analysis. The Lancet Global Health. 2017;5(12):e1221-e34.

118. Bastawrous A, Burgess Pl, Mahdi AM, Kyari F, Burton MJ, Kuper H. Posterior segment eye disease in sub-Saharan Africa: review of recent population-based studies. Tropical Medicine and International Health. 2014;19(5):600-9.

119. Charumathi Sabanayagam, Riswana Banu, Miao Li Chee, Ryan Lee, Ya Xing Wang, Gavin Tan, et al. Incidence and progression of diabetic retinopathy: a systematic review. Lancet Diabetes Endocrinology. 2018;7(7):P140-9.

120. Younis N, Broadbent DM, Vora JP, Harding SP. Incidence of sight-threatening retinopathy in patients with type 2 diabetes in the Liverpool Diabetic Eye Study: a cohort study. The Lancet. 2003;361(9353):195-200.

121. Burgess PI, Harding SP, García-Fiñana M, Beare NAV, Msukwa G, Allain TJ. First Prospective Cohort Study of Diabetic Retinopathy from Sub-Saharan Africa: High Incidence and Progression of Retinopathy and Relationship to Human Immunodeficiency Virus Infection. Ophthalmology. 2016;123(9):1919-25.

122. Burgess PI, Harding SP, García-Fiñana M, Beare NAV, Glover S, Cohen DB, et al. Incidence and progression of diabetic retinopathy in Sub-Saharan Africa: A five year cohort study. PloS one. 2017;12(8):e0181359.

123. Ahmed KR, Karim MN, Bhowmik B, Habib SH, Bukht MS, Ali L, et al. Incidence of diabetic retinopathy in Bangladesh: a 15-year follow-up study. Journal of Diabetes. 2012;4(4):386-91.

124. Burgess PI, Maccormick IJ, Harding SP, Bastawrous A, Beare NA, Garner P. Epidemiology of diabetic retinopathy and maculopathy in Africa: a systematic review. Diabetic Medicine. 2012:399-412.

125. Mengesha AY. Spectrum of eye disorders among diabetes mellitus patients in Gaberone, Botswana. Tropical Doctor. 2006;36(2):109-11.

126. Jingi AM, Nansseu JRN, Noubiap JJN, Bilong Y, Ellong A, Mvogo CE. Diabetes and visual impairment in sub-Saharan Africa: evidence from Cameroon. Journal of Diabetes & Metabolic Disorders. 2015;14(1):21.

127. Seyoum B, Mengistu Z. Retinopathy in patients of Tikur Anbessa Hospital diabetic clinic. Ethiopia Medical Journal. 2001;39:123-31.

128. Teshombe T, Melaku S. Pattern of retinal diseases at a teaching eye department, Addis Ababa. Ethiopia Medical Journal. 2004;42:185-93.

129. Gill GV, Gebrekidan A, English P, Wile D, Tesfaye S. Diabetic complications and glycaemic control in remote North Africa. Q J M. 2008;101:793-8.

130. Mwendwa FM, Otieno CF, Kayima JK, Amayo EO, Otieno PO. Risk factor profile and the occurence of microvascular complications in short-term Type 2 diabets mellitus at Kenyatta National Hospital, Nairobi. East African Medical Journal. 2005;82(12 Supple):5163-72.

131. Mwale C, Karimurio J, Njuguna M. Refractive errors in Type 2 diabetes. East African Medical Journal. 2007;84(6):259-63.

132. Glover SJ, Burgess PI, Cohen DB, Harding SP, Hofland HW, Zijlstra EE, et al. Prevalence of diabetic retinopathy, cataract and visual impairment in patients with diabetes in sub-Saharan Africa. Br J Ophthalmol. 2012;96(2):156-61.

133. Omolase C O, Adekanle O, Owoeye J F A, Omolase B O. Diabetic Retinopathy in a Nigerian community. Singapore Med J 2010;51 (1):56.

134. Onakpoya Oluwatoyin Helen, Adeoye Adenike Odunmorayo, Kolawole Babatope Ayodeji. Determinants of previous dilated eye examination among type II diabetics in Southwestern Nigeria. European Journal of Internal Medicine. 2010;21:176-9.

135. Rotchford AP, Rotchford KM. Diabetes in rural South Africa- an assessment of care and complications. South African Medical Journal. 2002;92(7):536-41.

136. Carmichael TR, Carp GI, Welsh ND, Kalk WJ. Effective and accurate screening for diabetic retinopathy using 60 degree mydriatic fundus camera. South African Medical Journal. 2005;95:57-61.

137. Mash B, Powell D, du Plessis F, van Vuuren U, Michalowska M, Levitt N. Screening for diabetic retinopathy in primary care with a mobile fundal camera – evaluation of a South African pilot project. South Africa Medical Journal. 2007;97(12):1281-8.

138. Reed O, Cook C. Retinopathy in diabetic patients evaluated at a primary care clinic in Cape Town. South African Medical Journal. 2007;97:941-4.

139. Mumba M, Hall A, Lewallen S. Compliance with Eye Screening Examinations among Diabetic Patients at a Tanzanian Referral Hospital. Ophthalmic Epidemiology. 2007;14(5):306-10.

140. Cleland Charles R, Burton Matthew J, Hall Claudette, Hall Anthony, Courtright Paul, Makupa William U, et al. Diabetic retinopathy in Tanzania: Prevalence and risk factors at entry into a regional screening programme. Tropical Medicine and International Health. 2016;21:417-26.

141. Lewis AD, Hogg RE, Chandran M, Musonda L, North L, Chakravarthy U, et al. Prevalence of diabetic retinopathy and visual impairment in patients with diabetes mellitus in Zambia through the implementation of a mobile diabetic retinopathy screening project in the Copperbelt province: a cross-sectional study. Eye (London, England). 2018;32(7):1201-8.

142. Burgess PI, Allain TJ, García-Fiñana M, Beare NAV, Msukwa G, Harding SP. High prevalence in Malawi of sight-threatening retinopathy and visual impairment caused by diabetes: identification of population-specific targets for intervention. Diabetic medicine : a journal of the British Diabetic Association. 2014;31(12):1643-50.

143. Wanjiku Mathenge, Andrew Bastawrous, Tunde Peto, Irene Leung, David Yorston, Allen Foster, et al. Prevalence and Correlates of Diabetic Retinopathy in a Population-based Survey of Older People in Nakuru, Kenya. Ophthalmic Epidemiology. 2014;21(3):169-77.

144. Mathenge W, Kuper H, Limburg H, Polack S, Onyango O, Nyaga G, et al. Rapid assessment of avoidable blindness in Nakuru district, Kenya. Ophthalmology. 2007;114(3):599-605.

145. Mathenge W, Bastawrous A, Foster A, Kuper H. The Nakuru posterior segment eye disease study: methods and prevalence of blindness and visual impairment in Nakuru, Kenya. Ophthalmology. 2012;119(10):2033-9.

146. Kuper H, Polack S, Limburg H. Rapid Assessment of Avoidable Blindness. Community Eye Health Journal. 2006;19(60):68-9.

147. Lin S, Ramulu P, Lamoureux EL, Sabanayagam C. Addressing risk factors, screening, and preventative treatment for diabetic retinopathy in developing countries: a review. Clin Exp Ophthalmol. 2016;October 2015:300-20.

148. Zhou Y, Wang C, Shi K, Yin X. Relationship between dyslipidemia and diabetic retinopathy: A systematic review and meta-analysis. Medicine. 2018;97(36):e12283.
149. Poore S, Foster A, Zondervan M, Blanchet K. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. Int J Health Policy Manag.
2015;4(1):19-28.

150. Kupitz DG, Fenwick E, Kollmann M, Holz FG, Finger RP. Diabetes and Diabetic Retinopathy Management in East Africa: Knowledge, Attitudes, and Practices of Hospital Staff in Kenya. Asia-Pacific Journal of Ophthalmology. 2014;3(5):271-6.

151. Ferris FL. How Effective Are Treatments for Diabetic Retinopathy? JAMA. 1993;269(10):1290-129.

152. American Diabetes Association. Standards of Medical Care in Diabetes. The Journal of Clinical Applied Research and Education. 2016;39(Supple 1):S1-S112.

153. Mwavua SM, Ndungu EK, Mutai KK, Joshi MD. A comparative study of the quality of care and glycemic control among ambulatory type 2 diabetes mellitus clients, at a Tertiary Referral Hospital and a Regional Hospital in Central Kenya. BMC Res Notes. 2016;9:12.

154. Chan JCN, Gagliardino JJ, Baik SH, et al. Multifaceted determinants for achieving glycemic control: the International Diabetes Management Practice Study (IDMPS). Diabetes Care. 2009;32:227–33.

155. Lee CC, Stolk RP, Adler AI, Patel A, Chalmers J, Neal B, et al. Association between alcohol consumption and diabetic retinopathy and visual acuity—the AdRem Study. Diabetic Medicine. 2010;27(10):1130-7.

156. Zhu W, Meng Y-F, Wu Y, Xu M, Lu J. Association of alcohol intake with risk of diabetic retinopathy: a meta-analysis of observational studies. Scientific Reports. 2017;7(1):4.

157. Klein R, Knudtson MD, Lee KE, Gangnon R, Klein BE. The Wisconsin
Epidemiologic Study of Diabetic Retinopathy: XXII the twenty-five-year progression of
retinopathy in persons with type 1 diabetes. Ophthalmology. 2008;115(11):1859-68.
158. Meta-Analysis for Eye Disease (META-EYE) Study Group. Global Prevalence

and Major Risk Factors of Diabetic Retinopathy. Diabetes care. 2012;35(3):556-64.

159. Sivaprasad S, Gupta B, Gulliford MC, Dodhia H, Mohamed M, Nagi D, et al. Ethnic variations in the prevalence of diabetic retinopathy in people with diabetes attending screening in the United Kingdom (DRIVE UK). PloS one. 2012;7(3):e32182.

160. Kostev K, Rathmann W. Diabetic retinopathy at diagnosis of type 2 diabetes in the UK: a database analysis. Diabetologia. 2013;56(1):109-11.

161. Zhang X, Saaddine JB, Chou C-F, Cotch MF, Cheng YJ, Geiss LS, et al. Prevalence of diabetic retinopathy in the United States, 2005-2008. JAMA. 2010;304(6):649-56.

162. Sivaprasad S, Gupta B, Crosby-Nwaobi R, Evans J. Prevalence of Diabetic Retinopathy in Various Ethnic Groups: A Worldwide Perspective. Survey of Ophthalmology. 2012;57(4):347-70.

163. Ozawa GY, Bearse MA, Adams AJ. Male–Female Differences in Diabetic Retinopathy? Current Eye Research. 2015;40(2):234-46.

164. Lee R, Wong TY, Sabanayagam C. Epidemiology of diabetic retinopathy, diabetic macular edema and related vision loss. Eye Vis (Lond). 2015;2:17-.

165. UK Prospective Diabetes Study (UKPDS) Group*. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). The Lancet. 1998;352(9131):837-53.

166. The ACCORD Study Group and ACCORD Eye Study Group*. Effects of Medical Therapies on Retinopathy Progression in Type 2 Diabetes. The New England Journal of Medicine. 2010;363(3):233-44.

167. Cai X, Chen Y, Yang W, Gao X, Han X, Ji L. The association of smoking and risk of diabetic retinopathy in patients with type 1 and type 2 diabetes: a meta-analysis. Endocrine. 2018;62(2):299-306.

168. Giloyan A, Harutyunyan T, Petrosyan V. The prevalence of and major risk factors associated with diabetic retinopathy in Gegharkunik province of Armenia: cross-sectional study. BMC Ophthalmol. 2015;15:46.

169. Lindenmeyer A, Sturt JA, Hipwell A, Stratton IM, Al-Athamneh N, Gadsby R, et al. Influence of primary care practices on patients' uptake of diabetic retinopathy screening: a qualitative case study. Br J Gen Pract. 2014;64(625):e484-92.

170. Zheng Yingfeng, He Mingguang, Congdon Nathan. The worldwide epidemic of diabetic retinopathy. Indian Journal of Ophthalmology. 2012;60(5):428-31.

171. Cavan D. The diabetes epidemic and its implications for eye health. Community Eye Health. 2015;28(92):61-3.

172. Scanlon PH. Diabetic Retinopathy Medicine. 2019;47(2):1-4.

173. Lamoureux EL, Tai ES, Thumboo J, Kawasaki R, Saw SM, Mitchell P, et al. Impact of diabetic retinopathy on vision-specific function. Ophthalmology. 2010;117(4):757-65.

174. Gupta P, Aravindhan A, Gan A, Man R, Fenwick, Mitchell P, et al. JAMA Ophthalmology. 2017;135(12):1410-6.

175. Fenwick E, Rees G, Pesudovs K, Dirani M, Kawasaki R, Wong TY, et al. Social and emotional impact of diabetic retinopathy: a review. Clin Exp Ophthalmol. 2012;40(1):27-38.

176. Khoo K, Man REK, Rees G, Gupta P, Lamoureux E, Fenwick E. The relationship between diabetic retinopathy and psychosocial functioning: a systematic review. Quality of Life Research. 2019;28(8):2017-39.

177. Chireh B, Li M, D'Arcy C. Diabetes increases the risk of depression: a systematic review, meta-analysis and estimates of population attributable fractions based on prospective studies. Preventive Medicine Reports. 2019;10(14):100822.

178. Rees G, Xie J, Fenwick EK, Sturrock BA, Finger R, Rogers SL, et al. Association Between Diabetes-Related Eye Complications and Symptoms of Anxiety and Depression. JAMA Ophthalmol. 2016;134(9):1007-14.

179. Köberlein J, Beifus K, Schaffert C, Finger RP. The economic burden of visual impairment and blindness: a systematic review. BMJ Open. 2013;3(11):e003471.
180. Chu Y, Jacobs P, Johnson JA. Productivity Losses Associated With Diabetes in

the U.S. Diabetes Care. 2001;24(2):257-61.

181. Cavan D. The diabetes epidermic and its implications for eye health. Community Eye Health Journal. 2015;28(92):61-3.

182. Foster A. Planning a programme to prevent visual loss from diabetic retinopathy. Community Eye Health Journal. 2015;28(92):s01-s5.

183. Matuszewski W, Bandurska-Stankiewicz E, Modzelewski R, Kamińska U, Stefanowicz-Rutkowska M. Diagnosis and treatment of diabetic retinopathy — historical overview. Clinical Diabetology 2017. 2017;6(5):182-8.

184. Wilson JMG, Jungner G. Principles and practice of screening for disease. Geneva: World Health Organization; 1968.

185. Pieczynski J, Grzybowsk A. Review of Diabetic Retinopathy Screening Methods and Programmes Adopted in Different Parts of the World European Ophthalmic Review. 2015;9(1):49-55.

186. Piyasena MMPN, Murthy GVS, Yip JLY, Gilbert C, Peto T, Gordon I, et al. Systematic review and meta-analysis of diagnostic accuracy of detection of any level of diabetic retinopathy using digital retinal imaging. Systematic Reviews. 2018;7(1):182.

187. Andermann A, Blancquaert I, Beaucham, Déry V. Revisiting Wilson and Jungner in the genomic age: a review of screening criteria over the past 40 years. Bull World Health Organ. 2008;86(4):241-320.

188. University of Winsconsin. The Winsconsin Epidemiologic Study of Diabetic Retinopathy (WESDR) 2015 [Available from: <u>www.epi.ophth.wisc.edu</u>.

189. Scanlon PH, Dirani M, van Wijngaarden P. Screening for sight-threatening diabetic retinopathy: An update. Egyptian Retina Journal. 2014;2(1):3-18.

190. International Council for Ophthalmology. ICO Guidelines for Diabetic Eye Care February 2014. 2014.

191. Swanson M. Retinopathy screening in individuals with type 2 diabetes: who, how, how often, and at what cost--an epidemiologic review. Optometry. 2005;76(11):636-46.

192. Lewis K. Improving patient compliance with diabetic retinopathy screening and treatment. Community Eye Health. 2015;28(92):68-9.

193. Cavan D, Makaroff L, da Rocha Fernandes J, Sylvanowicz M, Ackland P, Conlon J, et al. The Diabetic Retinopathy Barometer Study: Global perspectives on access to and experiences of diabetic retinopathy screening and treatment. Diabetes Res Clin Pract 2017;129:16-24.

194. Piyasena MMPN, Murthy GVS, Yip JLY, Gilbert C, Zuurmond M, Peto T, et al. Systematic review on barriers and enablers for access to diabetic retinopathy screening services in different income settings. PLOS ONE. 2019;14(4):e0198979.

195. International Council of Ophthalmology. ICO Guidelines for Diabetic Eye Careupdated 2017. San Francisco, California: International Council of Ophthalmology; 2017.

196. Stewart MW. Treatment of diabetic retinoapthy: recent advances and unresolved challenges. World Journal of Diabetes. 2016;7(16):333-41.

197. Mapa Mudiyanselage Prabhath Nishantha Piyasena, Gudlavalleti Venkata S. Murthy, Jennifer L. Y. Yip, Clare Gilbert, Maria Zuurmond, Tunde Peto, et al.

Systematic review on barriers and enablers for access to diabetic retinopathy screening services in different income settings. PLoS ONE. 2019;14(4):e0198979.

198. Scanlon PH, Aldington SJ, Leal J, Luengo-Fernandez R, Oke J, Sivaprasad S, et al. Development of a cost-effectiveness model for optimisation of the screening interval in diabetic retinopathy screening. Health Technology Assessment. 2015;19(74):1-116.

199. Mead A, Burnett S, Davey C. Diabetic retinal screening in the UK. Journal of the Royal Society of Medicine. 2001;94(3):127-9.

200. Benbassat J, Polak C, J. J. Objectives of teaching direct ophthalmoscopy to medical students. Acta Ophthalmol. 2012;90:503-7.

201. Raman R, Srinivasan S, Roy R. Screening practices for diabetic retinopathy. Expert Review of Ophthalmology. 2015;10(6):519-21.

202. Nanda D, Sarkar M, Sahu V, Chandrakar AK, Garg M, Pandey N, et al. Comparison between slit-lamp biomicroscopy and fluorescein angiography in diagnosing diabetic retinopathy. Scholars Journal of Applied Medical Sciences 2017;5(1B):108-11.

203. Bragge P, Gruen RL, Chau M. Screening for presence or absence of diabetic retinopathy: a meta-analysis. Arch Ophthalmol. 2011;129.

204. Scanlon PH. The English National Screening Programme for diabetic retinopathy 2003–2016. Acta Diabetologica. 2017;54(6):515-25.

205. Shi L, Wu H, Dong J, Jiang K, Lu X, Shi J. Telemedicine for detecting diabetic retinopathy: a systematic review and meta-analysis. British Journal of Ophthalmology. 2015;99:823–31.

206. Kurji K, Kiage D, Rudnisky CJ, Damji KF. Improving diabetic retinopathy screening in Africa: patient satisfaction with teleophthalmology versus ophthalmologist-based screening Middle East African Journal of Ophthalmology. 2013;20:56-60.

207. Norgaard MF, Grauslund J. Automated Screening for Diabetic Retinopathy- A systematic Review. Ophthalmic Research. 2018;60(1):9-7.

208. Hansen MB, Abràmoff MD, Folk JC, Mathenge W, Bastawrous A, Peto T. Results of Automated Retinal Image Analysis for Detection of Diabetic Retinopathy from the Nakuru Study, Kenya. PloS one. 2015;10(10):e0139148-e.

209. Bellemo V, Lim SW, Lim G, et al. Artificial intelligence using deep learning to screen for referable and vision-threatening diabetic retinopathy in Africa: a validation study. Lancet Digital Health. 2019; ;1:e35-e44.

210. Bilong Y, Katte JC, Koki G, Kagmeni G, Obama OPN, Fofe HRN, et al. Vaidation of smartphone-based retinal photography for Diabetic Retinopathy Screening. Ophthalmic Surgery, Lasers and Imaging Retina. 2019;50(5):S18-S22.

211. Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. Diabetic Medicine. 2010;27(3):249-56.

212. Royle P, Mistry H, Auguste P, Shyangdan D, Freeman K, Lois N, et al. Panretinal photocoagulation and other forms of laser treatment and drug therapies for non-proliferative diabetic retinopathy: systematic review and economic evaluation. Health Technol Assess. 2015;19(51).

213. Vashist P, Singh S, Gupta M, Saxena R. Role of Early Screening for Diabetic Retinopathy in Patients with Diabetes Mellitus: An Overview. Indian Journal of Community Medicine. 2011;36(4):247-54.

214. Day TE, Ravi N, Xian H, Brugh A. Sensitivity of diabetic retinopathy associated vision loss to screening interval in an agent-based/discrete event simulation model. . Comput Biol Med. 2014;47:7-12.

215. Scanlon P. Diabetic Retinopathy Screening: Progress or Lack of Progress. 2012:17-29.

216. Scanlon PH. Screening Intervals for Diabetic Retinopathy and Implications for Care. Current Diabetes Reports. 2017;17(10):96.

217. Klein R. Screening interval for retinopathy in type 2 diabetes. Lancet. 2003;361(9353):190–1.

218. Royle P, Mistry H, Auguste P, Shyangdan D, Freeman K, Lois N, et al. Panretinal photocoagulation and other forms of laser treatment and drug therapies for non-proliferative diabetic retinopathy: systematic review and economic evaluation. Health Technol Assess. 2015;19(51):v-xxviii, 1-247.

219. Diabetic Retinopathy Study Research Group. Photocoagulation treatment of proliferative diabetic retinopathy; Clinicla application of Diabetic Retinopathy Study (DRS) finidnfs, DRD report no 8. Ophthalmology. 1981;88(7):583-600.

220. Resnikoff S, Lansingh VC, Washburn L, Felch W, Gauthier T-M, Taylor HR, et al. Estimated number of ophthalmologists worldwide (International Council of Ophthalmology update): will we meet the needs? British Journal of Ophthalmology. 2019:314336.

221. Fulton BD, Scheffler RM, Sparkes SP, Auh EY, Vujicic M, Soucat A. Health workforce skill mix and task shifting in low income countries: a review of recent evidence. Hum Resour Health. 2011;9:1.

222. American Diabetes Association. Standards of Medical Care in Diabetes-2017. Diabetes Care. 2017;40(Suppl 1).

223. Klein B, Klein R. Diabetic Retinopathy. In: Johnson G, Minnasian D, Weale R, West SK, editors. The Epidemiology of Eye Disease. 3rd Edition ed. London: Imperial College Press; 2012.

224. Early Treatment Diabetic Retinopathy Study Research Group. Grading diabetic retinopathy from stereoscopic color fundus photographs--an extension of the modified Airlie House classification. ETDRS report number 10. Ophthalmology. 1991;98:786-806.

225. Evans J R, Michelessi M, Virgili G. Laser photocoagulation for proliferative diabetic retinopathy. The Cochrane database of systematic reviews. 2014(11):CD011234.

226. Virgili G, Parravano M, Menchini F, Evans JR. Anti-vascular endothelial growth factor for diabetic macular oedema. Cochrane Database of Systematic Reviews. 2014;24(10):CD007419.

227. Grover DA, Li T, Chong CCW. Intravitreal steroids for macular edema in diabetes. Cochrane Database of Systematic Reviews. 2008;1(1):CD005656.

228. Diabetic Retinopathy Network. DR-NET Toolkit London: DR-NET; 2016 [Available from: <u>https://sites.google.com/site/drnetcomm/resources</u>.

229. World Health Organization. NCD Global Monitoring Framework: Ensuring progress on non-communicable diseases in countries. Geneva: WHO; 2011.

230. Commonwealth Eye Health Consortium. Building a network to combat diabetic retinopathy across the Commonwealth London: CEHC; 2018 [Available from: https://cehc.lshtm.ac.uk/dr-links/drnet/.

231. Ramke J, Zwi AB, Silva JC, Mwangi N, Rono H, Gichangi M, et al. Evidence for national universal eye health plans. Bull World Health Organ. 2018;96(10):695-704.
232. World Health Organization. The World Health Report 2000: improving performance. Geneva: WHO; 2000.

233. World Health Organization. Everybody's business: Strengthening health systems to improve health outcomes: WHO's framework for action. Geneva: WHO; 2007.

234. Wanjiku Mathenge. Artificial intelligence for diabetic retinopathy screening in Africa. Lancet Digital Health. 2019;1:e6-e7.

Section B Health System Assessment

'An epidemic is the wrong time to discover our preparations have not been sufficient, or that we have overlooked crucial components of the response.'

> Dr Tedros Adhanom Ghebreyesus, 2018 Director General, World Health Organization

'The pathway to UHC is not something you reach by investing in something called UHC. You make that pathway by building parts of the health systems, addressing the health care needs of a specific population and by addressing specific conditions. It will inevitably involve a mix of focusing on specific segments of populations, specific diseases and functional components of the system.'

> Peter Sands, Executive Director, 2019 The Global Fund to Fight AIDS, Tuberculosis and Malaria

Chapter 3: Health system assessment for diabetes and diabetic retinopathy

3.0 Overview

Health systems play an integral role in the response to the diabetes and DR epidemic, as described in chapter two. The concept of health system assessment (HSA) has gained significant emphasis in research and policy, because it can identify important areas of action to influence the achievement of health outcomes.¹ Chapter three reports on the health system assessment for diabetes and diabetic retinopathy in Kenya. This includes a summary of the purpose and research questions (3.1), the conceptual framework (3.2), an overview of theoretical frameworks for HSA (3.3), adaptation of a framework for this study (3.4), and a summary of study methods (3.5). Section 3.6 outlines the results, culminating in the identification of priority interventions. A detailed description of study methods and additional results are provided in *Research Papers 1 and 2* in the next chapter.

3.1 Purpose of the health system assessment

Building on the WHO definition of a health system,² the system for diabetes and DR can be defined as 'all resources, institutions and activities whose purpose is to promote, restore or maintain health for PLWD, including eye health'. The assessment of the strengths and weaknesses of the system in its ability to respond to the rising burden of these conditions is a lever that countries can use to strengthen the health system.^{1, 3} There is broad commitment at the global level to strengthen health systems as an approach to addressing emerging health needs,⁴ but there is no consensus on the optimal combination of strategies to improve the response of health systems to diabetes and DR. Examples of interventions that have been applied to ensure that effective diabetes and DR services are available include diabetes registers, task shifting, telemedicine, mobile screening clinics, integrated models of care, photographic retinal screening and artificial intelligence.⁵⁻⁷ However, it is uncertain whether any of those interventions would be feasible and appropriate in the context of Kenya. Therefore, there is a need to identify the local needs and resources, suitable interventions to address the needs, and suitable methods to test these interventions. This HSA is designed to identify the gaps and strengths in the health system, as a first step towards addressing the gaps.

A general HSA,⁸ and an Eye Health System Assessment (EHSA) were conducted in 2010 and 2017 respectively.⁹ Both HSAs showed that the health system is highly pluralistic, with formal and informal public and private sectors. However, the two HSAs did not provide evidence of an interaction between the two systems, or the specific issues relating to diabetes and DR services. Given that these conditions are a priority, the need for such evidence to inform the next eye health plan for Kenya has been articulated.¹⁰ This chapter describes a HSA for diabetes and DR, which sought to answer the following research questions:

- 1. What are the strengths and weaknesses in the health system for diabetes and diabetic retinopathy?
- 2. What priority intervention(s) can address the key gaps in the services for diabetic retinopathy?

3.2 Conceptual Framework of the health system

The conceptual framework of the health system for diabetes and DR, Figure 3-1, corresponds to the typical care pathway that a PLWD in Kenya would follow. In the *community*, there are health promotion activities and screening camps for diabetes.

Persons who do not access screening in the community are encouraged to visit a health facility for testing. Those diagnosed with diabetes are linked to the nearest *diabetes clinic* for metabolic control, follow-up and screening for complications. Assessment for metabolic control, through HbA1c, is a key service indicator for all PLWD. Diabetes services should also refer all PLWD for screening for DR, which is available at the *eye clinics*. Attendance at DR screening is another key service indicator. The eye clinic also provides treatment for VTDR. Strengths and weaknesses can exist at any or all of the three levels (the community, diabetes clinic or eye clinic), which has impact on the burden of visual impairment from DR.



Figure 3-1: Conceptual Framework for health system assessment

3.3 Health system frameworks

The purpose of a HSA is to take stock of a specific health system.¹ There are several frameworks for health system assessment in the literature. Hoffman and colleagues identified 41 different health systems frameworks.¹¹ The volume and diversity of existing frameworks highlights several things. First, the variation in the way different people or institutions conceptualize health systems. Second, the high interest around the concept of health systems. Third, the difficulty in capturing all the attributes of health systems in one framework, since health systems are complex. Fourth, the evolution of the understanding of health systems over time. Surprisingly, although many frameworks have been described in the literature, examples of how they have been applied in health system assessments are relatively few, with some of the frameworks being dominant.¹²

In 2007 the WHO described the health systems building blocks framework, which is ubiquitous and highly influential in the literature.¹³ In this framework, the health system is composed of six interacting building blocks, and each of them is relevant to diabetes and DR: – a) service delivery, b) health workers, c) health information, d) medical products, technology and vaccines, e) health finance, and f) governance and leadership. The framework has been used in studies relating to DR, vaccines, measles, polio, HIV and malaria, although it was originally designed for resource mobilization, not for research.^{4, 14}The widespread use of the framework for assessing health systems fosters a shared understanding among researchers and the users of research evidence.¹⁵ Kenya has adopted this framework, and in 2010 used it for a general HSA.⁸ Subsequently Kenya added two building blocks to the framework (infrastructure and research), to make eight blocks, Figure 3-2.¹⁶

The limitations of using the building blocks framework in research have previously been described.^{4, 17} In particular, it does not capture the experiences of PLWD along the care pathway, a parameter of interest in this study. Additionally, there is need to capture evidence on the strengths and weaknesses on both the supply-side and the demandside of the health system. As the framework is predominantly a supply-side model ¹⁷ it is necessary to complement it with other approaches that take into account the demandside.

In 2015 the WHO developed a specific tool for the assessment of diabetic retinopathy and diabetes management systems (TADDS).¹⁸ This tool is based on the WHO building blocks framework. Its stated focus is to assess the collaboration between diabetes and DR services. It has seven elements a) service delivery, b) health workers, c) health information, d) health technology, e) health finance, and f) policies, priorities and programs g) health promotion for diabetes and DR. It has the advantage of focusing the HSA on the condition of interest, but it narrows the scope of the "leadership and governance" building block to 'policies, priorities and program' and the medicines and other products building block to health technology. At the time of this study, this tool was new and had not yet applied in practice. Figure 3-2 compares the building blocks in these frameworks.

WHO health systems building blocks	Health systems building blocks (Kenya's modification)	WHO TADDS tool key sections
Leadership and Governance	Leadership and Governance	Priorities, policies and programs
Service delivery	Service delivery	Service delivery
Health workforce	Health workforce	Health workforce
Medical products and technologies	Medical products and technologies	Health technology
Health information management system	Health information management system	Health information management system
Health financing	Health financing	Health financing
	Infrastructure	Health promotion for diabetes and DR
	Research	

Figure 3-2: Comparison of building block focused frameworks

Another approach to HSA is the use of tracer conditions. Kessner and colleagues in 1973 were the first persons to describe the use of tracer conditions to assess health systems.¹⁹ A tracer condition is a carefully selected health problem that makes it possible to capture differences in the performance of a health system.²⁰ There are six criteria for a tracer condition.^{21, 22} Table 3-1 shows how the criteria apply to diabetes and DR. The combination of the two conditions is suitable as a tracer for the responsiveness of a health system to chronic diseases in general and NCDs in particular, for three reasons. First, they occur in association with other chronic comorbidities, such as hypertension.^{23, 24} Second, for people with NCDs and other chronic diseases, preventing complications is an important aim of the health system.²⁵ As DR is a specific complication of diabetes, the two conditions can mirror the subtleties of the

progression of disease. Thirdly, chronic diseases require an integrated health system,²⁵ which can be exemplified by the level of integration between services for diabetes and DR. A HSA using this approach therefore provides useful evidence for the health system for chronic diseases, including NCDs. Diabetes has been used as a tracer condition for health systems in several studies.^{20, 22, 26-28} This is the first study to consider the combination of diabetes and DR as a tracer condition.

Criteria	How diabetes and DR fit the criteria
Disease has a known epidemiology	The epidemiology of both conditions globally, in SSA and in Kenya has been described in chapter 2
Disease is well defined and easy to diagnose	The definition and criteria for diagnosis are provided in chapter 1 (diabetes) and chapter 2 (DR)
Its prevalence in the population is large enough to enable adequate data to be collected	Both population-based and clinic-based surveys have shown that that the prevalence is sufficient to enable collection of data that can be used for planning (chapter 2)
Its natural history is known, and it varies with the utilization and effectiveness of health care	The predictors of the development of complications in both conditions, which include metabolic control, have been described in chapter 2
It requires specific treatment, in the absence of which functional impairment results	Hypoglycaemic drugs and lifestyle measures are required for glycaemic control, and the treatments for DR have been described in Chapter 2, without which visual impairment results
Available and well-defined techniques of medical management exist for at least one of the following: prevention, diagnosis, treatment or rehabilitation	Prevention, diagnosis and treatment apply to both diabetes and DR. Rehabilitation is provided for those who develop severe visual impairment and blindness

Table 3-1: The suitability of diabetes and DR as tracer conditions

Nolte and colleagues assessed the performance of health systems using diabetes type 1

as a tracer condition in 29 countries.²⁰ Using routine data, they demonstrated a

remarkable variation in case-fatality across these countries, suggesting differences in
the ability of the various health systems to respond to diabetes. In this study, it is envisaged that using diabetes as a tracer will provide a broad understanding of the health system response to the needs identified along the care continuum. We did not use mortality as an indicator due to lack of suitable registers for diabetes and death in Kenya.

Diabetes type 1 has also been used as a tracer condition to investigate the capacity of the health system in Mozambique, Zambia and Mali, using a rapid assessment protocol for insulin access (RAPIA).^{20, 29} The RAPIA method is a rapid and low cost method of situation analysis, centred on access to insulin. The RAPIA study was conducted in three areas; the capital city, a large urban centre and a rural area, to provide a picture of the national health system. It involved studying the path of insulin from its arrival in the country to the point of the end-user, including identifying determinants along the pathway. In Kenya a RAPIA study to assess insulin access in two counties of Kenya (Meru and Trans-Nzoia) found that a public-private partnership (the *Base of the pyramid* project) helped to stabilize the cost of insulin but was not sufficient to improve access to diabetes care.²⁷

3.4 Adaptation of the WHO health system framework for this study

Each of the frameworks provides conceptual contributions and a general direction for health system assessment. However they do not serve all purposes and require adaptation for use in different settings.^{4, 30} This study is anchored on the WHO framework as the main assessment framework, since all the building blocks are relevant to diabetes and DR. Since the study is in Kenya, the eight building blocks model was considered appropriate for facilitating shared understanding with Kenyan policy-makers and for future comparison with other local studies. The tracer approach, used in combination, helped to illuminate how the various building blocks of the health system are responsible for health system performance along the care pathway. This combination has the benefit of focusing the study on issues pertinent to diabetes and DR whilst retaining the benefits of a common framework. It is important to note, for replication of this approach in other studies, that familiarity with the typical patient pathway is a key prerequisite for adapting the framework for this purpose.

3.5 A summary of study methods

Bennett and Peters have proposed a set of three key principles to guide HSAs: relevance, coherence and trustworthiness.¹ These principles were operationalized by consulting local stakeholders and obtaining consensus on: the purpose of HSA and the choice of the appropriate study counties (relevance), the fitness of the HSA framework for the purpose (coherence), and the choice of study procedures (trustworthiness). The stakeholders included the NCD division and the Ophthalmic Services Unit at the Ministry of Health, the national DR working group and the national eye care coordinating committee.

Three counties were purposively selected for the study, based on geographic context (rural/urban), location within the diabetes belt (Figure 3-3) and other unique characteristics highlighted below. The 'diabetes belt' is an area that stretches from the coast to the Lake Victoria region, corresponding to the area along the railway (marked red). However, there are also additional pockets outside this belt that also have high prevalence (also marked red). It was named the 'diabetes belt' based on a historical perception of high prevalence as reported from some screening camps in this region. Nairobi (population 3.6 million, 100% urban) is a cosmopolitan city with a unique social structure compared to the rest of the country (high population, heterogeneous social strata including the affluent, the middle class and the urban poor) and a unique health system structure (a large number of both private and public health facilities, including primary, secondary and tertiary centres). Nakuru (population 1.8 million, 46% urban) has been the focus of previous DR prevalence and incidence studies reported in chapter two, and it is of interest to relate these research efforts to the national health system.³¹ Kirinyaga (population 0.6 million, 16% urban) is considered to have a relatively high prevalence of diabetes compared to other counties within the 'diabetes belt'. Through simple random sampling from a sampling frame of clinics, three diabetes clinics were identified in each county (public, private-for-profit and private-not-for-profit facilities), giving nine diabetes clinics. Eye clinics were also identified using the same procedure. Data for this mixed-methods analytical cross-sectional study was collected through:

a) Patient interviews at the diabetes clinic (n=270)

b) Interviews with service providers for diabetes (n=18) and eye care (n=9)
c) Interviews with key informants who were familiar with the organization and delivery of healthcare at the national or at county level, with particular reference to diabetes and eye care (n=18)

d) Review of documents provided by key informants

The data collection tools were designed to collect information on the eight building blocks, with reference to the community, diabetes and eye clinic settings. This data was then triangulated. Research papers 1 and 2 provide additional details on the methods. The London School of Hygiene and Tropical Medicine (LSHTM) and the African Medical Research Foundation (AMREF) granted ethics approval.



Figure 3-3: Study counties and the diabetes belt

3.6 Results

Based on the data from all the interviews and the document review, the findings for each building block are discussed concisely below. Additional results are presented in the research papers in chapter 4.

Service delivery

Community

Strengths: Local radio and television programs have health programs that include diabetes education for the public. This reflects recognition that before people become patients, they need to be empowered to protect their health.³² Testing for diabetes is available at outreach screening camps. Community health volunteers (CHVs) link those diagnosed with diabetes to primary care clinics and diabetes support groups (DSGs) in the community. These DSGs and CHVs are important examples of linkage between patients and community resources, one of the elements of the chronic care model (CCM).^{25, 33} At the DSGs, PLWD receive self-management education, glucose-monitoring services and peer support. DSGs value disclosure and PLWD discuss their illness and treatment freely. There is no stigma associated with diabetes in adults in the community and in the DSGs. This is important because it influences the capacity of PLWD to take on self-care tasks, including attending appointments and adhering to treatment.³⁴ Weakness: Public education through mass media is inadequate and mainly concentrated around advocacy events such as the World Diabetes Day, World Health Day and World Sight Day. This portrays a lack of emphasis on prevention of both diabetes and DR. A high proportion of adults have not been tested for diabetes, therefore many PLWD remain undiagnosed. The first point of care for the community is local clinics and pharmacies, many of which do not offer routine blood glucose testing, causing delay in

diagnosis. Children with Type 1 diabetes often avoid disclosure of their condition to the school community for fear of stigmatization. This may reduce their capacity to enact self-care within the school context, and hence undermine health outcomes.³⁵ Similarly, formal and informal places of employment do not offer any type of diabetes services, and employees may not disclose their illness to the employer since they do not perceive this as beneficial. Work sites and schools therefore require strategic engagement.³⁶ Although DSGs exist in the three counties, they are not uniformly active. For example, DSGs are very active in Kirinyaga and Nairobi, and less active in Nakuru, which is attributed to variation in DSG leadership capacity at county level.

Diabetes clinic

Strengths: There are dedicated diabetes clinics in all the counties, where PLWD are booked to attend appointments for follow-up. All the clinics in Kirinyaga and Nakuru run twice a week. In Nairobi, some clinics run five days a week but most run two days. On the day of appointment, PLWD are seen on a 'first in first out' basis. PLWD are free to switch clinics or use multiple clinics. This aspect of autonomy is valued by PLWD, but its effect on quality of care in LMICs has not been investigated. Diabetes clinics work closely with DSGs, which ensures that PLWD in the community are linked with diabetes clinics. DSGs also mobilize communities for screening camps organized by diabetic clinics.

Weaknesses: Most of the diabetes clinics run only two days a week, rather than the five working days. This results in a high volume of patients and long waiting queues at the clinics. The result is short consultation interactions that focus on the immediate complaint, and comorbidities such as depression and hypertension may be missed. This differs from the productive patient-provider interactions envisaged in the CCM.³⁷

113

Outpatient and inpatient services are delinked, such that it is not possible to track the frequency of hospitalizations of PLWD attending the clinic. Other supply side barriers to services are: inefficiencies such loss of patient files, lack of facilities for HbA1c testing, stock out of insulin, oral drugs and glucose test strips. PLWD receive care from multiple diabetes clinics which is likely to deter continuity of care.^{5, 38}

Demand side barriers include long distances to diabetes clinics, with its associated transport costs and travel inconveniences. Some of the PLWD have to connect two or three routes to reach the health facilities, and they are typically required to arrive by 8am in order to be attended that day. The cost of tests and medication, the opportunity cost of waiting time at the clinic and lack of permission from employers to attend clinic appointments are additional problems, similar to the experience in other settings.³⁸

Barriers to accessing HbA1c services

Only 27% of PLWD have had an HbA1c test in the preceding 12 months, and only 40% have ever had the test, with Kirinyaga county having the lowest indices, Figure 3-4. HbA1c testing is often not available in diabetes clinics due to malfunctioning equipment or lack of reagents. Therefore PLWD have to get HbA1c done privately, incurring high costs. In addition, health workers often use fasting blood sugar to monitor glycaemic control, rather than send PLWD for HbA1c test outside the health facility. Considering that HbA1c measures the glycaemic control over three months, is a proxy for the quality of care and a predictor for complications, it is a more suitable test for chronic care.³⁹⁻⁴¹ This reflects that resource challenges in the health system are hindering adoption of chronic care.⁴²



Figure 3-4: Access to HbA1c testing by county in 270 PLWD (90 in each county)

Barriers to accessing DR screening services among PLWD attending diabetes clinics

DR screening rates are low among PLWD attending diabetes clinics in all counties, with the lowest indices being in Kirinyaga, Figure 3-5 and 3-6. The predictors of DR screening are discussed in research paper 1. Supply side barriers include failure to refer PLWD for screening.⁴³ Service providers attribute this to forgetting, heavy clinic workload, diabetes being a complex disease that requires too many tests, more pressing concerns such as renal failure, lack of guidelines on DR screening and lack of point of use materials such as checklists to ensure screening is not forgotten. Further, they cite the lack of coordinated links with eye care services. The diabetes and eye clinics are located in different buildings in most of the facilities and sometimes PLWD who are referred for DR screening are "lost" along the facility corridors, and do not arrive at the eye clinic. Diabetes clinics do not have a formal way of following up PLWD to monitor uptake of referral or to obtain feedback from eye care services. Demand side barriers include low levels of awareness of PLWD on the need for DR screening and failing to take up the appointment, especially if they did not have symptoms.⁴³ This delay or refusal to take up screening services even when they are available has been noted in the literature on diabetes care in LMICs.³⁸ It points to a level of screening hesitancy (delay in acceptance or refusal of a service despite its availability), mostly associated with lack of awareness, low perceived risk of DR or perceived inconveniences of screening services or undervaluing of the services. This suggests a need for PLWD education about DR and its treatment as well as research on effective strategies for risk communication.



Figure 3-5: Access to DR screening by county among 270 PLWD (90 per county)



Figure 3-6: DR screening rate in different categories of diabetes clinics

Eye clinic

Strengths: There are written information brochures on DR, which are provided for PLWD and general health care providers. Opportunistic screening services and case finding for symptomatic PLWD are available at the static eye clinics and outreach eye care camps on a walk-in basis. Tertiary eye clinics have a retina clinic at least once a week. There is no waiting list for screening in any of the eye clinics. There are waiting lists for laser photocoagulation in some clinics, but the waiting time is only one week. *Weaknesses*: Eye care services are underutilized by asymptomatic PLWD. This is paradoxical, since screening services provide significant benefit for early detection of DR.⁴⁴ Most of the PLWD attending diabetes services do not attend the eye clinic for screening, because of the demand side and supply-side barriers highlighted above. There are missed opportunities for screening at the eye clinics as well, mainly attributed to lack of clinical guidelines to standardize care. Common with diabetes in other SSA countries, patients with DR present late when they have advanced DR and vision loss.⁴⁵ The eye clinics do not engage with the DSGs.

Opportunities

Service provider and key informants suggested the need to integrate diabetes and DR services to improve service coordination on the long-term. The rationale for this integration is discussed in research paper 2. Previous studies have identified a positive association between different integrated models of care and diabetes outcomes.⁵ Progress with such integration will require time and capacity development across the system.

National DR Guidelines are needed to ensure services are available to PLWD, and that all actors in eye health services for PLWD speak from a common guideline. There are opportunities for peer-led education on diabetes eye health in the community and for engaging DSGs to link PLWD with eye clinics for DR screening. These interventions will be discussed in subsequent chapters of this thesis.

Health workforce

Community

Strength: Community health volunteers (CHVs) link community members with primary care facilities, for instance for testing for diabetes. Peer supporters in DSGs carry out health education and link PLWD with diabetes clinics, which is a form of task shifting. Previous studies have identified task shifting as a successful intervention in some parts of SSA, which has potential to influence diabetes care.^{5, 45}

Weakness: The peer supporters and the CHVs are not trained to provide education on the component of diabetes eye health, resulting in low heath literacy on this aspect. **Diabetes clinic**

Strengths: All clinics have a multidisciplinary team available, including a physician, diabetes educator, pharmacist, nurses, nutritionists, records clerks, medical officers, clinical officers and social workers. PLWD are referred to additional specialists such as surgeons based on need. Most of the team is trained locally, though some have foreign training. This is important because multidisciplinary care is a facilitator for effective diabetes care within a chronic care model.^{5, 46} The extent to which the different cadres provide care as a coordinated team rather than separately is uncertain. This is an important area for research — as diabetes care by uncoordinated specialists may overwhelm both health workers and PLWD, given its effect of increasing workload.³⁴ *Weaknesses*: The number of health workers is inadequate for the population, and the distribution is generally skewed in favour of Nairobi. Table 3-2 provides an example of the distribution of physicians and diabetes educators. Service providers identify limited training as a major hindrance to providing services, coupled with frequent transfers of trained workers. This points to the need for effective and constantly available training opportunities.^{38, 45}

	Kirinyaga	Nakuru	Nairobi	National
Total pop (millions)	0.6	1.8	3.6	55
Physicians - N	3	33	120	501
(per 100,000 pop)	(0.5)	(1.8)	(3.3)	(9.1)
Diabetes educators - N	5	50	2500	3000
(Per 100,000 pop)	(0.8)	(2.8)	(69.4)	(54.5)

Table 3-2: Health workforce for diabetes services - physicians and diabetes educators

Eye clinic

Strengths: The eye care workforce includes ophthalmologists, ophthalmic clinical officers and ophthalmic nurses, mostly trained in two training institutions in the country. These cadres provide service delivery but also have roles as educators, administrators and researchers.

Weaknesses: There are no optometrists, patient counsellors, equipment maintenance technicians or records clerks in most of the public service eye clinics. There is only one equipment technician at the Ophthalmic Services Unit, and another at one tertiary hospital, which leads to delays in equipment repairs. The eye care workforce is insufficient in number and distribution, as in other LMICs.^{47, 48} Table 3-3 shows the cadre to population ratio. The training capacity of the training institutions is insufficient to meet the needs. Although the pre-service training is perceived as high quality, the lack of opportunities for continuous professional development especially regarding eye care for PLWD is a major hindrance to service delivery. Thus the capacity of both individuals and institutions influence the service delivery.⁴²

	Kirinyaga	Nakuru	Nairobi	National	VISION 2020 recommendations (ratio by 2020)
Total pop (millions)	0.6	1.8	3.6	55	-
Ophthalmologists	1	3	46	130	1:250,000
(Per 100,000)	(0.2)	(0.2)	(1.3)	(2.4)	(0.4:100,000)
Ophthalmic	3	5	20	200	1:100,000
(per 100,000)	(0.5)	(0.3)	(0.6)	(3.6)	
Ophthalmic	0	3	21	180	1:100,000
(per 100,000)	-	(0.2)	(0.6)	(3.3)	

Table 3-3: Distribution of eye care workers

Opportunities

Empowering peer supporters in DSGs to deliver education on eye health to PLWD may be a useful form of task shifting, given the workforce shortage. An integrated openaccess training program for diabetes and eye heath workers may address the need for a training program that is flexibly available.

Health Management Information Systems (HMIS)

Community

Strengths: DSGs keep paper-based records of PLWD, recording sociodemographic details, comorbidity with hypertension and the measurements taken at every DSG meeting: blood sugar, blood pressure and weight or body mass index (BMI). PLWD keep a copy of this data (patient-held records).

Weaknesses: The data collected is not shared or used to inform services beyond the DSG. There is no system of risk factor surveillance, which is an important target for countries looking to improve their health system performance for diabetes.⁴⁹

Diabetes clinic

Strengths: There is a gradual shift from paper-based records to electronic records, though currently both forms are used. Aggregate facility data (such as numbers attended) is transmitted to the national HMIS system.

Weaknesses: Individual PLWD do not have access to the comprehensive medical record from the diabetes clinics. There is no diabetes registry with call and recall mechanism and there is no way of tracking the movement of PLWD along the care pathway. The Chronic Care Model recommends that HMIS systems foster the formation and use of diabetes registries, to enhance access to clinical data, patient engagement and generation of evidence.^{46, 50, 51} The facility and county data is not used for decisionmaking at the local level. Without complete electronic medical records (EMR), it is difficult to use the HMIS to track inputs and outputs such as medicines used, to avoid stock-outs.^{45, 46} The data transmitted to the national HMIS is mainly workload data, and some facilities do not submit this data, which limits the use of the data.

Eye clinic

122

Strengths: Sociodemographic and clinical data is collected at the eye unit mainly using paper-based systems.

Weaknesses: Individual PLWD do not have patient-held records. Indicators for DR, such as number screened for DR are not reported. There is no mechanism for tracking PLWD who have been screened, referred or treated, and most clinics do not have infrastructure for capturing and storing retinal images. There is no surveillance system for DR. The data collected also has missing/incomplete information, as most eye clinics do not have a trained and dedicated records officer. Training the people entering data to also analyse it and present it can improve the performance.⁵²

Opportunities:

There is a need to develop a diabetes database to provide longitudinal information on individual PLWD. The HMIS system should alert the provider to the needed tests and provide tracking. Given the mobile phone penetration of about 77% in Kenya, mhealth interventions such as patient reminders through short message service (SMS) would be useful for scheduling follow-ups or educating PLWD.⁵³

Health Financing

Community:

Strengths: In some counties, public-private partnerships (PPPs), such as the 'base of the pyramid program' by Novo Nordisk have helped to stabilize the cost of insulin at about \$5 per vial of Mixtard insulin, compared to \$18 in other counties. This is because they supply the insulin at subsidized cost to some facilities, introducing competition among retail suppliers, and hence prices come down. ²⁷

Weaknesses: Less than 20% of the population has any form of insurance coverage, which implies that a correspondingly small proportion of PLWD have insurance.⁵⁴ DSGs have not been adequately sensitized on the need for PLWD to register with the national hospital insurance fund (NHIF), the largest provider of health insurance. Comorbidity with hypertension or depression increases the financial burden on PLWD.⁵⁵ In addition, the transport burden is high for patients in rural areas, which may impede access to services.^{27, 56} The other weakness is that peer supporters and CHVs work on a volunteerbasis, but the community is responsible for keeping them incentivized. This often involves a form of payment in kind, but this is largely invisible and unaccounted for in the health financing indicators. The sustainability of PPPs, as mentioned above, is in doubt, since they are not integrated within national plans.²⁷

Diabetes clinic:

Strengths: NHIF covers some of the costs for laboratory tests and treatments for diabetes, within the chronic disease package of care.⁵⁷ Some commodities and services are available at subsidized cost, due to PPPs mentioned earlier.²⁷ There is a centralized payment system at each health facility for financial accountability.

Weaknesses: Most of the services have to be paid for at the point of care (out of pocket payment) as only a small proportion of PLWD have NHIF, mainly those in formal employment. This means that the unemployed PLWD are more likely to suffer financial barriers to access, leading to inequity. In addition, some of the services are not available at the diabetes clinics, and have to be sourced from external pharmacies and laboratories, at a high cost.⁵⁷ Further, only 40% of health facilities in the country are covered by NHIF, hence PLWD attending facilities outside this network have to pay for services out of pocket.⁵⁴

124

Eye clinic

Strengths: The cost of screening is included in the consultation fee at the eye clinic, hence there is no extra cost to the patient requiring DR screening. NHIF has recently expanded coverage to include treatments for DR: laser, anti-VEGF and vitrectomy. This incremental coverage has resulted from lobbying by interest groups especially the Ophthalmologic Society of Kenya, as has been the experience of other countries.⁵⁸ Some of the outreach screening programs provide transport for PLWD so that they can obtain treatment at static eye clinics. This innovation to reduce the transport burden is in keeping with the delivery support element of CCM.³⁴

Weaknesses: A consultation fee is paid out-of-pocket for each clinic visit, with PLWD attending both diabetes and eye clinics having to pay a double fee. Most PLWD present to the eye clinic late, when DR is advanced, yet the cost of treatment would be lower if DR is detected and treated early.^{7, 59} Without NHIF cover, PLWD have to pay for treatment out of pocket. The financial burden for DR treatments is considerable, reflecting that universal health coverage has not yet been achieved.^{57, 60} On the other hand the sustainability of financing UHC through NHIF is doubtful, given that the current premium rates are quite low (averaging 2.4% of gross pay for formal sector workers) and yet the benefit package is very generous.⁶¹ The country is currently facing these sustainability concerns.

Opportunities

CHVs and DSGs can be instrumental in ensuring that their members are registered with NHIF, if they are adequately sensitized.

125

Medicines and health technologies

Community

Strengths: DSGs have glucometers and blood pressure machines and sometimes they have some drugs at subsidized cost. There are some mobile phone applications for health promotion, such as *Afya Pap*, but their use among PLWD has been limited to a pilot project in Nairobi, perhaps because the highest coverage with mobile phones is in urban areas.^{53, 62}

Weaknesses: Technology to aid lifestyle modification such as wearable devices are not commonly available. Most PLWD do not have devices for home glucose monitoring. DSGs have the devices for group use but often run out of strips for testing blood glucose. PLWD have difficulty obtaining insulin syringes, as they sometimes have to buy them in private pharmacies. This is of concern because it limits glycaemic monitoring and increases the risk of complications.

Diabetes clinic

Strengths: Access to diabetes medicines is facilitated by two main factors. Firstly, the Essential Drugs List for county hospitals includes key medications, such as insulin and metformin. Secondly, some clinics have insulin at subsidized cost, through arrangements with stakeholders in the supply chain management, such as Novo Nordisk, the International Diabetes Federation, the Changing Diabetes in Children program, Medicins Sans Frontieres, and distributors of drugs.

Weaknesses: Clinics experience shortages of drugs at different points in the year, mostly due to procurement delays. As the cost of diabetes medicines is unaffordable to many PLWD, non-adherence is likely to result.^{45, 57} Clinics sometimes run out of glucose test strips, and most of them do not offer HbA1c testing.

Eye clinic

Strengths: Treatment services (laser photocoagulation) for DR are available in at least one facility in most counties.

Weaknesses: Eye clinics sometimes run out of supply of mydriatic drugs. Anti-VEGF is not routinely available in government clinics, hence PLWD are asked to buy it and bring it for injection. This negatively affects compliance with anti-VEGF treatment. The multidose vial of anti-VEGF is expensive, and can lead to drug wastage.

Opportunities: The use of artificial intelligence and telemedicine in DR screening has been tested in research settings, but is not yet applied in routine use.⁶³ There is potential for automated AI-based grading systems to improve efficiency in DR screening.

Governance

Stewardship for diabetes care is under the auspices of the Division of Non-Communicable Diseases (NCDS) of the Ministry of Health. There is a comprehensive national strategy for the control of NCDs 2015-2020 which is actively used.⁶⁴

Community

Strengths: The Kenya Defeat Diabetes Association, a civil society organization responsible for the DSGs, is strongly engaged in advocacy with the government and other stakeholders at national level, such as the Diabetes Management and Information Centre, Diabetes Kenya Association, World Diabetes Federation, and International Diabetes Federation. These stakeholders, coordinated by the Division of NCDs, mobilize communities and schools for events such as annual diabetes walks, community screening and health promotion activities especially around annual advocacy weeks. At facility level, local health committees provide a forum for community participation in governance. The national strategy for control of NCDs includes a strategic objective on implementing legislation regarding the content of processed foods and drinks (refined sugar, salt, saturated and trans fats), as well as the labelling and marketing of these products.⁶⁴ Another strategic objective in the strategy relates to implementing economic incentives for encouraging consumption of healthy foods, such as taxes and subsidies.⁶⁴ *Weaknesses*: The component of eye health is often missing in the health promotion program. Action on the social and economic determinants of health, such as engagement with the agriculture, food industry and urban planning sectors has been challenging, as has been enacting legislation on processed food products, because these require lengthy political processes.

Diabetes clinic

Strengths: There is a national diabetes strategy, albeit it requires updating.³⁶ The Ministry of Health has a focal person for NCDs in each county who provides leadership. The NCD division is in the process of establishing some diabetes clinics as centres of excellence for diabetes care in children. Diabetes services actively engage DSGs, nongovernmental organizations (NGOs), and pharmaceuticals as stakeholders. *Weaknesses*: The NCD and diabetes strategies do not include eye health. The strategies are not available in some of the diabetes clinics outside Nairobi, which hinders their use, though they have potential to improve diabetes care.^{46, 65} The diabetes strategy does not provide guidance on managing prediabetes, and most of the people with prediabetes do not attend diabetes clinics. Although there is a focal NCD lead, coordination between public, private-for-profit and private-not-for-profit actors is challenging as they have different lines of accountability. In addition, the role of the focal NCD lead in fostering implementation of the chronic care model is not explicit.

Eye clinic

Strengths: The national level has strong links with professional associations, academia, NGOs and other sectors besides the Ministry of Health. Decentralisation of governance to county level has increased efficiency, for instance in procurement of supplies. Each county has a focal eye care person who provides leadership. Some non-governmental organizations such as the Fred Hollows Foundation and Operation Eyesight Universal support DR services in some counties.

Weaknesses: There is no long-term strategy for control of DR, and no national guidelines for DR. There have been several attempts to develop the guidelines in the past, but the process was not completed. A recent systematic review identified clinical guidelines as one of the interventions that have improved diabetes care in SSA.^{38, 45} This might be because guidelines articulate a consistent set of priorities and attract stakeholder support.⁴²

Opportunities

The upcoming updating of the diabetes strategy is an important window of opportunity to include eye health in the strategy and to implement CCM in diabetes care. Leveraging on the stakeholder support can facilitate the development of clinical guidelines for DR.

Infrastructure

Community

Strengths: DSGs can use public facilities, such as bus parks, churches or schools for outreach screening camps.

Weaknesses: There is no infrastructure and equipment for screening for DR at the community level.

Diabetes clinic

Strengths: In a few facilities, diabetes clinics and eye care clinics are located within the same building. This proximity is advantageous for linking PLWD to the two services. One diabetes clinic in Nairobi has a retinal camera for DR screening, facilitating screening at the point of diabetes care.¹⁴

Weaknesses: Often the clinics are in separate buildings and do not have sufficient space hence they are over-crowded. Equipment inventories are not regularly done. Most clinics reported that the equipment was insufficient, reflecting suboptimal infrastructure for diabetes.^{40, 41}

Eye clinic

Strengths: Some of the eye clinics have renovated buildings and vehicles provided by partners, especially NGOs.

Weaknesses: The equipment for screening is insufficient, and most eye clinics do not have retinal cameras. There are not enough treatment facilities including space and equipment for laser, as well as theatre for vitrectomy or administration of anti-VEGF. Some of the equipment such as lasers and vitrectomy machines are broken down and not functional. Spare parts are often unavailable locally.

<u>Research</u>

Community

Strengths: Previous population-based research includes epidemiologic studies such as the STEPS survey⁶⁶ and the Nakuru cohort study.^{63, 67-70} Some studies have investigated access to medicines.^{27, 71} Kenya has adopted risk factor surveillance based on the STEPwise approach. *Weaknesses*: Public engagement and dissemination of research to rural communities is insufficient.

Diabetes clinic

Strengths: There are strong national research partnerships, such as the Kenya Diabetes Research Group. The types of research that have been done include operational research and epidemiological studies. A recent study has examined the cost of diabetes care to the patient.⁵⁷

Weaknesses: Most of the research is quantitative, and there is a need for more qualitative research, for example on patient-related outcomes such as quality of life and treatment burden. Appraising the evidence is not systematically performed, and it is unclear how much this research has been used to inform policy and practice.

Eye clinic

Strengths: There is a national DR working group coordinated by the Ministry of Health, which is interested in collating all the research on DR and conducting collaborative research. There is a strong link between the national level and research institutions, both local and international. Some studies have investigated the effectiveness of a DR screening program,⁷²the use of telemedicine,⁷³ and the use of AI in DR screening.⁶³ The most recent Kenya Demographic and Health Survey (KDHS) has included data on visual impairment.⁷⁴

Weaknesses: The research conducted does not include patient-related outcomes. There is a shortage of skills in implementation research.

Opportunities

The research endeavours should contribute to developing local research capacity as robust research can provide evidence to improve DR services.

There is a need for research that focuses on the experience of patients along the care pathway and patient-related outcomes, such as financial protection.

3.7 Synthesis of gaps and strengths

Although there are some differences between the counties, the findings revealed gaps in health systems in all the counties, suggesting the need for health system strengthening (HSS) in the country. Both supply-side and demand-side barriers lead to gaps in access and continuity of care. There is low access to both HbA1c and DR screening. As DR is asymptomatic, without screening there are long delays before detection and treatment, with worse health and cost outcomes.

Three key strengths of the health system should be harnessed. First, there are community resources such as DSGs whose task profile can be expanded to include educating PLWD on eye health and linking them to screening services. Second, there is high stakeholder support for strengthening services. Third, key informants and service providers recognize the need to foster collaboration between diabetes and DR services, and they could provide support for collaboration and integration.

3.8 Priority-setting – health system strengthening

The results of the health system assessment were shared with the national DR working group. Taking consideration of the main gaps, and the strengths, as well as the resources available, the DR working group prioritized two HSS actions:

- (a) the development of national clinical guidelines for diabetic retinopathy
- (b) the development of a flexible training program on eye health for DR

Guidelines were considered a priority because they would influence multiple interacting factors to improve services, such as mobilise resources for DR, strengthen a referral system and provide a quality assurance mechanism for DR services. Guidelines would also have long-term effects. Training was considered important because health workers identified it as a need. These HSS activities are discussed in chapter 5 and 6 of this thesis.

3.9 Priority-setting- Intervention to improve access to DR screening

DR screening is the entry point to DR services, and since attendance at screening is low, increasing the attendance at screening is a priority. The HSS action points above target the supply-side barriers to screening. It is unclear whether a peer supporter-led intervention within DSGs can help to address the demand side barriers and increase demand for screening. A clinical trial to test the effectiveness of this intervention was designed with input from stakeholders in diabetes and DR care. As Kirinyaga county has the lowest uptake of DR screening, and the most active DSGs among the three counties, it was selected for this intervention. Additional information on DSGs is provided below, while Chapter 7, 8 and 9 of this thesis describe this clinical trial.

Diabetes Support Groups

The Kenya Defeat Diabetes Association (KDDA) is a not-for-profit civil society organization that brings together the DSGs in the country. It has a national and county level governance structure and works in partnership with the Ministry of Health, particularly the Division of Non-Communicable Diseases. KDDA trains peer educators, mobilizes resources for DSGs (glucometers, glucose test strips, blood pressure machines, weighing scales, stationery), and organizes community diabetes screening camps. These activities are funded by partners such as Non-Governmental Organizations that work with the county health services. KDDA also engages in advocacy for improved access and quality of services at the facility, county and national level.

133

PLWD become aware of DSGs through community events such as health talks and announcements provided at various meetings – at village barazas (public meetings), places of worship, women's groups, door to door visits by community health volunteers and community diabetes screening camps. Peer educators also give health talks at the diabetes and nutrition clinics in health facilities, and recruit PLWD to join DSGs. A DSG is established when at least 15-20 PLWD in a particular location have been recruited and two DSG leaders (1 male, 1 female) have been selected from among them. The target size of DSGs is 50-100 PLWD. Formal membership status is obtained by payment of a one-off requisite fee (Ksh 1500, \$15), upon which one receives a KDDA membership card. However attendance and participation in DSG activities is open even to those who do not have formal membership status. DSGs meet once a month at a designated venue agreed on by the group e.g. at a church, school, market or health facility. The activities at the meeting include health talks by peer educators, measurement of weight, blood pressure and blood glucose. PLWD pay Ksh 100 (\$1) at each meeting to support the purchase of resources such as glucose test strips. The role of the DSG leader is administrative – organizing the DSG meetings, stewardship of DSG resources and linking the DSG with county and national leadership of KDDA. The DSG leaders are volunteers and do not receive formal training for that role. Peer educators on the other hand are trained by KDDA to deliver health talks on selfmanagement, and to provide group/individualized peer support; they also are volunteers. In some DSGs, one individual has the dual role of leader and peer educator. Being volunteers, they do not receive formal remuneration for their work, and the DSG activities are not expected to interfere with their usual economic activities.

REFERENCES

1. Bennett S, Peters DH. Assessing National Health Systems: Why and How. Health Systems & Reform. 2015;1(1):9-17.

2. World Health Organization. The World Health Report 2000: improving performance. Geneva: WHO; 2000.

3. Smith PC, Mossialos E, Papanicolas I, Leatherman S. Performance measurement for health system improvement: experiences, challenges, prospects. Mossialos E, editor. New York: Cambridge University Press; 2009.

4. Mounier-Jack S, Griffiths UK, Closser S, Burchett H, Marchal B. Measuring the health systems impact of disease control programmes: a critical reflection on the WHO building blocks framework. BMC Public Health. 2014;14(1):278.

5. Ong SE, Koh JJK, Toh S-AES, Chia KS, Balabanova D, McKee M, et al. Assessing the influence of health systems on Type 2 Diabetes Mellitus awareness, treatment, adherence, and control: A systematic review. PLOS ONE. 2018;13(3):e0195086.

6. Wanjiku Mathenge. Artificial intelligence for diabetic retinopathy screening in Africa. Lancet Digital Health. 2019;1:e6-e7.

7. Scanlon PH. The English National Screening Programme for diabetic retinopathy 2003–2016. Acta Diabetologica. 2017;54(6):515-25.

8. Luoma M, Doherty J, Muchiri S, Barasa T, Hofler K, Maniscalco L, et al. Kenya Health System Assessment 2010. Bethesda, MD: Health Systems 20/20 project, Abt Associates Inc; 2010.

9. Ministry of Health. Kenya Eye Health System Assessment. Nairobi: Ministry of Health; 2017.

10. Ramke J, Zwi AB, Silva JC, Mwangi N, Rono H, Gichangi M, et al. Evidence for national universal eye health plans. Bull World Health Organ. 2018;96(10):695-704.

11. Hoffman SJ, Røttingen JA, Bennett S, Lavis JN, Edge JS, Frenk J. Background paper on conceptual issues related to health systems research to inform a WHO Global strategy on health systems research Geneva: World Health Organization; 2012 [updated 2012. Available from: <u>http://www.who.int/alliance-</u>

hpsr/alliancehpsr_backgroundpaperhsrstrat1.pdf.

12. Pyone T, Smith H, van den Broek N. Frameworks to assess health systems governance: a systematic review. Health Policy and Planning. 2017;32(5):710-22.

13. World Health Organization. Everybody's business: Strengthening health systems to improve health outcomes: WHO's framework for action. Geneva: WHO; 2007.

14. Poore S, Foster A, Zondervan M, Blanchet K. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. Int J Health Policy Manag. 2015;4(1):19-28.

15. Blanchet K, Gilbert C, de Savigny D. Rethinking eye health systems to achieve universal coverage: the role of research. British Journal of Ophthalmology. 2014;98:1325-8.

16. Ministry of Health Kenya. Kenya Health Policy Framework 2014-2030: Towards attaining the highest standards of health Nairobi: Ministry of Health; 2014.

17. Sherr K, Fernandes Q, Kanté AM, Bawah A, Condo J, Mutale W, et al. Measuring health systems strength and its impact: experiences from the African Health Initiative. BMC Health Services Research. 2017;17(3):827.

18. World Health Organization. TADDS-Tool for the assessment of diabetic retinopathy and diabetes management systems Geneva: WHO; 2015.

19. Kessner D, Kalk C, Singer J. Assessing health quality: the case for tracers. N Engl J Med. 1973:189-94.

20. Nolte E, Bain C, McKee M. Diabetes as a Tracer Condition in International Benchmarking of Health Systems. Diabetes Care. 2006;29(5):1007-11.

21. Papanicolas I, Smith PC. Health system performance comparison: an agenda for policy, information and research. New York: Open University Press; 2013.

22. Smith RD, Hanson K. Health systems in low and middle income countries: an economic and policy perspective. New York: Oxford University Press; 2012.

23. Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556-64.

24. Cheung Ning, Mitchell Paul, Wong Tien Yin. Diabetic retinopathy. The Lancet. 2010;376:124-36.

25. Beran D. Health systems and the management of chronic diseases: lessons from Type 1 diabetes Diabetes Management. 2012;2(4):323–35.

26. Azevedo M, Alla S. Diabetes in sub-Saharan Africa: Kenya, Mali, Mozambique, Nigeria, South Africa and Zambia. International Journal of Diabetes in Developing cities. 2008;28(4):101-8.

27. Shannon GD, Haghparast-Bidgoli H, Chelagat W, Kibachio J, Skordis-Worrall J. Innovating to increase access to diabetes care in Kenya: an evaluation of Novo Nordisk's base of the pyramid project. Global health action. 2019;12(1):1605704-.

28. Vasoontara Yiengprugsawan, Judith Healy, Hal Kendig, Malinee Neelamegam, Palitha Karunapema, Vijj Kasemsup. Reorienting Health Services to People with Chronic Health Conditions: Diabetes and Stroke Services in Malaysia, Sri Lanka and Thailand. Health Systems & Reform. 2017;3(3):171-81.

29. Beran D, Yudkin JS, de Courten M. Assessing health systems for type 1 diabetes in sub-Saharan Africa: developing a ' Rapid Assessment Protocol for Insulin Access'. BMC Health Services Research. 2006;6(17).

30. van Olmen J, Marchal B, van Damme W, Kegels G, Hill PS. Health systems frameworks in their political context: framing divergent agendas. BMC Public Health. 2012;12(774).

31. Jones K. Strategies for globalizing research in the educational sciences. 42nd annual conference of the Japan Society for Science Education (JSSE); Shinshu University, Nagano, Japan2018.

32. Sheikh K, Ranson MK, Gilson L. Explorations on people centredness in health systems. Health policy and planning. 2014;29(Suppl 2):ii1-ii5.

33. Stuckey HL, Adelman AM, Gabbay RA. Improving care by delivering the Chronic Care Model for diabetes. Diabetes Management. 2011;1(1):37-52.

34. Boehmer KR, Abu Dabrh AM, Gionfriddo MR, Erwin P, Montori VM. Does the chronic care model meet the emerging needs of people living with multimorbidity? A systematic review and thematic synthesis. PLOS ONE. 2018;13(2):e0190852.

35. Nyblade L, Stockton MA, Giger K, Bond V, Ekstrand ML, Lean RM, et al. Stigma in health facilities: why it matters and how we can change it. BMC Medicine. 2019;17(1):25.

36. Ministry of Health. Kenya National Diabetes Strategy 2010-2015. Nairobi: Ministry of Health; 2010.

37. Wagner EH, Austin BT, Davis C, Hindmarsh M, Schaefer J, Bonomi A. Improving chronic illness care: translating evidence into action. Health Aff (Millwood). 2001;20(6):64-78.

38. Beran D. The Impact of Health Systems on Diabetes Care in Low and Lower Middle Income Countries. Current Diabetes Reports. 2015;15(4):20.

39. American Diabetes Association. Standards of Medical Care in Diabetes-2017. Diabetes Care. 2017;40(Suppl 1).

40. Pastakia SD, Nuche-Berenguer B, Pekny CR, Njuguna B, O'Hara EG, Cheng SY, et al. Retrospective assessment of the quality of diabetes care in a rural diabetes clinic in Western Kenya. BMC Endocrine Disorders. 2018;18(1):97.

41. Pastakia SD, Karwa R, Kahn CB, Nyabundi JS. The Evolution of Diabetes Care in the Rural, Resource-Constrained Setting of Western Kenya. Annals of Pharmacotherapy. 2011;45(6):721-6.

42. Balabanova D, McKee M, Mills A. Good Health at Low Cost 25 Years On: What Makes a Successful health System? London: London School of Hygiene and Tropical Medicine; 2011.

43. Mwangi N, Macleod D, Gichuhi S, Muthami L, Moorman C, Bascaran C, et al. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Tropical Medicine and Health. 2017;45(41).

44. International Diabetes Federation, Fred Hollows Foundation. Diabetes eye health: a guide for health professionals. Brussels, Belgium: International Diabetes Federation; 2015.

45. Nuche-Berenguer B, Kupfer LE. Readiness of Sub-Saharan Africa Healthcare Systems for the New Pandemic, Diabetes: A Systematic Review. Journal of Diabetes Research. 2018;2018:12.

46. Schmittdiel JA, Gopalan A, Lin MW, Banerjee S, Chau CV, Adams AS. Population Health Management for Diabetes: Health Care System-Level Approaches for Improving Quality and Addressing Disparities. Current diabetes reports. 2017;17(5):31.

47. Palmer JJ, Chinanayi F, Gilbert A, Pillay D, Fox S, Jaggernath J, et al. Mapping human resource for eye health in 21 countries of sub-Saharan Africa: current progress towards VISION 2020. Human Resources for Health. 2014;12(44).

48. Palmer JJ, Chinanayi F, Gilbert A, Pillay D, Fox S, Jaggernath J, et al. Trends and implications for achieving VISION 2020 human resources for eye health targets in 16 countries of sub-Saharan Africa by the year 2020. Human Resources for Health. 2014;12(1):45.

49. Manne-Goehler J, Geldsetzer P, Agoudavi, K A-BG, Aryal KK BB, et al. Health system performance for people with diabetes in 28 low- and middle-income countries: A cross-sectional study of nationally representative surveys. PLoS Med. 2019;16(3):e1002751.

50. Ku GM, Kegels G. Adapting chronic care models for diabetes care delivery in low-and-middle-income countries: a review. World J Diabetes. 2015;6:566-75.

51. Stellefson M, Dipnarine K, Stopka C. The Chronic Care Model and Diabetes Management in US Primary Care Settings: A Systematic Review. Preventing Chronic Disease. 2013;10(120180).

52. Faal H, Cook C, Thulasiraj R. Managing information in eye care programmes: the health systems perspective. Community Eye Health. 2010;23(74):50-2.

53. Omae Malack Oteri, Langat Philip Kibet, Ndung'u Edward N. Mobile Subscription, Penetration and Coverage Trends in Kenya's Telecommunication Sector. International Journal of Advanced Research in Artificial Intelligence (IJARAI). 2015;4(1):1-7.

54. Barasa E, Rogo K, Mwaura N, Chuma J. Kenya National Hospital Insurance Fund Reforms: Implications and lessons for Universal Health Coverage. Health Systems and Reforms. 2018;4(4):346-61.

55. Subramaniam M, Sum CF, Pek E, Stahl D, Verma S, Liow PH, et al. Comorbid depression and increased health care utilisation in individuals with diabetes. Gen Hosp Psychiatry. 2009;31(3):220-4.

56. O'Hara EG, Nuche-Berenguer B, Kirui NK, Cheng SY, Chege PM, Buckwalter V, et al. Diabetes in rural Africa: what can Kenya show us? The Lancet Diabetes & Endocrinology. 2016;4(10):807-9.

57. Subramanian S, Gakunga R, Kibachio J, Gathecha G, Edwards P, Ogola E, et al. Cost and affordability of non-communicable disease screening, diagnosis and treatment in Kenya: Patient payments in the private and public sectors. PLOS ONE. 2018;13(1):e0190113.

58. Reich MR, Harris J, Ikegami N, Maeda A, Cashin C, Araujo EC, et al. Moving towards universal health coverage: lessons from 11 country studies. The Lancet. 2016;387(10020):811-6.

59. Scanlon PH, Aldington SJ, Leal J, Luengo-Fernandez R, Oke J, Sivaprasad S, et al. Development of a cost-effectiveness model for optimisation of the screening interval in diabetic retinopathy screening. Health Technology Assessment. 2015;19(74):1-116.

60. World Health Organization. Executive summary in: The World Health Report: Health Systems Financing –The Path to Universal Coverage Geneva: WHO; 2010.

61. Okungu V, Chuma J, McIntyre D. The cost of free health care for all Kenyans: assessing the financial sustainability of contributory and non-contributory financing mechanisms. International Journal for Equity in Health. 2017;16(39).

62. Wesolowski A, Eagle N, Noor AM, Snow RW, Buckee CO. Heterogeneous Mobile Phone Ownership and Usage Patterns in Kenya. PLOS ONE. 2012;7(4):e35319.

63. Hansen MB, Abràmoff MD, Folk JC, Mathenge W, Bastawrous A, Peto T. Results of Automated Retinal Image Analysis for Detection of Diabetic Retinopathy from the Nakuru Study, Kenya. PloS one. 2015;10(10):e0139148-e.

64. Ministry of Health. Kenya national strategy for prevention and control of noncommunicable disease 2015-2020. 2015.

65. Atieno-Jalango G, Tsolekile LP, Puoane T. Do healthcare workers adhere to diabetes clinical care guidelinees? A study at a national hospital, Kenya. Journal of Hypertension. 2014;3(6):1-4.

66. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases; 2015.

67. Bastawrous A, Mathenge W, Peto T, Weiss HA, Rono H, Foster A, et al. The Nakuru eye disease cohort study: methodology & rationale. BMC Ophthalmol. 2014;14:60.

68. Bastawrous Andrew, Mathenge Wanjiku, Wing Kevin, Bastawrous Madeleine, Rono Hillary, Weiss Helen A, et al. The incidence of diabetes mellitus and diabetic retinopathy in a population-based cohort study of people age 50 years and over in Nakuru , Kenya. 2017:1-14.

69. Kuper H, Mathenge W, Macleod D, Foster A, Gichangi M, Rono H, et al. Mortality during 6 years of follow-up in relation to visual impairment and eye disease: results from a population-based cohort study of people aged 50 years and above in Nakuru, Kenya. BMJ Open. 2019;9(6):e029700.

70. Mathenge Wanjiku, Bastawrous Andrew, Peto Tunde, Leung Irene, Yorston David, Foster Allen, et al. Prevalence and Correlates of Diabetic Retinopathy in a Population-based Survey of Older People in Nakuru, Kenya. Ophthalmic Epidemiology. 2014;21(3):169-77.

71. Onyango MA, Vian T, Hirsch I, Salvi DD, Laing R, Rockers PC, et al. Perceptions of Kenyan adults on access to medicines for non-communicable diseases: A qualitative study. PLOS ONE. 2018;13(8):e0201917.

72. Gichuhi S, Gichangi M, Nyamori J, Gachago M, Nyenze EM, Nyaga PT, et al. Evaluation of the Kenyatta National Hospital diabetic retinopathy screening program 2015-2016. Journal of Ophthalmology of Eastern Central and Southern Africa. 2017;21(2):40-4.

73. Kurji K, Kiage D, Rudnisky CJ, Damji KF. Improving diabetic retinopathy screening in Africa: patient satisfaction with teleophthalmology versus ophthalmologist-based screening Middle East African Journal of Ophthalmology. 2013;20:56-60.

74. Kenya National Bureau of Statistics, Ministry of Health/Kenya, National AIDS Control Council/Kenya, Kenya Medical Research Institute, Population NCf, Development/Kenya. Kenya Demographic and Health Survey 2014. Rockville, MD, USA; 2015.

Chapter Four: Additional findings from health system assessment

4.0 Overview

This chapter includes two results papers. Research paper 1 is on the predictors of uptake of DR screening (DRS). This paper has been included since it describes the barriers and enablers for screening in the specific context of PLWD in Kenya. These results provide the basis of thinking for developing the interventions described throughout the rest of the thesis.

Research paper 2 is on the rationale for integration of diabetes and DR services, as proposed by key informants and service providers who participated in the health system assessment. The health system context influences the nature, extent and success of integration. This paper has been included because it provides evidence on how integration might be implemented and includes a contextual framework that may be generalizable to similar settings.

4.1 Research paper 1

Preamble

The barriers and enablers that influence the access and uptake of screening for DR among people living with diabetes (PLWD) are poorly understood, and there is little evidence on how these factors vary by context. Understanding the predictors for attendance to DRS is essential to refine strategies to improve access to DRS. There are no known studies which have looked at the specific predictors to attendance at DRS in Kenya, and this study addresses this gap. The evidence from this study has been used to inform the development of interventions to increase attendance at DRS, which are discussed in subsequent chapters. This paper was published in the journal BMC Tropical Medicine and Health in December 2017 after peer review.

BMC Tropical Medicine and Health – Licences and Copyright

The following terms apply to this publication

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy**, **distribute, and use their attributed research**, **free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago.

All articles published by *Tropical Medicine and Health* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access

https://tropmedhealth.biomedcentral.com/about

4.2 Research paper 2

Preamble

The lack of linkage between diabetes services and eye care services for PLWD is a challenge for the health system in Kenya. PLWD in diabetes clinics receive diabetes care, largely without an eye heath component e.g. eye health education, screening or referral for treatment. PLWD attending eye clinics for DR care may receive services to address this ocular complication but without adequate attention to other aspects of diabetes management. This limits the continuity of care for PLWD, who also have difficulty navigating the health system.

To strengthen health systems and to achieve continuity of high quality care, some integration between diabetes and DR services is important. Experts recommend tailoring integrated care interventions to the local context, which highlights the potential value of this paper's focus on understanding integration of care within Kenya's health system.

The evidence presented in this paper is part of the results of the health system assessment for diabetes and diabetic retinopathy. Integration of DR into diabetes services emerged as a key theme during interviews with key informants and service providers, as they described the factors related to lack of integration, ways to remedy the lack of integration, and the priorities for integration.

The paper has been submitted to BMC Health Services Research.

Once published, the following terms will apply:

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy**, **distribute, and use their attributed research**, **free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago.

All articles published in *BMC Health Services Research* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access

https://bmchealthservres.biomedcentral.com/about



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?	BMC Tropical Medic	cine and Health	
When was the work published?	Dec 2017		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	Not applicable		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D – Multi-authored work
Student Signature:	Nyawira	 à	
Supervisor Signature:			

Date:	17 SEPT 2019
-------	--------------

Date: SEPT 17, 2019

RESEARCH

Open Access



Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya

Nyawira Mwangi^{1,2*}, David Macleod¹, Stephen Gichuhi³, Lawrence Muthami⁴, Consuela Moorman⁵, Covadonga Bascaran¹ and Allen Foster¹

Abstract

Background: Diabetic retinopathy (DR) is a significant public health concern that is potentially blinding. Clinical practice guidelines recommend annual eye examination of patients with diabetes for early detection of DR. Our aim was to identify the demand-side factors that influence uptake of eye examination among patients already utilizing diabetes services in three counties of Kenya.

Methods: We designed a clinic based cross-sectional study and used three-stage sampling to select three counties, nine diabetes clinics in these counties and 270 patients with diabetes attending these clinics. We interviewed the participants using a structured questionnaire. The two outcomes of interest were 'eye examination in the last 12 months' and 'eye examination ever'. The exposure variables were the characteristics of participants living with diabetes.

Results: The participants had a mean age of 53.3 years (SD 14.1) and an average interval of 4 months between visits to the diabetes clinic. Only 25.6% of participants had ever had an eye examination in their lifetime, while 13.3% had it in the preceding year. The independent predictors of uptake were referral by diabetes services, patient knowledge of diabetes eye complications, comorbid hypertension and urban or semi-urban residence.

Conclusions: We conclude that access to retinal examination for DR is low in all three counties. An intervention that increases the knowledge of patients with diabetes about eye complications and promotes referral of patients with diabetes for eye examination may improve access to annual eye examination for DR.

Keywords: Diabetes, Diabetic retinopathy, Eye examination, Access, Screening, Kenya, Sub-Saharan Africa

Background

Diabetes mellitus (DM) causes visual impairment and blindness through diabetic eye disease, which includes cataract and diabetic retinopathy. Diabetic retinopathy (DR) is a progressive microvascular complication of diabetes. Approximately one third (34.6%) of people living with diabetes (PLWD) have DR and 10% have sightthreatening DR (STDR) [1]. The increasing magnitude of DM and DR is a significant public health challenge [2, 3]. There is strong evidence for the cost-effectiveness of screening for DR in prevention of blindness [4, 5].

* Correspondence: Nyawira.Mwangi@lshtm.ac.uk

¹London School of Hygiene and Tropical Medicine, London, UK

²Kenya Medical Training College, Nairobi, Kenya

Full list of author information is available at the end of the article



© The Author(s). 2017 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.



are at risk of developing STDR and needing treatment [10, 11]. Third, treatment of patients who have STDR reduces the risk of vision loss [12–14].

The determinants of access to retinal examination are complex and include both supply and demand factors [9, 15]. Understanding the demand-side factors facilitates the development of targeted demand-side interventions that reduce the barriers and support the enablers to increase the uptake of eye examination. Several studies have examined the use of eye care among patients with diabetes in America, Asia, Europe, and Oceania [16–21]. Many studies in Africa have focused on access to eye care for cataract but not DR [22–30]. In this paper, we report on factors influencing the uptake of eye examination for DR in PLWD. We define this test as a retinal examination through a dilated pupil conducted by an eye care worker using either an ophthalmoscope or retinal camera.

Research in context panel

Evidence before this study

We searched Ovid MEDLINE, Cochrane Library, and EMBASE (2000–2016) using the terms 'diabetes' and 'diabetic retinopathy' in combination with the following terms: 'access', 'screening' and 'eye examination'. We also searched cited references in articles identified by this search strategy. The evidence is that uptake of annual retinal examination is low in resource-poor settings (Table 1). However, the predictors of uptake of retinal examination have not been documented.

Added value of this study

We found the uptake of retinal examination among patients utilising diabetes services in three counties of Kenya to be even lower than documented in other studies. Predictors of uptake of retinal examination were (a) referral from diabetes services, (b) knowledge of diabetes eye complications and (c) comorbid hypertension. About half of the patients had the perception that a retinal

Table 1 Summary of other studies in developing countries

examination was not necessary in the absence of ocular symptoms. Using this evidence, we present a conceptual model on how to improve uptake of retinal examination.

Implications of all the available evidence

An intervention to reverse low uptake of retinal examination should include both health education and referral pathway interventions. From our findings, the education component should prioritize two aspects of knowledge: (1) information on diabetes eye complications and (2) information on eye examinations (importance and frequency). The referral intervention should address barriers to uptake of examination. These interventions are potentially cost-effective and may also strengthen integration of diabetic retinopathy screening into diabetes services.

This study was conducted when Kenya has just completed a STEPwise survey on risk factors for noncommunicable diseases and determined the prevalence of DM. It could form the baseline from which trends in uptake of retinal examination can be compared as prevalence of DM increases in the next decade.

Methods

Study setting

This study was part of a cross-sectional health system assessment for diabetes and diabetic retinopathy. A three-stage sampling process was used. Three counties were purposively selected to represent different environments and populations within the diabetes belt in Kenya: Kirinyaga (predominantly rural), Nakuru (semi-urban) and Nairobi (urban). Three diabetes outpatient clinics were selected in each county. A list of public, private and faith-based clinics in each country was obtained, and 1 clinic was selected in each category through random sampling. In each of these nine diabetes clinics, 30 patients were selected by random sampling from the PLWD attending the clinic on the day of interview. The list of male and female patients was used as the

	/	1	2					
Study	Current study	Mumba et al. [32]	Onakpoya et al. [36]	Njambi, L [33]	Adriono et al. [19]	Wang et al. [20]	Shivashankar et al. [34]	GV Murthy et al. [35]
Country	Kenya	Tanzania	Nigeria	Kenya	Indoneshia	China	Delhi, India	11 cities, India
Year	2016	2009	2009	2012	2011	2010	2016	20
Target PLWD population	Adults in nine diabetes clinics	Adults in one diabetes clinic in a tertiary hospital	Adults in one diabetes clinic	Adults attending a diabetes clinic in one hospital	Adults in three clinics	Adults attending health facilities	Adults attending 23 primary care clinic	Adults attending diabetes hospitals/clinics
Sample size	270	316	84	253	196	824	406	285
Screening rate (last 12 months)	13.3%	28%	Not reported	Not reported	15.3%	33.3%	7.4%	Not reported
Screening rate (ever)	25.6%	59.1%	28.9%	29%	Not reported	56.8%	Not reported	67.7%

sampling frame, with a random starting point and a regular sampling interval of between three and five depending on the volume of the patients attending each clinic. This procedure made it possible to recruit an equal number of men and women. A minimum sample size of 73 per county (thereafter increased to 90) was determined based on the estimate that 5% of the population of PLWD attending diabetes clinics have an annual dilated eye examination, with the desirable degree of accuracy set at 0.05.

The study followed the tenets of the World Medical Association's Declaration of Helsinki. The London School of Hygiene and Tropical Medicine and African Medical Research Foundation granted ethical approval. All participants gave written informed consent.

Participants

Eligible persons included those 18 years of age or older, known to have diabetes, resident in the county, receiving services at participating outpatient diabetes clinics, and willing to participate in the study. Non-residents in the county and acutely ill patients were excluded.

The primary investigator and research assistants interviewed the participants in English or Kiswahili using a pretested structured questionnaire. Prior to data collection, the questionnaire was reviewed by local diabetologists, ophthalmologists and statisticians. A pilot test with diabetes patients was conducted in two different diabetes clinics within the study area (which were not part of the study sample).

Participation was voluntary and participants did not receive any financial incentives. The questionnaire had four broad categories of questions for PLWD: (a) sociodemographic characteristics, (b) experience with diabetes services, (c) knowledge of complications of diabetes and (d) experience with examination for complications of diabetes (including DR). All subjects reporting previous eye examinations were questioned as to whether the eye care worker instilled eye drops to dilate the pupils before the eye examination. This differentiated a regular eye examination and a dilated eye examination.

Statistical analysis

STATA version 14 was used for data analysis [31]. The study had two outcomes of interest: 'eye examination in the last 12 months' and 'eye examination ever'. Both were dichotomous 'yes' and 'no' variables. The exposure variables were characteristics of participants in relation to living with diabetes.

Descriptive statistics were shown as counts and percentages for categorical variables, and means and standard deviations for continuous variables. For each of the two outcomes of interest, tests of crude association were performed using chi-square tests for categorical exposure variables and *t* tests for continuous variables. Univariate logistic regression was used to identify exposure variables that were predictors of uptake of examination in the last year, and in analysing the ever had an eye exam outcome, all logistic regressions were adjusted for age. Multivariable analysis was performed using forward stepwise selection where exposure variables with the lowest *p* value were sequentially added to the regression model, using a cutoff for inclusion in the model of *p* < 0.05.

Role of funding source

The funders did not participate in study design, data collection, analysis, writing of the paper or submission for publication.

Results

Outcome variables: uptake of dilated eye examination

Ninety participants were interviewed in each county (n = 270). None of the participants declined to participate, and data for all variables was collected for all participants. Only 25.6% (n = 69) had ever had fundoscopy, while only 13.3% (n = 36) had been examined in the preceding year. The uptake of eye examination in other resource-poor settings is shown in Table 1.

Exposure variables: participant characteristics

The mean age of participants was 52.3 years (SD 14.1, range 25–88 years). Approximately 47% were male, 23.7% had a family history of diabetes and 37.4% had comorbid hypertension. The mean duration of diabetes was 7.3 years (SD 5.5), and participants attend the diabetes clinic every 4 months (SD 1.5) on average. The main reason for that frequency is the physician's recommendation. The other variables are shown on the first column of Table 2.

Determinants of eye examination

Table 2 also shows the patient-level determinants for fundoscopy. Only 24.4% had been referred from the diabetes clinic for a retinal examination, and 13.3% had taken this examination (fundoscopy) in the last 12 months. Variables that had the strongest evidence of an association with having had the exam in the last 12 months were (a) referral for an eye examination (p < p0.001), (b) knowledge of diabetes eye complications (p =0.002), (c) comorbid hypertension (p = 0.02) and (d) county of residence (p = 0.07) (Table 3). Participants referred for an eye exam had almost eight times the odds of having attended an eye exam in the last 12 months compared to those who had not been referred (OR 7.9, 95% CI 3.7-16.4, p < 0.001). Participants who had a knowledge of diabetes eye complications had four times the odds (OR 3.9, CI 1.6–9.1) of attending as those who had no knowledge of eye complications. Hypertensive

Table 2 Patient characteristics and association with eye examination

Variable	Summary of participants characteristics	Retinal exam last 12 months			Retinal exam ever		
	N (%) Mean (SD)	Had eye exam	No eye exam	p value	Had eye exam	No eye exam	p value
Number (%) in each category	270	36 (13.3%)	234 (86.3%)		69 (25.6)	201 (74.4)	
Number (%) by county				0.07			0.002
Kirinyaga	90	6 (6.7%)	84 (93.3%)		11 (12.2)	79 (87.8)	
Nairobi	89	14 (15.7%)	75 (84.3%)		29 (32.6)	60 (67.4)	
Nakuru	91	16(17.6%)	75 (82.4%)		29 (31.9)	62 (68.1)	
Age (mean years, SD)	53.3 (14.1)	57.1 (11.7)	52.7 (14.4)	0.08	60.5 (13.8)	50.8 (13.4)	< 0.0001
Sex (no. %)				0.7			0.5
Men	127 (47%)	18 (14.2)	109 (85.8)		35 (27.6)	92 (72.4)	
Women	144 (53%)	18 (12.5)	126 (87.5)				
Literacy				0.3			0.05
Primary or below	88 (32.8%)	13 (14.8)	75 (85.2)		30 (34.1)	58 (65.9)	
Secondary	111 (41.4%)	11 (9.9)	100 (90.1)		21 (18.9)	90 (81.1)	
Post-secondary	69 (25.8%)	12 (17.4	57 (82.6)		18 (26.1)	51 (73.9)	
Occupation				0.4			0.014
Unemployed	70 (25.9%)	6 (3.6)	64 (91.4)		19 (27.1)	51 (72.9)	
Low skilled	70 (25.9%)	9 (12.9)	61 (87.1)		14 (20)	56 (80)	
Professional	90 (33.3%)	13 (14.4)	77 (85.6)		18 (20)	72 (80)	
Retired	40 (33.3%)	8 (20)	32 (80)		18 (45)	22 (55)	
Duration of diabetes (mean years, SD)	7.3 (5.5)	8.9 (4.5)	7.1 (5.6)	0.06	9.4 (5.5)	6.6 (5.3)	0.0002
Interval of diabetes clinic visits (months)	4.0 (1.5)	4.3 (1.3)	4.0 (1.5)	0.4	4.3 (1.4)	3.9 (1.5)	0.08
Referred for eye examination				< 0.001			< 0.001
Yes	66 (24.4%)	23 (34.9)	43 (65)		47 (68.1)	19 (28.8)	
No	204 (75.6%)	13 (6.4)	191 (93.6)		22 (10.7))	182 (89.2)	
Perceived level of glucose control				0.02			0.4
Very good	10 (3.7%)	0	10 (100)		2 (20)	8 (80)	
Well	73 (27%)	17 (23.3)	56 (76.7)		23 (31.5)	50 (68.5)	
Adequate	107 (39.6%)	9 (8.4)	98 (91.6)		24 (22.4)	83 (77.6)	
Poor	68 (25.2%)	10 (14.7)	58 (85.3)		19 (27.9)	49 (72.1)	
Very poor	12 (4.4%)	0	12 (100)		1 (8.3)	11 (91.7)	
Diabetes in family member				0.8			0.6
Yes	64 (23.7%)	8 (12.5)	56 (87.5)		18 (28.1)	46 (71.9)	
No	206 (76.3%)	28 (13.6)	178 (86.4)				
Information on diabetes given at health facility				0.3			0.8
Yes	205 (75.9%)	30 (14.6)	175 (85.4)		53 (25.9)	152 (74.2)	
No	65 (24.1%)	6 (9.2)	59 (90.8)		16 (24.6)	49 (75.4)	
Knowledge of diabetes complications				0.4			0.9
Yes	103 (38.1%)	16 (15.5)	87 (84.5)		26 (25.2)	77 (74.8)	
No	167 (61.9%)	20 (12)	146 (88)		43 (25.8)	124 (74.3)	
Knowledge of diabetes eye complications				0.001			0.001
Yes	150 (55.6%)	29 (19.3)	121 (80.7)		50 (33.3)	100 (66.7)	
No	120 (44.4%)	7 (5.8)	113 (94.2)		19 (15.8)	101 (84.2)	

Variable	Summary of participants characteristics	Retinal exam last 12 months			Retinal exam ever		
	N (%) Mean (SD)	Had eye exam	No eye exam	p value	Had eye exam	No eye exam	p value
Comorbid hypertension				0.02			0.04
Yes	101 (37.4%)	20 (19.8)	81 (80.2)		33 (32.7)	68 (67.3)	
No	169 (62.6%)	16 (9.5)	153 (90.5)		36 (21.3)	133 (78.7)	
Opinion on need for an eye examination				<i>P</i> < 0.001			P < 0.001
No need	51 (18.9%)	1 (2.0)	50 (98)		6 (11.8)	45 (88.2)	
Only for ocular symptoms	115 (42.6%)	15 (13)	100 (87)		27 (23.5)	88 (76.5)	
Acceptable	80 (29.6%)	13 (16.3)	67 (83.8)		25 (28.8)	57 (71.3)	
Already doing it	9 (3.3%)	5 (55.6)	4 (44.4)		9 (100)	0	
Other opinion	15 (5.6%)	13 (13.3)	2 (86.7)		4 (26.7)	11 (73.3)	

Table 2 Patient characteristics and association with eye examination (Continued)

individuals had twice the odds of attendance, compared to those with normal blood pressure (OR 2.3, CI 1.1–4.7). The PLWD in Kirinyaga (rural) were the least likely to have had an eye examination in the last 12 months, with PLWD in Nairobi (urban) having 2.6 times the odds (CI 1.1–7.1) and PLWD in Nakuru (semi-urban) having three times the odds (CI 1.1–8.0).

The main predictors for having *ever* had fundoscopy included (a) referral for eye examination (OR 20.5, CI 10.2–40.9, p < 0.001), (b) knowledge of diabetes eye complications (OR 2.7, CI 1.5–4.8, p < 0.001), (c) county (p = 0.02) and (d) comorbid hypertension (OR 1.8 CI 1.0–3.1 p = 0.02). The PLWD in Nakuru or Nairobi had three times the odds of attendance as

compared in Kirinyaga (OR 3.4, CI 1.6–7.5 and OR 3.5, CI 1.6–7.5) (Table 3).

There was strong evidence of association of having a dilated eye examination (ever) with both increasing age and duration of diabetes (p < 0.0001), but the effect size was quite small, with the odds increasing by 1.1 times each year (thus, 2.6 times every decade), Table 3. In multivariable analysis, (a) referral, (b) knowledge of diabetes eye complications and (c) county of residence remained independent predictors for fundoscopy. Referral and knowledge of diabetes eye complications had the strongest relationship with uptake of eye examination and thus were included in the final multivariable analysis model. Interaction between referral and knowledge of

Table 3 Predictors of eye examination last 12 months

Variable	Eye exam last 12 mor	nths	Eye exam ever		
	OR (95% CI)	p value	OR (95% CI)	<i>p</i> value	
Demographic factors					
Increasing age (every year)	1.2 (1.1–1.6)	0.08	1.1 (1.0–1.1)	< 0.001	
Male gender	1.1 (0.6–2.3)	0.7	1.2 (0.7–2.1)	0.5	
County of residence (compared to Kirinyaga)					
Nakuru	3.0 (1.1-8.0)	0.03	3.4 (1.6–7.5)	0.02	
Nairobi	2.6 (1.1–7.1)	0.06	3.5 (1.6–7.5)	0.02	
Education					
Post-secondary education	1.1 (0.5–2.8)	0.8	0.7 (0.3–1.4)	0.3	
Occupation (as compared to the unemployed)				
Professional	1.8 (0.6–5.0)	0.3	0.7(0.3–1.5)	0.3	
Retired	2.7 (0.9–8.3)	0.09	2.2 (1.0-5.0)	0.06	
Duration of diabetes	1.1 (1.0–1.1)	0.06	1.0 (1.0–1.1)	< 0.001	
Referral for eye examination	7.9 (3.7–16.4)	< 0.001	20.5 (10.2–40.9)	< 0.001	
Knowledge of diabetes complications	3.9 (1.6–9.2)	0.002	2.7 (1.5–4.8)	0.001	
Comorbid hypertension	2.3 (1.1–4.7)	0.02	1.8 (1.0–3.1)	0.04	

diabetes eye complications was tested, and these remained significant independent predictors (p < 0.0001).

As referral for examination (ever) was the strongest predictor of uptake of examination, the variables associated with referral were analysed. The main exposure variables positively associated with referral (Table 4) were (a) increasing age (p < 0.0001), (b) longer duration of diabetes (p = 0.0005), (c) knowledge of diabetes eye complications (p = 0.003), (d) positive opinion on need for an eye examination (p < 0.001), (e) retirement (p = 0.01) and (f) residence in Nairobi or Nakuru (p = 0.03).

For the 109 (40.4%) who had knowledge that diabetes causes complications, the complications that were of concern were losing a leg 34%, kidney failure 31.2%, stroke 22% and blindness 9%. Although 150 (55. 6%) knew that diabetes can affect the eye, 18.9% of the participants felt that there was no need for an eye

examination and 42.6% would only go for an examination if they developed ocular symptoms.

Discussion

The results indicated that both initiation and maintenance of annual fundoscopy is low. This may be due to the lack of systematic DR screening programmes in the country. Similar findings have been documented in other resource-poor settings (Table 1) [19, 20, 32–36]. The findings can be generalised to examination for DR in adult PLWD populations in Kenya since the study included any PLWD above 18 years in three geographical locations representing the rural-urban continuum within the diabetes belt. The lowest uptake was in Kirinyaga, suggesting that the macro environment in which PLWD live is a determinant of uptake [15]. Referral by the diabetic clinic for an eye examination, positive opinion on need for an eye examination and knowledge of diabetes

Table 4 Variables associated with referral for eye examination

Variable	Referred	Not referred	<i>p</i> value
Number (%) in each category	66 (24.4)	204 (75.6)	
Number (%) by county			0.03
Kirinyaga	15 (16.7)	75 (83.3)	
Nairobi	30 (33.8)	59 (66.3)	
Nakuru	21 (23.1)	70 (76.9)	
Age mean years, SD	59.8 (13.3)	51.2 (13.8)	< 0.0001
Sex <i>N</i> (%)			0.09
Male	37 (29.1)	90 (70.9)	
Female	29 (20.3)	114 (79.7)	
Occupation N (%)			0.01
Unemployed	17 (24.3)	53 (75.7)	
Low skilled	13 (18.6)	57 (81.4)	
Professional	18 (20)	72 (80)	
Retired	18 (45)	22 (55)	
Literacy N (%)			0.6
Primary or below	24 (27.3)	64 (72.7)	
Secondary education	24 (21.6)	87 (78.4)	
Post-secondary education	18 (26.1)	51 (73.9)	
Duration of diabetes years: mean, SD	9.3 (5.4)	6.6 (5.4)	0.0005
Diabetes in family member N (%)			
Yes	16 (25)	48 (75)	0.9
No	50 (24.3)	156 (75.7)	
Comorbid high BP N (%)			
Yes	31 (30.7)	70 (69.3)	0.07
No	35 (20.7)	134 (79.3)	
Knowledge of diabetes eye complications N (%)			0.003
Yes	47 (31.3)	133 (68.7)	
No	19 (28.8)	101 (84.2)	

eye complications are the modifiable factors that were positively associated with uptake of examination.

Sociodemographic attributes of patients were found to affect uptake of examination. The heterogeneity by county reflects geographic, social, cultural and/or economic influences. Rural populations are known to have low access to screening services [20]. This could be related (in part) to a rural-urban gap in awareness, resources or empowerment [37]. Paksin-Hall et al. [38] found income level, education level and health insurance status to be important determinants of annual dilated eye examinations, but these were not significant independent predictors in this study.

In previous studies, increasing age was a predictor for having an eye examination [39, 40]. In our study, the evidence for this association with strong for eye examination ever (p < 0.0001) and weak for an examination in the last year (p = 0.08). Although the effect size was small, the findings of an association are consistent with an increased likelihood of examination with age. Given that the risk of developing DR increases with age, older adults, more than any other age group, need to have regular eye examination, and as the population is aging, an expanding need for retinal examination in the country is predictable. Duration of diabetes is an important predictor for incidence and prevalence of DR, [40–42] so as more people live longer with diabetes, the need for an annual eye examination will increase.

Gender was not a predictor of uptake of examination, although there was very weak evidence that male gender was a predictor for referral (p = 0.09). A positive family history of diabetes was similarly not a predictor of uptake of examination, which suggests that barriers to access are not just at the individual level but also within households [15].

Hypertension in PLWD was a positive predictor of uptake of eye examination in this study, as also reported in another study [19]. Comorbidity is known to increase health care utilization in diabetes, [43] and hypertension is a common vascular comorbidity [11, 33, 35, 37, 40]. Uncontrolled hypertension is a risk factor for development of DR. There was weak evidence that PLWD with hypertension were more likely to be referred (p = 0.07) than normotensive PLWD, perhaps because the diabetes is considered more severe. This association strengthens the case for integration of eye care into noncommunicable disease care.

There was very strong evidence that knowledge of any diabetes eye complication increases the uptake of examination (Table 3). Other studies have also found that knowledge is a predictor for uptake of screening [19, 20, 42]. However, in this study, only 9% listed blindness as a complication that they were concerned about. Another finding in this study is that PLWD need knowledge

about the necessity and the frequency of eye examinations. Nearly half (42.6%) of the participants thought that DR screening should be symptom-led, which is a misconception that can lead to delay in getting an examination and treatment resulting in visual loss. Educational messages need to be tailored to an awareness of eye complications from diabetes and the need for diabetics to have the eyes examined once a year. This tallies with the finding that the most frequently reported suggestion for improvement given by PLWD was the need for more information/education. Thus, there exists a real opportunity for demand-driven health education.

Although there was strong evidence that knowledge of diabetes eye complications is a predictor of examination, there exists a gap between possessing this knowledge and the uptake of examination. About 56% of PLWD knew that diabetes causes eye complications, but only 25% of all PLWD had ever received an eye examination. Similarly, although approximately 25% were given a referral, only 13% had actually gone for the examination in the last year. This suggests that there are additional factors besides knowledge and referral that influence uptake.

The health belief model (HBM) is a widely used theoretical framework for understanding health behaviours within public health. Weiss et al. have previously shown that behavioural interventions can improve uptake of eye examination [44]. Taking the predictors found in this study into consideration, and using HBM as a theoretical framework, we conceptualise that self-efficacy is on the pathway between knowledge, referral and uptake of examination (Fig. 1). Research has shown that health behaviours such as taking an eye examination are associated with self-efficacy. In turn, self-efficacy can be increased in four ways: performance accomplishment, vicarious experience, verbal persuasion and psychological cues [45]. We postulate that interventions that increase knowledge, referral and self-efficacy can increase uptake of eye examination. Our conceptual model captures these different aspects (Fig. 1).

Only a quarter of PLWD had received a referral for DR screening. Similarly, in other studies in China and India, less than a half had been referred [19, 20], al-though in one study in India, over 60% had a referral [35]. We found that the strongest predictor for having an eye examination was referral from diabetes services. Participants already attend the diabetes clinic every 4 months because of the recommendation of the diabetes services. As there is no systematic DR screening programme, a referral to the eye clinic is a crucial bridge. These three visits a year are missed opportunities for referral for eye examination. Lack of a diabetes provider's recommendation has been documented as a barrier in Germany [9] and Paraguay [42], diabetes services being gate keepers to other services required by PLWD.



Written communication from the patient's ophthalmologist to the primary care provider has also been found to increase adherence to future dilated eye examination [46]. Conversely, as entry to the eye clinic in Kenya does not actually require a formal referral note from diabetes services, an intervention that empowers patients for 'self-referral' might increase uptake of the examination.

There was strong evidence that older people with diabetes, those with longer duration of diabetes and those with a knowledge of diabetes eye complications were more likely to be referred. There was evidence that PLWD in Kirinyaga were less likely to be referred than those in Nakuru or Nairobi. Interventions to strengthen the use of clinical guidelines can ensure that all PLWD get a referral for an annual eye check.

Limitations

This study has several limitations. First, as this is a cross-sectional study, a temporal relationship cannot be established between the predictor factors and the uptake of screening. In addition, the association between the variables is still subject to residual confounding by unmeasured variables such as distance from home to the eye clinic, medical conditions such as depression, disability and membership of diabetes support groups. Self-reported data was used and is prone to recall bias and social desirability bias. Under reporting of health behaviours, such as the duration since the participant had the last eye examination, may introduce information bias. This is a clinic-based study and did not include PLWD

not attending diabetes services; however, we presume that they would have an even lower uptake of screening for DR.

Conclusions

There is poor compliance with recommendations for annual eye examination among PLWD who have access to diabetes services. An intervention targeted at motivating adherence is essential. Such an intervention should empower PLWD to request/demand an eye examination and strengthen knowledge, referral and self-efficacy.

The opportunity to increase uptake of eye examination is also a valuable avenue for integrating diabetes care and eye care. Programmes to increase awareness regarding the importance of eye examinations can be combined with interventions to improve blood pressure monitoring and other aspects of diabetes management.

Implications

Our study has demonstrated the low uptake of screening for DR by PLWD and described the attributes associated with uptake of eye examination. Low uptake has adverse effects at individual level and at the health system level because of the associated increased risk of blindness from DR. The low uptake highlights barriers in the link between diabetes services and eye care services. There is need to integrate screening for DR within the routine diabetes services and to implement interventions to increase uptake of screening.

Future work

As the burden of diabetes grows over the next decade, there is a need to investigate the trend in uptake of annual eye examination and to examine sustainable interventions that can maximise increase uptake for eye examination. There is also need to investigate why there is a lack of attention to DR screening among diabetes clinicians and to evaluate the effect of providing them with clinical decision-making tools such DR guidelines.

Abbreviations

CI: Confidence interval; DM: Diabetes mellitus; DR: Diabetic retinopathy; HBM: Health Belief Model; PLWD: Persons living with diabetes; SD: Standard deviation; STDR: Sight-threatening diabetic retinopathy

Acknowledgements

The Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium funded the study and the publication. Lindsay Hampejsková reviewed the manuscript before submission.

Funding

The funding for this study was provided by The Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium.

Availability of data and materials

The data that support the findings of this study are available from the authors, but restrictions apply to the availability of these data, which were used under license for the current study, and so, are not publicly available. Data are however available from the authors upon reasonable request.

Authors' contributions

NM, CB and AF designed the study, which was then reviewed critically by SG, DM, CM and LM. NM and LM obtained the data and analysed it with DM. All authors were involved in the data interpretation. NM had full access to the data. NM drafted the manuscript. All authors reviewed it and approved the final version.

Ethics approval and consent to participate

The ethics review committees of the London School of Hygiene and Tropical Medicine and the African Medical Research Foundation granted ethics approval for the study. All participants gave informed consent for participation.

Consent for publication

Not applicable

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹London School of Hygiene and Tropical Medicine, London, UK. ²Kenya Medical Training College, Nairobi, Kenya. ³University of Nairobi, Nairobi, Kenya. ⁴Kenya Medical Research Institute, Nairobi, Kenya. ⁵Oxford University Hospitals NHS Trust, Oxford, UK.

Received: 10 October 2017 Accepted: 28 November 2017 Published online: 21 December 2017

References

- Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556–64.
- Zheng Y, He M, Congdon N. The worldwide epidemic of diabetic retinopathy. Indian J Ophthalmol. 2012;60(5):428–31.

- Burgess PI, MacCormick IJ, Harding SP, Bastawrous A, Beare NA, Garner P. Epidemiology of diabetic retinopathy and maculopathy in Africa: a systematic review. Diabet Med. 2013;30(4):399–412.
- Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. Diabet Med. 2010;27(3):249–56.
- Sloan Frank A, Grossman Daniel S, Lee PP. Effects of receipt of guidelinerecommended care on onset of diabetic retinopathy and its progression. Ophthalmology. 2009;116:1515–21. e3
- 6. International Diabetes Federation. IDF diabetes atlas 7th edition. 2015.
- Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Nairobi: Ministry of Health, Division of Non-Communicable Diseases; 2015.
- 8. Ning C, Paul M, Wong TY. Diabetic retinopathy. Lancet. 2010;376:124-36.
- Baumeister SE, Schomerus G, Andersen RM, Tost F, Markus MR, Volzke H, et al. Trends of barriers to eye care among adults with diagnosed diabetes in Germany, 1997–2012. Nutr Metab Cardiovasc Dis. 2015;25(10):906–15.
- International Council of Ophthalmology. ICO guidelines for diabetic eye care—updated 2017. San Francisco, California: International Council of Ophthalmology, 2017 23 February 2017. Report No.
- American Diabetes Association. Standards of medical care in diabetes— 2017. Diabetes Care. 2017;40(Suppl 1):S91-S93.
- Evans JR, Michelessi M, Virgili G. Laser photocoagulation for proliferative diabetic retinopathy. Cochrane Database Syst Rev. 2014;24(11):Cd011234. doi:10.1002/14651858.CD011234.pub2.
- Grover DA, Li T, Chong CCW. Intravitreal steroids for macular edema in diabetes. Cochrane Database Syst Rev. 2008;23(1):CD005656. doi:10.1002/ 14651858.CD005656.pub2.
- Virgili G, Parravano M, Menchini F, Evans JR. Anti-vascular endothelial growth factor for diabetic macular oedema. Cochrane Database Syst Rev. 2014;24(10): CD007419. doi:10.1002/14651858.CD007419.pub4.
- Levesque JF, Harris MF, Russell G. Patient-centred access to health care: conceptualising access at the interface of health systems and populations. Int J Equity Health. 2013;12:18.
- Dickey H, Ikenwilo D, Norwood P, Watson V, Zangelidis A. Utilisation of eyecare services: the effect of Scotland's free eye examination policy. Health Policy. 2012;108:286–93.
- Maclennan PA, Mcgwin G, Heckemeyer C, Lolley VR, Hullett S, Saaddine J, et al. Eye care utilization among a high-risk diabetic population seen in a public Hospital's clinics. JAMA Ophthalmol. 2014;132:162–7.
- Lee PP, Feldman ZW, Ostermann J, Brown DS, Sloan FA. Longitudinal rates of annual eye examinations of persons with diabetes and chronic eye diseases. Ophthalmology. 2003;110:1952–9.
- Adriono G, Wang D, Octavianus C, Congdon N. Use of eye care services among diabetic patients in urban Indonesia. Arch Ophthalmol (Chicago, III : 1960). 2011;129:930–5.
- Wang D, Ding X, He M, Yan L, Kuang J, Geng Q, et al. Use of eye care services among diabetic patients in urban and rural China. Ophthalmology. 2010;117(9):1755–62.
- Glasson NM, Larkins SL, Crossland LJ. What do patients with diabetes and providers think of an innovative Australian model of remote diabetic retinopathy screening? A qualitative study. BMC Health Serv Res. 2017;17:158.
- 22. Aboobaker S, Courtright P. Barriers to cataract surgery in Africa: a systematic review. Middle East Afr J Ophthalmol. 2016;23(1):145–9.
- Kessy JP, Lewallen S. Poverty as a barrier to accessing cataract surgery: a study from Tanzania. Br J Ophthalmol. 2007;91(9):1114–6.
- 24. Briesen S, Roberts H, Ilako D, Karimurio J, Courtright P. Are blind people more likely to accept free cataract surgery? A study of vision-related quality of life and visual acuity in Kenya. Ophthalmic Epidemiol. 2010;17:41–9.
- Briesen S, Geneau R, Roberts H, Opiyo J, Courtright P. Understanding why patients with cataract refuse free surgery: the influence of rumours in Kenya. Trop Med Int Health. 2010;15:534–9.
- Geneau R, Lewallen S, Bronsard A, Paul I, Courtright P. The social and family dynamics behind the uptake of cataract surgery: findings from Kilimanjaro region, Tanzania. Br J Ophthalmol. 2005;89:1399–402.
- Geneau R, Massae P, Courtright P, Lewallen S. Using qualitative methods to understand the determinants of patients' willingness to pay for cataract surgery: a study in Tanzania. Soc Sci Med. 2008;66:558–68.
- 28. Abubakar T, Gudlavalleti MV, Sivasubramaniam S, Gilbert CE, Abdull MM, Imam AU. Coverage of hospital-based cataract surgery and barriers to the

uptake of surgery among cataract blind persons in Nigeria: the Nigeria National Blindness and Visual Impairment Survey. Ophthalmic Epidemiol. 2012;19(2):58–66.

- Lewallen S, Roberts H, Hall A, Onyange R, Temba M, Banzi J, et al. Increasing cataract surgery to meet vision 2020 targets: experience from two rural programmes in east Africa. Br J Ophthalmol. 2005;89:1237–40.
- Syed A, Polack S, Eusebio C, Mathenge W, Wadud Z, Mamunur AK, et al. Predictors of attendance and barriers to cataract surgery in Kenya, Bangladesh and the Philippines. Disabil Rehabil. 2013;35(19):1660–7.
- StataCorp. Stata statistical software: release 14. College station, TX: StataCorp LP; 2015.
- Mumba M, Hall A, Lewallen S. Compliance with eye screening examinations among diabetic patients at a Tanzanian referral hospital. Ophthalmic Epidemiol. 2007;14(5):306–10.
- Njambi L. Prevalence of diabetic retinopathy and barriers to uptake of diabetic retinopathy screening at Embu Provincial General Hospital, Central Kenya. East Afr J Ophthalmol. 2012;16:5–11.
- Shivashankar R, Bhalla S, Kondal D, Ali MK, Prabhakaran D, Narayan KV, et al. Adherence to diabetes care processes at general practices in the National Capital Region-Delhi, India. Indian J Endocrinol Metabol. 2016;20(3):329–36.
- MVS G, Anchala R, Gudlavalleti AS, Ramachandra SS, Shukla R, Jotheeswaran AT, et al. Perceptions and practices related to diabetes reported by persons with diabetes attending diabetic care clinics: the India 11-city 9-state study. Indian J Endocrinol Metabol. 2016;20(Suppl 1):S26–32.
- Onakpoya Oluwatoyin Helen, Adeoye Adenike Odunmorayo, Kolawole Babatope Ayodeji. Determinants of previous dilated eye examination among type II diabetics in Southwestern Nigeria. Eur J Int Med. 2010;21:176–9.
- Kyari F, Tafida A, Sivasubramaniam S, Murthy GV, Peto T, Gilbert CE, et al. Prevalence and risk factors for diabetes and diabetic retinopathy: results from the Nigeria national blindness and visual impairment survey. BMC Public Health. 2014;14:1299.
- Paksin-Hall A, Dent ML, Dong F, Ablah E. Factors contributing to diabetes patients not receiving annual dilated eye examinations. Ophthalmic Epidemiol. 2013;20(5):281–7.
- Bastawrous A, Mathenge W, Wing K, Bastawrous M, Rono H, Weiss HA, Macleod D, Foster A, Peto T, Blows P, Burton M, Kuper H. The incidence of diabetes mellitus and diabetic retinopathy in a population-based cohort study of people age 50 years and over in Nakuru, Kenya. BMC Endocr Disord. 2017;17(19). ISSN 1472-6823. doi:10.1186/s12902-017-0170-x.
- Cleland CR, Burton MJ, Hall C, Hall A, Courtright P, Makupa WU, et al. Diabetic retinopathy in Tanzania: prevalence and risk factors at entry into a regional screening programme. Trop Med Int Health. 2016;21:417–26.
- Mathenge W, Bastawrous A, Peto T, Leung I, Yorston D, Foster A, et al. Prevalence and correlates of diabetic retinopathy in a population-based survey of older people in Nakuru, Kenya. Ophthalmic Epidemiol. 2014;21(3):169–77.
- 42. Cano MR. Prevalence of diabetic retinopathy and barriers to uptake of eye care services by diabetic patients at the Social Security Institute Central Hospital in Asunción, Paraguay. Commun Eye Health J. 2007;20(61):10. 1
- Struijs JN, Baan CA, Schellevis FG, Westert GP, van den Bos GA. Comorbidity in patients with diabetes mellitus: impact on medical health care utilization. BMC Health Serv Res. 2006;6:84.
- 44. Weiss DM, Casten RJ, Leiby BE, Hark LA, Murchison AP, Johnson D, et al. Effect of behavioral intervention on dilated fundus examination rates in older African American individuals with diabetes mellitus: a randomized clinical trial. JAMA Ophthalmol. 2015;133(9):1005–12.
- Bandura, A. Self-efficacy. In V. S. Ramachaudran (Ed.), Encyclopedia of human behavior (Vol. 4, pp. 71-81). New York: Academic Press. (Reprinted in H. Friedman [Ed.], Encyclopedia of mental health. San Diego: Academic Press; 1998).
- Storey PP, Murchison AP, Pizzi LT, Hark LA, Dai Y, Leiby BE, et al. Impact of physician communication on diabetic eye examination adherence: results from a retrospective cohort analysis. Retina. 2016;26(1):20–7.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research



Additional notes on methods reported in Paper 1

The study was conducted in three counties that vary in size and demographic characteristics. Disproportionate equal allocation was used in sampling, with an equal number of facilities and participants being recruited for the study in each of the counties. Proportionate sampling was not feasible as we do not have the county-level data on prevalence of diabetes and distribution of PLWD which would be required to calculate the sample size in each county.

Additional information on Table 2* in Paper 1

*Mwangi, N., Macleod, D., Gichuhi, S. et al. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Trop Med Health **45**, 41 (2017). <u>https://doi.org/10.1186/s41182-017-0080-7</u>

Table 2*: Participant characteristics and association with eye examination

Variable	Sample statistics	Retinal exam (last 12 months)			Retinal examination (ever)		
		Had eye exam	No eye exam	P value	Had eye exam	No eye exam	P value
Number (%) in each category	270 (100%)	36(13.3%)	234 (86.3)		69(25.6)	201(74.4)	
Number (%) by county				0.07			0.002
Kirinyaga	90	6 (6.7)	84 (93.3)		11(12.2)	79(87.8)	
Nairobi	89	14 (15.7)	75 (84.3)		29(32.6)	60(67.4)	
Nakuru	91	16(17.6)	75 (82.4)		29(31.9)	62(68.1)	
Age (mean years, SD)	53.3 (14.1)	57.1(11.7)	52.7(14.4)	0.08	60.5(13.8)	50.8(13.4)	<0.0001
Age (median, IQR)	52(43-65)	56.5(49-65.5)	50.5(41-65)		61(51-71)	48(41-62)	
Sex (no, %)				0.7			0.5
Men	127(47%)	18(14.2)	109 (85.8)		35(27.6)	92(72.4)	

Women	144(53%)	18 (12.5)	126(87.5)		34(23.8)	109 (76.2)	
Literacy				0.3			0.05
Primary or below	88(32.85)	13(14.8)	75(85.2)		30(34.1)	58(65.9)	
Secondary	111(41.4%)	11(9.9)	100(90.1)		21(18.9)	90(81.1)	
Post-secondary	69(25.8%)	12(17.4	57(82.6)		18(26.1)	51(73.9)	
Occupation				0.4			0.014
unemployed	70(25.9%)	6(3.6)	64(91.4)		19(27.1)	51(72.9)	
Low-skilled	70(25.9%)	9(12.9)	61(87.1)		14(20)	56(80)	
Professional	90(33.3%)	13(14.4)	77(85.6)		18(20)	72(80)	
Retired	40(14.8%)	8(20)	32(80)		18(45)	22(55)	
Duration of diabetes (mean years, SD)	7.3 (5.5)	8.9 (4.5)	7.1(5.6)	0.06	9.4(5.5)	6.6(5.3)	0.0002
Duration of diabetes (median, IQR)	6(3-7)	8(5-12)	6(3-10)		9 (5-13)	5(3-9)	
Interval of diabetes clinic visits (months) – mean (SD)	4.0 (1.5)	4.3(1.3)	4.0(1.5)	0.4	4.3(1.4)	3.9(1.5)	0.08
Interval of diabetes clinic visits (months) -median and IQR	4 (3-5)	4(3.5-5)	4 (3-5)		4 (3-5)	4 (3-5)	
Referred for eye examination				<0.001			<0.001
yes	66(24.4%)	23(34.9)	43(65)		47(68.1)	19(28.8)	
no	204(75.6%)	13(6.4)	191(93.6)		22 (10.7))	182 (89.2)	
Perceived level of glucose control				0.02			0.4
Very good	10(3.7%)	0	10(100)		2(20)	8(80)	
well	73(27%)	17(23.3)	56(76.7)		23(31.5)	50(68.5)	
adequate	107(39.6%)	9 (8.4)	98(91.6)		24(22.4)	83(77.6)	
Poor	68(25.2%)	10(14.7)	58(85.3)		19(27.9)	49(72.1)	
Very poor	12(4.4%)	0	12(100)		1(8.3)	11(91.7)	

Diabetes in family member				0.8			0.6
yes	64(23.7%)	8(12.5)	56(87.5)		18(28.1)	46(71.9)	
no	206(76.3%)	28(13.6)	178(86.4)		51(24.8)	155(75.2)	
Information on diabetes given at health facility				0.3			0.8
Yes	205(75.9%)	30(14.6)	175(85.4)		53 (25.9)	152 (74.2)	
No	65(24.1%)	6(9.2)	59(90.8)		16(24.6)	49(75.4)	
Knowledge of diabetes complications				0.4			0.9
Yes	103(38.1%)	16(15.5)	87(84.5)		26(25.2)	77(74.8)	
No	167(61.9%)	20(12)	146(88)		43(25.8)	124(74.3)	
Knowledge of diabetes eye complications				0.001			0.001
yes	150(55.6%)	29(19.3)	121(80.7)		50(33.3)	100(66.7)	
no	120(44.4%)	7(5.8)	113(94.2)		19(15.8)	101 (84.2)	
Comorbid hypertension				0.02			0.04
yes	101(37.4%)	20(19.8)	81(80.2)		33(32.7)	68(67.3)	
no	169(62.6%)	16(9.5)	153(90.5)		36 (21.3)	133 (78.7)	
Opinion on need for an eye examination				P<0.001			P<0.001
No need	51(18.9%)	1 (2.0)	50 (98)		6(11.8)	45(88.2)	
Only for ocular symptoms	115 (42.6%)	15(13)	100 (87)		27(23.5)	88(76.5)	
Acceptable	89(29.6%)	13(16.3)	67(83.8)		25(28.8)	57(71.3)	
Already doing it	9(3.3%)	2 (13.3)	13 (86.7)		9(100)	0	
Other opinion	15(5.6%)	36(13.3)	234 (86.7)		4(26.7)	11(73.3)	

London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT www.lshtm.ac.uk



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?			-
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	BMC Health Services Research
Please list the paper's authors in the intended authorship order:	Nyawira Mwangi, Covadonga Bascaran, Jacqueline Ramke, Lawrence Muthami, Stephen Gichuhi, David Macleoed, Consuela Moormon, Mathew Kipturgo, Allen Foster
Stage of publication	Undergoing revision

SECTION D – Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I designed the study, conducted the research, analysed and interpreted the data, drafted the
	consideration of the comments from the co-
	athors and peer reviewers

Student Signature:	Nyawira	\cap	 Date:	17 SEPT 2019	
Supervisor Signature:	_, _ (Date:	SEPT 17	2019

Paper 2

Rationale for integration of services for diabetes mellitus and diabetic

retinopathy in Kenya

Nyawira Mwangi^{1, 4}, Covadonga Bascaran¹, Jacqueline Ramke^{1, 6}, Lawrence Muthami², Stephen Gichuhi³, David Macleoed⁴, Consuela Moormon⁵, Mathew Kipturgo¹, Allen Foster⁴

¹Kenya Medical Training College

²Kenya Medical Research Institute

³Department of Ophthalmology, University of Nairobi

⁴London School of Hygiene & Tropical Medicine

⁵Oxford University Hospitals NHS Trust

⁶University of Auckland, New Zealand

Abstract

Background: Good diabetes mellitus (diabetes) and diabetic retinopathy (DR) management depends on the strength of the health system, prompting us to conduct a health system assessment for diabetes and DR in Kenya. We used diabetes and DR as tracer conditions to assess the strengths and weaknesses in the health system, and potential interventions to strengthen the health system. In this paper, we report on the need and relevance of integration to strengthen diabetes and DR care.

Methods: Using a mixed methods study design, we collected data from service providers in diabetes clinics and eye clinics in three counties, from key informants and we reviewed documents.

Results: There is interest in integration of diabetes and DR services to address discontinuity of care. We report the findings describing the context of integration, why integration is a goal and how these services can be integrated. We use the results to develop a conceptual framework for implementation.

Conclusions: The principal rationale for integrated service provision is to address service gaps and to prevent complications of diabetes and DR. The stakeholder interest and the existing infrastructure can be leveraged to improve these health outcomes.

Key Words: diabetes mellitus, diabetic retinopathy, non-communicable diseases, screening, integration, health systems, service delivery, Kenya, Africa

Abbreviations

DR	Diabetic Retinopathy
DR NET	Diabetic Retinopathy Network
KI	Key Informant
NCD	Non-Communicable Disease
NGO	Non-Governmental Organization
PLWD	People Living With Diabetes
SARAM	Service Availability and Readiness Assessment Mapping
WHO	World Health Organization

Introduction

Integration of services is a strategy for improving the performance of health systems and achieving clinical outcomes. The World Health Organization's (WHO) working definition of "integrated service delivery" is: "the management and delivery of health services so that clients receive a continuum of preventive and curative services, according to their needs over time and across different levels of the health system."^[1] This definition posits integration as a composite construct with continuity and quality of care being essential components. The aim is to provide services that meet the needs of the user: services that are not disjointed, that are easy to navigate and that provide a smooth link to specialist services, if required.^[1, 2] Although there is consensus on the desirable outcomes of integration, and its importance for universal health coverage in every country, the rationale and the operational models remain contextual.^[3, 4]

Diabetes Mellitus (hereafter referred to as 'diabetes') is associated with the development of organ damage, leading to multiple morbidity. Providing care for people living with diabetes (PLWD) thus requires balancing diabetes management with management of its chronic complications. Diabetic retinopathy (DR) is the major ocular morbidity in diabetes, and there is strong epidemiologic evidence that its prevalence is increasing.^[5, 6] In common with other chronic diseases, the management of both conditions requires: promotion of healthy lifestyle, early detection, compliance to treatment, regular monitoring of treatment outcomes, and active involvement of the patient and family in the care. An integrated approach is an efficient and effective method of addressing inter-related chronic diseases.^[4, 7, 8] At present, diabetes and DR care are provided in diabetes and eye clinics respectively, with minimal collaboration between them. Given that the patient with DR also requires diabetes services, and the similarities in the approach to management, it is appropriate to explore the extent to which

diabetes and DR services provided in a comprehensive or integrated manner.^[9] This provides an opportunity to consider how integration would intersect with the need for specialist services.

The literature on integration of diabetes services has largely focused on integration with HIV, tuberculosis and hypertension.^[8, 10-12] The paucity of literature on integration with services for diabetic retinopathy services might be based on the assumption that these services are automatically integrated, since they are intricately linked. However, the point of entry into integration is often unspecified. Further, the interventions that should be integrated, and in which ways and by whom, is not explicit even in clinical guidelines. The evidence on what diabetes practitioners and eye care practitioners think of the integration, or of their professional relationship is also sparse.^[13]

Proactive prevention and early detection of DR is an important best practice that is often missing in the services for PLWD. As this population has regular contact with diabetes services, this platform is a good entry point to bring DR services to where the patient is, or to link the patient to the eye service, where the DR services are. Innovative approaches such as integration can augment access, quality and continuity of care for PLWD. In this paper, we explore the interface between diabetes services and eye care services in Kenya, as an unexploited area for integrated care for DR. We use our results to develop an operational framework for integration.

Methods

Study design and theoretical approach

A mixed methods cross-sectional health system assessment for diabetes and diabetic retinopathy was conducted in three counties of Kenya, guided by the WHO's health system building blocks framework and the tracer condition approach.^{[14],[15]} The aim of focusing the assessment on diabetes and DR was to provide evidence relevant to services for the two conditions that may be missed in a general health system assessment. Both conditions meet the criteria for a tracer condition, Table 4-1. In line with the WHO framework, we defined a good service as one which delivers effective, safe, quality, personal and non-personal health interventions to PLWD, when and where needed, with minimum waste of resources.^[16] In this paper we report on integration as a theme that emerged from interviews with key informants and service providers.

Criteria	How diabetes and DR fit the criteria
Disease has a known epidemiology	The epidemiology of both conditions globally,
	in SSA and in Kenya has been described
Disease is well defined and easy to	The definition and criteria for diagnosis of
diagnose	each condition is well-established
Its prevalence in the population is	Both population-based and clinic-based
large enough to enable adequate	surveys have shown that the prevalence is
data to be collected	sufficient to enable collection of data that can
	be used for planning services
Its natural history is known, and it	The natural history of diabetes and DR
varies with the utilization and	including the predictors of the development
effectiveness of health care	of complications is known
It requires specific treatment, in the	Hypoglycaemic drugs and lifestyle measures
absence of which functional	are required for glycaemic control, and the
impairment results	treatments for DR have been described,
	without which visual impairment results
Available and well-defined	Prevention, diagnosis and treatment apply to
techniques of medical management	both diabetes and DR. Rehabilitation is
exist for at least one of the	provided for those who develop severe visual
	impairment and blindness

Table 4-1: Diabetes and	l diabetic	retinopathy a	s tracer	conditions
-------------------------	------------	---------------	----------	------------

following: prevention, diagnosis,	
treatment or rehabilitation	

Sampling and data collection

Kirinyaga (predominantly rural), Nakuru (semi-urban) and Nairobi (urban) counties were selected through stratified purposive sampling to represent these different regions within the diabetes belt in Kenya. Three health facilities providing outpatient diabetes services in each county were identified by simple random sampling from a sampling frame of the clinics. Two clinicians who provide diabetes services were interviewed (n= 3 counties*2 clinicians*3 facilities=18). Three eye care workers providing services in the county were also interviewed (n=3 workers*3 facilities=9). The primary investigator and research assistants interviewed the 27 service providers at the clinics using a structured questionnaire with both closed-ended and open-ended questions.

The key informants were representatives of stakeholders in these services, who were familiar with the organization and delivery of healthcare at the national or at county level, with particular reference to diabetes and eye care, but whose principal role in the health system is non-clinical. The key informants (n=18) were identified using an initial sampling frame and subsequently through snowballing from those interviewed. These included eight health service managers, four NGO program leaders, four policy makers and two members of the umbrella PLWD body that represents patients. The primary investigator interviewed the key informants at their work sites or preferred locations. Interviews lasted 45-60 min, were audio-recorded and extensive field notes were taken. We also conducted document review of health system documents provided by the key informants and service providers.

We collected data at national level (key informant interviews) and county level (interviews with service providers, key informant interviews). The structured questionnaire and the topic guide had questions on the strengths and weaknesses of the health system for diabetes and DR, and potential interventions to strengthen the health system.

Ethics

The study was conducted according to the tenets of the World Medical Association's Declaration of Helsinki. The London School of Hygiene and Tropical Medicine and African Medical Research Foundation (AMREF) granted ethical approval. All participants gave written informed consent. Participation was voluntary and participants did not receive any financial incentives.

Data analysis

Audio records were transcribed verbatim. All qualitative data (from interviews and documents) were analysed using thematic content analysis, guided by the theoretical frameworks.^[17] The primary investigator and a second independent coder read and coded the transcripts independently section by section, after agreeing on a coding structure. Where clarifications with participants were required, they were contacted by telephone. The codes were then grouped together into subthemes. These were later collapsed into themes within the six building blocks of the health system, as well as within the areas of prevention, diagnosis, treatment, and rehabilitation. We reviewed themes repeatedly across all transcripts. Quantitative data and data from document review was summarized using descriptive statistics and summary tables respectively. Triangulation of different types and sources of data was useful for elaboration and providing complementary insights.

Results

Characteristics of participants

We interviewed 18 key informants and 27 service providers from diabetes services (n=18) and eye care (n=9). None of the participants invited declined to participate. Of the 45 participants, 25 (56%) were male, the median age and duration of employment being 41 years and 15 years respectively. We examined 22 documents, which were strategies and strategic plans, reports, policies, published literature and meeting presentations related to diabetes and DR in Kenya.

Integration as an emerging theme in participant interviews

When the participants were prompted to discuss potential interventions to strengthen the health system for diabetes and DR, integration emerged as a dominant theme. Table 4-2 shows sample quotes within the integration theme.

Table 4-2: Integration as a theme in the different building blocks

Leadership and Governance

We have been working very closely (with ophthalmic services) at national level...the next step is integration of diabetes eye care services into comprehensive diabetes services (Key informant, diabetes care)

Our policies, which include DR care, fit into the NCD policies...but in practice they do not seem to work in an integrated way (Key informant, eye care)

Service delivery

We offer a wide range of services in the diabetes outpatient clinics...but there is a missing link with the eye services...you know, for the annual eye examination (Key informant, diabetes care)

Eye care services for DR are part of the wider community of diabetes services, and also part of the wider community of eye care services (Key informant, eye care)

Human resources for health

We need integration of the training on comprehensive diabetes management...we have to integrate the eye component into it, and we have to integrate this training in the preservice curriculum of colleges and universities...this is actually a low-hanging fruit (Key informant, diabetes care)

Nurses in the diabetes clinic and in the eye clinic should also be trained as trainers of trainers in diabetes eye care...they need to be part of the team (Service provider, eye care)

Medicines and health technologies

NHIF [National Hospital Insurance Fund] caters for the costs of inpatient care for both diabetes and diabetic retinopathy, now we need to include all tests and medicines for both conditions in this cover (Key informant, eye care)

We value integration of services,...it may help to ensure we don't lose PLWD to informal services... we would not integrate herbal medicine into our services, but this is a cultural and social issue that cannot be addressed by us alone (Key informant, diabetes services)

Health Management Information System

Surveillance of chronic illnesses like diabetes and diabetic retinopathy is difficult. The solution is an integrated electronic medical records system (Key informant, diabetes care)

Even though we do not use the same reporting system or software, it should still be possible for us to have access to relevant data from the diabetes system, and vice versa....we are not talking of a merger (Key informant, eye care)

Health financing

I would suggest that an integrated implementation framework be developed at the county level, and it should have a dedicated budget (Service provider, diabetes care)

DR being principally a diabetes issue, we need to present the case for financing for DR by NHIF as part of diabetes services (Key informant, eye care)

Integration as envisaged in government policies and plans

Integration is a key policy objective as reflected in a sample of the documents, Table 4-3.

Possible integration with HIV, Tuberculosis and Malaria programs is envisaged, though how it

should be done is not explicit. Integration of diabetes and DR is not mentioned.

Table 4-3: Examples of concepts of integration in a sample of health system plans and strategies

Kenya Health Sector Strategic and Investment Plan 2013-2017

The health sector will focus on:

• Integrating health service provision tools, mechanisms and processes for responding to NCDs

• Establishing screening programs at community level and in health facilities for major NCDs (Page 25)

Kenya Essential Package for Health

Institutional screening for NCDs is one of the KEPH interventions for reversing the rising burden of NCDs. The services targeted are routine BP, routine BMI and blood sugar testing.

Kenya National Strategy for the Prevention and Control of Non-Communicable Diseases 2015-2020

Several bottlenecks of NCD prevention and control have been identified and addressed in this strategy, including: "Silo" nature of the health system with minimal opportunities of integrating NCDs in well-established public health care platforms like HIV, TB, family planning, maternal and child health. (Page 31 and 32)

Strategic Objective 1 of the strategy: To establish mechanisms to raise the priority accorded to NCDs at national and county level...The interventions for this objective include integrating NCD prevention and control into policies across all government sectors.

Kenya National Diabetes Strategy (2010-2015)

The objectives of the Kenya diabetes strategy include:

- To improve early detection for diabetes and its complications through screening
- To network and integrate diabetes care with other national programs e.g. HIV/AIDS, TB and malaria (Page 9)

One of the activities in the resource mobilization strategy is:

• Integrate diabetes prevention and control into the national and district health plans (Page 11)

Kenya Service Availability and Readiness Assessment Mapping (SARAM) Report 2013

- General service readiness for provision of NCD services is 34% (for the KEPH defined NCD services)
- There is an overall limitation on the availability of KEPH services contributing to reversing the burden of non-communicable diseases (Page 112-126)

Norms and Standards for Health Service Delivery 2006

Integration of care: Every contact with individuals, households and communities is used to ensure that a comprehensive set of defined services is made available. This is different from using "every opportunity to do everything". (Page 4)

How integration of diabetes and eye care services can be implemented

Participants described a positive existing relationship between diabetes and eye care services

in the context of DR, and envisioned a closer and newer way of 'mutual accommodation':

In the DR NET [Diabetic Retinopathy network] program, we have worked very well as

physicians and ophthalmologists (Key informant, diabetes service)

Diabetes services need to accommodate us more, it seems that DR gets forgotten

(Service provider, DR)

We identified three points of emphasis regarding how the integration of the two services should be implemented. Firstly, is that DR should be integrated into diabetes services. This is because of the pre-requisite for a functional service, such as the diabetes services, to which the DR service can be integrated. Policy documents recognize that services should be integrated into existing well-established health services or programs, Table 4-3.

Sometimes we forget the eye, because there are too many different things that have to be done for the patient... but we need to make sure it is not" (Service provider, diabetes) We would like DR to be seen as a diabetes issue, not an eye care issue (Key informant, eye care)

When we review the diabetes guidelines, DR will take centre stage (Key informant, diabetes service)

Of the 18 key informants, 17 (94.4%) believed that DR services should be integrated with diabetes care. 61.1% of key informants (n=11) reported that diabetes services should lead in the integrated service because they have a stronger infrastructure and accessibility to PLWD. However, 33% of key informants (n=6) indicated that eye care infrastructure in some hospitals is stronger than the diabetes infrastructure, but diabetes services should lead the integration because they have a stronger reach to the PLWD. One key informant (5.5%) felt that the discourse on the relative merit of integration should not focus on the infrastructure but should strengthen links between the services.

The second point of emphasis is that eye care workers have a role in enhancing care for diabetes care, as well as care for other non-communicable diseases:

Using the eye examination, eye clinicians can monitor diabetes, and hypertension...because the finding of diabetic or hypertensive retinopathy is useful information (Key informant, diabetes service) Eye care workers should also ask patients about diabetes control (Service provider, diabetes)

Thirdly, both diabetes and eye care services need to work together:

I see that sometimes they {eye care services} will just examine the eye and not be interested in the medical management of the diabetes...we should all be seen to be involved with this (Service provider, diabetes) Those of us on the ground...we know that DR is being missed in diabetes services...I think we need to go to the diabetes clinic...get involved with diabetes and get to look into the eye (Service provider, DR)

Key informants identified that referral and screening for DR might be strengthened through integration:

Although we have links with diabetes services, there are unexploited opportunities to integrate DR, particularly screening, with routine diabetes services (Key informant, eye care)

There is no follow up system so that even if a diabetes patient is referred to eye clinic and disappears, there is no system of follow up or feedback, as currently we work as separate services (Service provider, diabetes)

Of the 18 diabetes clinicians, only one had received a patient referred from eye services in the preceding month, while four of the nine eye care clinicians had received a referral from diabetes services in the same period. This implies that cross-referral is ineffective; there are missed opportunities for referral or patients are lost in transit. Furthermore, none of the diabetes clinics had visual acuity charts or ophthalmoscopes, and none of the eye clinics had a functional glucometer, which shows there is a big role for referral. The lack of readiness of health facilities for provision of NCD services is also noted in the Service Availability and Readiness Mapping (SARAM) report, Table 4-3.

Benefits of integration

We identified three main benefits of integration. First, integration can help to address service fragmentation, Table 4-3, as well as other gaps identified by participants, Table 4-4. Participants suggested that integration may provide opportunities for joint on-the-job training for staff, which is a priority because 12/18 diabetes clinicians and 4/9 eye care clinicians had not had a recent training update on DR and diabetes respectively. Secondly, participants suggested that integration might enhance continuity of care and increase awareness of DR among diabetes care providers. Thirdly, participants also identified that integration can attenuate potential problems, such as conflicting clinical recommendations that confuse PLWD and staff. However, none of the participants suggested that integration would have an economic benefit.

Table 4-4: Benefits of integration

Purpose of integration	1
------------------------	---

1. To solve current gaps in care

Why integration is required

Services are fragmented
Services are duplicated
Patients are lost between the diabetes clinic and eye clinic
Patients do not get referred appropriately
No screening program for DR
Glycaemic control is not monitored
DR is not detected early enough
Staff in each service work in isolation
There are missed opportunities for adequate assessment
Patients are not empowered for selfmanagement

2. To obtain benefits of integration

To increase first contact of PLWD with eye examination

To facilitate continuity of care

To increase staff awareness of DR

To provide follow up system for patients who are referred

To increase the chances of timely detection rates of DR

To increase coverage of DR screening

Eye care workers can monitor and report diabetes and comorbidities through an eye exam

To emphasise health promotion

To improve equity by increasing access to services by the poor

To avoid burdening the patient with multiple visits

To have a platform for monitoring health outcomes

To use the available resources for maximum benefit

To have the two services 'speak the same language'

To develop joint clinical guidelines and tools

To make the service user-friendly

3. To avoid potential problems

To avoid conflicting recommendations for patient care

To avoid overloading the clinicians with different information

Even patients who have a retinal image taken in the diabetes clinic will still need a comprehensive eye examination, this will be missed if there is no integration *To reduce the number of PLWD going blind from diabetes*

Steps towards implementing integration

We found that the policy documents do not elaborate how integration should occur. However, the norms and standards document states that integration "*does not mean 'doing everything'…*". This implies the need to establish the priorities. The participants identified four main priorities: referral (n=34), retinal screening for DR (n=23), patient monitoring (n=19) and patient education for self-management (n=16). Seven participants remarked that the interaction between diabetes and eye clinics must be continuous, particularly through bidirectional referral; otherwise, "*integration will be ineffective*". Five participants indicated that the integration should be gradual, and preceded by a pilot.

The inputs that will be needed to achieve integration were listed: joint planning, joint training of health workers on diabetes and DR, equipment for monitoring diabetes (glucometer, test strips), DR screening equipment, a database that includes both diabetes and DR, clinical checklists and guidelines. Key informants suggested that financing for the additional inputs and processes would be sourced from the government and partners. All diabetes clinicians indicated that they would be happy to have a retinal camera situated in their clinic, though they had space constraints. Six of the nine eye care service-providers were willing to hold regular outreach clinics to screen PLWD for DR.

All participants concurred that they would have roles in the integration, which include; getting buy-in from all staff and administrators, facilitating or participating in joint planning, obtaining the resources and supporting implementation. Participants identified that integration should be led by the team leads in diabetes and eye clinics, and should prioritize strengthening

referral, metabolic control, self-management and screening for DR. Based on the findings in this study, we present a conceptual framework for integration, Figure 4-1, showing the inputs required to make integration feasible and the outcomes envisaged in successful integration.

Discussion

Integrated health systems have been promoted as a means to build a more effective, efficient and patient-focused health system that better meets the needs of the populations served.^[4] Integrated DR services can blur the boundary between diabetes services and eye care services, to create a shared repertoire and synergy for the investments made in these services. Such synergy is vital for strengthening the health system responsiveness to the rising burden of diabetes and DR.^[7, 13] It can ensure equity by reducing the exclusion and difficulty in navigating the services by PLWD, since comprehensive diabetes care would include DR services. Further, it provides a unique opportunity to integrate primary, secondary (early detection) and tertiary prevention (treatment to prevent complications), Figure 1. This can lead to improved health outcomes and therefore more cost-effective use of health system resources by PLWD. For example, screening and detection of DR might improve adherence to self-management and optimal metabolic control, which would prevent additional complications of diabetes that require expensive treatment.

The endorsement of integration in the health policies is relevant to its sustainability, because it implies long-term government commitment.^[7] Integration of comprehensive diabetes care with HIV, tuberculosis and malaria services would entail investment at all levels of health care, as services for these communicable diseases are offered across the continuum of primary, secondary and tertiary care. To ensure that comprehensive diabetes care includes DR, we

propose a service-level model of integration at the diabetes clinics and eye clinics, which are usually located at secondary and tertiary hospitals. This is expedient for three reasons. One, the resources for integration are already available at this level of the health system, where the two clinics are already functioning. This would avoid aggravating existing resource challenges, such as health workforce shortages.^[18] Two, the integrated service removes the complexity that patients face while navigating the care pathway, and which often presents a barrier to access to DR services.^[1, 19] Three, the integrated services includes specialist diabetes and eye care services, which shows that integration does not imply compromising specialist functions.^[1, 2] Such a fear can cause resistance by specialists, although this was not evident in our study.^[1, 19]

Furthermore, there was high level of interest on integration among all participants, which is likely to facilitate successful implementation of integration. This is important because reluctance, opposition or lack of ownership by the service providers would lead to poor integration results.^[20] In other studies, service providers have been concerned about the likely increase in workload.^[11, 20] In this study, staff shortages, inadequacy of space for additional services, lack of equipment and weak referral linkages were identified as potential challenges but not as deterrent to the integration. Although we did not investigate the reasons for this enthusiasm, it might be because diabetes and eye care services target the same population (PLWD) and have a converging goal in relation to diabetic retinopathy (prevention of blindness). It might also have resulted from several system antecedents: (1) An ongoing pilot program of the DR NET hospital-twinning initiative, which is a link program involving both diabetes and eye care stakeholders, with the aim of strengthening DR services. (2) A recent national STEPwise survey for risk factors of non-communicable diseases, and (3) Sensitization of participants on DR as a potentially blinding condition.

The reasons for integration nominated by participants reflect the perceived differences between integrated and non-integrated care. The main impetus is local service gaps, such as fragmentation, and missed opportunities for early detection of DR or inefficiency and discontinuity of care, which concurs with the drivers cited in other literature.^[2, 7, 10, 13, 20] These are typical barriers to access to care that will be addressed through integration.^[2] The necessity for integration has also been recognised in a previous study in Kenya.^[13] Cost-control was not identified as the major driving force for integration in this study, unlike in other contexts.^[19] However, integration is likely to reduce costs by reducing duplication of services and multiple client visits.^[1, 12, 20] In addition, early detection or DR is a sound economic investment because timely treatment is cost-effective.^[21, 22]This shows that the interest in integration among these participants was predicated on improved services outcomes and not as an end in itself.

The conceptualisation of integration around screening is significant because Kenya does not have a systematic screening program for diabetes or DR.^[13] It reveals an excellent opportunity to develop an effective screening program inclusive for all PLWD attending diabetes services. The bidirectional referral strategy shows the pertinence of organising integration as a process of mutual but not symmetric accommodation. It is not symmetric because the entry point is the diabetes services, which PLWD are already accessing even without integration; hence, it is the primary service. An excellent example of how synergy might be realised is that eye care workers can identify and monitor comorbidities. Ocular findings in hypertension, hyperlipidaemia, and other medical conditions may be the first sign of these diseases, and these can be identified during the screening examination. Medications used to treat these comorbidities might also have ocular adverse effects that can be identified upon ocular

examination. A comprehensive dilated eye examination can be a radar for detection or monitoring several comorbidities and medications.

The integration could be operationalized through co-implementation of the key interventions, which are self-management, glycaemic control, DR screening and referral. This scope focuses on prevention of complications, rather than treatment.^[23] Some integration models, such as sexual and reproductive health programs integrated into HIV programs, have focused on clinical services, such as testing or prescription of treatment, rather than lifestyle modification.^[11, 24] Conversely, other integration models for diabetes, HIV and hypertension have emphasised adherence counselling for medication and lifestyle modification.^[25] Still other models have included a mix of patient education and prescription of medications for PLWD.^[10, 26, 27]No method is inherently good or bad, the scope largely depends on the objectives of the two services.^[1, 7, 8] Given the priority for diabetes and DR is to prevent progression or complications, bidirectional referral and health promotion approaches would be useful.^[13, 21] In the event of future integration with HIV, Tuberculosis and Malaria programs, a treatment component may be added, since the main priority for these conditions is universal coverage to treatment for those eligible.^[7, 27]

Several inputs are required: inter-professional collaboration, joint planning, clinical governance, training, clinical tools, database and equipment. This means that supply-side resources are required, and need to be allocated differently.^[7, 13] Integration is not monolithic but encompasses all building blocks of the health system and hence requires resource mobilization. Given that integration cannot mitigate against lack of necessary resources or infrastructure, failure to invest in it would hamper the desired benefits .^[1, 23] For example, the lack of a monitoring and evaluation component has been identified as a weakness in previous integration initiatives.^[28] This being one of the first studies to discuss this context of
integration, we have proposed a conceptual framework for integration, which can be used by policy-makers for planning, Fig 4-1.

What are the expected effects of integration? Investing in this integrated service delivery system creates distinct deliverables, such as increased demand for the specified services and reducing the unmet need for DR screening.^{[7], [12, 21]}Integration should translate to prevention of complications of diabetes and DR, which is a widely agreed priority of health systems.^[21] To monitor whether integration confers these benefits, an appropriate metric will need to be jointly determined.

Our study has several strengths. Geographical variability (three counties) accentuated the external validity of the study. The inclusion of clinicians from both diabetes and eye care services, as well as patient representatives, enabled us to obtain unique perspectives of service providers. The data is subject to social desirability bias as the participants are directly involved in the services, however we used triangulation to mitigate this. This is the first study to document the interventions and the platforms for integration of these services in the region. The main limitation is the novelty of the concept of integration with respect to diabetes and DR but this shows that this health system is dynamic, and it may jumpstart the process of broader integration of diabetes services.

Conclusion

Integration, as envisaged in this paper, is relevant to the goals of the health system and congruent to the existing health system for diabetes and DR and to the broader health strategies in Kenya. The purpose of integration is to address service gaps, ensure universal access to a range of services and prevent complications of diabetes and DR. This is evidence

that the health system is not static in its response to NCDs, and integration may be applicable to other countries with similar health systems.

Future research

Further research is required to test and refine this empirical conceptual framework. The impact

of integration on the following parameters will also need to be evaluated: performance of

health workers, service utilization, patient satisfaction and cost of the services. The effect of

integration on equity may be determined by disaggregating the health outcomes of PLWD by

gender, social strata and other indices of vulnerability.

REFERENCES

- 1. World Health Organization, *Integrated Health Systems: What and Why*. 2008, World Health Organization: Geneva.
- 2. Natasha Curry and Chris Ham, *Clinical and service integration: The route to improved outcomes*. 2010, The King's Fund: London.
- 3. World Health Organization, *WHO global strategy on people-centred and integrated health services: interim report*. 2015: <u>www.who.int</u>.
- 4. Gail D. Armitage, et al., *Health systems integration: state of the evidence.* International Journal of Integrated Care Vol. 9, 2009. **9**(17).
- 5. Burgess, P.I., et al., *Epidemiology of diabetic retinopathy and maculopathy in Africa: a systematic review.* Diabet Med, 2013. **30**(4): p. 399-412.
- 6. Zheng, Y., M. He, and N. Congdon, *The worldwide epidemic of diabetic retinopathy.* Indian Journal of Ophthalmology, 2012. **60**(5): p. 428-431.
- 7. Atun, R., et al., *Improving responsiveness of health systems to non-communicable diseases.* The Lancet, 2013. **381**(9867): p. 690-697.
- 8. Haregu, T.N., et al., *Integration of HIV/AIDS and noncommunicable diseases in developing countries: rationale, policies and models.* International Journal of Healthcare, 2015. **1**(1).
- 9. Poore, S., et al., *Planning and developing services for diabetic retinopathy in Sub-Saharan Africa*. Int J Health Policy Manag, 2015. **4**(1): p. 19-28.
- 10. Leung, C., et al., *Preparedness of HIV care and treatment clinics for the management of concomitant non-communicable diseases: a cross-sectional survey.* BMC Public Health, 2016. **16**(1): p. 1002.
- 11. Workneh, M.H., G.A. Bjune, and S.A. Yimer, *Assessment of health system challenges* and opportunities for possible integration of diabetes mellitus and tuberculosis services

in South-Eastern Amhara Region, Ethiopia: a qualitative study. BMC Health Serv Res, 2016. **16**: p. 135.

- 12. Divala, O.H., et al., *The burden of hypertension, diabetes mellitus, and cardiovascular risk factors among adult Malawians in HIV care: consequences for integrated services.* BMC Public Health, 2016. **16**(1): p. 1243.
- 13. Kupitz DG, et al., *Diabetes and Diabetic Retinopathy Management in East Africa: Knowledge, Attitudes, and Practices of Hospital Staff in Kenya.* Asia-Pacific Journal of Ophthalmology, 2014. **3**(5).
- 14. Health Systems 20/20, *The Health System Assessment Approach: A how-to manual Version 2.0.* 2012: <u>www.healthsystemassessment.org</u>.
- 15. Nolte, E., C. Bain, and M. McKee, *Diabetes as a Tracer Condition in International Benchmarking of Health Systems*. Diabetes Care, 2006. **29**(5): p. 1007-1011.
- 16. World Health Organization, *Everybody's business: Strengthening health systems to improve health outcomes: WHO's framework for action.* 2007, WHO: Geneva.
- 17. Braun, V. and V. Clarke, *Using thematic analysis in psychology*. Qualitative Research in Psychology, 2006. **3**(2): p. 77-101.
- 18. Luoma M, et al., *Kenya Health System Assessment 2010*. 2010, Health Systems 20/20 project, Abt Associates Inc: Bethesda, MD.
- 19. Esther Suter, et al., *Ten Key Principles for Successful Health Systems Integration*. Healthc Q., 2009. **13**: p. 16-23.
- 20. Levitt, N.S., et al., *Chronic noncommunicable diseases and HIV-AIDS on a collision course: relevance for health care delivery, particularly in low-resource settings--insights from South Africa.* Am J Clin Nutr, 2011. **94**(6): p. 1690S-1696S.
- 21. Atun, R., et al., *Diabetes in sub-Saharan Africa: from clinical care to health policy.* The Lancet Diabetes & Endocrinology, 2017.
- 22. Solomon, S.D., et al., *Diabetic Retinopathy: A Position Statement by the American Diabetes Association*. Diabetes Care, 2017. **40**(3): p. 412-418.
- 23. Nigatu, T., Integration of HIV and Noncommunicable Diseases in Health Care Delivery in Low- and Middle-Income Countries. Preventing Chronic Disease, 2012.
- 24. Rebecca Hope, et al., *Health Systems Integration of Sexual and Reproductive Health and HIV Services in Sub-Saharan Africa: A Scoping Study.* Journal of Acquired Immune Deficiency Syndrome, 2014. **67**(4).
- B Janssens, et al., Offering integrated care for HIV/AIDS, diabetes and hypertension within chronic disease clinics in Cambodia. Bulletin of World Health Organization, 2007.
 85: p. 880-885.
- 26. Gucciardi, E., et al., *Exploring interprofessional collaboration during the integration of diabetes teams into primary care.* BMC Fam Pract, 2016. **17**: p. 12.
- 27. Tilahun Nigatu Haregu, et al., *National responses to HIV/AIDS and non-communicable diseases in developing countries: analysis of strategic parallels and differences.* Journal of Public Health Research, 2014. **3**(99): p. 27-36.
- 28. Dehne KL, Snow R, and O'Reilly KR, *Integration of prevention and care of sexually transmitted infections with family planning interventions: what is the evidence for public health benefits?* Bulletin of the World Health Organization, 2000. **78**(5): p. 628-639.



VTDR: vision-threatening diabetic retinopathy

Section C Health System Strengthening

"Mosquito [...] had asked Ear to marry him, whereupon Ear fell on the floor in uncontrollable laughter. "How much longer do you think you will live?" she asked. "You are already a skeleton." Mosquito went away humiliated, and any time he passed her way he told Ear that he was still alive."

"You do not know me,' said Tortoise. 'I am a changed man. I have learned that a man who makes trouble for others makes trouble for himself."

"Eneke the bird says that since men have learned to shoot without missing, he has learned to fly without perching."

Prof Chinua Achebe 1930-2013 Nigerian novelist, poet, professor of literature, and critic In his book 'Things Fall Apart'

Chapter Five

Health system strengthening: Development of clinical guidelines for diabetic retinopathy in Kenya

5.0 Overview

Multiple supply and demand factors interact to influence health systems. These factors are responsible for low access to diabetic retinopathy (DR) services in Kenya (chapter 3). *Health system strengthening* involves broad interventions designed to permanently make the health system function better in relation to the long-term goals for population health, including prevention of vision loss and blindness from DR.

Chapter 5 and 6 address Objective 3 of the thesis: To use the evidence from the literature and evidence from the health system assessment as a platform for health system strengthening. This chapter focuses on this health system strengthening initiative and includes two papers.

Evidence-based clinical guidelines for diabetic retinopathy define what services people living with diabetes (PLWD) should receive for prevention, screening, diagnosis and treatment of DR. The health system assessment identified that lack of national clinical guidelines for DR in Kenya leads to lack of standardization of services. Addressing this supply-side barrier can ensure that the required resources, tools and services for DR control are available and aligned to the local needs. This aspect of health system strengthening was considered critical at national level. (Chapter 3). National clinical guidelines were therefore developed through a process of adaptation from generic guidelines.

The adaptation process followed internationally defined good practice, which include defining the PICO framework (population, intervention, comparator and outcome), determining the

clinical questions, using a systematic approach for literature search, appraising the quality of generic guidelines using validated instruments, identifying contextual considerations and following established guideline adaptation frameworks.

The adaptation process requires high-level skills in finding, appraising and interpreting evidence, in stakeholder engagement for a participatory process as well as in writing the guideline documents. The PhD provided opportunity to develop these skills, while guideline development provided opportunity to apply the skills in a real-life situation, and to participate in knowledge translation to close the gap between evidence and practice.

5.1 Research Paper 3

Research paper 3 is on the process and outputs of guideline development. This paper has been included since it describes the opportunities, challenges and lessons learnt that might need to be considered in future guideline development. Given that Information on guideline development in LMICs is sparse, the paper facilitates understanding of the feasibility and implications of this initiative in similar contexts. This paper was published in the journal BMC Implementation Science in December 2018 after peer review.

BMC Implementation Science – Licences and Copyright

The following terms apply to this publication

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy, distribute, and use their attributed research, free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago. All articles published by *BMC Implementation Science* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access https://implementationscience.biomedcentral.com/about

5.2 Research Paper 4

Research paper 4 provides a summary of the content of the guidelines. This paper has been included because it provides an understanding of the recommendations for DR care in Kenya, and because the <u>guideline document</u> is too large to include in this chapter. This paper was published in the Journal of Ophthalmology of Eastern, Central and Southern Africa in December 2017 after peer review.

The following terms apply:

"All articles published in *Journal of Ophthalmology of Eastern, Central and Southern Africa* are made freely and permanently accessible online on the JOECSA website immediately upon publication and without subscription charges. "

This journal provides immediate open access to its content on the principle that making research freely available to the public supports a greater global exchange of knowledge.

http://joecsa.coecsa.org/index.php/JOECSA/about/editorialPolicies#openAccessPolicy



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?	BMC Implementation	Research	
When was the work published?	2018		et / A
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	Not applicable		*
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D – Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I led the work described in the paper, drafted the manuscript and prepared subsequent revisions in consideration of the comments from the co-athors and peer reviewers
--	---

Student Signature:

Nyawira

17 SEPT 2019 Date:

Date: SEP- 17, 2019.

Improving health worldwide

www.lshtm.ac.uk

RESEARCH

Open Access



Adapting clinical practice guidelines for diabetic retinopathy in Kenya: process and outputs

Nyawira Mwangi^{1,11*}, Muchai Gachago^{2,4}, Michael Gichangi³, Stephen Gichuhi², Kibata Githeko⁴, Atieno Jalango⁵, Jefitha Karimurio², Joseph Kibachio⁶, Lawrence Muthami⁷, Nancy Ngugi⁸, Carmichael Nduri⁹, Patrick Nyaga⁸, Joseph Nyamori², Alain Nazaire Mbongo Zindamoyen¹⁰, Covadonga Bascaran¹¹ and Allen Foster¹¹

Abstract

Background: The use of clinical practice guidelines envisages augmenting quality and best practice in clinical outcomes. Generic guidelines that are not adapted for local use often fail to produce these outcomes. Adaptation is a systematic and rigorous process that should maintain the quality and validity of the guideline, while making it more usable by the targeted users. Diverse skills are required for the task of adaptation. Although adapting a guideline is not a guarantee that it will be implemented, adaptation may improve acceptance and adherence to its recommendations.

Methods: We describe the process used to adapt clinical guidelines for diabetic retinopathy in Kenya, using validated tools and manuals. A technical working group consisting of volunteers provided leadership.

Results: The process was intensive and required more time than anticipated. Flexibility in the process and concurrent health system activities contributed to the success of the adaptation. The outputs from the adaptation include the guidelines in different formats, point of care instruments, as well as tools for training, monitoring, quality assurance and patient education.

Conclusion: Guideline adaptation is applicable and feasible at the national level in Kenya. However, it is labor- and time -intensive. It presents a valuable opportunity to develop several additional outputs that are useful at the point of care.

Keywords: Clinical practice guidelines, Diabetes mellitus, Diabetic retinopathy, Guideline development, Guideline adaptation, Kenya

Background

The first definition of clinical practice guidelines (CPG), hereafter referred to as "guidelines," was provided by the Institute of Medicine (IOM) in the USA in 1990: "systematically developed statements to assist practitioners and patient decisions about appropriate healthcare for specific circumstances" [1]. Guidelines-related initiatives have subsequently increased globally since the 1990s. This definition was revised in 2011 to: "statements that include recommendations to optimize patient care that are informed by a systematic review of the evidence and an assessment of the benefits and harms of alternative care

* Correspondence: Nyawira.Mwangi@lshtm.ac.uk

¹Kenya Medical Training College, Nairobi, Kenya ¹¹London School of Hygiene and Tropical Medicine, London, UK

Full list of author information is available at the end of the article



options" [2]. Guidelines constitute one tool for good decision-making in clinical practice, which has potential to reduce variations in health care and its cost. Although a plethora of barriers may compromise their effectiveness, guidelines are instruments to improve the quality of care.

Guideline adaptation is potentially an efficient alternative to de novo guideline development, particularly in resource-constrained contexts [3]. Adapting guidelines to suit a local context may also improve local uptake of the guidelines [4, 5]. Adaptation requires an active, systematic, and participatory process [4] that preserves the integrity of the transferable evidence-based recommendations. Although this adaptation process is context-specific and may not be transferable or generalizable, it needs to be systematic, explicit, transparent, rigorous, and reproducible. The ADAPTE and Practice Guideline Evaluation

© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated. and Adaptation Cycle (PGEAC) framework of adaptation are validated approaches to conduct and document this process [4].

The Institute of Medicine [2] has described eight attributes of good guideline development. These are (a) validity, (b) reliability and reproducibility, (c) clinical applicability, (d) clinical flexibility, (e) clarity, (f) documentation, (g) development by a multidisciplinary process, and (h) plans for review. Guidelines are likely to reflect these attributes when they are developed via a transparent process by a multidisciplinary team without potential bias and conflicts of interest, and supported by a systematic review of the evidence [2].

This paper describes the process involved in adapting the diabetic retinopathy (DR) guidelines for Kenya, in order to assist others undertaking a similar endeavor.

The STEPwise survey [6] for risk factors of non-communicable diseases in 2015 reported that diabetes mellitus (DM) affects an estimated 2% of the Kenyan population aged 18–69 years, with the highest proportion (5%) being in the 45–59 years age group. Every person living with diabetes (PLWD) is at risk of potentially blinding diabetic retinopathy (DR). In turn, visual loss from DR is associated with additional morbidity, such as falls, fractures, and difficulties with taking medications. Both DM and DR are associated with significant morbidity, mortality, and excess health care costs. The prevalence of DM is predicted to rise steeply over the next decade [7], and consequently DM and DR are important public health concerns.

Effective and quality service delivery in relation to DR in Kenya is required within the existing health system [8–10]. Currently, there are notable gaps in DR screening, diagnosis, referral, treatment, and follow-up. Although screening and laser treatment are cost-effective interventions for prevention of blindness from DR [11], there are inequities in access to them. Some of the services are underutilized for a variety of reasons, while some of the services delivered are of insufficient quality. This disparity is linked to multiple supply and demand factors, such as variation in referral practices of eye care providers, integration of services, and level of awareness of patients [9, 12, 13].

Clinical guidelines offer recommendations to improve service delivery, advocate for resources, leverage existing resources, and improve outcomes. Implementing evidence-based practice guidelines for DR is thus vital to address the gaps and prevent blindness from DR. International guidelines for this purpose exist, but there are no published local guidelines. This guideline adaptation aimed to address this lack of national guidelines. We envisaged providing a user-friendly guideline that describes appropriate care based on the best available scientific evidence.

Methods

We relied on adaptation instead of de novo development of the guideline in order to avoid duplication of effort, to use the available resources cost-effectively, and to facilitate customization of the guidelines prepared for other income and health system settings to reflect local context.

The process of standardizing clinical practice recommendations for DR in Kenya began over a decade ago. Several guideline documents have been produced although none has been formally published as a national guideline. Our reflection was that opportunity costs, turnover of experts involved in the process, and other contextual factors might have slowed down further development of the guidelines. The methodology discussed here is that followed over the last 2 years leading to the production of the published guideline. However, we expect that a similar process was undertaken in the previous period.

The adaptation process has been systematic, consultative, and guided by a technical working group (TWG). Several widely used toolkits and guidelines provided a point of entry [14–20]. We followed the tasks of adaptation, as applied within the ADAPTE framework, although some of the tasks were synchronized and we often had to return to previously completed steps. The ADAPTE process is a well-known framework for guideline adaptation, which consists of 3 phases and 24 steps. Seven core principles underpin this framework, and the TWG adopted them for this adaptation [14]. Figure 1 provides a simplified schema of our methods.

The identification of the need for the DR guidelines by stakeholders prompted the Ophthalmic Services Unit to constitute a steering team of five members. This team prepared terms of reference and a list of potential members for the TWG. At the first few meetings of the TWG, we discussed the following: the need for the guidelines, the feasibility of guideline development, the required expertise, funding, work plan, outputs, and role definition of the members. In subsequent meetings, the topic and clinical questions were defined.

We identified the methodological resources, the clinical guidelines, and the evidence for effectiveness of various DR interventions through a literature search on various databases including Cochrane Library, ELDIS, Embase, Global Health, and PubMed. We also searched the websites of agencies that develop these resources. The search strategy (Additional file 1) was limited to reports published in English from 2000 up to date.

Two TWG members conducted the literature search and recorded the characteristics and content of retrieved guidelines. Guidelines that did not meet the predefined inclusion and exclusion criteria were eliminated. Two reviewers assessed the quality of the retrieved guidelines using the AGREE II instrument and presented the findings in a TWG meeting. All TWG members participated

Determin	ation of the need for a national guideline
	↓
Establishi	ng a multidisciplinary Technical Working Group (TWG)
	↓
Defining t	the scope and the clinical questions
	↓
Review of including	f toolkits or manuals used for guideline development, the ADAPTE framework ↓
Gathering	g the evidence and guidelines through a literature search $\mbox{\downarrow}$
Assessing using AGF	the guidelines for quality, relevance and applicability REE II instrument ↓
Preparing	draft local guidelines
	↓
External r	eview and feedback
	↓ ↓
Stakehold	der consensus development
	↓ ↓
Pilot-testi	ing with end-users
	•
Revision o	of draft guideline based on findings of pilot-testing
Official ac	doption and launch of the guidelines
Scaled up	dissemination, training and implementation
ia 1 Scho	ma for the methods used in guideline adaptation

in the assessment for currency, content, consistency, acceptability, and applicability of the recommendations. Following consensus on the results of the assessment, guidelines suitable for adaptation were selected.

The clinical guidelines that were selected for reference were the American Diabetes Association (ADA) Standards for Medical Care in Diabetes [21], International Council of Ophthalmology (ICO) guidelines for diabetic eye care [22], the Royal College of Ophthalmologists' diabetic retinopathy guidelines [23], and the Canadian Diabetes Association's retinopathy guidelines [24]. We chose the ADA and ICO guidelines as the prototypes for DM and DR guidelines respectively and collated evidence from Cochrane systematic reviews relevant to diabetic retinopathy. None of the guidelines contained an adaptation template for different contexts.

We utilized the AGREE II (Appraisal of Guidelines Research and Evaluation II) instrument to assess the quality of the clinical guidelines. This instrument consists of 23 items grouped into six domains: (a) scope and purpose, (b) stakeholder involvement, (c) rigor of development, (d) clarity and presentation, (e) applicability, and (f) editorial independence.

The following additional guidelines were also reviewed, so as to identify any potential conflict in recommendations for care of diabetes and other comorbidities: Kenya national guidelines for management of diabetes [25], Kenya national strategy for the prevention and control of non-communicable diseases [26], and International Diabetes Federation's diabetes eye health guide for health professionals [27]. Similarly, we reviewed previous drafts of local DR guidelines.

Draft guidelines were prepared by the TWG and circulated via email to all members for review. Three drafts were circulated, with the final draft also being circulated to external reviewers to assess content validity, clarity, and applicability. The TWG evaluated the final draft guidelines for the quality requirements of the AGREE II instrument prior to release. A consensus stakeholders meeting approved the final draft.

Pilot testing was conducted in different health care settings in purposively selected counties that differed in characteristics that may influence applicability. We collected feedback on the usability through interviews, reports, and observation. Dissemination of guidelines was done through county coordinators, conferences, training institutions, professional associations, social media, and distribution of print material.

Results

Table 1 shows the results of the adaptation process in each of the steps of the ADAPTE framework. We applied the guiding principles as exhibited in Table 2. We further applied the AGREE II instrument to ensure quality of our draft guideline as reported below.

Scope and purpose

The TWG's first task was to define the scope. The main options were to include only DR or diabetic eye disease as a whole. The consensus was to limit the scope to DR because of its unique natural history and public health implications. The overall objective of the guideline is to reduce the proportion of PLWD who go blind due to diabetic retinopathy in Kenya through interventions for prevention, early detection, and effective treatment of DR. The adaptation process aimed to reduce inappropriate variation in screening and treatment, to provide a rational guide for referral, and to use the diabetes care and eye care resources efficiently to meet these goals. The recommendations needed to be germane to the social context, the patient pathway, and the referral systems in addition to being capable of integration into the routine workflow. The Population, Intervention, Professions, Outcomes and Health care system (PIPOH) summary (Table 3) defined the clinical questions addressed in the guidelines.

Table 1	I Adap	ting the	guidelines	using the	ADAPTE	process
---------	--------	----------	------------	-----------	--------	---------

Table	able 1 Adapting the guidelines using the ADAPTE process				
Step	Activity	Result			
Phase	l Set up				
1.	Establish a resource team	The Ophthalmic Services Unit constituted a steering team of five members which developed the terms of reference and prepared a list of 25 potential members for the technical working group (TWG), who were subsequently invited to the group			
2.	Determine criteria for selection and select a topic using criteria	DR was selected because it is a public health concern and there is variation in standards of care			
3.	Check if adaptation is feasible	Evidence-based guidelines were already in use internationally, and there was high interest from the Ministry of Health, clinicians, and other users to develop guidelines			
4.	Identify necessary resources and skills	There was high level of commitment by members of the TWG. The Fred Hollows Foundation committed to provide funds, and the required expertise was available: retinal specialists, public eye health specialists, endocrinologists, diabetes educators, epidemiologists, search and retrieval of information, critical appraisal, research, policy, guideline development, and eye health systems. A need for input from other professions in the multidisciplinary care team for type 1, type 2, and gestational diabetes was identified			
5.	Complete tasks of the set-up phase	Members of the group decided to function as a working group coordinated by the Ophthalmic Services Unit. A set of guiding principles to foster development of the guidelines was adopted (Table 2). Potential conflicts of interest were explored, and there were none to declare			
6.	Write the plan for adaptation	A timeline for completion, list of additional resource persons to be included, list of outputs to be developed in conjunction with the guidelines and task allocation among the TWG members were agreed upon			
Phase	II Adaptation				
7.	Determine and clarify the question	A PIPOH summary was prepared (Table 3). The areas of interest for standards of care were determined as screening, diagnosis and management of DR, and the management of DM in relation to DR, within the existing care pathway for PLWD in the Kenyan health system			
8.	Search for guidelines and other relevant documentation	The TWG searched for relevant DR guidelines and evidence on DR interventions in systematic reviews			
9.	Screen the retrieved guidelines and record their characteristics and content	The recommendations of the guidelines for screening, diagnosis and management of DR, and the management of DM in relation to DR was reviewed, extracted, and compiled in summary tables. Evidence from Cochrane systematic reviews was also reviewed			
10.	Eliminate a large number of the retrieved guidelines using the AGREE instrument	The rigor dimension of the AGREE II tool was utilized to eliminate guidelines that did not meet the stipulated criteria			
11.	Assess the quality of the guideline	The AGREE II instrument was used to scrutinize the quality of the guidelines			
12.	Assess the currency of the guideline	The guidelines retrieved were sufficiently current, and we did not identify any new evidence			
13.	Assess the content of the guideline	Recommendations for screening, diagnosis, grading, referral and treatment were examined and did not differ significantly between guidelines			
14.	Assess the consistency of the guideline	There was clear consistency between the evidence from systematic reviews, the interpretation of the evidence, and the recommendations in the guidelines in all the areas of interest			
15.	Assess the acceptability and applicability of the recommendations	Care was taken to ensure the recommendations are not in conflict with other local guidelines and to appraise the implications of the guidelines on health service delivery			
16.	Review assessments	The results of the assessment of the guidelines were discussed in meetings of the TWG			
17.	Select among guidelines and recommendations to create an adapted guideline	The ICO guideline for DR was the main guideline used because the recommendations compared well with the other high-quality DR guidelines and the practice-based recommendations were well-stated			
18.	Prepare a draft of the adapted guideline	The facilitators of the working group compiled the results of the deliberations and wrote the draft guideline document			
Phase	III Finalization				
19.	Seek feedback on the draft guideline from those who would be using it	Three revisions of the draft were circulated for comment to TWG members as well as surgeons, pediatricians, ophthalmologists, Kenya Defeat Diabetes Association, vitreoretinal surgeons, physicians, diabetes educators for agreement and identification of gaps			
20.	Consult with endorsement bodies	The Ministry of Health adopted the guidelines			
21.	Consult with developers of guidelines used as sources	No substantive changes were made to recommendations so this step was not undertaken			

Table 1 Adapting the guidelines using the ADAPTE process (Continued)

Step	Activity	Result
22.	Acknowledge source documents	The key guideline documents and other resources used have been acknowledged through attribution
23.	Plan for aftercare of the adapted guideline	A review date was planned for 5 years. Monitoring indicators were also identified. Pilot- testing has been used to check for usability. Distribution will be through electronic and print copies
24.	Produce a final document of the guideline and other outputs	The following additional outputs were produced (along with the guideline): posters and brochures for patient information, posters, brochures and checklist to be used by clinicians, workshop slides for training health workers, quality assurance guidelines

Stakeholder involvement

The Ophthalmic Services Unit at the Ministry of Health convened a steering group of five members. They drafted the terms of reference for a task-oriented TWG, which were to (1) determine the scope and focus of the required guidelines, (2) appraise the evidence and recommendations in existing DR guidelines, (3) develop the national guideline, and (4) craft messages to be used at the point of care, to influence practice.

The steering group identified 25 potential members of the TWG, based on the criteria of diverse expertise, experience, representation of multiple stakeholder groups, and commitment to the process, all aimed at increasing both internal and external validity of the guidelines. These members received personal invitations to participate. An average of 15 were active members of the TWG at any given time, but the others remained involved on the periphery and received frequent reminders to participate remotely. Participation was through attending meetings, email and telephone correspondence, face-to-face consultations, availing resource documents, reviewing drafts, providing evidence, and informal consultations. This proactive integrative and flexible approach was designed to ensure ownership, external

validity, and the involvement of end users of the guideline.

The TWG members were all volunteers with other clinical, educational, administrative, and policy roles related to DM and DR in public, private, or faith-based health facilities, academia, ministry of health, and professional organizations. Participation on a volunteer basis inferred limitation of availability, though additionally, it implied indirect institutional participation of the employer. They had diverse expertise including clinical, public health, research, epidemiology, literature search, systematic reviews, and health systems. Differences in opinion were encountered in the deliberation of some recommendations, particularly regarding the role of different cadres in making DR treatment decisions and delivering treatment. This was resolved through varied strategies: expressing judgements about values and risks, making reference to regulation, reviewing the evidence for role specification, and adapting the role definition prescribed by the source guidelines and informal consensus techniques.

We did not employ a research assistant, because the team had skills in literature review, recent systematic reviews on interventions for DR were available and the

Table 2	Guiding	principles	for guideline	adaptation
---------	---------	------------	---------------	------------

	Guiding principle	Indicator
1.	Respect for evidence-based principles in the development of guidelines	The evidence on which the recommendations are based is included in the guidelines
2.	Ensuring that the quality of guidelines is high	Well-known frameworks for guideline development were used to guide and assess the quality of the adaptation process
3.	Participation of key stakeholders to foster acceptance and ownership of the adapted guideline and ultimately promote its use	The involvement of stakeholders was acknowledged in reports of the adaptation process
4.	Consideration of context during adaptation to ensure relevance for local practice and policy	The context of application of the guidelines has been explicitly stated and the content adapted for the Kenyan health system
5.	Transparency to promote confidence in the guideline development process	The methodology in the adaptation process has been documented so that it is accessible and reproducible
6.	Flexibility to accommodate specific needs and circumstances in the health system	The guideline presents recommendations for diverse categories of PLWD (such as those with different stages of DR or comorbidities) who receive service in different clinical settings
7.	Respect for and acknowledgement of guideline materials used as sources	Citation and referencing have been used to acknowledge all source documents

	,	1
	Parameter	Specification
Р	Population	All patients with diabetes mellitus who are aged \geq 12 years
1	Intervention	Screening, diagnosis, referral, and management of diabetic retinopathy
Ρ	Professions (target users)	Primary care workers, diabetes care providers, eye care workers, administrators, policy-makers
0	Outcomes	All persons living with diabetes are screened for DR at least annually and blindness from DR is prevented
Н	Health care setting	Community, Primary, Secondary, and Tertiary level health care settings

Table 3 PIPOH summary of the clinical questions

existing guidelines were current. The team did not have a health economist and did not conduct an economic appraisal. As the guidelines were in English, we did not require expertise in foreign languages.

The TWG considered it is important to include patients' values and perspectives in the guidelines. A patient group was invited and PLWD who are clinicians were included, but despite our efforts, we did not succeed in having patients directly participate in the adaptation. We also aspired to have the participation of large groups of PLWD in a way that adequately represents the diversity of perspectives of PLWD from different geographical locations, social strata, and stages of disease. Since we do not have a comprehensive database of PLWD in the country, this was not feasible.

Rigor of development

We obtained high quality and current international guidelines. We examined the methods and the quality of the evidence used to formulate the recommendations for interventions for DR. We also considered the implications for resources and health service delivery in Kenya. Further, we searched for any recent evidence from systematic reviews and for relevant domestic research. In the absence of this, and judging the recommendations current and evidence-based, we incorporated them in our guidelines. The draft guidelines were subsequently reviewed by external multidisciplinary reviewers and pilot-tested in various health facilities. The guideline will be updated in 5 years to incorporate any new evidence that will have emerged.

Clarity and presentation of the guideline

We used the Conference for Guideline Standardization (COGS) checklist [20] as a guide to the content that needed documentation, although we excluded those items on the list that we did not consider necessary. The adapted guideline also includes additional information that was not in the international guidelines, such as the pattern of diabetes in Kenya, integrating DM and DR services, dissemination, and review plans.

In writing the guidelines, we avoided vague, nonspecific, or ambiguous terms and phrases. We aimed to produce a user-friendly guideline in which the precise recommendations are easily identifiable and clear, and the formatting is appropriate.

Applicability

We recognized the facilitators and barriers to the application of this guideline in the Kenyan health system. To overcome the barriers, the guideline provides tools to facilitate its implementation at the point of use. These inworkshop clude slides for training guideline implementers. Flexible 1- day training programs have been executed at implementing health facilities, conferences and training institutions, in conjunction with guidelines dissemination. The potential resource implications (equipment, staff, and training) and resultant work burden of applying the recommendations were considered. A monitoring and evaluation plan has also been included to assess adherence to recommendations and the outcomes of the implementation.

We required data on the costs of DR services in Kenya, but we did not undertake this as the Division of Non-Communicable Diseases had recently undertaken costing for diabetes services, including DR services. We lacked a costing model for guideline adaptation at the start of the exercise, but in our experience, the largest cost of the adaptation process was the production, pilot-testing, dissemination, and implementation of the print outputs. This may be reduced with progressive enhancement of digital literacy of the users and increased utilization of the electronic resources.

Editorial independence

The basic logistic needs of the adaptation process (administrative and meeting costs), as well as the implementation costs, were funded by The Fred Hollows Foundation. The funders did not influence the content of the guideline. There was a 100% consensus on the desired outcome of the guidelines, which is prevention of blindness from DR. Members and funders did not have any conflict of interest with respect to this outcome.

Context-specific modifications

Unlike the reference guidelines, the Kenyan guideline is designed for use by both diabetes and eye care clinicians, as well as other stakeholders in eye care. The population of interest is all PLWD aged 12 years and over, without any exceptions. The guideline attempts to take care of various complexities of service delivery. The role of different cadres in the screening, diagnosis, and treatment of DR is highlighted. Noting the variability in access to required equipment and skills, a referral mechanism has been determined through mapping of the services available in different facilities in the country. We found it practical to constantly relate the guideline to the patient pathway. Additional specifications have been made on linking diabetes care and eye care services, clinical governance for the services, and using the health information management systems (HIMS) to monitor the effect of implementation of the guidelines.

Outputs

This process has led to several outputs: (i) the national guidelines in various formats-print copies, electronic version, and an executive summary of the recommendations. The guideline has a national coverage and applies to persons with any type of diabetes.' (ii) quality assurance guidelines; (iii) mapping of DR services in the country; (iv) posters and leaflets for patients; (v) posters and checklists for clinicians; (vi) workshop slides for training health workers; and (vii) monitoring and evaluation tool. These outputs are to be used at the point of care by diabetes and eye care clinicians, as well as by administrators and policy makers. They were chosen because they were perceived to increase convenience of users and to intensify user adoption of the guidelines. They are in English, but they can be translated. An executive summary of the guidelines is published in a separate paper [12].

Feedback from pilot-testing indicated that the guideline is useful in various clinical and geographic settings in the country. It served an educational role for clinicians and reduced missed opportunities for screening and referral. The demand for the print outputs continued after pilot-testing. The point-of-use outputs were reported to boost user satisfaction because they contained simplified key messages for different users. During prospective collection of feedback, the lack of a tool to guide integration of diabetes and eye care services has been identified as a gap, and its development is being considered.

Discussion

The process of DR guideline development in Kenya has taken several iterative episodes. This trajectory may reflect the intensive work that guideline adaptation entails, as well as the capacity building that has resulted over that process. This experience is not unique to this initiative; long timelines have been reported in other guideline initiatives in the same context [28]. Contextual factors such as transitions in the guideline development team or critical leadership may result in delays, repetition of effort, and modification of approach.

We did not experience a shorter time scale for adaptation compared to the 2–3 year period suggested for de novo development or shorter timelines for adaptation [29]. This could be because we did not conduct this guideline development process continuously and the team of experts had other primary engagements. However, it is evident that the adaptation approach also requires a heavy time commitment. From our experience, which concurs with the literature, it is an essential pre-requisite to realistically determine the workload, resources, access to expertise, and the need for dedicated leadership [29].

The diverse expert skills and commitment of our multidisciplinary TWG are a recognizable success factor for our initiative. This is pertinent for both the internal and external validity (generalizability) of the guidelines [4, 17]. As adaptation requires significant investment from this team, the selection of potential TWG members is a pivotal priority step. Kenya being one of the countries with facing health workforce crisis [30], the major drawback we faced was availability of the TWG members to attend face to face meetings, as they had competing clinical and managerial responsibilities. This was predictable and inexorable, necessitating strategies to ensure group functioning was not interrupted.

The integrative participation method, which allowed both in-person participation and remote participation of the working group, helped to mitigate this constraint. This may have provided impetus to the process and achievement of the outcome. The enabling factor was that both electronic communication and face-to-face meetings were feasible, allowing for flexible engagement.

Involving patients in the process of guideline development is recommended [1, 29], because their opinions about the process of investigation and treatment and their outcomes are often quite contrary to the views of professionals. The lack of practical methods for engagement of PLWD precluded it. Although there are resources to guide this, such as the toolkit from Guideline International Network [16], local literature or precedence to guide such a process is lacking. Despite invitation of patient representatives to attend meetings and to review the drafts, we did not get this direct input. This may be because this type of involvement is not instinctively consistent with patient expectations or felt needs in our setup. Recruiting diverse groups of PLWD in a representative manner in a setting without comprehensive databases also requires contextual strategies. These limitations call for further local research.

We found that the guideline adaptation had the pattern of back and forth interlinked steps. We did not follow the ADAPTE steps sequentially as a stringently linear progressive and prescriptive tool. Further, even with the use of methodological tools, we found that there is need to maintain focus so that the process is not staggered. Clarity of the scope and significance to the patient pathway helped to maintain focus and continuity. Focus helped to avoid attrition, considering that universal completion of guideline development is not the norm [1].

Similar to de novo development, adaptation requires a review of the evidence and explicit use of valid evidence [15]. It would have enriched the process if we had additional domestic evidence. There is need for local research to fill gaps in scientific knowledge regarding interventions for DR in this population and health system. Local evidence on economic analysis such as cost-minimization and cost-utility evidence of the interventions is also necessary.

Adaptation itself has cost implications, and although we did not have a large dedicated budget for the process, we found it important to have a budget for the variable costs. In the absence of funding for fixed costs, we cannot provide an estimate of the funding required. Such estimates would help to calculate the cost-effectiveness of the process, considering the opportunity costs, and comparison with the cost of de novo development. The interventions described in the guidelines are clinically effective, but we need to investigate whether these interventions and the process of guideline development are also cost-effective in our setting. At present, we assume that guidelines augment the efficiency of DR services and optimize value for every shilling invested in the health system. In order to balance cost and accessibility of services, the guidelines promote the use of existing resources while aspiring to progressively mobilize the range of resources that are recommended.

National guidelines for various conditions may contain conflicting recommendations, which can be confusing for clinicians. This is especially the case for PLWD, as they often have comorbidities. We avoided such discrepancies by reviewing the other diabetes-related guidelines, and we recommend this as an important step in adaptation.

The output from the adaptation process is a guideline that is different from the generic guidelines (or source guidelines) and contains additional information that will be useful for the target user. An additional benefit is the production of additional tools for use at the point of care. This shows that the role of guideline adaptation is not limited to endorsing generic recommendations.

This initiative has coincided with other DR activities, such as the initial steps towards implementing regular retinal screening for PLWD attending diabetes services. This may have contributed to its success, and we can further leverage on this to market the guidelines. Given that countries in the African region, particularly the Eastern, Central, and Sothern African region may face similar needs to develop guidelines, a network or collaboration of sharing and learning may be an efficient approach to develop them.

Conclusions

Guideline adaptation is a structured investment -intensive process that is feasible at the country level. Rigor and focus are important in this process. The ADAPTE process and AGREE II instrument are valuable tools for this process, though it would be helpful for the generic guidelines to have an adaptation template for other contexts.

Multiple informational, technological, economic, social, and professional variables influence the effectiveness of guideline adaptation. Beyond the utility of this process in producing the outputs we required, it could also be useful to inform the development of other guidelines in similar contexts. Our experience has helped to provide insights on the use of the adaptation methodology in the African context. We have also identified guideline development as a potential area for collaboration.

Involvement of the end user of the guidelines (diabetes and eye health clinicians) in this adaptation process aims to increase adherence to the guidelines. We expect that DR services that were not routinely available to PLWD in Kenya will now become accessible as a response to the guidelines.

Implications for practice

Availability of a national guideline is a necessary but not sufficient impetus to standardize patient care. The extent to which the prevention of blindness from DR is realized will depend on the effectiveness of guideline dissemination and implementation, in tandem with other interventions.

Future research

An economic analysis is required to determine whether guideline adaptation is cost-effective. Research evidence is also required to determine the effective methods of involving patients, such as DR patients, in the adaptation process. In addition, the effectiveness of the guidelines in reducing DR blindness will need evaluation.

Additional file

Additional file 1: Search strategy for "Adapting Clinical Practice Guidelines for Diabetic Retinopathy in Kenya" (PDF 568 kb)

Abbreviations

ADA: American Diabetes Association; AGREE: Appraisal of Guidelines for Research and Evaluation; COGS: Conference for Guideline Standardization; CPG: Clinical practice guideline; DM: Diabetes mellitus; DR: Diabetic retinopathy; ICO: International Council of Ophthalmology; IOM: Institute of Medicine; PGEAC: Practice Guidelines Evaluation and Adaptation Cycle; PIPOH: Population, Intervention, Profession, Outcomes, Health care system; PLWD: People living with diabetes; TWG: Technical Working Group

Acknowledgements

The Ophthalmic Services Unit, Ministry of Health, Kenya and the Technical Working Group for DR guidelines provided the data on which this paper is

based. We are also grateful to the stakeholders, external reviewers, and end users who participated in the adaptation process.

Funding

The process of guideline adaptation was funded by The Fred Hollows Foundation and the Ministry of Health, Kenya. The publication costs are funded by the Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium.

Availability of data and materials

The data that support the findings of this study is available from the Ophthalmic Services Unit, Ministry of Health, Kenya. The data is available from the authors upon reasonable request to the corresponding author and with permission of the Ophthalmic Services Unit, Ministry of Health, Kenya.

Authors' contributions

NM drafted the manuscript, and had the responsibility for the decision to submit it for publishing. NM, SG, MGi, LM, CB and AF conceptualized the design. All authors except CB and AF were members of the Technical Working Group that collected data. All authors participated in interpretation of the data and critically reviewed the manuscript. NM revised the paper in consideration of feedback from co-authors. All authors read and approved the final manuscript.

Ethics approval and consent to participate

Not applicable

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Kenya Medical Training College, Nairobi, Kenya. ²Department of Ophthalmology, University of Nairobi, Nairobi, Kenya. ³Ophthalmic Services Unit, Ministry of Health, Nairobi, Kenya. ⁴Upper Hill Eye and Laser Centre, Nairobi, Kenya. ⁵Kabarak University, Nakuru, Kenya. ⁶Division of Non-Communicable Diseases, Ministry of Health, Nairobi, Kenya. ⁷Kenya Medical Research Institute, Nairobi, Kenya. ⁸Kenyatta National Hospital, Nairobi, Kenya. ⁹The Fred Hollows Foundation, Nairobi, Kenya. ¹⁰PCEA Kikuyu Eye Hospital, Kikuyu, Kenya. ¹¹London School of Hygiene and Tropical Medicine, London, UK.

Received: 31 July 2017 Accepted: 30 May 2018 Published online: 15 June 2018

References

- Field MJ, Lohr KN, Committee to advise the public health service on clinical practice guidelines for Institute of Medicine. Clinical practice guidelines: directions for a new program. Washington, DC: National Academy Press; 1990.
- 2. Institute of Medicine. Clinical practice guidelines we can trust. Washington DC: National Academies Press; 2011.
- Fervers B, Burgers JS, Haugh MC, Latreille J, Mlika-Cabanne N, Paquet L, Coulombe M, Poirier M, Burnard B. Adaptation of clinical guidelines: literature review and proposition for a framework and procedure. Int J Qual Health Care. 2006;18(3):167–76.
- Harrison MB, Legare F, Graham ID, Fervers B. Adapting clinical practice guidelines to local context and assessing barriers to their use. CMAJ. 2010; 182(2):E78–84.
- Dizon JM, Machingaidze S, Grimmer K. To adopt, to adapt, or to contextualise? The big question in clinical practice guideline development. BMC Res Notes. 2016;9(1):442.
- Ministry of Health, Kenya National Bureau of statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Nairobi: Ministry of Health, division of noncommunicable Diseases; 2015.
- 7. International Diabetes Federation. IDF Diabetes Atlas 7th Edition. 2015.

- Poore S, Foster A, Zondervan M, Blanchet K. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. Int J Health Policy Manag. 2015;4(1):19–28.
- Kupitz DG, Fenwick E, Martin Kollmann KH, Holz FG, Finger RP. Diabetes and diabetic retinopathy management in east Africa: knowledge, attitudes, and practices of hospital staff in Kenya. Asia-Pac J Ophthalmol. 2014;3(5):271–6.
- Bastawrous A, Wanjiku M, Kevin W, Madeleine B, Hillary R, Weiss Helen A, Macleod D, Foster A, Peto T, Blows P, Burton M, Kuper H. The incidence of diabetes mellitus and diabetic retinopathy in a population-based cohort study of people age 50 years and over in Nakuru. Kenya. 2017:17(19).
- 11. Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. Diabet Med. 2010;27(3):249–56.
- 12. Nyawira Mwangi, Muchai Gachago, Michael Gichangi, Stephen Gichuhi, Kibata Githeko, Atieno Jalango, Karimurio J, Kibachio J, Ngugi N, Nyaga P, Nyamori J, Zindamoyen ANM, Bascaran C, Foster A, For the technical working group. Clinical practice guidelines for diabetic retinopathy in Kenya: an executive summary of the recommendations. Journal of ophthalmology of eastern central and southern Africa 2017; 27(2) pp 33–39.
- Mwangi N, Macleod D, Gichuhi S, Muthami L, Moorman C, Bascaran C, Foster A. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Trop Med Health. 2017;45(41).
- The ADAPTE Collaboration. The ADAPTE process: resource toolkit for guideline adaptation ver 2.0, available at http://www.g-i-n.net. Accessed 10 May 2016.
- South African Medical Research Council. Guideline Toolkit by the SAGE Project [cited 2018]. Available from: https://guidelinetoolkit.org.za/gt.
- Guidelines International Network. G-I-NPublicToolkit: Patient and Public Involvement in Guidelines 2015 [cited 2017]. Available from: http://www.g-i-n. net/document-store/working-groups-documents/g-i-n-public/toolkit/toolkit-2015.
- 17. Graham ID, Morrison MB. Evaluation and adaptation of clinical practice guidelines. Evid Based Nurs. 2005;8:68–72.
- The AGREE Research Trust. AGREE 11 instrument, available at http://www. agreetrust.org. Accessed 20 May 2016.
- World Health Organization. WHO Handbook for Guideline Development, 2nd edition, 2014, available at http://www.who.int. Accessed 5 Apr 2016.
- Richard N, Shiffman MDMCIS, Paul Shekelle MP, Marc Overhage J, PhD MD, Jean Slutsky PAMSPH, Jeremy Grimshaw MP, Aniruddha M, Deshpande MD. Standardized reporting of clinical practice guidelines: a proposal from the conference on guideline standardization. Ann Intern Med. 2003;139:493–8.
- 21. American Diabetes Association. Standards of medical care in diabetes-2017. Diabetes Care. 2017;40(Suppl 1):S91–S93.
- International Council of Ophthalmology. ICO guidelines for diabetic eye careupdated 2017. San Francisco: International Council of Ophthalmology. 2017. http://www.icoph.org/downloads/ICOGuidelinesforDiabeticEyeCare.pdf.
- Royal College of Ophthalmologists. Diabetic Retinopathy guidelines, available at www.rcophth.ac.uk. Accessed 15 May 2016.
- Canadian Diabetes Association Clinical Practice Guideline Expert Committee, Boyd SR, Advani A, Altomare F, Stockl F. Retinopathy. Can J Diabetes 2013; 37(Suppl 1):S137–S141.
- 25. Ministry of Public Health and Sanitation. National Clinical Guidelines for management of diabetes mellitus. 2010.
- 26. Ministry of Health. Kenya national strategy for Prev Control of noncommunicable disease 2015-2020. 2015.
- International Diabetes Federation, Foundation TFH. Diabetes eye health: a guide for health professionals, available at https://www.idf.org/e-library/ guidelines/76-diabetes-eye-health-a-guide-for-health-professionals-en.html. Accessed 15 May 2016.
- English M, et al. Arch Dis Child. 2017;102:846–51. https://doi.org/10.1136/ archdischild-2017-312629.
- Harrison MB, Graham ID, van den Hoek J, Dogherty EJ, Carley ME, Angus V. Guideline adaptation and implementation planning: a prospective observational study. Implement Sci. 2013;8(49).
- World Health Organization. World Health Report 2006: Working together for health. 2006.



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?	Journal of Ophthalmology of Eastern Central and Southern Africa		
When was the work published?	2017		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	Not applicable		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

SECTION D – Multi-authored work

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I led the work described in the paper, drafted the manuscript and prepared subsequent revisions in consideration of the comments from the co-athors and peer reviewers
--	---

Student Signature:

Nyawira

Date: 17 SEPT 2019

Improving health worldwide

Date: SEP. 17, 2019

Clinical guidelines for diabetic retinopathy in Kenya: an executive summary of the recommendations

Nyawira M^{1,9}, Muchai G^{2,4}, Gichangi M³, Gichuhi S², Githeko K⁴, Atieno J⁵, Karimurio J², Kibachio J⁶, Ngugi N⁷, Nyaga P⁷, Nyamori J², Zindamoyen ANM⁸, Bascaran C⁹, Foster A⁹ for the Technical Working Group

¹Kenya Medical Training College, Nairobi, Kenya
²Department of Ophthalmology, University of Nairobi, Kenya
³Ophthalmic Services Unit, Ministry of Health, Kenya
⁴Upper Hill Eye and Laser Center, Kenya
⁵Kabarak University, Nakuru, Kenya
⁶Division of Non-Communicable Diseases, Ministry of Health, Nairobi, Kenya
⁷Kenyatta National Hospital, Nairobi, Kenya
⁸PCEA Kikuyu Eye Hospital, Kikuyu, Kenya
⁹London School of Hygiene and Tropical Medicine, London, United Kingdom

Corresponding author: Dr Nyawira Mwangi, Kenya Medical Training College, Nairobi, Kenya. Email: Nyawira. Mwangi@lshtm.ac.uk

ABSTRACT

All persons living with Diabetes Mellitus (DM) have a lifetime risk of developing Diabetic Retinopathy (DR), a potentially blinding microvascular complication of DM. The risk increases with the duration of diabetes. The onset and progression of DR can be delayed through optimization of control of blood glucose, blood pressure and lipids. The risk of blindness from DR can be reduced through cost-effective interventions such as screening for DR and treatment of sight-threatening DR with laser photocoagulation and anti-VEGF medications.

Several factors make it important to provide guidance to clinicians who provide services for diabetes and diabetic retinopathy in Kenya. First, the magnitude of both DM and DR is expected to increase over the next decade. Secondly, as the retina is easily accessible for examination, the early signs of retinopathy may provide clinicians with the first evidence of microvascular damage from diabetes. This information can be used to guide subsequent management of both DM and DR. Thirdly, there are notable gaps in service delivery for the detection, treatment and follow-up of patients with DR, and the services are inequitable. Strengthening of service delivery will require close collaboration between diabetes services and eye care services.

Following a systematic and collaborative process of guideline development, the first published national guidelines for the management of diabetic retinopathy have been developed. The purpose of this paper is to highlight the recommendations in the guidelines, and to facilitate their adoption and implementation.

Key words: Clinical practice guidelines, Diabetic retinopathy, Kenya

INTRODUCTION

Diabetes Mellitus (DM) is a priority non-communicable disease that requires multidisciplinary care and continuity of care. Its prevalence and incidence is increasing in every country. According to the STEPwise survey¹ for risk factors of non-communicable diseases in 2015, DM affects an estimated 2% of the Kenyan population aged 18-69 years with the highest proportion (5%) being in the 45-59 years age group. Every patient with diabetes is at risk of potentially blinding ocular complications, particularly Diabetic Retinopathy (DR). In turn, visual loss from diabetic retinopathy is associated with additional morbidity, such as falls, fractures and difficulties with seeing and taking medications. Both DM and DR are silent diseases that patients may be unaware of until they

cause complications. Clinicians attending to patients with these conditions have a role to reduce the associated morbidity, disability and mortality.

Diabetic Retinopathy (DR) is the leading cause of blindness in diabetes, and for this reason it warrants specific attention. It is estimated that a third of the people with diabetes have diabetic retinopathy, and one third of the latter (or 10% of those with diabetes) have vision threatening DR². Early signs of retinopathy or maculopathy in a patient with diabetes, identified on retinal examination, may be the first evidence of generalised microvascular damage from poor control of diabetes. This information would be very useful in planning subsequent holistic management of the patient. There is therefore need to strengthen links between eye care services and diabetes services.

Journal of Ophthalmology of Eastern Central and Southern Africa

The risk factors for DR include both modifiable and non-modifiable factors. Epidemiologic studies have identified the non-modifiable risk factors to include increasing duration of diabetes and genetic factors^{2,3}. The leading modifiable risk factors include poor control of blood sugar, poor control of blood pressure and dyslipidaemia^{2,4-6}. Service providers need to pay attention to these factors, as well as to other lifestyle factors associated with diabetes in order to delay onset or progression of DR. These are interventions for primary prevention of blindness from DR.

Stronger service delivery for DR is required within the existing health system⁷⁻⁹. Currently there are notable gaps in the screening, diagnosis, referral, treatment and follow-up. Although screening for DR and laser treatment are cost-effective interventions for prevention of blindness from DR¹⁰, there are inequities in access to them. Some of the services are underutilised and of insufficient quality. DR guidelines and use of clinical guidelines is an important step towards ensuring that all people with diabetes have access to quality DR services.

The purpose of these clinical practice guidelines is to give guidance regarding screening and diagnosis of DR, management of diabetes as it pertains specifically to DR, and treatment of DR. These guidelines apply to all patients with type 1 or type 2 diabetes who are at least 12 years old, who receive care at the primary, secondary or tertiary level of the health system. They should be used by health workers providing diabetes services and eye care services, as well as by administrators and policy-makers who plan for the resources for these services.

METHODOLOGY FOR GUIDELINES DEVELOPMENT

The development of these guidelines was a systematic, widely consultative process guided by an expert technical group over a lengthy period and involving many stakeholders. The process was guided by the use of several toolkits and guidelines which include: ADAPTE toolkit¹¹, PGEAC framework¹², AGREE 11 instrument¹³ and WHO handbook for guidelines development¹⁴.

The guidelines were adapted from existing relevant standards and guidelines, particularly the American Diabetes Association standards for medical care in diabetes¹⁵ International Council of Ophthalmology guidelines for diabetic eye care¹⁶ Canadian Diabetes Association's retinopathy guidelines¹⁷ and the Royal College of Ophthalmologists' diabetic retinopathy guidelines¹⁸. The following guidelines were also reviewed, and the recommendations are in line with their provisions: Kenya national guidelines for management of diabetes¹⁹, Kenya national strategy for the prevention and control of non-communicable diseases²⁰, and International Diabetes Federation's diabetes eye health guide for health professionals²¹. The adaptation strategy was chosen instead of de novo development in order to avoid duplication of effort, to use the available resources cost-effectively and to facilitate customization of the guidelines to reflect local context. These guidelines were identified through a literature search followed by application of the AGREE 11 instrument to evaluate the quality of the guidelines. Previous drafts of local DR guidelines were also reviewed. Care was taken to ensure that the guidelines are evidence-based, locally applicable, of high quality, and that the process of adaptation was consultative. The guidelines were subjected to external review by a multidisciplinary team as well as pilot-testing in various health facilities.

The process of guideline development is discussed in detail in a separate paper.

Key messages

- 1. Stronger service delivery is needed for People Living with Diabetes (PLWD) in Kenya in relation to DR. There is need to develop strong links between diabetes services and eye care services within the existing health system.
- 2. Blindness from DR is avoidable, but only if diabetes care givers and eye health professionals perform their roles in ensuring early detection and treatment of DR.
- 3. Screening is important for early detection of treatable diabetic retinopathy. It is also a cost-effective intervention for reducing blindness from DR. All patients with diabetes aged 12 years and above should have a retinal examination (usually a dilated eye examination or a retinal photograph) once a year or more frequently if recommended by the eye specialist.
- 4. Consistent and appropriate metabolic control reduces the onset and progression of sight-threatening diabetic retinopathy.
- 5. Laser photocoagulation therapy, local intraocular pharmacological therapy and surgery reduce the risk of significant visual loss.

Recommendations

Domain: Strengthening links between diabetes services and eye care services at primary, secondary and tertiary level of care

- 1. All health workers providing diabetes services should raise awareness of PLWD on diabetic retinopathy and support them to access eye examination (Panel 1).
- 2. The health worker attending to a PLWD at any health care level should use a checklist (Panel 2) to identify whether the patient has had a retinal examination in the preceding 12 months. Any patient who has not should be referred to the nearest facility for screening for DR.

- 3. The screening examination for DR should be performed by trained personnel (health care worker, eye care workers or technicians) and should consist of a minimum of:
 - a. A retinal examination appropriate for DR, which would include dilated direct or indirect ophthalmoscopy, slit lamp bio-microscopic examination of the retina or retinal photography.
 - b. Visual acuity test using distant and near charts. If the visual acuity is reduced, then pin hole test is also performed."
- 4. All PLWD require follow up, therefore service providers in both eye clinics and diabetes clinics should encourage compliance with follow-up.
- 5. A service charter for diabetes services and eye care services should be available in all clinics offering these services to inform PLWD of the minimum care they should expect and to highlight the importance of yearly comprehensive risk assessment including dilated eye examination.
- 6. The ophthalmologist and the physician/diabetologist/ endocrinologist will provide clinical governance as the team leads for diabetes services and eye care services. The clinical governance team would also include: (a) liaison diabetes nurse (from the diabetes clinic), (b) liaison ophthalmic nurse (from the eye clinic), (c) designated medical records officer and (d) a biomedical / equipment maintenance technician.
- 7. The physician/diabetologist/endocrinologist and the ophthalmologist would be responsible for oversight on the screening for DR, training of health workers and data management.
- 8. The Ophthalmic Services Unit would be responsible for clinical governance for DR services at national level.
- 9. Data would be systematically collected in both diabetes clinics and eye clinics. The liaison nurse in the diabetic clinic and eye clinic would forward the captured data to the designated records officer for reporting.
- 10. The Ophthalmic Services Unit would coordinate the use of this data to inform decisions on service improvement strategies.

Panel 1: Key messages for health workers attending to *PLWD*

1 in 3 of patients with diabetes has diabetic retinopathy. Act to save their vision today!

1	Send all newly diagnosed patients with diabetes for a baseline retinal examination.
2	Ensure all patients with diabetes have an annual retinal examination.
3	Refer any patient with diabetes who has poor vision to an eye specialist urgently.

4	Do not wait for visual loss to refer patients with diabetes to an eye specialist- 'prevention is better than cure.'
5	Glycated hemoglobin (HbA1c) is a good indicator for long-term sugar control and should be done at least annually for all patients. Good glycemic control prevents or delays diabetic retinopathy.
6	Assess for other end organ damage- oral examination, diabetic foot review, renal and cardiac function, tests for neuropathy at least annually
7	Assess for hypertension, hyperlipidemia and other co-morbidities as these may impact negatively on diabetes and diabetic retinopathy.
8	Send all pregnant patients with diabetes for a retinal examination at least each trimester and post-delivery, or more frequently if recommended by the eye specialist.

Panel 2: Checklist for screening for DR

Patients name:
Date of birth:
Date of screening:
Duration of diabetes:

Have you ever had an examination of the back of the eye (either a photograph was taken or drops were instilled into the eye before examination)? a. YES b. NO

How long ago was the last examination of the back of the eye?

a. < 12 months b. ≥ 12 months

Where was the eye examination done? facility

Domain: Eye examination for DR

Annual dilated and comprehensive eye examination is recommended for all patients with diabetes, aged 12 years and above, starting at the time of diagnosis of diabetes mellitus, unless the eye specialist recommends a different frequency. For type 1 diabetes, an eye exam should be done at diagnosis, at 5 years of diagnosis and annually thereafter.

Journal of Ophthalmology of Eastern Central and Southern Africa

Domain: Patient-centred care

- 1. Health workers should provide verbal and written information on diabetes, diabetic retinopathy, and on the health care that is needed, including self-management.
- 2. All PLWD should receive regular and individualised self-management support on healthy diet, appropriate physical activity and weight reduction if they are overweight. All smokers should be encouraged to quit smoking.
- Patient education materials such as posters, leaflets, booklets and flyers on DR should be available (Panel 3) to patients and to peer support groups.
- 4. Diabetes support and structured self-management education should also be provided to family members of PLWD.
- 5. PLWD should be treated with dignity and involved in decision-making for their care. The results of their examination and the implications should be explained to them, and they should be encouraged to ask questions. The presence of co-morbidities should be taken into consideration in the care of each patient.

Panel 3: Key messages for patients

DID YOU KNOW THAT DIABETES AFFECTS THE EYES?

What can you do to prevent blindness?

- 1 Diabetes mellitus is marked by high sugar levels in blood. High blood sugar destroys small blood vessels in the body including those at the back of the eyes, leading to a condition called diabetic retinopathy.
- 2 Damage to the eyes is slow, painless, gets worse with time and finally leads to blindness if not treated in good time.
- 3 The damage to the eyes needs to be detected early, before permanent damage occurs.
- 4 An eye check by an eye specialist can detect damage to the eyes before symptoms develop. During the examination, the eye specialist will check vision, and instil an eye drop to assess the damage in the eye. Both eyes need to be examined.
- 5 For prevention and treatment of diabetic retinopathy, the eye specialist may advise on sugar, blood pressure, and lipid control.
- 6 For treatment of diabetic retinopathy, the eye specialist may perform laser or administer injections in the eye or perform eye surgery.
- 7 All persons with diabetes should have their eyes checked once every year by an eye specialist, even before any symptoms or poor vision develop or as frequently as recommended by the eye specialist.

- 8 A child with diabetes should have the eyes checked annually from the age of 12 years, or more frequently if recommended by the specialist.
- 9 A pregnant mother with diabetes should undergo an eye check by an eye specialist at least once every trimester, and soon after delivery, or as frequently as recommended by the eye specialist.
- 10 If the eyes are found to be normal at your eye check by an eye specialist, please continue with an eye check annually. If you notice any abnormality with your eyes, visit the eye specialist as soon as possible.

Domain: Metabolic control

- 1. All PLWD should be asked about the level of control of glucose, blood pressure and lipids.
- 2. To prevent the onset and delay the progression of diabetic retinopathy, people with diabetes should be treated to achieve optimal control of blood glucose.
- 3. Regular monitoring of blood sugar at home should be encouraged.
- 4. Regular monitoring of blood pressure in a health care setting or at home should be encouraged. Target blood pressure is 140/90 mmHg. Drugs blocking the Renin-Angiotensin System (RAS) may have benefits, particularly for mild retinopathy, but should be discontinued during pregnancy.
- 5. A comprehensive biochemical profile (risk assessment) should be done at least annually, and include fasting lipid profile, HbA_{1c}, urine microalbumin among other tests.
- 6. Aim for a target glycosylated haemoglobin (HbA $_{1c}$) of <7%.
- 7. Serum fasting lipid profile should be assessed at diagnosis and annually. Consider statins in primary and secondary prevention of DR but discontinue statins in pregnancy.

Domain: Pregnancy

- 1. All female PLWD of reproductive age should be asked if they are pregnant.
- 2. Patients should be assessed for diabetic retinopathy before pregnancy, at least once every trimester of pregnancy, as well as within 6 months after delivery or more frequently if recommended by the eye specialist.
- 3. Statins and angiotensin inhibitors should be discontinued in patients who are planning for pregnancy.
- 4. All women of child-bearing age who have diabetes should be educated that pregnancies should be planned.

Domain: Screening programs for DR

- 1. All PLWD should be screened for DR at least once a year, irrespective of whether they have ocular symptoms or not.
- 2. Screening programs can utilise whatever screening method is available (ophthalmoscopy, slit-lamp biomicroscopy and retinal photography), and should be conducted by a suitably trained person. Pupil dilation is recommended. Visual acuity should be assessed before pupil dilation.
- 3. Photography based screening
 - a. Where a fundus camera is available, ideally the fundus camera should be located in the diabetes clinic.
 - b. Fundus photographers should be trained to identify cataract, other causes of media haziness and glaucoma on the images. Patients who have these pathologies should be referred to an ophthalmologist. Ultrasonography may be useful in assessing the posterior segment in the presence of cataract or vitreous haemorrhage.
 - c. Regular retraining in form of short courses (either online courses or standard contact courses) should be provided for screeners.
- 4. The following patients should be referred to an ophthalmologist:
 - a. Where the screening examination is unsuccessful, or the results of the visual acuity test or retinal examination are unclear
 - b. Where the retinal examination is unsuccessful, for example due to additional pathologies
 - c. Any grade of retinopathy, except mild nonproliferative retinopathy
 - d. Visual acuity worse than 6/12 and all patients with ocular symptoms
- 5. Screening should identify true positives (patients with DR). For this to be achieved, it is important to use the correct equipment, adhere to the standards of practice, make correct diagnosis and have a quality assurance mechanism. The guidelines for quality assurance are provided as an addendum to the guidelines.
- 6. All health workers have a role in ensuring patients undergo screening. All health workers should also document and collect data on screening activities, as the data is useful for planning and monitoring services. Health workers at each health facility will collect this data using standard monthly data collection forms. The Ophthalmic Services Unit in the Ministry of Health will coordinate the collection of data.

Domain: Diagnostic evaluation of patients at the eye clinic

Once the person with diabetes has been referred to an eye specialist, he or she should undergo a complete ophthalmic assessment. Ophthalmic evaluation by an eye care worker is available at the secondary level of the health system. This should include taking medical history, assessing visual acuity, and identifying and grading DR or Diabetic Macula Oedema (DME), using standard procedures described in the guidelines.

Domain: Referral pathway

Once a decision for referral for evaluation or treatment has been made, it should be carried out as soon as possible. The nearest health facility offering DR services will be identified (mapping of services has been conducted, and this information is provided as an addendum to the guidelines), and patients will be referred to reach the facility on the designated days that the services can be provided.

Domain: Treatment interventions

The ophthalmologist will make the final diagnosis and the decision on the treatment that the patient should receive. There is evidence from Cochrane systematic reviews included in Panel 4 to support the use of laser photocoagulation in proliferative diabetic retinopathy²² anti-VEGF injections in diabetic macula oedema²³ and intravitreal steroids in refractory diabetic macula oedema²⁴. Laser or intravitreal injections can be administered by the ophthalmologist at secondary level. Surgical interventions for DR will be provided by the vitreo-retinal surgeon at tertiary level. Practical information on these procedures is provided in the guidelines.

Panel 4. Evidence from Cochrane systematic reviews for Interventions used to treat diabetic retinopathy.

Laser photocoagulation is beneficial in reducing the risk of severe visual loss and the risk of progression 12 months after treatment in patients with proliferative diabetic retinopathy compared to no treatment or deferred treatment. However most trials here are old and the quality of evidence is judged as low²².

There is very low or low quality evidence from randomized controlled trials that anti-VEGF injections are effective in patients with proliferative diabetic retinopathy but they prevent intraocular bleeding²⁵.

There is high quality evidence that anti-VEGF injections are effective in preserving and improving vision in patients with diabetic macula oedema compared to grid laser²³.

Intravitreal steroids delivered either by injection or implants may improve visual outcomes in patients with persistent or refractory diabetic macula oedema but it is unclear whether they are beneficial in other earlier stages²⁴.

Domain: Follow up

All PLWD screened for DR will require follow up. The frequency of follow-up depends on the clinical findings, and the grading/severity of DR, as described in the guidelines.

Domain: Patients with low vision

Refer the patient with low vision (best corrected visual acuity of <6/18) for rehabilitation. PLWD who would benefit from counselling and social services should be referred as appropriate.

Domain: Monitoring DR services

Specific process and outcome indicators will be used to monitor services on a quarterly basis at each level of service delivery, using the hospital health management and information system. Health workers at each health facility will therefore collect this data using standard monthly data collection forms. The Ophthalmic Services Unit will coordinate the collection of data. The indicators of interest are listed in the guidelines.

CONCLUSIONS

Patients with diabetes require specific care relevant to diabetic retinopathy, which includes patient education, screening, referral, treatment and follow-up. These are the first published national clinical practice guidelines for the screening and management of diabetic retinopathy. Their goal is to ensure best practice throughout the whole pathway from primary care to tertiary care. Implementation of the guidelines has potential to reduce blindness from DR.

The Working Group welcomes feedback from all users of these guidelines. In particular, data on the enablers and challenges experienced in the use of the guidelines would be very useful in informing revisions on the guidelines. Please email feedback to the Ophthalmic Services Unit, Ministry of Health, through ophthalmicserviceske@ gmail.com.

ACKNOWLEDGEMENTS

We acknowledge The Fred Hollows Foundation for funding the process of guidelines development.

Conflict of interest: Nothing to declare.

REFERENCES

- 1. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases, 2015.
- 2. Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, *et al.* Global prevalence and major risk factors of diabetic retinopathy. *Diabetes Care.* 2012; **35**:556-564.
- 3. University of Winsconsin. The Winsconsin Epidemiologic Study of Diabetic Retinopathy (WESDR) 2015 [cited 2015 151215]. Available from: www.epi.ophth.wisc.edu.
- 4. The Diabetes Control and Complications Trial Research Group. The effect of intensive treatment of diabetes on the development and progression of longterm complications on insulin-dependent diabetes mellitus. *New Engl J Med.* 1993; **329**(14):
- 5. King P, Peacock I, Donnelly R. The UK Prospective Diabetes Study (UKPDS): clinical and therapeutic implications for type 2 diabetes. *Br J Clin Pharmacol.* 1999; **48**:643-648.
- 6. The ACCORD Study Group. Effects of intensive blood-pressure control in type 2 diabetes mellitus. *New Engl J Med.* 2010; **362**:1575-585.
- Poore S, Foster A, Zondervan M, Blanchet K. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. *Int J Health Policy Manag.* 2015; 4(1):19-28.
- 8. Kupitz DG, Fenwick E, Kollmann KHM, Holz FG, Finger RP. Diabetes and diabetic retinopathy management in East Africa: Knowledge, Attitudes, and Practices of hospital staff in Kenya. *Asia-Pacific Ophthalmol.* 2014; **3**(5):
- 9. Bastawrous A, Mathenge W, Wing K, Bastawrous M, Rono H, Helen WA, *et al.* The incidence of diabetes mellitus and diabetic retinopathy in a populationbased cohort study of people aged 50 years and over in Nakuru , Kenya. 2017:1-14
- 10. Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. *Diabetic Med.* 2010; **27**(3):249-256.
- 11. The ADAPTE Collaboration. The ADAPTE process: resource toolkit for guideline adaptation ver 2.0, available at http://www.g-i-n.net. 2009.
- 12. Graham ID, Morrison MB. Evaluation and adaptation of clinical practice guidelines. *Evidence Based Nursing (EBN)*. 2005; **8**:68-72
- 13. The AGREE Research Trust. AGREE 11 instrument, available at http://www.agreetrust.org. 2009.
- 14. World Health Organization. WHO Handbook for guideline development, available at http://www.who. int. 2012.

- American Diabetes Association. Standards of Medical Care in Diabetes-2017. *Diabetes Care*. 2017; 40(Suppl 1):
- International Council of Ophthalmology. ICO Guidelines for Diabetic Eye Care-updated 2017. San Francisco, California: International Council of Ophthalmology, 2017 23 February 2017. Report No.
- Canadian Diabetes Association Clinical Practice Guideline Expert Committee, Boyd SR, Advani A, Altomare F, Stockl F. Retinopathy. *Can J Diabetes*. 2013; **37** (Suppl 1):S137-41.
- 18. Royal College of Ophthalmologists. Diabetic Retinopathy guidelines, available at www.rcophth. ac.uk. 2012.
- 19. Ministry of Public Health and Sanitation. National Clinical Guidelines for management of diabetes mellitus. 2010.
- 20. Ministry of Health. Kenya National Strategy for Prevention and Control of Non-communicable Disease 2015-2020. 2015.
- 21. International Diabetes Federation, Foundation TFH. Diabetes eye health: a guide for health professionals, available at www.idf.org/eyecare. 2015.

- 22. Evans JR, Michelessi M, Virgili G. Laser photocoagulation for proliferative diabetic retinopathy. *The Cochrane Database of Systematic Reviews*. 2014; **24**(11):Cd011234.
- 23. Virgili G, Parravano M, Menchini F, Evans JR. Anti-vascular endothelial growth factor for diabetic macular oedema. *Cochrane Database of Systematic Reviews.* 2014; (10)
- 24. Grover DA, Li T, Chong CCW. Intravitreal steroids for macular edema in diabetes. *Cochrane Database of Systematic Reviews*. 2008; (1)
- 25. Martinez-Zapata MJ, Martí-Carvajal AJ, Solà I, Pijoán JI, Buil-Calvo JA, Cordero JA, *et al.* Antivascular endothelial growth factor for proliferative diabetic retinopathy. *Cochrane Database of Systematic Reviews.* 2014; (11)

Chapter Six

Health system strengthening: The development of an open online training course on control of DR

6.0 Overview

The unavailability of educational opportunities for health workers providing care to people living with diabetes (PLWD) hampers the capacity of the workforce for control of DR (chapter three). Health systems strengthening through health workforce capacity building is therefore a critical factor for expanding availability of comprehensive DR services. Online courses provide a useful model allowing flexible access to training for health workers providing services in different institutions while minimizing work-place disruption. Although much attention has been given to the number and distribution of health workers, little focus has been directed at developing models of training that address these real-life workplace needs.

This chapter describes the experience and lessons learnt from developing an online course to strengthen capacity of health workers in Kenya to deliver services for DR. Section 6.1 highlights how online learning may address the needs for continuing education for health workers. Section 6.2 describes the role of mentorship in the development of the course. Section 6.3 and 6.4 describe the systematic process of course development following the theories of instructional design and adult learning. The course will be launched in late 2019 and the implications are discussed.

6.1 Online learning for continuous professional development of health workers

Continuous professional development (CPD) is required because preservice training alone is not sufficient to keep health workers equipped to respond to the rapidly evolving changes in health care.¹⁻³ Importantly, CPD has been linked to professional competence, health worker motivation and job satisfaction.^{4, 5}Traditional forms of CPD include educational meetings with face-to-face contact, such as conferences, symposia and workshops, but the literature points out numerous challenges. These include difficulties juggling clinical workload and time-off for CPD, especially with inflexible CPD sessions that do not consider work commitments.^{1, 6, 7} In low-resource settings, where the need for training and CPD may be highest, the limited CPD sessions available often occur off-site, in urban areas, at tertiary training institutions or central teaching hospitals. For those living remotely, the expenses in terms of registration fees, accommodation and transport are often prohibitive.

Online learning can enhance education experience, support professional development, ease time constraints, overcome geographic limitations and offer greater flexibility in CPD.⁸ Online platforms for CPD may enable care providers to remain at their place of employment while continuing their education.⁷ Richmond et al in their systematic review found that online methods may be as effective as alternative methods for training health workers.⁹ Such platforms are gaining in popularity because of their relatively low costs, high flexibility, and reduced dependence on geographical or site boundaries.¹⁰ Studies have shown that online CPD opportunities are acceptable across a diverse group of health care workers in sub-Saharan Africa.^{6, 11}

Despite these benefits, there are also barriers to the use of online CPD such as cost of developing programmes, possible technology problems, lack of computers, lack of continuous

internet and data costs. Sub-Saharan Africa has a lower level of internet use than other regions, but this is rapidly changing in East Africa.¹² Kenya is one of the countries with the highest smart phone penetration and mobile adoption.^{13, 14} This presents an opportunity for online CPD learning to take place. However investment in developing CPD courses is a complex undertaking, this is likely to remain a significant barrier.¹⁵ Documenting experiences related to course development would therefore be of interest to policy makers, medical educators, regulators and front line workers involved in designing or supporting implementation of CPD interventions.

Developing CPD courses is particularly pertinent in the subject of diabetes eye health because of the dynamic epidemiology of diabetes-related vision loss, and the technology and competencies required for addressing it. ¹⁶⁻¹⁸ Studies from Africa have reported insufficient training about diabetes, on specific procedures such as examination of the retina (fundoscopy) and on the management of DR.¹⁷ In addition, the recent advances in knowledge and technology in the field of diabetes eye health will only offer benefits to PLWD if health workers remain up to date in the field. At present, there is a paucity of local online CPD courses on this subject. However, there is evidence from a free online course 'Global Blindness: Planning and managing eye care services' that appropriately designed online courses can deliver training at scale, support educational capacity building and impact health practice in LMICs.¹⁹ Another major weakness is the insufficient information on how the few courses available were developed, the contextual factors considered and the instructional design methodology used.^{9,} ^{15, 20}For that reason, the existing literature does not adequately advance an understanding of how courses can be developed, hence failing to promote interpretation and replicability. To address this gap, the process of developing this open online course on control of DR, and the lessons learnt from this process are described in this chapter.

6.2 Adult learning theory

In order for knowledge transfer to translate into effective CPD for health workers, there is a need to leverage adult learning theories, because they provide the conceptual frameworks for the acquisition of knowledge, skills, and attitudes to achieve behaviour change.²¹ Adult learning theories hold a set of assumptions about how adults learn and emphasize the value of the process of learning. They postulate that adults are independent and self-directing, have prior knowledge and experience, apply learning to real life, are interested in problem solving and have internal motivation for learning.^{15, 22, 23} Although the theories were developed for traditional classroom education, the basic principles effectively transfer to the online learning environment. Online CPD programmes take into account these principles through integrating the following factors: - voluntary participation, autonomy of learning.^{7, 24} Examples of how these approaches were applied in the development of the course are provided in the sections below.

6.3 The ADDIE instructional design framework

The use of an instructional design framework in course design helps to ensure that the process is credible and transparent, and that the course is likely to be relevant and effective in professional improvement.^{25, 26} A literature search for evidence on effective models of online course development identified at least 30 models, but the ADDIE (acronym for <u>A</u>nalysis, <u>D</u>esign, <u>D</u>evelopment, <u>I</u>mplementation, and <u>E</u>valuation) model is the most ubiquitous, with good evidence of its effectiveness.^{27, 28} The framework was first used during the World War II when the United States military developed strategies for rapidly training people to perform complex technical tasks.²⁹ It has remained the most influential instructional design framework in the

literature.^{30, 31} It involves analysis of learning needs, the design and development of a curriculum, and the implementation and initial evaluation of the training programme.

The ADDIE framework has previously been used to design high quality educational programmes, both print-based and online.³² It is particularly useful where the focus of the programme is targeted toward changing participant behaviour.³³ It takes into account learning theories, the learner's needs and context, and can be applied to teach knowledge, skills, or attitudes.^{28, 30} In health and medicine, it has been used to design training interventions to change practice behaviours in the management of various medical conditions.^{26, 28, 33-35}

6.4 Mentorship for course development

Improving clinical care rapidly, comprehensively, at large scale, and sustainably is a fundamental challenge for health systems.³⁶ Mentorship is an effective approach for developing knowledge and skills, and it could potentially be used to enhance knowledge transfer between researchers, educators and clinicians.³⁷ For example, mentoring can build the expertise and self-efficacy of mentees from LMICs to develop online courses.³⁸ However, little empirical research in healthcare settings has specifically described the application of mentorship for this purpose.^{36, 37}

Insights from mentoring initiatives can inform the conduct of future mentoring initiatives to optimize chances of success. An overview of mentorship to develop this course within the UNESCO *Open Education for a Better World* (OE4BW) is documented and discussed below, so that future efforts towards developing similar mentoring programmes could draw from this learning and experience.³⁹ This is an international online mentoring programme run by the United Nations Educational, Scientific and Cultural Organization (UNESCO) to unlock the

potential of open education in achieving the United Nations' 2030 Sustainable Development Goals (SDGs).⁴⁰ Mentees develop course materials that address one or more SDGs, finding a solution with impact on society and ensuring quality of the course materials. The mentor's role is to advise on the design of the course, and to ensure that the course is supportively but critically reviewed. At the end of the six-month mentoring period, the mentees and mentors attend a face-to-face workshop organized by UNESCO to present the outputs and lessons learnt. The formal mentoring structure provided by UNESCO ends at six months, but mentoring may extend post-programme, given that courses often take a long time to develop. Some mentoring programmes have reported difficulty maintaining the mentoring relationship during this transition period, but I did not experience this.⁴¹ UNESCO provides a website to host information on the course and mentorship.⁴⁰ This is particularly helpful because it helps to form a community of practice for professional networking.

6.5 Developing the online course on control of DR using the ADDIE framework

Figure 6-1 depicts the conceptual model of the five phases of the ADDIE framework as operationalized in the development of this course. The analysis, design, development and implementation phases are cyclical and iterative. Each phase depends upon the successful completion of the preceding phase.²⁶ Formative evaluation provides feedback specific for each phase, facilitating refinement of the outputs.²⁵ Table 6-1 shows the tasks and outputs in each phase. Experts on diabetes, ophthalmology and medical education reviewed the outputs from each of the phases for quality assurance.



Figure 6-1: Conceptual model of the ADDIE framework as applied in the development if this course

Table 6-1:	Tasks and output	s within the ADDIE	framework in this course

Phase	Sample tasks	Sample outputs
<i>Analysis</i> Defining what is to be learnt	 Identification of target learners Identification of needs and problems of target learners Analyse subject field Work with content experts to determine the content Search for evidence Search for materials 	 Target learners identified Description of current constraints Current environment described Course goal and objectives

Design

Determine the learning objectives and strategies

Development Prepare course materials

Implementation Course delivery

.

Evaluation

Formative and summative evaluation of the course

- Identify optimal methods of content delivery
- Identify the learning theory
- Identify the media for course delivery
- Consult with subject experts
- Identify the creative commons licence
- Write objectives
- Plan instruction
- Identify resources
- Identify course site
- Vetting of design document by key stakeholders
- Work with multimedia team, content experts and instructional designers
- Develop learning materials
- Peer review
- Work with institutional accreditation team
- Upload content on the course site
- Testing the prototype
- Official launch
- Course advertisement
- Support learners
- Pre-course user survey
- Post-course user survey
- Monitoring course analytics

- Measurable objectives
- Instructional strategyProduct prototype
- specificationsStoryboard

- Slides
- Scripts
- Videos
- Graphics
- Exercises
- Course link
- User feedback
- Course analytics
- Course evaluation
 reports
- Revised prototype based on the evaluation
6.5.1 Analysis

The first step in course development is to analyse the need and context of the learners. The health system assessment (chapter three) provided the evidence on the need for the course. The literature provides evidence of other courses developed as a result of needs identified from health system assessment.³³ The health system assessment also identified the population (P) of interest for the training, the intervention (I), which is the development of the course, the comparator (C), which is the current situation with few educational opportunities. The expected outcome (O) of the course development endeavour is the availability of the online course and the achievement of the learning outcomes. This analysis phase also involved searching for the relevant published evidence (E) and setting a timeframe (T). The EPICOT format⁴² provided the context for the course development, Table 6-2.

Evidence	Online courses are acceptable and feasible in this setting, as described within
	this chapter
Population	Heath workers providing care for PLWD
Intervention	Development of a flexible course on eye health for PLWD, which will enable
	learners to understand eye health complications in diabetes, especially DR,
	holistic care for PLWD, the role of the health care provider in the care
	pathway and how to strengthen health systems
Comparator	The current situation, where training opportunities are few, for both service
	providers and trainers
Outcome	Availability of a free online course that can be used on mobile devices or
	computers
Timeliness	Course should be ready for launching by World Diabetes Day, 2019

Table 6-2: EPICOT	description	for the (Control of	f Diabetic	Retinopathy	course
-------------------	-------------	-----------	------------	------------	-------------	--------

The target population is the heterogeneous and multidisciplinary group of health care providers who provide services along the care pathway described in chapter three. The topic of focus was chosen to address the training needs identified in the health system assessment. The needs of the target learners, including the working environment, technological and economic needs were identified. Analysing these factors helps to ensure the relevance of the course within the realities of the health system.^{12, 43} Over the long term, the course is expected to improve provision of DR services. Accreditation is an anticipated need, and this is expected to be provided through local training institutions in the future.⁴³ This is not easy to achieve, but experts from the two institutions that currently offer ophthalmology-related training in the country have reviewed the course materials, as a starting point for further engagement. In this way, health system strengthening (HSS) through development of an online course has potential to influence education services and health services, and support local institutions.^{44, 45} It also shows the need for sustainability and evolution of HSS interventions over time, in response to local needs.^{44, 46} However, this long-term impact of the course will need to be evaluated.

Although the course was developed for the local context, it was also intended to have an appeal to an international audience. As other researchers have documented, developing an online course from a LMIC is challenging because the audience might undervalue the quality of the course content, since the MOOC originated from an LMIC.^{47, 48} To address this potential challenge, quality assurance through peer review from experts from a mix of income settings was planned. It is significant that institutions in LMICs offer only 3% of the health and medicine-related online courses, and this is partially attributable to this barrier to entry into knowledge production and curation.^{49, 50} Another factor might be challenges related to weak health systems, where health workers are few and there is lack of time and incentives to

engage in knowledge production.⁴⁵ What is critical is that the unequal production and accumulation of knowledge is as worrying as the inequitable distribution of wealth and health.^{48, 51} Academic and professional communities in LMICs may be missing opportunities to address this gap.

Subject matter analysis is essential to defining the content of a course. Experts (ophthalmologists and physicians) who are active in both clinical and training services were consulted and clinical guidelines on diabetes eye care were reviewed.⁵²⁻⁵⁵ Based on this consultation, the literature review on the subject (chapter two), the gaps identified in the health system assessment as well as the standards set in clinical guidelines, there was consensus that the course should be organized in four units (Box 1). In consideration of the principles of adult learning, the course will be available throughout the year, to provide flexible access to learners.

A search for existing online resources identified a few courses relevant to diabetes eye heath, such as from the International Centre for Eye Health⁵⁶ and the International Diabetes Federation.⁵⁷ However these courses are only available during two or three runs in a year to participants who register, which is a limitation to access for the target population.⁵⁸ Using best practices to design, develop, implement, and evaluate courses designed locally can provide creative solutions to this accessibility gap.⁴³ At the same time, the local courses can include links to the external online courses, thus maximising access to all the available learning opportunities. The strategic linking of the course to other online courses offers an opportunity to expand the educational scope of this HSS initiative. This might also trigger the development of partnerships to offer additional courses.

Table 6-3 Course Units

Unit 1	Understanding the disease
Unit 2	Understanding the patient
Unit 3	Understanding the health care provider
Unit 4	Understanding the health system

6.5.2 Course Design

The contextual realities of the target learners were considered as design decisions were made. English was chosen as the preferred language because it is the language used in educational programmes in Kenya. Language appropriateness and tailoring materials to reflect countryspecific health care realities in LMICs are crucial to successful CPD.¹² In addition, the use of English will reduce barriers to access by an international audience.^{49, 58} The technological requirements for this course are access to a computer or a smart phone and internet connection. The course is designed to be compatible with all internet browsers at no cost to the learner, in order to promote access. In line with the principle of autonomy, participation is voluntary, self-paced, and without rigid deadlines, which allows learners to revisit the materials whenever they need to. Learners from LMICs are known to favour these attributes.^{6, 47}

The instructional materials include different media elements such as text, graphics, images, research papers, and short .mp4 videos that are downloadable for offline use. Simplicity in course design and low technological demand are imperative given the uneven technological resources of learners.^{15, 58} All videos are captioned and text description is provided for non-text visuals, to create engaging learning experiences for adult learners with different skill-sets,

interests or disabilities. Adult learners also learn better where they can see the immediate value and application of content, hence real world case scenarios are included. Useful hyperlinks to relevant external resources are also provided to facilitate autonomous exploration and curation, without overwhelming the learner with too many of them.¹⁵ To allow the learners to reuse the materials, all the course materials are published under a creative commons licence.⁵⁹

Social media use is integrated in the course design, since learners are likely to have used this strategy to accumulate experience in digital literacy, and this can stimulate active engagement of adult learners.²⁴ The course uses Google Drive and Dropbox to manage documents, Flickr for photographs, and Youtube for videos. Twitter and LinkedIn are used to connect with the course instructor. Pre- and post-course surveys will further enhance learner-instructor communication.

Open-source online platforms for hosting the course were investigated, and the free version of WordPress was chosen for the course website. WordPress has also been used before for educational purposes.⁶⁰ Some of the attributes of the platform are popularity with users, free access to learners, user-friendly content editing options, ease of administration, large variety of plugins, allowance to export content to other online platforms and opportunity for use on computers, tablets or smartphones. The setup of the website requires technical expertise, therefore it is recommended that a website designer be included in the course development team.²⁴

Storyboarding is the systematic and iterative process where all the elements to be included in each unit are mapped.³⁰ Storyboarding is helpful for pointing out steps that might be overlooked in instructional sequence, and allows for modification of the course structure.^{30, 32}

Storyboards shared with collaborators, such as mentors and web designers, enhance shared understanding. The storyboard was prepared on a Microsoft Word template, since it is predominantly text-based. Each of the four units is designed to take a maximum of 1 week if learners allocate 3-4 hours per week, which can be spread within the week. The course duration is four weeks. This is important because the most prevalent point of dropout from online learning occurs after week 4, which suggests that a four-week duration is more convenient for learners than longer courses.⁵⁸ Each unit is composed of several topics, which are chunked into short sections. This signposting makes it easier to navigate the course.³¹ The stakeholders reviewed the outputs from the design phase (Table 1) for quality assurance.

6.5.3 Developing course content

The storyboards created in the design phase were used to develop learning materials. This process required thinking through each step of the storyboard, with an unexpectedly high investment in time and labour. Preparing content for each week took several weeks, and as other researchers have found, the time required and the opportunity cost are often underestimated.^{50, 61} This challenge was overcome by dedicating blocks of time to content development, and working with mentors, in addition to utilizing the evidence synthesis and writing skills developed over the PhD. It is advisable not to underestimate the amount of workload involved, especially for course developers with other responsibilities.⁴⁷ Collaborative creation of course materials by a team of experts may alleviate this challenge, as the heavy workload can be shared.¹² However feasibility may be constrained by existing health workforce challenges in LMICs and course development may divert time from other activities such as service delivery, research, teaching or administration.⁵⁰ On the other hand, online courses may

enhance the effectiveness of the limited faculty.¹² This raises the question as to how best to calibrate participation such that LMICs contribute and gain from online learning.

Preparing instructional audio-visual content was challenging because it required access to studios, compressing the content into short lectures, recording in front of cameras and editing the multimedia materials. Like in other LMICs, access to resources and expertise for digital capture was limited at local level.⁴⁷ Once the slides and scripts were ready, the multimedia team at LSHTM filmed the narration and provided video editing services. Access to these institutional resources was an opportunity provided by being a student at the school and by linkage from the mentors. Building on such educational presence and networks is therefore pivotal to successful completion. Recording was challenging because speaking in front of a camera with no audience did not feel natural. In addition, listening to playback of my own voice was quite disconcerting, as the voice did not sound authentic. The literature reports similar experiences with digital production and these can potentially slow progress.⁶¹ However mentorship helped to manage expectations and acclimatize to the experience, hence increasing efficiency.³⁷

Developing course materials requires balancing the use of multiple formats to enhance usability and ensuring scientific accuracy and clarity.¹⁵ Metadata such as an overview of the course and strategic instructions that have been provided at the start of the course help to enhance navigation. To facilitate comprehensibility, precise language has been used. Sentences have been formulated in a way that is not too abstract or scientific. The webpages are not text-heavy, as this often has an adverse effect on readability. Medical educators and subject experts have reviewed the materials and their suggestions for improvement, such as including definitions of important concepts, have been incorporated.

Adult learners are self-directed, so although the four course units are arranged in sequence, learners can complete the units in any order. Further, adult learners are goal and relevancy-oriented, so the practical application to the care of PLWD is emphasised. In order to provide for personalized learning, which is important for adult learners, 'to do' prompts, questions for reflection and links to optional resources are provided for those interested in deeper study of particular concepts. This enables learners to apply autonomy and personalize learning through weekly course announcements and emails that may make learners feel more personally connected.^{15, 60, 61} However, such a high level of personalization was beyond the scope and feasibility of this course, since it would require extensive instructor support all year round.

6.5.4 Implementation

The course content was uploaded to the WordPress course site:

<u>https://oerdiabeticretinopathy.wordpress.com/</u>. The course instructor, the mentors, and a learning designer have done usability testing of the site on a variety of browsers and devices. Feedback from this usability testing helped to fix errors (duplicate pages and inactive hyperlinks) in navigation and improve user experience.

For the course to have acceptability among learners, it should be user-friendly and meet the expectations of the users. Pilot testing with users can be helpful to establish the content validity (the utility of the course for training on the different components of control of DR) and face validity (the appropriateness and relevance to real life needs of the users of the course).⁶² For that reason, after testing usability the course was piloted by ten potential learners and ten potential instructors. They were asked to run through the course slowly, identifying unforeseen practical difficulties and providing feedback. Most of the feedback related to the format of the

website, and not to the content. In addition, users recommended accreditation and certification by an academic institution, which is not currently available but is planned for the future, as already discussed. Instructors recommended that such certification should be based on a pre-test and post- test scores so that educational effectiveness can be measured.

6.5.5 Evaluation

In each stage of the ADDIE framework, formative evaluation was constantly applied to identify limitations in course outputs (Table 6-1), and to modify them appropriately. Once the course is launched, learning analytics from the course site will be used to track learner engagement. In addition, Kirkpatrick's framework for evaluating training will be used to evaluate the effectiveness of learning on the course. This framework focuses on evaluation at four levels: immediate reaction to the training, learning obtained from the training, behaviour change and results (effect on the initial training gap).⁶³ The first two levels (reaction and learning) will be assessed using the post-course survey as well as pre-test and post-test scores. To evaluate the extent to which the course supports behaviour change in clinical practice, a questionnaire will be emailed to learners three months after the course. The effect on training needs and performance will be assessed in a subsequent needs assessment.

6.6 Potential challenges in running the online course

The literature reports a variety of barriers to implementation of online courses.^{45, 47, 49, 50} These include existing norms and behaviours, the lack of appreciation of investment in online learning by the academic and health community, and lack of corresponding alignment of resources. The official launch of the course as part of the activities organized by the Ministry of Health for the World Diabetes Day 2019, and course advertisement through the mailing lists of

professional groups will promote engagement with learners and educators. This integration into local structures and processes is expected to facilitate local ownership and implementation of the course at scale in Kenya.⁴⁵

An educational website needs to be maintained over time and content updated, if it is to be a sustainable solution.¹⁵ The instructor will update content annually, and collaboration with institutions and the Ministry of Health will help to overcome the funding and opportunity costs. Other researchers have found that support for the long-term implication of maintaining, updating and upgrading the site is often overlooked, yet it is vital to the quality of the online courses.^{12, 15, 50} Institutional and country-level support will be critical for sustainability.³⁶

Adapting to the online environment can be a challenge for both learners and facilitators.⁴⁵ The WordPress interface used in this course is simple to use, therefore users are unlikely to experience technical problems with the site, however this will be monitored and technical support provided.⁶⁰ The popularity of WordPress increases the risk of academic fraud, viruses or breaches of data protection, but this is not considered a major concern as all the materials are open access.^{60, 64}

Other researchers have reported the inherent challenges in the evaluation of online courses. For example, high dropout rates and the need to adjust for contextual factors affect the validity and complexity of assessment of online learners.^{61, 65} These are challenges that the online course community must learn to overcome. Although there are no easy solutions, the use of diverse methods of evaluation in this course will provide evidence on the effectiveness of learning.⁶⁵ The long-term impact of these evaluation strategies will need to be evaluated.

As mentioned earlier, certification is an important motivator for learners in online courses in health and medicine in LMICs.^{6, 49, 61} Assessments for accreditation will not be provided on the

platform, but through institutional websites once the course is accredited. Educational institutions accredit courses through a rigorous quality assurance, which will provide further quality assurance, in addition to the peer review already mentioned. The accreditation process is likely to be a time-consuming and complex process, which is a major challenge. However, the engagement of stakeholder institutions may provide an enabling environment.

6.7 Lessons learnt during course development

The learning gained from the PhD research was a significant starting point in terms of subject domain knowledge. Reflecting on the principles of control of DR gained from the PhD was a useful prompt for knowledge transfer within the health system. The potential impact of public health training on health system strengthening in LMICs has previously been examined.^{66, 67} Given that health workers in Kenya also often work as educators in training institutions (chapter three), health service delivery and education are closely connected. Considering the unique demands of this dual role, it is important to understand how learning from a PhD might interact with the health system and the education context, thereby enhancing capacity for health system strengthening. This is of interest because success of training researchers for Africa should not only be measured by research outputs, but also by the ability to execute other programmes that will advance population health in Africa.⁶⁷

Being a mentee from Kenya (LMIC) and my mentors being from other income settings provided rich diversity within the mentorship process. The diversity fostered access to learning from contrasting perspectives and led to a concrete conception of the detailed task of course development in the early stages of course development. This was useful because at the beginning I was unfamiliar with the online course design process, leading to underestimating the complexity of the design task. Mentorship illuminated the strategic opportunities, for

example, the digital technologies for online courses. It also allowed the recognition of contextual complexities that needed to be overcome, such as how to make a business case for free online courses, which is a nascent concept in LMIC training institutions.⁴⁷ In addition, mentoring helped to contextually apply theories of learning and theories of course design, either explicitly or implicitly, to the designing of the course.⁶⁸ The literature reports that interactions between foreign mentors and local researchers may create tension or conflicts due to variations in mentoring culture, but this was not experienced.^{41,69}

6.8 Conclusion

Reflecting on the synergy between the academic skills important in both PhD and course development can provide an understanding of the value that can be leveraged from a doctoral programme. The academic discourse around the potential contribution of PhD studies in African health systems is ongoing.⁶⁷ There is growing recognition that such training should provide the skills to apply theories to inter-relationships between health and other factors, such as education.^{51, 70} In particular, doctoral researchers need to be agents of change and seek solutions to the challenges in education, research and health caused by scarce resources and other contextual factors. This is because they have the advantage of familiarity with local challenges, context-specific determinants, and the expertise to apply relevant global concepts to local problems. As an example, effective course development requires a facilitator who is familiar with the subject field and possesses the requisite expertise, which a PhD provides.^{12, 62} There is potential for doctoral programmes to contribute to capacity building for the health workforce. This model can be replicated in other settings in sub-Saharan Africa with similar health workforce challenges.

REFERENCES

1. Field S, Abrahams Z, Woods DL, Turner R, Onah MN, Kaura DK, et al. Accessible continued professional development for maternal mental health. 2019. 2019;11(1).

2. Clark E, Draper J, Rogers J. Illuminating the process: Enhancing the impact of continuing professional education on practice. Nurse Education Today. 2015;35(2):388-94.

3. Golnik K. Why should we continue to learn? Community Eye Health Journal. 2017;30(97):1-3.

4. Eddy A, Eddy D, Doughty J. Evidencing Continual Professional Development: Maximising Impact and Informing Career Planning. Journal of Medical Imaging and Radiation Sciences. 2015;46(4):361-4.

5. International Council of Ophthalmology. ICO Guide to Effective CPD/CME. 2016.

6. Feldacker C, Jacob S, Chung MH, Nartker A, Kim HN. Experiences and perceptions of online continuing professional development among clinicians in sub-Saharan Africa. Human resources for health. 2017;15(1):89-90.

7. Berndt A, Murray CM, Kennedy K, Stanley MJ, Gilbert-Hunt S. Effectiveness of distance learning strategies for continuing professional development (CPD) for rural allied health practitioners: a systematic review. BMC Medical Education. 2017;17(1):117.

8. Reeves S, Fletcher S, McLoughlin C, Yim A, Patel KD. Interprofessional online learning for primary healthcare: findings from a scoping review. BMJ Open. 2017;7(8):e016872.

9. Richmond H, Copsey B, Hall AM, Davies D, Lamb SE. A systematic review and metaanalysis of online versus alternative methods for training licensed health care professionals to deliver clinical interventions. BMC Medical Education. 2017;17(1):227.

10. Vaona A, Banzi R, Kwag KH, Rigon G, Cereda D, Pecoraro V, et al. E-learning for health professionals (Review). Cochrane Database of Systematic Reviews. 2018;1.

11. Kyalo IW, Hopkins S. Exploring the Acceptability of Online Learning for Continuous Professional Development at Kenya Medical Training Colleges. The Electronic Journal of e-Learning. 2013;11(2):82-90.

12. Frehywot S, Vovides Y, Talib Z, Mikhail N, Ross H, Wohltjen H, et al. E-learning in medical education in resource constrained low- and middle-income countries. Human resources for health. 2013;11:4-5.

13. Communications Authority of Kenya. First quarter sector statistics report for the financial year 2018/2019. Nairobi: Communications Authority of Kenya; 2019.

14. Munyua AW. Exploring the multi-stakeholder experience in Kenya. Journal of Cyber Policy. 2016;1(2):206-21.

15. Sisson SD, Hill-Briggs F, Levine D. How to improve medical education website design. BMC Medical Education. 2010;10(1):30.

16. Courtright P, Mathenge W, Kello AB, Cook C, Kalua K, Lewallen S. Setting targets for human resources for eye health in sub-Saharan Africa: what evidence should be used? Human Resources for Health. 2016;14(1):11.

17. Burgess PI, Msukwa G, Beare NAV. Diabetic retinopathy in sub-Saharan Africa: meeting the challenges of an emerging epidemic. BMC Medicine. 2013;11(1):157.

World Health Organization Regional Office for Africa. Core competencies for the eye health workforce in the WHO African Region. Brazzaville: WHO Regional Office for Africa; 2019.
 Barsley S, Lock A, Mwangi N, Batel D, Open borders for eye health education. Human

19. Parsley S, Leck A, Mwangi N, Patel D. Open borders for eye health education. Human Resources for Health Global Forum Dublin, Ireland.2017.

20. Khalil MK, Elkhider IA. Applying learning theories and instructional design models for effective instruction. Advances in Physiology Education. 2016;40(2):147-56.

21. Mukhalalati BA, Taylor A. Adult Learning Theories in Context: A Quick Guide for Healthcare Professional Educators. J Med Educ Curric Dev. 2019;6:2382120519840332-.

22. Chacko T. Emerging pedagogies for effective adult learning: From and ragogy to heutagogy. Archives of Medicine and Health Sciences. 2018;6(2):278-83.

23. Abela J. Adult learning theories and medical education: a review. Malta Medical Journal. 2009;21(01):11-8.

24. Palis AG, Quiros PA. Adult learning principles and presentation pearls. Middle East African journal of ophthalmology. 2014;21(2):114-22.

25. Durak G, Ataizi M. The ABC's of Online Course Design According to Addie Model. Universal Journal of Educational Research. 2016;4(9):2084-91.

26. Almomen RK, Kaufman D, Alotaibi H, Al-Rowais NA, Albeik M, Albattal SM. Applying the ADDIE—Analysis, Design, Development, Implementation and Evaluation—Instructional Design Model to Continuing Professional Development for Primary Care Physicians in Saudi Arabia. International Journal of Clinical Medicine 2016;7:538-46.

27. Donmez M, Cagiltay K. A Review and Categorization of Instructional Design Models. World Conference on e-learning; Washigton DC, United States Of America. 2016.

28. Patel SR, Margolies PJ, Covell NH, Lipscomb C, Dixon LB. Using Instructional Design, Analyze, Design, Develop, Implement, and Evaluate, to Develop e-Learning Modules to Disseminate Supported Employment for Community Behavioral Health Treatment Programs in New York State. Front Public Health. 2018;6:113 -5.

29. Clark D. ADDIE timeline 2015. [Available from:

http://www.nwlink.com/~donclark/history_isd/addie.html.

30. Branch RM. Instructional Design: The ADDIE Approach. New York: Springer Science 2009.

31. Cheung L. Using the ADDIE Model of Instructional Design to Teach Chest Radiograph Interpretation. Journal of Biomedical Education. 2016;2016:6.

32. Aldoobie N. ADDIE Model. American International Journal of Contemporary Research. 2015;5(6):68-72.

33. Malan Z, Mash B, Everett-Murphy K. Development of a training programme for primary care providers to counsel patients with risky lifestyle behaviours in South Africa. Afr J Prim Health Care Fam Med. 2015;7(1):819.

34. Mash B. Development of the programme mental disorders in primary care as internetbased distance education in South Africa. Medical Education. 2001;35(10):996-9.

35. Mash B. Diabetes education in primary care: a practical approach using the ADDIE model. Contin Med Educ 2010;28:485-7.

36. Wensing M, Grol R. Knowledge translation in health: how implementation science could contribute more. BMC Medicine. 2019;17(1):88.

37. Gagliardi AR, Webster F, Straus SE. Designing a knowledge translation mentorship program to support the implementation of evidence-based innovations. BMC Health Services Research. 2015;15(1):198.

38. Thompsona L, Jeffriesa M, K T. E-mentoring for e-learning development. Innovations in Education and Teaching International. 2010;47(3):305–15.

39. UNESCO - Open Education for a Better World. OE4BW Project-Control of Diabetic Retinopathy 2018 [Available from: <u>http://oe4bw.ijs.si/project/diabetic-retinopathy/</u>.

40. United Nations Educational Scientific and Cultural Organization. OE4BW Mentoring Programme 2017-2018 Paris, France: UNESCO; 2017 [Available from:

https://unesco.ijs.si/oebw-2017/, https://unesco.ijs.si/project/open-education-for-a-better-world/.

41. Bennett S, Paina L, Ssengooba F, Waswa D, M'Imunya JM. Mentorship in African health research training programs: an exploratory study of fogarty international center programs in Kenya and Uganda. Education for Health. 2013;26(3):183-7.

42. Brown P, Brunnhuber K, Chalkidou K, Chalmers I, Clarke M, Fenton M, et al. How to formulate research recommendations. BMJ (Clinical research ed). 2006;333(7572):804-6.

43. Filipe HP, Silva ED, Stulting AA, Golnik KC. Continuing professional development: best practices. Middle East African journal of ophthalmology. 2014;21(2):134-41.

44. Swanson RC, Atun R, Best A, Betigeri A, de Campos F, Chunharas S, et al. Strengthening health systems in low-income countries by enhancing organizational capacities and improving institutions. Globalization and Health. 2015;11(1):5.

45. O'Doherty D, Dromey M, Lougheed J, Hannigan A, Last J, McGrath D. Barriers and solutions to online learning in medical education – an integrative review. BMC Medical Education. 2018;18(1):130.

46. Rwabukwisi FC, Bawah AA, Gimbel S, Phillips JF, Mutale W, Drobac P, et al. Health system strengthening: a qualitative evaluation of implementation experience and lessons learned across five African countries. BMC Health Services Research. 2017;17(3):826.

47. Pasha A, Abidi SH, Ali S. Challenges of offering a MOOC from an LMIC. International Review of Research in Open and Distributed Learning. 2016;17(6).

48. Cole DC, Johnson N, Mejia R, McCullough H, Turcotte-Tremblay, Barnoya J, et al. Mentoring health researchers globally: Diverse experiences, programmes, challenges and responses. Global Public Health. 2016;11(9):1093-108.

49. Liyanagunawardena TR, Williams SA. Massive open online courses on health and medicine: review. J Med Internet Res. 2014;16(8):e191-e.

50. Lucas H, Kinsman J. Distance- and blended-learning in global health research: potentials and challenges. Global Health Action. 2016;9(1):33429.

51. Oni T, Yudkin JS, Fonn S, Adongo P, Kaseje M, Ajuwon A, et al. Global public health starts at home: upstream approaches to global health training. The Lancet Global Health. 2019;7(3):e301-e2.

52. International Council of Ophthalmology. ICO Guidelines for Diabetic Eye Care-updated 2017. San Francisco, California: International Council of Ophthalmology; 2017.

53. American Diabetes Association. Standards of Medical Care in Diabetes-2017. Diabetes Care. 2017;40(Suppl 1).

54. Ministry of Health. Kenya National Diabetes Strategy 2010-2015. Nairobi: Ministry of Health; 2010.

55. International Diabetes Federation, Fred Hollows Foundation. Diabetes eye health: a guide for health professionals. Brussels, Belgium: International Diabetes Federation; 2015.

56. International Centre for Eye Health. Open Education – Online courses & resources for eye health 2019 [Available from: <u>http://iceh.lshtm.ac.uk/oer/</u>.

57. International Diabetes Federation. Diabetic Retinopathy 2018 [Available from: https://www.idfdiabeteschool.org/Short-Course/diabetic-retinopathy/en.

58. Höfler E, Zimmermann C, Ebner M. A case study on narrative structures in instructional MOOC designs. Journal of Research in Innovative Teaching & Learning. 2017;10(1):48-61.

59. Creative Commons. Creative Commons Licences 2017 [Available from: https://creativecommons.org/licenses/.

60. Avila J, Sostmann K, Breckwoldt J, Peters H. Evaluation of the free, open source software WordPress as electronic portfolio system in undergraduate medical education. BMC medical education. 2016;16(157).

 Hew KF, Cheung WS. Students' and instructors' use of massive open online courses(MOOCs): Motivations and challenges. Educational Research Review. 2014;12:45-58.
 Parson R, Danilovich N, Lochnan H, Kitto S, Delva D, Viner G, et al. Twelve tips for bringing competencies into continuing professional development: Curriculum mapping. MedEdPublish. 2019;8(2):75.

63. Smidt A, Balandin S, Sigafoos J, Reed VA. The Kirkpatrick model: A useful tool for evaluating training outcomes. Journal of Intellectual & Developmental Disability. 2009;34(3):266-74.

64. Wynter L, Burgess A, Kalman E, Heron JE, Bleasel J. Medical students: what educational resources are they using? BMC Medical Education. 2019;19(1):36.

65. Sherr K, Fernandes Q, Kanté AM, Bawah A, Condo J, Mutale W, et al. Measuring health systems strength and its impact: experiences from the African Health Initiative. BMC Health Services Research. 2017;17(3):827.

66. Zwanikken PAC, Alexander L, Scherpbier A. Impact of MPH programs: contributing to health system strengthening in low- and middle-income countries? Human Resources for Health. 2016;14(1):52.

67. Manabe YC, Katabira E, Brough RL, Coutinho AG, Sewankambo N, Merry C. Developing independent investigators for clinical research relevant for Africa. Health Research Policy and Systems. 2011;9(1):44.

68. Smith CF, Martinez-Álvarez C, McHanwell S. The context of learning anatomy: does it make a difference? Journal of Anatomy. 2014;224(3):270-8.

69. Lescano AG, Cohen CR, Raj T, Rispel L, Garcia PJ, Zunt JR, et al. Strengthening Mentoring in Low- and Middle-Income Countries to Advance Global Health Research: An Overview. American Journal of Tropical Medicine and Hygiene. 2019;100(1_Suppl):3-8.

70. Bullin C. To what extent has doctoral (PhD) education supported academic nurse educators in their teaching roles: an integrative review. BMC Nursing. 2018;17:6-8.

Section D DURE Cluster randomized controlled trial

"I stand before you and the world humbled by this recognition and uplifted by the honor of being the 2004 Nobel Peace Laureate. As the first African woman to receive this prize, I accept it on behalf of the people of Kenya and Africa, and indeed the world. I am especially mindful of women and the girl child. I hope it will encourage them to raise their voices and take more space for leadership."

Prof Wangari Muta Maathai (1940-2011) Environmentalist, professor of veterinary anatomy and political activist



The DURE trial

Chapter Seven

Effectiveness of peer support to increase attendance at DR screening - a pragmatic cluster randomized clinical trial Study protocol

7.0 Overview

A prospective cluster randomised clinical trial (RCT) was conducted to investigate whether a peer support can increase attendance to diabetic retinopathy screening (DRS). Members of diabetes support groups who have never had DRS were recruited into the study. Diabetes support groups were randomised on a 1:1 basis to one of the following arms: 1) Intervention arm, to receive a peer led health education intervention, in addition to usual care provided in support groups 2) Control arm, to receive usual care alone. The proportion of participants attending DRS was evaluated over a six months period.

7.1 Intervention development

The intervention was developed following the guidance provided by the United Kingdom Medical Research Council's (MRC) guidelines on developing and evaluation complex interventions.¹

 Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008;337:a1655. <u>https://doi.org/10.1136/bmj.a1655</u>.

The MRC recommends that interventions are developed systematically, using the best available evidence and appropriate theory to understand the likely process of change. Interventions

should then be tested in a pilot phase targeted at feasibility objectives before full-scale implementation. A clear description of the intervention and a detailed report of both process and outcome evaluations of the trial are important to enable replication and synthesis of evidence. Figure 1 outlines the focus of each of the four phases of the trial, based on the MRC guidance.



Figure 7-1: Development, piloting, implementation and evaluation of the clinical trial

Research paper 5 gives a detailed description of the methods for this clinical trial in Kirinyaga, Kenya. This paper was published in the journal BMC Implementation Science in December 2018 after peer review. Subsequent chapters provide the results for the pilot trial (chapter 8), process evaluation (chapter 9) and outcome evaluation (chapter 10).

7.2 Research Paper 5

BMC Public Health – Licences and Copyright

The following terms apply to this publication

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy, distribute, and use their attributed research, free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago.

All articles published by *BMC Public Health* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access

https://bmcpublichealth.biomedcentral.com/about



Registry T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi		
Principal Supervisor	Professor Allen Foster		
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access		

<u>If the Research Paper has previously been published please complete Section B, if not please move to</u> <u>Section C</u>

SECTION B – Paper already published

Where was the work published?	BMC Public Health		
When was the work published?	2018		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion	Not applicable		
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	
Please list the paper's authors in the intended authorship order:	
Stage of publication	Choose an item.

<u>SECTION D – Multi-authored work</u>

For multi-authored work, give full details of your role in the research included in the paper and in the preparation of the paper. (Attach a further sheet if necessary)	I designed the study, wrote this study protocol, drafted the manuscript and prepared subsequent revisions in consideration of the comments from the co- athors and peer reviewers
--	---

Student Signature:

Nyawira

Date: 5287 17, 2019

Open Access



Effectiveness of peer support to increase uptake of retinal examination for diabetic retinopathy: study protocol for the DURE pragmatic cluster randomized clinical trial in Kirinyaga, Kenya

Nyawira Mwangi^{1,2*}, Mark Ng'ang'a³, Esbon Gakuo⁴, Stephen Gichuhi⁵, David Macleod², Consuela Moorman⁶, Lawrence Muthami⁷, Peter Tum¹, Atieno Jalango⁸, Kibata Githeko⁹, Michael Gichangi¹⁰, Joseph Kibachio¹¹, Covadonga Bascaran² and Allen Foster²

Abstract

Background: All patients with diabetes are at risk of developing diabetic retinopathy (DR), a progressive and potentially blinding condition. Early treatment of DR prevents visual impairment and blindness. The natural history of DR is that it is asymptomatic until the advanced stages, thus annual retinal examination is recommended for early detection. Previous studies show that the uptake of regular retinal examination among people living with diabetes (PLWD) is low. In the Uptake of Retinal Examination in Diabetes (DURE) study, we will investigate the effectiveness of a complex intervention delivered within diabetes support groups to increase uptake of retinal examination.

Methods: The DURE study will be a two-arm pragmatic cluster randomized clinical trial in Kirinyaga County, Kenya. Diabetes support groups will be randomly assigned to either the intervention or usual care conditions in a 1:1 ratio. The participants will be 700 PLWD who are members of support groups in Kirinyaga. To reduce contamination, the unit of randomization will be the support group. Peer supporters in the intervention arm will receive training to deliver the intervention. The intervention will include monthly group education on DR and individual member reminders to take the eye examination. The effectiveness of this intervention plus usual care will be compared to usual care practices alone. Participant data will be collected at baseline. The primary outcome is the proportion of PLWD who take up the eye examination at six months. Secondary outcomes include the characteristics of participants and peer supporters associated with uptake of eye examination for DR. Intention-to-treat analysis will be used to evaluate the primary and secondary outcomes.

Discussion: Eye care programs need evidence of the effectiveness of peer supporter-led health education to improve attendance to retinal screening for the early detection of DR in an African setting. Given that the intervention combines standardization and flexibility, it has the potential to be adopted in other settings and to inform policies to promote DR screening.

Trial registration: Pan African Clinical Trial Registry PACTR201707002430195, registered 25 July 2017, www.pactr.org

Keywords: Diabetes mellitus, Diabetic retinopathy, Diabetes support groups, Retinal screening, Blindness, Health education, Self-efficacy theory, Peer-support, Kenya, Africa

* Correspondence: Nyawira.Mwangi@lshtm.ac.uk

¹Kenya Medical Training College, Nairobi, Kenya

²International Centre for Eye Health, London School of Hygiene and Tropical Medicine, Keppel Street, WC1E 7HT, London, UK

Full list of author information is available at the end of the article



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

The global prevalence of diabetes has escalated in recent decades, with important implications on the health system. In 2015 the International Diabetes Federation estimated that there were 415 million people with diabetes aged 20-79 years (global prevalence of 8.8%), and this is predicted to increase to 642 million by 2040 (global prevalence of 10.4%). [1] The incidence and prevalence of diabetes is increasing disproportionately faster in resource-poor regions and 75% of people living with diabetes currently reside in low- and middle-income countries. [1-3] This dramatic increase in incidence is occurring in both rural and urban areas. [4] The regional prevalence for Africa was 3.8% in 2015, and the number of people with diabetes in this continent is expected to increase by 140% between 2015 and 2040. [1] In Kenya, the STEPwise survey for risk factors of non-communicable diseases in 2015 found a diabetes prevalence of 2% in the population 18-69 years, and 5.4% in the population 45–59 years. [5]

All patients with diabetes are at risk of developing diabetic retinopathy (DR), the most severe and progressive ocular complication of diabetes. One third of patients with Type 2 diabetes have DR while 10% of them have sight-threatening DR, which represents a significant public health concern. [6, 7]. A population-based study in Nakuru county, Kenya found that 35.9% of people living with diabetes (PLWD) have DR [8]. Visual impairment and blindness from DR is preventable mainly through early detection and timely treatment. Since DR is asymptomatic until the advanced stages, regular retinal screening is of paramount importance. DR meets the Wilson and Jungner (1968) criteria for screening, and current clinical guidelines support annual screening [9–12]. Participation of PLWD in regular retinal screening, has been shown to be clinically effective in preventing blindness and is also cost-effective. [13, 14]

In developed countries, health systems have formal surveillance programmes for detection of DR. Kenya does not have a national population-based DR screening service where PLWD are systematically invited for screening, but opportunistic screening is available in various hospitals. Importantly, participation and re-participation rates in screening for DR are sub-optimal in Kenya and other resource poor settings. [15-22]. The determinants of the attendance to retinal examination are complex and include both supply and demand factors. [23, 24] For instance, a Tanzanian study found that PLWD also have limited awareness on diabetic retinopathy, particularly on the need for annual eye examination [25]. This is a barrier that appropriate demand side interventions could address. A health system assessment conducted before this study has shown that 87% of PLWD in Kenya have an unmet need for *annual* retinal screening. [22] One of the gaps associated with this is the lack of strong links between diabetes services and eye care services. There is need for context-specific pragmatic solutions to address this gap.

A systematic review of interventions to increase diabetic retinopathy screening attendance reported that several strategies are effective, including those targeting the patient (e.g. increasing patient awareness), the health care practitioner (e.g. improving adherence to recommendations) or the organization (e.g. improving patient records) [26]. Members of diabetes support groups (DSGs) are a population subgroup that might benefit from additional support to initiate screening, and adhere to re-screening. Targeting screening interventions towards PLWD in DSGs provides a timely opportunity for three reasons. Firstly this is a community resource that is already available in the community setting. Secondly this population is likely to consider health as an important rationale for behaviour change and the health seeking behaviour of members is potentially malleable to change through peer support. Peer support refers to the provision of emotional and informational support from a created social network member who is considered an equal and who has characteristics similar to the target population. [27] Thirdly it is an economical, culturally-sensitive and flexible intervention for improving diabetes care and outcomes. [28, 29]

Self-efficacy is a direct predictor of health behaviour, according to the social cognitive theory and the self-efficacy theory [30] [31]. Self-efficacy is a predictor of uptake of screening for DR among PLWD. [32] Interventions that improve patients' self-efficacy decrease perceived barriers and improve the likelihood of initiating the desired health behaviour. There are four main sources of self-efficacy [31]: (i) Successful performance accomplishments (e.g. having attended a previous eye examination) (ii) vicarious experience (e.g. learning that peers have successfully attended an eye examination) (iii) verbal persuasion (e.g. encouragement and recommendation to go for an eye examination by a trusted person, such as a peer or a health care worker) and (iv) psychological cues (decreased sense of isolation of PLWD interacting with a peer, or increased awareness of the risk of DR after receiving educational messages on DR). Fig. 1 shows how self-efficacy for taking a retinal examination might improve through peer support in the Uptake of Retinal Examination in Diabetes (DURE) study.

Rationale

Although there is evidence that peer support improves glycaemic control and quality of life among adult PLWD, and that peer support is both cost-effective and flexible [33], evidence on whether or not it would increase uptake of eye examination in an agrarian African population is lacking. DURE study aims to provide this evidence.

The development of the intervention has been informed by the following:



- A review of the literature on peer support in diabetes [34–42] and other chronic conditions in resource poor settings. [43–51] There is evidence that peer approach is widely used in the management of diabetes, to promote physical activity, healthy eating and improvemt in glycaemic control.
- ii. The results of a recent published meta-analysis of randomized clinical trials on effectiveness of peer-support for glycaemic control in Type 2 diabetes [52] which concluded that peer support had a significant impact on improving HbA_{1c} levels in patients with poor glycaemic control.
- iii. A health system assessment in three counties of Kenya conducted before this study showed that services for DR are underutilised: 74% of PLWD have *never* had a retinal examination in their lifetime, and 76% have *never* had a recommendation for an eye examination by their diabetes care provider [22].
- iv. Evidence that improving health literacy, provider patient interaction and linking patients to health care improves patients' self-efficacy and glycaemic control. [42, 53, 54]

Aim

To evaluate, by means of a pragmatic cluster randomized controlled trial, the effectiveness of a peer supporter-led community education programme in Kirinyaga county, Kenya.

Research questions

- 1. To what extent can health education delivered by peer supporters increase the demand for annual retinal examination among PLWD?
- 2. What are the contextual factors that determine the effectiveness of the intervention?

Hypothesis

The hypothesis is that the proportion of PLWD having a retinal examination for DR is higher in diabetes support groups (DSGs) allocated to the peer supporter-led educational package than in DSGs randomized to the usual standard of care.

Methods

Design

This is a two-arm pragmatic cluster randomized controlled trial with additional process evaluation. It is a complex intervention to empower patients to undergo an annual eye examination. It is complex because those delivering and receiving the intervention require to demonstrate different behaviours and to engage in multiple interactions. [55] Its design is guided by the Medical Research Council framework for complex interventions,

available at https://mrc.ukri.org/documents/pdf/complex-interventions-guidance/ [55].

The study will be conducted in accordance with the Consolidated Standards of Reporting Trials (CONSORT) 2010 statement and its extension to cluster randomized clinical trials (cRCTs). [56, 57] The cRCT design is adopted for the following reasons: (i) to reduce the effect of intervention contamination, as compared to an individually randomised trial, as patients in the same DSG often interact with one another (ii)to make it feasible to study the effect of the intervention at the individual level and the cluster level.

Definition of eye examination for DR

We define this test as: measurement of visual acuity and a retinal examination through a dilated pupil conducted by an eye care worker (using either an ophthalmoscope, a slit lamp or a retinal camera). Retinal examination for DR and DR screening in this protocol are used interchangeably.

Study setting

This trial will be conducted among the DSGs in Kirinyaga county, Kenya. The target population is members of the 16 support groups and volunteer peer supporters within these groups. Eye examination will be conducted at Kerugoya County Referral Hospital.

Sample size calculation

We aim to randomize seven diabetes patient support groups (clusters) with an average membership of 50 each to each arm. The study thus has two arms of equal size (350 participants in each arm). This sample size has been calculated using standard formula for sample size for cRCTs and taking into consideration the primary outcome of interest [58, 59]. A 15% loss to follow-up contingency has been built into the sample size calculation. This sample size would have 80% power to detect a two-fold difference in the proportion of PLWD who take up eye exam, with a 5% level of significance. Member registers of the DSGs will be obtained from the team lead. These registers will be the frame for identification of participants for the study.

Pilot study

A pilot study will be conducted in two clusters with 50 PLWD in each arm (intervention arm and control arm), selected through convenience sampling. The pilot will be conducted for 3 months and will involve: Testing study operational procedures; Implementation of the intervention in the intervention clusters; Testing study instruments for quantitative and qualitative data collection (questionnaires, observation sheets and topic guides); Outcomes evaluation. The primary outcome will be the proportion of participants in each arm that take up eye examination.

Inclusion and exclusion criteria

Participants will be included if they are PLWD aged 18+ years, will reside in Kirinyaga for the next 12 months, are members of DSGs in Kirinyaga, have a mobile phone and are willing to participate in the study. In addition to these criteria, peer supporters will be selected from those willing to participate as peer supporters, willing to commit two days for training and many hours of peer support, fluent in Kikuyu or Kiswahili, and have had a retinal examination for DR before the start of the study. PLWD who will be excluded are those already attending annual retinal screening, have a severe debilitating medical condition, are already on treatment for DR or do not meet the inclusion criteria.

Recruitment

Eligible participants will be recruited into the study by the research nurse, who will also obtain informed consent at the cluster level using the consent form approved for the study. Participants will be asked for consent to receive the intervent ion and for follow up. If the patient does not consent, reasons will be sought and recorded. After recruitment, a unique identifier number will be issued. All those recruited will be given an identification card which contains the name and a unique study number. They will be required to present this card at the eye clinic when they go for retinal examination. The flow diagram for the study is presented in Fig. 2.

Randomization

Randomization will be done after recruitment. The randomization will be through computer generated random numbers prepared by a statistician (DM) using STATA version 15 (StataCorp 2017), away from the project site. The allocation sequence will be concealed from the other trial personnel. Block randomization with block sizes of two or four will be used to ensure that the two arms are balanced over time, and to maintain unpredictability of allocation. Masking will not be possible but only the research team will have formal knowledge of the allocation.

Intervention

Two peer supporters will be recruited from each cluster in the intervention arm (one male and one female). These peer supporters will be selected from volunteers who meet the specified criteria. They will receive structured training in a two-day workshop. The content of the training sessions includes: an introduction to the project, the role of the peer supporter, diabetic eye disease and DR, retinal examination for DR, communication skills, managing groups, confidentiality and behaviour change. The training team will include a certified diabetes educator.

They will receive support to retain them in the study (airtime vouchers for delivering the telephone reminders).



Table 1 Key messages to be delivered to participants

Messages on di	abetic eye disease
1	Diabetes causes several complications in the eye, including DR
2	DR is a progressive condition that leads to blindness if treatment is not provided in good time
3	DR has no symptoms until the advance stages
4	An eye check by an eye specialist can detect damage to the eyes before symptoms develop
5	All persons with diabetes should have their eyes checked once every year by an eye specialist, even before any symptom or poor vision develops
6	Do not wait for your vision to get worse or for any other symptom to occur before you see an eye specialist
7	If eyes are found to be normal at your eye check by an eye specialist, please continue with an eye check annually
8	If you notice any abnormality with your eyes between your clinic appointments, visit the eye specialist as soon as possible. It may not mean that you have diabetic eye disease, it may be a simple problem that requires treatment.
9	The eye check may help to determine if your sugar, blood pressure, and lipid control needs to be re-assessed. Good control of your blood sugar levels, blood pressure and cholesterol reduces the risk of diabetes-related sight loss.
10	If you are found to have DR the eye specialist will inform you about the diagnosis, and how it will be treated.
Messages on re	tinal examination for DR
1.	Ensure you have a dilated eye examination at the eye clinic at least once a year.
2.	You do not need to have a referral note to go to the clinic. However we will give you a card to present at the clinic.
3.	At the eye clinic, your vision will be checked first.
4.	A examination for DR is different from any other type of eye examination. It is called a dilated eye examination.
5.	In this examination, the doctor puts eye drops into your eyes to dilate (widen) your pupils. This allows the doctor to have a good view of the back of the eye. Both eyes need to be examined.
6.	The examination is not painful. When the eye drops are first instilled, there may be a slight stinging sensation but this only lasts about a minute. You may feel uncomfortable because of light sensitivity and blurred vision once the pupils are dilated.
7.	Do not be afraid to ask the doctor questions about the examination or about diabetic eye disease.

They will also receive a weekly telephone call from the principal investigator for support. The intervention group will receive the usual care, a monthly group talk and a monthly individual telephone reminder to attend retinal examination. The key messages to be delivered in the group talk are shown in Table 1. The control group will receive the usual standard of care, which consists of ad hoc diabetes educational talks, blood sugar and blood pressure measurements during support group meetings.

Data collection

Standardized operating procedures will be used to collect data at baseline, using approved tools in the study proposal. Demographic and anthropometric (height, weight, waist circumference) as well as blood pressure, blood sugar and visual acuity will be recorded at baseline. Study participants will be given their body mass index and blood pressure measurements in the field, at the point of data collection; where these results are abnormal, participants will be referred to a health worker. Completed questionnaires will be monitored and data entry staff will be trained to minimize errors in data entry into computerized databases. Identifiers will be removed from participant data, and all paper data will be stored in locked cabinets. Electronic data will be password protected for confidentiality. A detailed data management plan is included in the study proposal.

Follow up

Participants in both arms will be followed up for six months to assess attendance to retinal examination. Participants who are lost to follow up will be identified at the monthly contact points with peer supporters. Three home visits will be made to trace participants who are lost to follow up. Characteristics of those lost to follow-up and reasons for loss will be evaluated.

At the end of six months two separate focus group discussions will be held with two participants from each intervention cluster in each arm (n = 14 for each focus group discussion) to explore the experience of the support groups with the intervention. A focus group discussion will also be held with peer supporters to explore the impact of 'peer supporting' on the management of their own diabetes, and their role in the health care team.

Process evaluation

A process evaluation will be conducted using qualitative interviews and non-participant observation (Table 2). The findings of the process evaluation will be evidence

Table 2 Domains and methods for process evaluation

Source	Domain	Data collection method	Stage of the trial
Trial registers	Recruitment Retention	Registers in the trial office	Throughout the trial $(n = 700)$
Participants	Fidelity Reach Dose received Effectiveness	Participant Questionnaire 2 Focus group discussions at 6 months	At recruitment (n = 700) Three months: n = 10% pf participants in each intervention cluster(35) Six months: $n = 10$ -% of participants in each cluster (35) N = 28
Non-participant observations by Pl	Recruitment Fidelity Dose delivered Context	PI Field notes	N = 2 group meetings per intervention cluster during the trial (14)
Peer supporters (PS)	Effectiveness Reach Fidelity Dose delivered Context	PS Questionnaires PS Diary for telephone calls PS Group session report form Focus group discussion at 6 months	After training (n = 14) Through the trial (n = 14) Throughout the study (1 report form per group meeting per PS) N = 7
Eye care workers Key informants	Context	In-depth interviews	At 6 months N = 3 N = 7
Research project manager	Reach Fidelity Dose delivered	Reports	At recruitment of PS At training the PS At 3 and 6 months
Research nurse	Outcome evaluation procedures	Report	At 3 months and 6 months
Study steering committee	Context	Spreadsheet of external events that may have affected study outcomes	At 6 months

PI Principal Investigator, PS Peer supporter

on why and how the intervention worked. The following domains of the intervention will be evaluated:

- 1) Whether the intervention activities are implemented as planned (fidelity).
- 2) The extent to which the intervention reaches the PLWD (reach).
- The degree to which PLWD are exposed to the intervention package (dose).
- 4) The extent to which the intervention is acceptable to PLWD and to eye care workers (acceptability).

5) The contextual factors that may have an influence on the theory of change (context).

Assessment of outcome

Primary outcome

Rates of eye examination in each arm will be assessed in each arm at the end of six months. This outcome will be assessed by an independent and masked research nurse who will review the eye clinic records of all participants. The outcome will be recorded on the outcome evaluation form for each participant. The form contains identification details of each participant recruited into the study (name, residence, telephone number) and will thus differentiate them from other patients who are examined in the eye clinic. The form does not contain information on the intervention arm to which the patient is allocated. The project manager will receive the completed outcome evaluation forms and link the data to the participant database for each arm.

Secondary outcomes

These outcomes will be assessed at six months:

- 1. Contextual factors that affect the effectiveness of the intervention
- 2. Characteristics of peer supporters associated with uptake of eye examination
- 3. Barriers to uptake of eye examination among PLWD.

These outcomes will be evaluated using the database for participants and peer supporters, as well as data from focus group discussions with peer supporters and in-depth interviews with eye care workers at six months.

Statistical analysis

Baseline comparability of the two groups will be assessed to check that the important confounders and baseline characteristics that would affect uptake of eye examination are balanced between the two arms through randomisation. If the arms are found to be substantially imbalanced an appropriately adjusted logistic regression model will be used.

Study-wide pooled analysis will be conducted for the primary outcome. Missing data will be reported using standard flow charts. Repeated measures mixed models regression with adjustment for age, sex, and baseline anthropometric measures will be used to compare the two groups for the primary outcome.

Analysis will be conducted on intention-to-treat basis. Regression analysis will be used to determine the extent to which individual and support group characteristics are associated with the primary outcome. Models for comparing continuous outcomes will use linear regression while models for categorical outcomes will use logistic regression. Kaplan-Meier analysis will be used to plot the survival curves for both treatment arms. Cox regression will be used to assess the impact of the intervention on time to first eye examination. The hazard ratio will be estimated with Cox regression, adjusting for substantial baseline imbalances if appropriate. Interim analysis is not planned.

Data monitoring

The principal investigator will coordinate and monitor all recruitment, intervention and follow up procedures. A data monitoring committee will not be required. There is no reason to expect significant adverse effects and there are no stopping rules. The principal investigator will have access to all the trial data sets.

Harms

Neither arm of the trial has serious anticipated harms. The retinal examination involves the use of mydriatic eye drops. This may cause temporary blurring of vision, but this is only expected to last for a few minutes or hours. In this trial the drugs will be instilled by highly experienced clinicians, and patients will be made aware of this effect beforehand. Any unexpected effects of the trial will be documented and reported to the sponsor and ethics committees.

Dissemination

The dissemination strategy will include a summary of the findings for support groups, a report to Kerugoya County government and the Ministry of Health Kenya, publications in peer-reviewed journals and presentations at national and international conferences.

Post-trial care

It is recommended that all PLWD have an annual retinal examination for DR, and more frequent examinations are required for those found to have any stage of DR. This is best practice that is recommended by the national guidelines for screening and management of DR in Kenya. [12] The service will continue to be available as routine care to PLWD at the Kerugoya County Referral Hospital beyond the study.

Discussion

This study is pragmatic in that it tests the effectiveness of this intervention in the real-world situation of the community and the health system in Kirinyaga. There is a strong need to develop interventions that can reach PLWD populations in real world settings to ensure that any effect found is generalizable.

Public health strategies to manage the diabetes and DR in sub-Saharan Africa (SSA) are known to be inadequate or non-existent. [60, 61] Given that we are at the emergence of the epidemic, this is an appropriate time to develop contextual interventions that will enable our health system to cope with this challenge. To our knowledge, this is the first study that has targeted the DSG population in DR research. The use of peer support in DR is a relatively new field and little has yet been published on the topic.

The trial is important for a number of reasons. For the *individuals with diabetes*, this trial is in line with the growing global focus on patient empowerment. The PLWD will be empowered to demand for retinal examination, thus reducing demand side barriers to uptake of the examination. These actively engaged PLWD will be linked to eye care providers by the peer supporters. The *Chronic Care Model*, which has been proposed as a suitable model for managing diabetes, emphasises on the need to implement such links between patients and the health system using community resources. [62] [47] As all PLWD are at risk of DR, empowerment to initiate and maintain screening will be beneficial to all.

For the *support groups*, if this intervention positively influences uptake of retinal examination, this could in turn influence how the DSGs define their role. It has potential to instigate a new agenda, making the groups key sites for preventive public health initiatives that are adaptable, feasible and embedded within support group culture. The peer supporters will remain a valuable resource in the DSG, which enhances sustainability of effect.

For *Kirinyaga county*, our study findings might help the county (formerly district) health services to develop initiatives to promote early detection of DR, by involving DSGs, empowering patients and developing effective referral systems for DR services. The role the support groups can play in strengthening the health system for diabetic retinopathy in the county will become explicit.

The intervention will be provided by trained peer supporters, which is a form of task-shifting. Task shifting is commonly applied in both diabetes and eye care services in our setting. It helps to address the severe shortage of human resources for health. The peer supporters will refer patients to the eye clinic, thus linking diabetes patients with eye clinics and strengthening the referral system in the county. In *national context*, Kenya aims to achieve universal eye health, which includes care for DR. This study provides a framework for the promotion of retinal screening in the population with the risk of developing DR. If effective, the intervention would be a sustainable and scalable to other countries.

In the *international contexts*, the DURE study has the potential to extend current evidence and inform the scientific debate as to whether embedding retinal screening into DSGs is an effective next step toward meeting health goals.

The explicit use of a theoretical construct (self-efficacy theory) to conceptualise the potential determinants that would influence attendance to DR screening is a key strength of the study. It enhances the understanding of the plausibility of the intervention. The intervention package combines both standardization and flexibility, which allows for scalability in diverse settings. A further strength of this study is the inclusion of process evaluation, which will assist in the interpretation of how and why the intervention did, or did not, bring about the predicted effects.

The study has potential limitations. There are only 16 support groups in the county, which limits the possibility of increasing the number of clusters to further enhance statistical power. Delayed recruitment of the required sample size and loss to follow-up during the trial may be a challenge. In mitigation, a 15% loss to follow-up contingency has been built into the sample size calculation. Sample attrition can result from any inaccuracies in the data collection, such as incorrect address and telephone number information. Other diabetes studies have documented that patients were unable or unwilling to participate due to transportation issues and lack of time or interest [63]. However this is not anticipated because: alternative contact information of participants will be documented, only one visit to the eye clinic is required of participants, and the intervention is expected to build participants' self-efficacy. Attrition bias may occur if whole clusters drop out, however this is not anticipated as the study period is short.

Despite these limitations, DURE study illustrates the tremendous potential of implementing pragmatic cluster RCTs in the diabetes support group setting. Implementing the trial in this at-risk population will be an invaluable learning opportunity. Many of the lessons learned from this experience could be useful to other research projects.

Trial status

At the time of submission, the trial is at the stage of enrolment.

Abbreviations

CONSORT : Consolidated Standards of Reporting Trials; cRCT: Cluster Randomized Clinical Trial; DR: Diabetic Retinopathy; DSG: Diabetes support group; DURE: Uptake of Retinal Examination in Diabetes; PACTR: Pan African Clinical Trials Registry; PLWD: People living with diabetes; SE: Self-Efficacy; STDR : Sight-threatening diabetic retinopathy

Acknowledgements

The authors acknowledge the Queen Elizabeth Diamond Jubilee Trust and the Commonwealth Eye Health Consortium for funding the trial. We are grateful to the following for their support with the preparatory work ahead of the study: Ophthalmic Services Unit, Ministry of Health, Kenya; the Division of Non-Communicable Diseases, Ministry of Health Kenya; Kenya Defeat Diabetes Association; Kerugoya County Government and Kenya Medical Training College. We thank the staff of the Disability and Eye Health Group, LSHTM for valuable input.

Funding

For the study is provided by The Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium, as a research grant awarded to Nyawira Mwangi at the London School of Hygiene and Tropical Medicine (LSHTM Grant code ITCRZ6816). The sponsor is the LSHTM, Keppel Street, WC1E 7HT London (Quality and Governance Manager) and the study has received peer review. The funders have no role in the design, execution and interpretation of study data, or in the decision to publish the study protocol.

Authors' contributions

NM, CB and AF conceptualised the study. SG, DM, CM and LM reviewed and revised the study design that was written by NM with the help of CB and AF. Thereafter MN, EG, PT, AJ, KG, MG and JK participated in the development of the study interventions as well as recruitment of the study centres to participate in the study. All authors critically reviewed of the protocol prepared by NM. Thereafter, NM obtained the ethics approvals and drafted the manuscript. All authors read and approved the final version, and agreed to participate in the study.

Ethics approval and consent to participate

Ethical approval has been granted by the Ethics Review Committees of London School of Hygiene and Tropical Medicine and the African Medical Research Foundation. Any protocol amendments will also be approved by these committees. Procedures and methods used in this study conform to the ethical guidelines defined by the World Medical Association's Declaration of Helsinki and are described in detail in the study protocol (version May 2017). Written informed consent will be sought at the cluster level (before randomization) from the leaders of the DSGs. Participants will be asked for verbal consent to receive the intervention (before randomization), and for verbal consent to follow-up.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Kenya Medical Training College, Nairobi, Kenya. ²International Centre for Eye Health, London School of Hygiene and Tropical Medicine, Keppel Street, WC1E 7HT, London, UK. ³Kerugoya County Referral Hospital, Kerugoya, Kenya. ⁴Kirinyaga County Health Services, Kerugoya, Kenya. ⁵Department of Ophthalmology, University of Nairobi, Kanya. ⁶Oxford University Hospitals NHS Trust, Oxford, UK. ⁷Kenya Medical Research Institute, Nairobi, Kenya. ⁸Kabarak University, Nakuru, Kenya. ⁹Upper Hill Eye and Laser Centre, Nairobi, Kenya. ¹⁰Ophthalmic Services Unit, Ministry of Health, Nairobi, Kenya. ¹¹Division of Non-Communicable Diseases, Ministry of Health, Nairobi, Kenya.

Received: 19 December 2017 Accepted: 26 June 2018 Published online: 13 July 2018

References

- 1. International Diabetes Federation. IDF diabetes atlas 7th edition. 2015.
- Peer N, Kengne AP, Motala AA, Mbanya JC. Diabetes in the Africa region: an update. Diabetes Res Clin Pract. 2014;103(2):197–205.

- Guariguata L, Whiting DR, Hambleton I, Beagley J, Linnenkamp U, Shaw JE. Global estimates of diabetes prevalence for 2013 and projections for 2035. Diabetes Res Clin Pract. 2014;103(2):137–49.
- Ogurtsova K, da Rocha Fernandes JD, Huang Y, Linnenkamp U, Guariguata L, Cho NH, Cavan D, Shaw JE, Makaroff LE. IDF diabetes atlas: global estimates for the prevalence of diabetes for 2015 and 2040. Diabetes Res Clin Pract. 2017;128(2017):40–50.
- Ministry of Health, Kenya National Bureau of statistics, World Health Organization: Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. In.: Ministry of Health, division of non-communicable Diseases. 2015;20(2):44–51.
- UK Prospective Diabetes Study (UKPDS) Group*. Intensive blood-glucose control with sulphonylureas or insulin compared with conventional treatment and risk of complications in patients with type 2 diabetes (UKPDS 33). Lancet. 1998;352(9131):837–53.
- Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, Chen S-J, Dekker JM, Fletcher A, Grauslund J, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556–64.
- Wanjiku M, Andrew B, Tunde P, Irene L, David Y, Allen F, Hannah K. Prevalence and correlates of diabetic retinopathy in a population-based survey of older people in Nakuru, Kenya. Ophthalmic Epidemiol. 2014;21(3): 169–77.
- American Diabetes Association. Standards of Medical Care in Diabetes-2017Diabetes Care. 2017;40(Suppl 1)
- International Council of Ophthalmology. ICO guidelines for diabetic eye care-updated 2017. San Francisco: California: international council of Ophthalmology; 2017.
- 11. Wilson JMG, Jungner G. Principles and practices for screening for disease. Public Health Papers of the World Heath Organization. 1968;34
- 12. Ministry of Health: Guidelines for screening and management of Diabetic Retinopathy. In. Nairobi: Minstry of Health Kenya; 2017.
- 13. Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. Diabet Med. 2010;27(3):249–56.
- Sloan Frank A, Grossman Daniel S, Lee Paul P. Effects of receipt of guidelinerecommended care on onset of diabetic retinopathy and its progression. Ophthalmology. 2009;116:1515–21. e1513
- Mumba M, Hall A, Lewallen S. Compliance with eye screening examinations among diabetic patients at a Tanzanian referral hospital. Ophthalmic Epidemiol. 2007;14(5):306–10.
- Njambi L. Prevalence of diabetic retinopathy and barriers to uptake of diabetic retinopathy screening at Embu provincial general hospital, Central Kenya. East African J Ophthalmology. 2012;16:5–11.
- 17. Gitalisa A, Dandan W, Cosmos O, Nathan C. Use of eye care services among diabetic patients in urban Indonesia. Archives of ophthalmology (Chicago, III : 1960). 2011;(129):930–5.
- Roopa S, Sandeep B, Dimple K, Ali Mohammed K, Dorairaj P, Venkat Narayan KM, Nikhil T. Adherence to diabetes care processes at general practices in the National Capital Region-Delhi, India. Indian J Endocrinology Metabolism. 2016;20(3):329–36.
- Wang D, Ding X, He M, Yan L, Kuang J, Geng Q, Congdon N. Use of eye care services among diabetic patients in urban and rural China. Ophthalmology. 2010;117(9):1755–62.
- Gudlavalleti Murthy VS, Raghupathy A, Venkat GAS, Ramachandra Srikrishna S, Rajan S, Jotheeswaran a T, Giridhara BR, Vivek S, Komal A, Jayanti S, et al. Perceptions and practices related to diabetes reported by persons with diabetes attending diabetic care clinics: the India 11-city 9-state study. Indian J Endocrinology Metab. 2016;20(Suppl 1):S26–32.
- Oluwatoyin HO, Odunmorayo AA, Ayodeji KB. Determinants of previous dilated eye examination among type II diabetics in southwestern Nigeria. European J Internal Med. 2010;21:176–9.
- Mwangi N, Macleod D, Gichuhi S, Muthami L, Moorman C, Bascaran C, Foster A. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Tropical Medicine and Health. 2017;45(41)
- 23. Levesque JL, L, Harris Mark F, Russell G. Patient-centred access to health care: conceptualising access at the interface of health systems and populations. Int J Equity Health. 2013;12:18.
- Baumeister SE, Schomerus G, Andersen RM, Tost F, Markus MR, Volzke H, Jurgens C. Trends of barriers to eye care among adults with diagnosed diabetes in Germany, 1997-2012. Nutr Metab Cardiovasc Dis. 2015;25(10):906–15.
- 25. Mafwiri MM, Mwakyusa N, Shilio B, Lutale JK. Health education and awareness about diabetic retinopathy among patients attending diabetic

clinics in tertiary and regional hospitals in Tanzania. J Ophthalmology of Eastern Central and Southern Africa. 2016;

- Zhang X, Norris SL, Saadine J, Chowdhury FM, Horsley T, Kanjilal S, Mangione CM, Buhrmann R. Effectiveness of interventions to promote screening for diabetic retinopathy. Am J Prev Med. 2007;33(4):318–35.
- 27. Dennis C-L. Peer support within a health care context: a concept analysis. Int J Nurs Stud. 2003;40(3):321–32.
- World Health Organization. Peer Support Programs in DiabetesDiabetes. Geneva: World Health Organization; 2007.
- Brownson CA, Heisler M. The role of peer support in diabetes care and selfmanagement. Patient. 2009;2(1):5–17.
- 30. Bandura A. Self efficacy: the exercise of control. New York: Freeman; 1997.
- Bandura A: Self efficacy. In: Encyclopedia of human behavior. Edited by Ramachaudran V S, vol. 4. New York: academic press. (reprinted in H. Friedman [Ed.], encyclopedia of mental health. San Diego: Academic Press, 1998). 1994: 71–81.
- Hall CE, Hall AB, Kok G, Mallya J, Courtright P. A needs assessment of people living with diabetes and diabetic retinopathy. BMC Res Notes. 2016;9:56.
- Riddell MA, Renwick C, Wolfe R, Colgan S, Dunbar J, Hagger V, Absetz P, Oldenburg B. Investigators TAPfPDP: cluster randomized controlled trial of a peer support program for people with diabetes: study protocol for the Australasian peers for progress study. BMC Public Health. 2012;12(843)
- 34. Bui TD, Kadzakumanja O, Munthali C. Mobilizing for the Lilongwe diabetes peer support Programme in Malawi. Malawi Med J. 2014;26(4):124–5.
- Baumann LC, Frederick N, Betty N, Jospehine E, Agatha N. A demonstration of peer support for Ugandan adults with type 2 diabetes. Int J Behav Med. 2015;22(3):374–83.
- 36. Whittle J. When does peer support improve glycemic control in persons with diabetes mellitus? JAMA Intern Med. 2014;174(6):982–3.
- Fisher EB, Earp JA, Maman S, Zolotor A. Cross-cultural and international adaptation of peer support for diabetes management. Fam Pract. 2010; 27(Suppl 1):i6–16.
- Ghorob A, Vivas MM, De Vore D, Ngo V, Bodenheimer T, Chen E, Thom DH. The effectiveness of peer health coaching in improving glycemic control among low-income patients with diabetes: protocol for a randomized controlled trial. BMC Public Health. 2011;11(208)
- Paul G, Smith SM, Whitford D, O'Kelly F, O'Dowd T. Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes. BMC Health Serv Res. 2007;7:136.
- Mash R, Kroukamp R, Gaziano T, Levitt N. Cost-effectiveness of a diabetes group education program delivered by health promoters with a guiding style in underserved communities in cape town, South Africa. Patient Educ Couns. 2015;98(5):622–6.
- Tang TS, Funnell MM, Gillard M, Nwankwo R, Heisler M. Training peers to provide ongoing diabetes self-management support (DSMS): results from a pilot study. Patient Educ Couns. 2011;85(2):160–8.
- Gao J, Wang J, Zheng P, Haardörfer R, Kegler MC, Zhu Y, Fu H. Effects of self-care, self-efficacy, social support on glycemic control in adults with type 2 diabetes. BMC Fam Pract. 2013;14(66)
- Ussher J, Kirsten L, Butow P, Sandoval M. What do cancer support groups provide which other supportive relationships do not? The experience of peer support groups for people with cancer. Soc Sci Med. 2006;62(10):2565–76.
- 44. Govindasamy D, Meghij J, Kebede Negussi E, Clare Baggaley R, Ford N, Kranzer K. Interventions to improve or facilitate linkage to or retention in pre-ART (HIV) care and initiation of ART in low- and middle-income settings–a systematic review. J Int AIDS Soc. 2014;17:19032.
- 45. Lewycka S, Mwansambo C, Rosato M, Kazembe P, Phiri T, Mganga A, Chapota H, Malamba F, Kainja E, Newell M-L, et al. Effect of women's groups and volunteer peer counselling on rates of mortality, morbidity, and health behaviours in mothers and children in rural Malawi (MaiMwana): a factorial, cluster-randomised controlled trial. Lancet. 2013;381(9879):1721–35.
- Elafros MA, Mulenga J, Mbewe E, Haworth A, Chomba E, Atadzhanov M, Birbeck GL. Peer support groups as an intervention to decrease epilepsyassociated stigma. Epilepsy Behav. 2013;27(1):188–92.
- Beaglehole R, Epping-Jordan J, Patel V, Chopra M, Ebrahim S, Kidd M, Haines A. Improving the prevention and management of chronic disease in lowincome and middle-income countries: a priority for primary health care. Lancet. 2008;372(9642):940–9.
- 48. Weaver MS, Lonnroth K, Howard SC, Roter DL, Lam CG. Interventions to improve adherence to treatment for paediatric tuberculosis in low- and

middle-income countries: a systematic review and meta-analysis. Bull World Health Organ. 2015;93(10):700–711B.

- Campbell HS, Phaneuf MR, Deane K. Cancer peer support programs-do they work? Patient Educ Couns. 2004;55(1):3–15.
- Kumakech E, Cantor-Graae E, Maling S, Bajunirwe F. Peer-group support intervention improves the psychosocial well-being of AIDS orphans: cluster randomized trial. Soc Sci Med. 2009;68(6):1038–43.
- Peterson JL, Rintamaki LS, Brashers DE, Goldsmith DJ, Neidig JL. The forms and functions of peer social support for people living with HIV. J Assoc Nurses AIDS Care. 2012;23(4):294–305.
- Qi L, Liu Q, Qi X, Wu N, Tang W, Xiong H. Effectiveness of peer support for improving glycaemic control in patients with type 2 diabetes: a metaanalysis of randomized controlled trials. BMC Public Health. 2015;15:471.
- Coleman C. Teaching health care professionals about health literacy: a review of the literature. Nurs Outlook. 2011;59(2):70–8.
- van Zyl DG, Rheeder P. Physician education programme improves quality of diabetes care. JEMSDA. 2005;10(3):86–90.
- 55. Medical Research Council. Developing and evaluating complex interventions: new guidance. London: Medical Research Council.
- Moher D, Hopewell S, Schulz KF, Montori V, Gotzsche PC, Devereaux PJ, Elbourne D, Egger M, Altman DG. CONSORT 2010 explanation and elaboration: updated guidelines for reporting parallel group randomised trials. Bmj. 2010;340:c869.
- Campbell MK, Piaggio G, Elbourne DR, Altman DG, Group C. Consort 2010 statement: extension to cluster randomised trials. Bmj. 2012;345:e5661.
- Hayes RJ, Bennett S. Sample size calculation in cluster randomized trials. Int J Epidemiol. 1999;28(319–326)
- McKenzie J, Ryan R, Di Tanna GL, Cochrane Consumers and Communication Review Group: Cochrane Consumers and Communication Review Group: cluster randomised controlled Trials In.; 2014.
- Kyari F, Tafida A, Sivasubramaniam S, M GVS, Peto T, Gilbert CE. Nigeria National Blindness and visual impairment study group: prevalence and risk factors for diabetes and diabetic retinopathy: results from the Nigeria national blindness and visual impairment survey. BMC Public Health. 2014; 14(1299):1–12.
- Burgess PI, Msukwa G, Beare NAV. Diabetic retinopathy in sub-Saharan Africa: meeting the challenges of an emerging epidermic. BMC Med. 2013; 11(157)
- 62. MacColl Center for Health Care Innovation. Chronic care model. In: Improving chronic illness care; 1997.
- Beaton SJ, Sperl-Hillen JAM, Von Worley A, Fernandes OD, Baumer D, Hanson AM, Parker ED, Busch ME, Davis HT, Victor Spain C. A comparative analysis of recruitment methods used in a randomized trial of diabetes education interventions. Contemporary Clinical Trials. 2010;31(2010):549–57.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions



Sample size calculation for DURE study

Based on this formula:

The formula for calculation of sample size for cluster-randomized trials whose primary outcome is a proportion is

$$\begin{split} \mathbf{c} &= 1 + (z_{\alpha/2} + z_\beta)^2 [\pi_0 (1 - \pi_0)/\mathbf{n} + \pi_1 (1 - \pi_1)/\mathbf{n} \\ &\quad + \mathbf{k}^2 (\pi_0^2 + \pi_1^2)] / (\pi_0 - \pi_1)^2 \end{split}$$

(Hayes and Bennett, 1999)

Parameter Zα/2=the normal distribution value corresponding to risk for Type 1 error (at 95% confidence level)	Value 1.96	Notes
Zβ = the normal distribution value corresponding to β for statistical power at 80%	0.84	
pi0 = true population proportion in control group	0.12	(using data from the health system assessment – 12.2% of PLWD in Kirinyaga had ever had a screening eye examination
pi1 = true population proportion in the intervention group	0.24	Reflecting an expected two- fold increase in the proportion that has ever had a screening eye examination
n = number of individuals in each cluster	45	Increased to 50 to account for potential loss to follow-up
k = coefficient of variation of the true proportions between clusters within each arm	0.25	Estimate gained from assessment of pattern of uptake of DR screening in LMICs
(Ζα/2+Ζβ)^2	7.84	
pi0(1-pi0)/n	0.1056	
pi1(1-pi1)	0.1824	
k^2(pi0^2+pi1^2)	0.0045	
(pi0-pi1)^2	0.0144	
c = number of clusters	6.9344	=7 clusters in each arm

Chapter Eight

Effectiveness of peer support to increase attendance at DR screening - a pragmatic cluster randomized clinical trial Pilot of the DURE trial

8.0 Overview

The previous chapter of this thesis (chapter 7) described the trial methods to addresses objective 4 in chapter 1 of the thesis: To develop and test an intervention through a randomized clinical trial (RCT) to improve eye care services for PLWD in Kenya.

This chapter consists of a research paper that describes the randomised pilot and feasibility study conducted in advance of the definitive RCT (Uptake of Retinal Examination in Diabetes, DURE trial), with the primary aim of assessing feasibility. The RCT is reported in adherence to the Consolidated Standards of Reporting Trials (CONSORT) 2010 statement and its extension to randomised pilot and feasibility trials (www.consort-statement.org).

Research paper 6 (submitted to BMC Pilot and Feasibility Studies) includes data on baseline characteristics of participants, feasibility outcomes as well as an interim measure of the effectiveness of the intervention. Publishing these results contributes to maximising the benefits of this research to other organisations or researchers seeking to understand the value of pilot and feasibility studies to the progression to full RCTs.
8.1 Research Paper 6

BMC Pilot and Feasibility Studies – Licences and Copyright

The following terms apply to this publication

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy, distribute, and use their attributed research, free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago.

All articles published by *BMC Pilot and Feasibility Studies* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access https://pilotfeasibilitystudies.biomedcentral.com/about



Registry T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?			
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	BMC Pilot and Feasibility Studies
Please list the paper's authors in the intended authorship order:	Nyawira Mwangi, Covadonga Bascaran, David Muraguri, Mark Ng'ang'a, Jacqueline Ramke, Mathew Kipturgo, Stephen Gichuhi, Min Kim, David Macloed, Consuela Moorman, Esbon Gakuo, Lawrence Muthami, Allen Foster
Stage of publication	Submitted

SECTION D – Multi-authored work

	I designed the study, conducted the research,
For multi-authored work, give full details of your role in	analysed and interpreted the data, drafted the
the research included in the paper and in the preparation	manuscript and prepared subsequent revisions
of the paper. (Attach a further sheet if necessary)	in consideration of the comments from the
1	co-athors and peer reviewers

Student Signature:

Supervisor Signature:



Date: _____17 SEPT 2019

Date: SEPT 17, 2019

Improving health worldwide

Chapter 8

Research paper 6

Feasibility of a cluster randomized controlled trial on effectiveness of peer –led health education interventions to increase uptake of retinal examination for diabetic retinopathy in Kirinyaga, Kenya: a pilot trial

Nyawira Mwangi^{1,2}, Covadonga Bascaran², David Muraguri³, Mark Nganga³, Jacqueline Ramke², Mathew Kipturgo¹, Stephen Gichuhi⁴, Min Kim², David Macleod², Consuela Moorman⁵, Esbon Gakuo³, Lawrence Muthami⁶, Allen Foster²

¹Kenya Medical Training College, Nairobi, Kenya

²London School of Hygiene and Tropical Medicine, London, United Kingdom

³Kerugoya County Referral Hospital, Kerugoya, Kenya

⁴University of Nairobi, Nairobi, Kenya

⁵Oxford Universities NHS Trust, Oxford, United Kingdom

⁶Kenya Medical Research Institute, Nairobi, Kenya

Corresponding author: Nyawira Mwangi, London School of Hygiene and Tropical Medicine,

Keppel Street, WC1E 7HT, London, United Kingdom. Email: <u>Nyawira.Mwangi@lshtm.ac.uk</u>

Abstract

Background

People living with diabetes can reduce their risk of vision loss from diabetic retinopathy by attending screening, which enables early detection and timely treatment. The aim of this pilot trial was to assess the feasibility of a full-scale cluster randomized controlled trial of an intervention to increase uptake of retinal examination in this population, as delivered within existing community-based diabetes support groups (DSGs).

Methods

All 16 DSGs in Kirinyaga county were invited to participate in the study. The first two groups recruited took part in the pilot trial. DSG members who met the eligibility criteria were recruited before the groups were randomized to the two arms. In the intervention group, two peer-supporters were trained to deliver monthly DSG-based eye health education and individual telephone reminders to attend screening. The control group continued with usual DSG practice, which is monthly meetings without eye health education. The recruitment team and outcome assessors were masked to the allocation. We documented the study processes to ascertain the feasibility, acceptability and potential effectiveness of the intervention. Feasibility was assessed in terms of clarity of study procedures, recruitment and retention rates, level of acceptability and rates of uptake of eye examination. We set the target feasibility criteria for continuation to the main study to be recruitment of 50 participants in the trial, 80% monthly follow-up rates for individuals and no attrition of clusters.

Results

Of the 122 DSG members who were assessed for eligibility, 104 were recruited and followed up; 51 (intervention) and 53 (control) arm. The study procedures were well understood and easy to apply. We learnt the DSG meeting days were the best opportunities for recruitment. The study

had a high acceptance rate (100% for clusters, 95% for participants) and high follow up and retention rate (100 % of those recruited). All clusters and participants were analysed. We observed that the rate of incidence of eye exam was about 6 times higher in the intervention arm as compared to the control arm. No adverse unexpected events were reported in either arm.

Conclusions

The study is feasible and acceptable in the study population. The results support the development of a full-scale cluster RCT, as the success criteria for the pilot were met.

Trial registration

Pan African Clinical Trials Registry PACTR201707002430195 Registered 25 July 2017 Keywords

Diabetes, Diabetic retinopathy, peer-support, Kenya, pilot, cluster randomised controlled trial

Background

The long-term complications of diabetes, such as diabetic retinopathy (DR), are a threat to health among people living with diabetes (PLWD). DR is a growing concern in global epidemiology due to the high proportion of DR that remains undetected. Vision loss from DR can be prevented through regular retinal screening (hereafter referred to as 'screening') and timely treatment.¹⁻³ There is notable geographic variation in the incidence and visual impairment burden of DR, both within and between countries, reflecting variation in access to health care.⁴⁻⁸ Services for DR prioritize early detection, metabolic control, regular monitoring and timely treatment. Access to these services is a significant challenge due to demand side barriers (such as low awareness of the need for services among PLWD) and supply side barriers (such as availability of clinical guidelines or screening services).⁹ There is a need for better evidence and patient empowerment to address the demand side barriers, as well as health system strengthening to address supply-side barriers.⁹⁻

In the *Global action plan for the prevention and control of non-communicable diseases(NCDs)* 2013-2020, the World Health Organization (WHO) highlighted the need to empower people with NCDs to seek early detection, and to provide them with appropriate education, incentives and tools for self-management.¹² The peer-support model has been used in diabetes and other chronic conditions to improve social support and self-management, with positive outcomes in other countries.¹³⁻¹⁷ In contrast, peer-support has not been used in diabetes eye health services, and subsequently there is a knowledge gap regarding its effectiveness to reduce vision loss from diabetic retinopathy. Leveraging on peer-support in a clinical or community setting might be a potential enabler for the adoption of healthy behaviours, such as screening.

Clinical guidelines target to have a 100% attendance to regular screening.¹⁸ Our health system assessment reported that PLWD in three counties of Kenya have low attendance to annual screening, which is the frequency recommended in this setting.¹⁹ This is consistent with findings that uptake of DR screening is low in many parts of the world, but more so where access to health care is generally limited.⁹ To address this deficit, the Uptake of Retinal Examination in Diabetes (DURE) trial²⁰ aims to test the effectiveness of peer-support in increasing the uptake of retinal examination among members of diabetes support groups (DSGs). Diabetes support groups are volunteer social groups of PLWD in which peers provide mutual support for improving diabetes care. The support may include information and skills for self-management, as well as emotional support. Given this objective, DSG members are likely to be health conscious and interested in adopting healthy behaviours. The intervention in this study is based on the self-efficacy theory²¹ and is targeted to PLWD who are already members of support groups and have not had screening in the previous 12 months or longer. Screening in this setting involves a visual acuity test and a retinal examination through a dilated pupil.^{18, 22}

The study setting is a rural county whose inhabitants are mainly small-scale farmers. The DSGs are spread over the 1200km² area of the county. Undertaking the DURE study raises important practical concerns. In this pilot study, our aim was to gain experience in delivering the intervention, and to assess if the DURE cluster randomized clinical trial (cRCT) is feasible by: (1) Testing clarity and ease of study procedures for enrolment and data collection. (2) Determining the potential for participant recruitment and retention. (3) Assessing the acceptability of the intervention, by considering the level of adoption of the study intervention on the primary outcome. Our hypothesis was that it is feasible to conduct the DURE study. We set the target feasibility criteria for continuation to the main study to be recruitment of 50 participants in each cluster, at least 80% follow-up rate of participants in each month of the trial and no attrition of clusters. The 90 - day duration of the pilot trial was considered sufficient for these feasibility objectives, while the main trial will take six months.

Methods

Study setting

The demographic and health statistics of Kirinyaga county are highlighted in Table 8-1. The study intervention was developed following a health system assessment for diabetes and DR in three counties of Kenya, which identified gaps in access to services for DR, as well as the need for health system strengthening. We found that only 7% of PLWD in this county had a DR screening exam in the preceding 12 months. The main barriers to access are lack of referral from diabetes services, lack of knowledge of diabetes eye complications among PLWD and the belief that a screening exam is only necessary once ocular symptoms develop.

An estimated 25-30% of the PLWD in the county are regular members of DSGs (with a registration number) while another 20% of PLWD attend some DSG meetings even though they are not members. Peer group leaders and community health volunteers recruit members as they give

group health talks at community meetings, churches, outreach camps, and diabetes clinics in health facilities. As membership is voluntary, the distribution of members by demographic parameters in different groups varies. All groups are under the Kenya Defeat Diabetes Association, which provides them with equipment for use within the group (such as a glucometer and a blood pressure machines). The association also trains peer-supporters and DSG leaders.

Parameter	Kirinyaga county	Kenya
Total population (estimates based on 2009 census)	595,379	48.5million
Females	50%	50%
Age > 18 years	409,995	22,005,235
Urban population	16%	29.9%
No of people with diabetes $(2\%)^{23}$	8,185	440,104
No of people needing an annual eye exam	8,185	440,104
No of eye care facilities	1	112
No of ophthalmologists	1	115
No of ophthalmic clinical officers	3	300
No of ophthalmic nurses	0	200

Table 8-1: Demographic and health statistics- Kirinyaga county

DSGs hold routine monthly meetings at a dedicated time and location in the community. An estimated 80% of the members attend at least two thirds of the meetings annually. The meetings are held in the morning, starting between 8 and 9am and last 2-3 hours. Each member's fasting blood sugar, blood pressure and weight are recorded. The group then shares a light meal. The cost of the blood sugar test strips and meal are met by a contribution of Kenya shillings 100 (the equivalent of 1\$ dollar) per PLWD attending the meeting. The other activities in the meeting include: group health talks delivered by peer-supporters; informal discussions among PLWD; planning for advocacy and awareness-raising activities. A record of these activities are captured in attendance registers and minutes of the meeting.

Sampling

All 16 DSGs in Kirinyaga were eligible for inclusion. We invited the DSG leaders to a meeting where we explained the objectives of the DURE study and invited all the groups to participate. The leaders then took time to discuss the study with their members before giving approval through signing consent forms. The first two DSGs to confirm willingness to participate were included in the pilot study, for simplicity, transparency and visibility to the DSG leaders. All sixteen DSGs consented to participate hence the remaining 14 will participate in the main study.

We aimed to recruit at least 50 members who met the eligibility criteria (Table 8-2) in each DSG (size of DSGs is 80-100 members). This is the same cluster size calculated for the main study, using the formula for sample size calculation provided by Hayes and Bennett.²⁴ A statistician not involved with the fieldwork conducted the sample size calculation. We also recruited two peer-supporters (1 male and 1 female) who met the eligibility criteria to deliver the intervention in the intervention cluster.

Design

The pilot study design mimics the design of the main study, being a two-arm cRCT with a 1:1 ratio. A research nurse who is a local health worker recruited participants who met the eligibility criteria during a DSG meeting. The list of existing DSG members was provided by the DSG lead. Participants were recruited through random selection. The research nurse assessed each potential participant for eligibility using the criteria in Table 8-2. Verbal consent for recruitment and follow-up was obtained at the time of recruitment, from individual participants who met the eligibility criteria (DSG leaders provided cluster-level consent). For those who did not consent, the reasons for non-consent were recorded. Baseline demographic, anthropometric and metabolic data was collected at the time of recruitment using standard operating procedures (appendix 3). All participants were given a study identifier card to present at the eye clinic at the time of eye exam. Two members of the research team observed the recruitment process to identify any difficulties with participant recruitment, eligibility criteria or completing the data collection tool. The recruitment nurse provided additional feedback on these critical components in a debriefing session after each recruitment session.

Criterion	Participants	Peer-supporters
Age > 18 years		
Member of a diabetes support group		
Will reside in the county for the next 12 months		
Has a mobile phone		
Willing to participate in the study		
Had not had a screening exam in the last 12 months		
Has had a screening exam in the preceding 12 months		
Willing to be a peer-supporter		
Willing to commit two days for training		
Willing to commit many hours to peer-support work		
Fluent in Kikuyu or Kiswahili		
Already attending DR screening		
Already receiving treatment for DR		
Has a debilitating illness		

Table 8-2: Eligibility criteria for participants and peer-supporters

KEY	Include	Exclude	Not applicable
-----	---------	---------	----------------

Following participant recruitment within each of the DSGs, random allocation of the intervention was through drawing of lots. A lay person not participating in the study picked one of four sealed and opaque envelopes from a container, in the presence of two members of the research team. Each envelope contained a card bearing the name of one DSG and either 'intervention' or 'control'. Opening the envelope revealed the arm allocation for one group, and by inference the allocation of the other group. This allocation was copied on the envelope and stored to provide a reference trail. The remaining envelopes were destroyed. A research assistant (not involved in recruitment or outcome assessment) followed up the participants in each group at monthly intervals for 90 days from the first group education session, to check retention rates.

The primary outcome was the feasibility of recruiting 50 participants in each cluster, and achieving at least 80% follow-up rate of participants in each month of the trial. The records at Kerugoya County Hospital eye clinic were monitored daily and the identifier cards of participants that attended screening were deposited in a specific container by the eye care team. These cards were then collected and given to the outcome assessment nurse (not involved in recruitment or follow-up). During the study, we also found that some eye care teams external to the county health services held outreach camps in the county and provided screening for some of the participants. As our team was alerted ahead of the outreach camps, we liaised with these teams and monitored the attendance of any study participants. As they used the same screening guidelines, any participant screened at the outreach site was taken to have the outcome of interest.

The recruitment nurse, research assistants and the outcome assessors had no training in eye care, and were masked to the cluster and participant allocation, to avoid contamination and bias. The eye care providers were also masked to the intervention allocation. It was not possible to mask the study participants or peer-supporters in the intervention arm, because the peer-supporters' activities within the DSG were overt. As the primary outcome was assessed from hospital records, we concluded that the lack of masking could not incentivise over-reporting or under-reporting in either arm.

Intervention

The DSGs in the intervention arm received the study intervention combined with usual care for 90 days, while the control group received the usual standard of care alone (Table 8-3). The intervention was a monthly group health talk and individual monthly telephone reminders to attend eye exam, delivered by two peer-supporters (1 male, 1 female) collaboratively. The

263

intended mechanism of the intervention and the key messages to be delivered in the health talk are described in the trial protocol.²⁰ Fidelity to the intervention and the influence of peer-supporter characteristics will be assessed in the process evaluation of the main trial.

We trained the peer-supporters for two days to deliver the intervention. They had already received previous training as peer-educators (training provided by the Kenya Defeat Diabetes Association), and had about four years' experience with providing peer education. We also supported them during the implementation phase in the following ways. (1) Telephone calls were made by the team before and after each group session to discuss the sessions and any challenges faced by the peer leaders. (2) Practical support was provided to help organize the logistics for local program delivery, such as provision of telephone airtime and reimbursement for transport costs for peer-supporters. (3) The research team attended two of the three DSG meetings. Peer-supporters kept logs of the DSG attendance during group sessions and the individual reminders.

Domain	Intervention group	Control group
Usual care	Monthly group meetings with general diabetes education talks, blood sugar and blood pressure measurements	Monthly group meetings with general diabetes education talks, blood sugar and blood pressure measurements
Intervention		
Peer-supporters training	Two days training following a structured curriculum regarding diabetes eye disease, retinal screening, role of peer-supporters, communication and other aspects specified in the protocol ²⁰	
Group education	Monthly group education provided by trained peer- supporters, with structured content on diabetic eye disease and retinal screening as specified in the protocol ²⁰	
reminders	reminder by peer-supporters	

Table 8-3: Trial Interventions

to participants to take a screening exam as specified in the protocol²⁰

Reporting and analysis

This study is reported in accordance with the Consolidated Standards of Reporting Trials (CONSORT) guidelines for pilot studies.²⁵ The CONSORT flow chart and the checklist are included as appendices. We documented recruitment procedures and rates, reasons for ineligibility and non-participation, and follow-up rates. We calculated descriptive statistics for each study arm at baseline. We also summarized survival outcomes at arm level as per intention-to-treat analysis and estimated hazard ratio for any differences. All data analysis was carried out using Stata version 15 (StataCorp, Texas, USA).

Results

The results are reported under four headings corresponding to the specific objectives of the pilot trial:

Study procedures for enrolment and data collection

We found that the enrolment rate was high during the DSG meetings. The peer group leaders could predict which meetings would be well attended, considering other concurrent community activities, and they mobilised members to attend. We found it helpful to liaise with these leaders in planning for the recruitment.

The study materials were easy to carry around and work with. We did not identify any practical, ethical or interpretation difficulties in the use of the eligibility criteria and the completion of the data collection tools. The eligibility criteria were therefore found to be appropriate and the data collection tools were well understood. We collected baseline data for all participants, before randomization. The time taken to complete the data collection processes per participant exceeded the planned time. Some of the participants had difficulty finding some of the data that we needed for the purpose of follow-up, such as the telephone numbers of the next of kin, but with assistance they were able to retrieve this from their phones or to contact other people who had the data. However, this added about 10 minutes per participant over the initial estimates for the recruitment process. This resulted in increased waiting time for the other persons awaiting recruitment. Learning from this, we trained three additional research nurses for the recruitment team, so that at least two of them could attend each recruitment meeting.

A research assistant checked the data collection tools for completeness at the recruitment site. All the data forms were also checked by the team lead for quality assurance after each recruitment meeting. Thereafter the forms were sent to the data entry assistant where further quality checks were carried out as data was entered into a database.

We used standardised protocols for all measurements, and the descriptive data is provided in Table 8-4. The two arms were balanced for most characteristics except age and gender. Females and older people were over-represented in the intervention arm. This difference reflects the variation in the existing composition of the DSGs.

Potential for recruitment and retention

The response rate of clusters was good (both accepted to participate). All 122 participants assessed for eligibility were willing to participate, but 6 (4.9%) withdrew during the recruitment process because of long waiting time. Of the 122 assessed for eligibility, 12 (9.8%) were ineligible as they were temporary visitors (non-resident) or were going to be absent from support groups during the study period (school and employment commitments).

By conducting recruitment at well-attended routine DSG meetings, all or most of the regular DSG members had an equal chance to be recruited in the study. We were able to recruit within the

anticipated time such that all the participants entered the study at time 0, and we had similar recruitment levels in both clusters. Follow-up rates were high in both arms. Both DSGs remained in the study and received the intended intervention (Fig 1). One peer-supporter fell sick over part of the period, but by this time most of participants in intervention arm already had the outcome of interest, and the other peer-supporter was able to carry on with the intervention. All participants were followed up for 90 days, and there was zero loss to follow up.

Table 8-4: Baseline characteristics				
	Intervention (N=51)	Control (N=54)	Total (N=104)	
Age (years)				
Median (IQR)	69 (60-72)	57 (57-62)	63 (54.5-70)	P=0.002
Sex				
Female (%)	41 (80.4)	30 (56.6)	71 (68.3)	P=0.009
Duration of diabetes (years)				
Median(IQR)	5.0 (2-10)	5.0 (2-8)	5.0 (2-10)	
Duration of support group membership (years)				
Median(IQR)	3.0 (1-3.5)	2.0 (1-4)	2.0 (1-4)	
Anthropometric measures Body Mass Index (kg/m ²)				
Male	25.0 (2)	25.3 (4.4)	25.2 (3.8)	
Female	25.4 (3.9)	26.4 (3.4)	25.8 (3.7)	
Waist circumference (cm)				
Male	96.5 (7.3)	96.0 (11.1)	96.1 (10)	
Female	97.0 (8.9)	100.0 (8.4)	98.3 (8.8)	
Metabolic measures				
Fasting blood sugar (g/dl)	7.4 (2.7)	9.1 (4.5)	8.3 (3.8)	
Systolic blood pressure (mmHg)	139.0 (20)	142.0 (26)	141.0 (23)	
Diastolic blood pressure (mmHg)	81.0 (7.6)	78.0 (13)	80.0 (11)	

Acceptability of the intervention

Before the start of the study, we had initial meetings with the Ophthalmic Services Unit at the national level, the Kirinyaga county director of health, eye care providers in the county, and the national officials of Kenya Defeat Diabetes Association (KDDA) that is the umbrella body for DSGs. We obtained their buy-in for the study, and they linked us with the DSGs.

Given the recruitment rates, the follow up rates and the attrition rates (Figure 8-1, Flow chart at the end of the paper), the study and the intervention were acceptable to participants and clusters. The peer-supporters attended all the sessions of the two-day training and gave the group talks as planned. For the individual telephone reminders, we found that peer-supporters supplemented this with face-to-face discussions with the individuals who were yet to take a retinal exam (in addition to the telephone reminders). This flexibility reflects a sense of ownership of the intervention by the peer-supporters.

We did not record any adverse event related to the intervention. Temporary blurring of vision after dilatation of the pupils during retinal examination (the outcome of interest) is a common undesirable effect but this was explained to the participants before examination and none of them declined to have the examination.

The 'drop-in' referral mode of patients from the DSGs for screening for DR was acceptable to both patients (who adopted it) and eye care providers (as they screened all who dropped in). The attendance to the eye clinic showed some peaks, and during these peaks the workload in the eye clinic was significantly increased, however all who turned up were screened. None of the participants who presented at the eye clinic had lost or forgotten the identifier card, which would suggest that the card was highly valued. Such participants would still have received screening and this would be captured in the eye clinic records.

Estimating the effectiveness of intervention

We estimated the tentative effectiveness of the intervention in both arms, although the study was not powered for hypothesis testing. Participation rates are presented without adjustment for clustering.

The intervention arm had a substantially higher uptake of eye exam during the trial (Fig. 8-1). Of the 104 participants, 31(29.8%) attended screening during the trial: 25/51 (49%) in the intervention arm, as compared to 6/53 (11.3%) in the control arm.

In the intervention arm, the rate at which participants attended an eye exam was high immediately after the start of the intervention and then it decreased (Figure 8-1). The highest rates were observed between day 10 and 20 following the first group education session, even though the education sessions continued on a monthly basis. The pattern suggests that most of the benefit of the intervention occurs early in the intervention. In the control arm, the rate of eye exam was nearly constant. Although it increased around day 50 (without any intervention) it didn't reach the rate in the intervention group.



Figure 8-2: Time to screening among PLWD in intervention and control arms

Discussion

We are the first to report a pilot cluster RCT on the feasibility of a full-scale RCT to assess the effectiveness of a community-based DSG intervention to increase uptake of DR screening among PLWD. Other studies have examined the effectiveness of such community-based groups on health outcomes such as maternal, neonatal and childhood survival.²⁶⁻²⁹

The study has several strengths. Firstly, a mixed methods health system assessment³⁰ preceded the trial. This helped to identify DSGs as a community resource that was a potential channel for increasing uptake of screening. Secondly, this study targets PLWD in a rural setting who have not had screening in the last 12 months. However, we found that our participants had never had screening, meaning the people who are the most vulnerable to DR-related blindness and who need the intervention most were included. It is known that screening programs are more cost-effective in people who derive more benefit from screening.^{2, 7}

Thirdly, it is known that demand-side behaviour changes alone may be insufficient to change the health outcomes being addressed, therefore health system strengthening is important before or within a trial.^{26, 27, 31}. Before this study, we developed and implemented national clinical guidelines for DR, and our training program for peer-supporters is currently being embedded in the Kenya Defeat Diabetes Association peer-support manual. Therefore, a further strength of the study is that we tested the intervention within a bigger health system strengthening context.^{32, 33} Fourth, the study setting, eligibility criteria, study design and amount of data collected in the pilot is by design, similar to what will be collected in the main study. This makes it easy to transfer the learning from the pilot to the main study.

Our findings show that the potential for recruitment and the feasibility of data collection, study implementation and follow-up is high. There was high acceptability of the study in general and the intervention by participants. This might be because the DSG members are already health conscious, or because of the community entry process that we followed. The top leadership of KDDA and the county director of health introduced us to the county support group leader, who in turn introduced us to the DSGs. We had strong liaison with these stakeholders and with local health care workers, which helped successful study implementation. We considered this to be important because the feasibility of the implementation and future scalability of the intervention depends on acceptability not only among the participants but also among these stakeholders in health care.³⁴ Further, the intervention itself requires constant engagement with the DSGs and the participants, which may have aided the acceptability and retention of participants in the study. Of note, we did not pay the participants—they participated voluntarily.

The recruitment process was embedded in DSG meetings, which was a critical factor for efficiency in recruitment. We learnt that recruitment required more time and more research nurses than initially planned, and this will be taken into account in the main study. The pilot findings suggest that the trial should achieve high recruitment and retention. We excluded PLWD who were temporary visitors to the DSGs, as they were not likely to stay long enough to receive the intervention. Other studies have used a similar approach to avoid contamination between clusters.³³ We monitored support groups attendance and did not have evidence of inter-cluster migration in our study. We also learnt the necessity of liaison with mobile eye care providers from other counties who visit Kirinyaga county on eye camps, as they provided screening to this population (besides the static eye clinic at Kerugoya county referral hospital).

When interventions are implemented in real-world settings, some degree of flexible adaptation of program components occurs.³⁴⁻³⁶ Although mobile phone interventions are useful due to their ubiquity even in this population³⁷, we found that face-to-face contact is valued and that peer-supporters still supplemented individual telephone reminders (prescribed in the protocol) with additional face-to-face reminders to persons who had not yet taken a retinal examination. This is perhaps because of the close residential proximity of the members and the existing personal relationships between them. It also reflects that in the 'real world' setting peer-support is not provided in tightly sequential or discrete categories. These flexible interactions may have contributed to the success of the intervention. The peer-supporters in this study were highly experienced with peer-support, having been peer-supporters for a long time and having had other trainings. This may have contributed to the success of the study, and it is not known whether if we have less experienced peer-supporters in the main study we will have different findings.

In the pilot study, we observed a much greater proportion of individuals attending retinal screenings in the intervention arm than in the control arm. Given only one cluster was randomised to each arm we cannot draw inferences from this, but it does suggest that the intervention has potential and is worth bringing forward to the full trial. Among those in the intervention arm who had an eye exam, there was a striking uptake of the exam in the first two weeks of the intervention. This means the benefit of the intervention was visible within a short period. Conversely, it also means that there is risk of eye care provider fatigue if the same pattern

272

of uptake is seen in the larger full trial. Kwaku et al³¹ have noted that such negative effects can be experienced in a clinical trial. We did not experience this in the pilot. In the full trial, we will stagger the intervention over time hence this challenge is unlikely to occur. As most of the PLWD only need one screening examination annually, we do not expect this to be a significant problem beyond the study.

Although the participants were aware of the risk of temporary blurring of vision during dilated eye exam, this was not a barrier to uptake of eye exam. However, since this is the first screening examination for the participants, we do not know whether it would be a barrier to future screening.

It is good practice to consider the attributes that contribute to the scalability of interventions, even at the stage of pilot trials.³⁸This pilot trial represents a step towards developing a scalable intervention because of its acceptability to participants. Acceptance by PLWD is necessary but not sufficient for scalability, since it also needs the support of service providers, administrators and policy-makers. Based on the evidence from the pilot trial, scalability might also be feasible because of (1) the acceptability and involvement of state and non-state stakeholders who run the support groups and health services (2) the trial is implemented within routine (pragmatic) conditions (3) we documented the processes involved in the trial.

The pilot study had some limitations. We only involved two DSGS in the study (out of the 16 DSGs in the county) but we considered this to be sufficient to address the issues of uncertainty in the feasibility of the study. For the main trial, we have recruited the other fourteen support groups (seven in each arm).

The first two DSGs that accepted to participate in the study were recruited in the pilot study. This convenience sample might mean that these initial findings are not generalizable to all DSGs. However, we considered that this method was helpful to demonstrate transparency to the DSG leaders and to meet the feasibility objectives. As women and older people were over-represented in the intervention arm, the extent to which the interim results of the primary outcome can be extended to men is not clear. This over-representation is a reflection of the composition of the two DSGs - it is a reflection of the participants that would receive this intervention if it became the usual standard of care. This composition was previously undocumented, therefore it is an important contribution of this pragmatic trial. With the larger sample size of the main trial (700 participants), we expect more balance, but if this is not the case we will conduct subgroup analysis.

Given that this is pragmatic trial²⁰, we anticipate that co-interventions may occur in the 'realworld' setting of the study. We found that there were external outreach eye camps in the county during the trial period. These camps have potential to introduce co-intervention bias, particularly if they include eye health promotion activities. However, the exposure to this co-intervention was likely balanced between the two arms, since the camps were widely publicised through community meetings and held at diverse locations across the county. In the main trial, we will identify any pre-specified or unplanned co-interventions, assess the risk of co-intervention bias and estimate its effect on the trial outcomes.

We did not perform a process evaluation at this stage, however this is planned as part of the main study,²⁰ when we will have data on the primary and secondary outcomes. The process evaluation will help us understand the way the intervention worked to lead to these outcomes.

Conclusion

The pilot study met our success criteria for the feasibility objectives. We conclude that a full trial of the intervention is feasible, and the results of this pilot will inform this full trial. The findings of this study may be relevant to other countries with a similar model of DSGs. Given the paucity of literature on implementing community-based interventions, the results of this pilot may be of interest to other researchers interested in addressing feasibility challenges in cRCT interventions targeting community groups.

Abbreviations

CONSORT	Consolidated Standards of Reporting Trials
cRCT	Cluster randomized controlled trial
DSG	Diabetes support groups
DR	Diabetic retinopathy
DURE	Uptake of retinal examination in diabetes trial
KDDA	Kenya Defeat Diabetes Association
NCDs	Non-Communicable Diseases
PLWD	People living with diabetes
RCT	Randomized controlled trial

Declarations

Ethics approval and consent to participate

The ethics review committees of the London School of Hygiene & Tropical Medicine (LSHTM) and the African Medical Research Foundation (AMREF) approved the DURE study, and this pilot falls within the scope of that trial. Informed written consent was obtained at the cluster level, while informed verbal consent was obtained at the individual level. The study adhered to the tenets of the Declaration of Helsinki and the principles of Good Clinical Practice.

Consent for publication

Not applicable

Availability of data and materials

The data that support the findings of this study are available from Ministry of Health, Kenya but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the Ministry of Health.

Competing interests

The authors declare that they have no competing interests.

Funding

The study is funded by the Queen Elizabeth Diamond Jubilee Trust, through the Commonwealth Eye Health Consortium in the form of a grant to the first author. The funding body did not participate in the design of the study, data collection and analysis, interpretation of data and writing the manuscript.

Acknowledgements

We thank the county government of Kirinyaga for supporting the study. We thank Michael

Gichangi (Ministry of health), Jefitha Karimurio (University of Nairobi), Alice Mwangi (Operation

Eyesight Universal) and Reuben Magoko (Kenya Defeat Diabetes Association) for assistance with

the community entry process, and for valuable discussion about the study intervention. We thank

the peer-supporters, participants, and the research teams for their participation.

REFERENCES

1. Shaw JE, Sicree RA, Zimmet PZ. Global estimates of the prevalence of diabetes for 2010 and 2030. Diabetes Research and Clinical Practice. 2010;87(1):4-14.

2. Pasqual FJ, Hendrick AM, Ryan M, Cason E, Ali MK, Venkat Narayan KM. Costeffectiveness of different diabetic retinopathy screening modalities. J Diabetes Sci Technol. 2016;10(2):301-7.

3. Jones S, Edwards RT. Diabetic retinopathy screening: a systematic review of the economic evidence. Diabetic Medicine. 2010;27(3):249-56.

 Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556-64.
Cockburn N, Steven D, Lecuona K, Joubert F, Rogers G, Cook C, et al. Prevalence, Causes

and Socio-Economic Determinants of Vision Loss in Cape Town, South Africa. PLoS One. 2012;7(2):e30718.

6. Casson RJ. Worldwide reduction in blindness: making progress? Lancet Global Health. 2013;1(6).

7. Ramke J, Zwi AB, Palagyi A, Blignault I, Gilbert CE. Equity and blindness: closing evidence gaps to support universal eye health. Ophthalmic Epidemiol. 2015;22(5).

8. Freeman EE, Roy-Gagnon MH, Samson E, Haddad S, Aubin MJ, Vela C, et al. The Global Burden of Visual Difficulty in Low, Middle, and High Income Countries. PLoS One. 2013;8(5):e63315.

9. Cavan D, Makaroff L, da Rocha Fernandes J, Sylvanowicz M, Ackland P, Conlon J, et al. The Diabetic Retinopathy Barometer Study: Global perspectives on access to and experiences of diabetic retinopathy screening and treatment. Diabetes Res Clin Pract 2017;129:16-24.

10. Courtright P, Mathenge W, Kello AB, Cook C, Kalua K, Lewallen S. Setting targets for human resources for eye health in sub-Saharan Africa: what evidence should be used? . Human Resources for Health 2016;14(11).

11. Mathenge Wanjiku, Bastawrous Andrew, Peto Tunde, Leung Irene, Yorston David, Foster Allen, et al. Prevalence and Correlates of Diabetic Retinopathy in a Population-based Survey of Older People in Nakuru, Kenya. Ophthalmic Epidemiology. 2014;21(3):169-77.

12. World Health Organization. Global action plan for the prevention and control of non communicable diseases 2013-2020. Geneva: WHO; 2013.

13. Whittle J. When does peer-support improve glycemic control in persons with diabetes mellitus? JAMA Intern Med. 2014;174(6):982-3.

14. Baumann LC, Frederick N, Betty N, Jospehine E, Agatha N. A demonstration of peersupport for Ugandan adults with type 2 diabetes. Int J Behav Med. 2016;22(3):374-83.

15. Gao J, Wang J, Zheng P, Haardörfer R, Kegler MC, Zhu Y, et al. Effects of self-care, self-efficacy, social support on glycemic control in adults with type 2 diabetes. BMC Fam Pract. 2013;14(66).

16. Ussher J, Kirsten L, Butow P, Sandoval M. What do cancer support groups provide which other supportive relationships do not? The experience of peer-support groups for people with cancer. Soc Sci Med. 2006;62(10):2565-76.

17. Peterson JL, Rintamaki LS, Brashers DE, Goldsmith DJ, Neidig JL. The forms and functions of peer social support for people living with HIV. J Assoc Nurses AIDS Care. 2012;23(4):294-305.

18. International Council of Ophthalmology. ICO Guidelines for Diabetic Eye Care-updated 2017. San Francisco, California: International Council of Ophthalmology; 2017.

19. Mwangi N, Macleod D, Gichuhi S, Muthami L, Moorman C, Bascaran C, et al. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Tropical Medicine and Health. 2017;45(41).

20. Mwangi N, Ng'ang'a M, Gakuo E, Gichuhi S, Macleod D, Moorman C, et al. Effectiveness of peer-support to increase uptake of retinal examination for diabeticretinopathy: study protocol for the DUREpragmatic cluster randomized clinical trialin Kirinyaga, Kenya BMC Public Health. 2018;18(871).

21. Bandura A. Self-Efficacy. S RV, editor. New York: Academic Press; 1994.

22. Ministry of Health. Guidelines for the screening and management of diabetic retinopathy in Kenya. Nairobi: Ministry of Health; 2017.

23. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases; 2015.

24. Hayes RJ, Bennett S. Sample size calculation for cluster randomized trials. International Journal of Epidemiology. 1999;28:319-26.

25. Eldridge SM, Chan CL, Campbell MJ, Bond CM, Hopewell S, Thabane L, et al. CONSORT 2010 statement: extension to randomised pilot and feasibility trials Pilot and Feasibility Studies. 2016;2(64).

26. Houweling TAJ, K A, Younes L, Kuddus A, Shaha, Haq B, et al. The effect of participatory women's groups on birth outcomes in Bangladesh: does coverage matter? Study protocol for a randomized controlled trial. Trials. 2011;12(208).

27. Fottrell E, Jennings H, Kuddus A, Ahmed N, MorrisonJ, Akter K, et al. The effect of community groups and mobile phone messages on the prevention and control of diabetes in rural

Bangladesh: study protocol for a three-arm cluster randomised controlled trial. . Trials. 2016;17(600).

28. Prost A, Colbourn T, Seward N, Azad K, Coomarasamy A, Copas A, et al. Women's groups practising participatory learning and action to improve maternal and newborn health in low-resource settings: a systematic review and meta-analysis. Lancet. 2013;381(9879):1736-46.

29. Sheela S Sinharoy, Wolf-Peter Schmidt, Ronald Wendt, Leodomir Mfura, Erin Crossett KAG, William Jack, et al. Effect of community health clubs on child diarrhoea in western Rwanda: cluster-randomised controlled trial Lancet Global health. 2017;5(7):PE699-E709.

30. Mwangi N, Bascaran C, Gichuhi S, Moorman C, Muthami L, Macleod D, et al. Health System Assessment for Diabetes and Diabetic Retinopathy in Kenya. 2016.

31. Kwaku Poku Asante, Caroline Jones, Sodiomon Bienvenu Sirima, Sassy Molyneux. Clinical Trials Cannot Substitute for Health System Strengthening Initiatives or Specifically Designed Health Policy and Systems Research. The American Journal of Bioethics. 2016;16(6):24-6.

32. Mwangi N, Gachago M, Gichangi M, Gichuhi S, Githeko K, Jalango A, et al. Adapting clinical practice guidelines for diabetic retinopathy in Kenya: process and outputs. Implementation Science. 2018;13(81).

33. Harris-Fry HA, Azad K, Younes L, Kuddus A, Shaha S, Nahar T, et al. Formative evaluation of a participatory women's group intervention to improve reproductive and women's health outcomes in rural Bangladesh: a controlled before and after study J Epidemiol Community Health 2016;70:663-70.

34. Andrew J Milat, Adrian Bauman, Sally Redman. Narrative review of models and success factors for scaling up public health interventions Implementation Science. 2015;10(113).

35. Dariotis JK, Bumbarger BK, Duncan LG, Greenberg MT. How do implementation efforts relate to program adherence? Examining the role of organizational, implementer, and program factors. Commun Psychol. 2008;36:744-60.

36. Monika Kastner, Radha Sayal, Doug Oliver, Sharon E, Straus, Lisa Dolovich. Sustainability and scalability of a volunteer based primary care intervention (Health TAPESTRY): a mixed-methods analysis. BMC Health Services Research 2017;17(514).

37. Bastawrous A, Armstrong MJ. Mobile health use in low- and high-income countries: an overview of the peer-reviewed literature. J R Soc Med 2013;106:130-42.

38. World Health Organization. Beginnining with the end in mind: planning pilot projects and other programmatic research for successful scaling up. Geneva: WHO; 2011.





Figure 8-1: Flow diagram for pilot study

Chapter Nine

Effectiveness of peer support to increase attendance at DR screening - a pragmatic cluster randomized clinical trial <u>Process Evaluation of the DURE trial</u>

9.0 Overview

The effectiveness of complex interventions is estimated through outcome evaluation of randomized clinical trials (RCTs). Chapter 10 will provide information on the effectiveness of the DURE trial in increasing attendance to DRS. There is also a need to understand the factors shaping implementation and outcomes of the trial in order to provide possible explanations for the effect of the intervention. Process evaluation of clinical trials provides evidence about what and how interventions are implemented, how interventions generate change in outcomes (mechanism of impact) and how the context affects the outcomes.¹

This chapter consists of a research paper that describes a mixed method process evaluation of the DURE full trial, guided by the Consolidated Framework for Implementation Research (CFIR).² This framework is an implementation science tool that is used to investigate the influence of the characteristics of the intervention, the individuals involved, the inner and outer setting of the intervention, and the implementation processes.

- Moore GF, Audrey S, Barker M, et al. Process evaluation of complex interventions: Medical Research Council guidance. BMJ 2015;350:h1258.doi:10.1136/bmj.h1258
- Damschroder LJ, Aron DC, Keith RE, et al. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implementation Science 2009;4:50.doi:10.1186/1748-5908-4-50

Research paper 7 (submitted to BMC Tropical Medicine and Health) includes our analysis of the quantity and quality of what was delivered, the role of context, why the intervention did or did not work, what elements appear to be the 'most essential components' of the intervention, barriers and facilitators, and implications for scale-up of the intervention. These findings can inform stakeholders on how to change or improve implementation of this intervention in the current settings or replication of the intervention in different settings.

9.1 Research Paper 7

BMC Tropical Medicine and Health – Licences and Copyright

The following terms will apply to this publication once it is published in this journal:

"The sustainable open access model that BMC pioneered has become the industry standard. We make research free to access for everyone, everywhere. Our **authors retain copyright** of their work through a Creative Commons attribution license that clearly states how **readers can copy, distribute, and use their attributed research, free of charge**. All of which helps make articles available to the widest audience, and contributes to the furthering of research in ways that would have seemed impossible two decades ago.

All articles published by *BMC Tropical Medicine and Health* are made freely and permanently accessible online immediately upon publication, without subscription charges or registration barriers. "

https://www.biomedcentral.com/about/open-access https://tropmedhealth.biomedcentral.com/about London School of Hygiene & Tropical Medicine Keppel Street, London WC1E 7HT www.lshtm.ac.uk



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi
Principal Supervisor	Professor Allen Foster
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?			0
When was the work published?			
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			-
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	BMC Tropical Medicine and Health
Please list the paper's authors in the intended authorship order:	Nyawira Mwangi, Covadonga Bascaran, Jacqueline Ramke, Mathew Kipturgo, Mark Ng'ang'a, Stephen Gichuhi, Min Kim, Dorothy Mutie, Consuela Moorman, Lawrence Muthami, Allen Foster
Stage of publication	Submitted

SECTION D – Multi-authored work

	I designed the study, conducted the research,			
For multi-authored work, give full details of your role in the research included in the paper and in the preparation	analysed and interpreted the data, drafted the			
	manuscript and prepared subsequent revisions			
of the paper. (Attach a further sheet if necessary)	in consideration of the comments from the			
	co-athors and peer reviewers			

www.lshtm.ac.uk

Supervisor Signature:

Date: 17 SEPT 2019

Date: SEPT 17, 2019

(2020) 48:1

RESEARCH

Tropical Medicine and Health

Open Access



Peer-support to increase uptake of screening for diabetic retinopathy: process evaluation of the DURE cluster randomized trial

Nyawira Mwangi^{1,2*}, Covadonga Bascaran¹, Jacqueline Ramke^{1,3}, Mathew Kipturgo², Min Kim¹, Mark Ng'ang'a⁴, Stephen Gichuhi⁵, Dorothy Mutie², Consuela Moorman⁶, Lawrence Muthami⁷ and Allen Foster¹

Abstract

Background: There is limited evidence on how implementation of peer support interventions influences effectiveness, particularly for individuals with diabetes. We conducted a cluster randomized controlled trial to compare the effectiveness of a peer-led health education package versus usual care to increase uptake of screening for diabetic retinopathy (DR).

Methods: Our process evaluation used a mixed-method design to investigate the recruitment and retention, reach, dose, fidelity, acceptability, and context of implementation, and was guided by the Consolidated Framework for Implementation Research (CFIR). We reviewed trial documents, conducted semi-structured interviews with key informants (n = 10) and conducted four focus group discussions with participants in both arms of the trial. Three analysts undertook CFIR theory-driven content analysis of the qualitative data. Quantitative data was analyzed to provide descriptive statistics relevant to the objectives of the process evaluation.

Results: The trial had positive implementation outcomes, 100% retention of clusters and 96% retention for participants, 83% adherence to delivery of content of group talks (fidelity), and 78% attendance (reach) to at least 50% (3/6) of the group talks (dose). The data revealed that intervention characteristics, outer setting, inner setting, individual characteristics, and process (all the constructs of CFIR) influenced the implementation. There were more facilitators than barriers to the implementation. Facilitators included the relative advantage of the intervention compared with current practice (intervention characteristics); awareness of the growing prioritization of diabetes in the national health policy framework (outer setting); tension for change due to the realization of the vulnerability to vision loss from DR (inner setting); a strong collective sense of accountability of peer supporters to implement the intervention (individual characteristics); and regular feedback on the progress with implementation (process). Potential barriers included the need to queue at the eye clinic (intervention characteristic), travel inconveniences (inner setting), and socio-political disruption (outer setting).

Conclusions: The intervention was implemented with high retention, reach, fidelity, and dose. The CFIR provided a valuable framework for evaluating contextual factors that influenced implementation and helped to understand what adaptations may be needed during scale up.

Trial registration: Pan African Clinical Trials Registry: PACTR201707002430195 registered 15 July 2017

Keywords: Diabetes, Diabetic retinopathy, Peer support, Cluster-randomized clinical trial, Consolidated Framework of Implementation Research (CFIR), Process evaluation

* Correspondence: Nyawira.Mwangi@lshtm.ac.uk

¹London School of Hygiene and Tropical Medicine, London, England

²Kenya Medical Training College, Nairobi, Kenya

Full list of author information is available at the end of the article



© The Author(s). 2020 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

Background

Early detection of diabetic retinopathy (DR) poses a significant medical and public health challenge, particularly because DR is asymptomatic until the advanced stages. The benefits of regular screening have been documented [1], but uptake remains low for people living with diabetes (PLWD) in settings without systematic DR screening programs [2]. Diabetes Support Groups (DSGs) provide an opportunity for demand-side interventions to increase attendance at screening and confront this inequity [3]. However, there is need for evidence on the factors that influence implementation and outcomes of DR interventions involving DSGs.

The DURE (Uptake of Retinal Examination in Diabetes mellitus) trial was a 6-month pragmatic cluster randomized controlled trial (cRCT) to evaluate the effectiveness of a complex intervention to promote screening for diabetic retinopathy among members of DSGs in Kirinyaga County, Kenya. The DURE trial interventions have been described in detail [3]. Briefly, the trial compared the proportion of PLWD who attended screening in seven DSGs that received the intervention with seven "usual care" DSGs that did not receive the intervention. The intervention consisted of (i) training of peer supporters; (ii) monthly group talks at the DSGs by peer supporters and referral of PLWD to the eye clinic; (iii) monthly individual reminders to PLWD (by peer supporters) to attend screening the eye clinic; and (iv) weekly telephone support to peer supporters from the research team.

The intervention was developed in accordance with the guidelines of the Medical Research Council (MRC) framework for complex interventions [4]. These guidelines recommend using appropriate theory to develop interventions that address the barriers to behavior change. Our formative research identified several barriers to uptake of DR screening [2]. Self efficacy is a strong precursor to behavior change [5, 6] including attendance to screening [7]. Based on the self-efficacy theory, we hypothesized that an intervention that increases selfefficacy can decrease the perceived barriers to attendance to screening. The theory proposes four methods of changing self-efficacy in order to change behavior: providing mastery experiences (e.g., recalling previous screening for other diabetes complications); vicarious learning (e.g., from hearing experiences of peers who have had DR screening); verbal persuasion (of the need for screening); and addressing psychological and affective states (such as anxiety about taking a screening exam) [6]. The theory-driven conceptual framework of intervention effect is illustrated in the protocol [3].

The MRC framework [4] emphasizes four key phases of interventions: intervention development; feasibility and piloting; implementation; and the evaluation of both outcomes and process. In this paper, we describe the results of the process evaluation. Process evaluation is a study which aims to understand the functioning of an intervention, by examining implementation, mechanisms of impact, and contextual factors [8]. The conduct of process evaluations alongside RCTs has been recommended, because they give insight into the "black box" of health care interventions; facilitate the interpretation of the findings; explain why, for whom and how a complex intervention had a particular impact; and determine whether a complex intervention should be scaled up or modified for other contexts [9, 10]. Process evaluations are particularly important in cRCTs, because of the potential for between-cluster differences that need to be understood [10, 11].

Theory-driven process evaluation necessitates that the designers make the theory explicit and then use it to identify how the intervention leads to the outcomes [12]. The Consolidated Framework for Implementation Research (CFIR) is a is a meta-theoretical framework that synthesizes constructs from multiple theories on implementation of interventions, in order to explain what works and why across multiple contexts [13]. The CFIR outlines five major factors that influence implementation of interventions: the characteristics of the intervention, the inner setting, the outer setting, the individuals involved, and the process of implementation (Table 1). By applying this framework to the process evaluation for this cRCT (Fig. 1), we aimed to (1) understand the determinants for the outcomes of the DURE intervention in Kirinyaga and (2) examine the context of implementation in terms of the intervention's recruitment and retention, reach, fidelity, dose, and acceptability (Table 2).

Methods

Ethics

The DURE trial and its process evaluation has ethics approval from the research ethics committees of the London School of Hygiene and Tropical Medicine in London and the African Medical Research Foundation in Nairobi. Trial registration: Pan African Clinical Trials Registry: PACTR201707002430195.

Setting

The setting of the study is described elsewhere [3]. Briefly, Kirinyaga county is a rural agrarian county in Central Kenya. The prevalence of diabetes in Kenya is estimated to be 2% in the population 18–64 years [14]. An estimated 40% of the PLWD in Kirinyaga are members of DSGs, and a health system assessment by our research group found that only 7% of them have had an annual DR screening exam as recommended. DSGs have monthly meetings in the community led by peer supporters, where they measure the weight, blood sugar, and blood pressure of attenders. They also engage in other activities relevant to health promotion and advocacy. At

Tab	e 1	Constructs of	f the	Consolidated	Framewor	k f	or Imp	lementation R	esearch
-----	-----	---------------	-------	--------------	----------	-----	--------	---------------	---------

Construct	How the construct relates to the DURE trial			
Characteristics of the intervention—core components and adaptable components	Intervention characteristics (adaptability complexity, relative advantage) can influence whether the intervention is adopted			
Inner setting—structural, political, and cultural context that directly affects the implementation	The context of the DSGs and the eye health system in Kirinyaga can influence how participants experience the intervention			
Outer setting—broader economic, political and social context	Broader political, economic, health policies, priorities, resources, incentives, and governance may impact trial activities			
Characteristics of individuals—people responsible for delivering the intervention (peer supporters)	Training, knowledge, perceptions, motivation, and leadership of peer supporters can influence extent of implementation of the intervention			
Implementation process—the activities involved in planning, engaging, execution, and evaluation of implementation process	The involvement of stakeholders in the planning, execution, and evaluation of progress of the trial may influence the acceptability of delivery and reception of the intervention			

DSG Diabetes Support Group, DURE Uptake of Retinal Examination in Diabetes study

the time of the DURE study, there were 16 DSGs active in Kirinyaga. Two of them participated in the pilot trial, while the other 14 were recruited into the main trial.

Two peer supporters were recruited in each DSG (1 male, 1 female), as per the eligibility criteria in Table 3, and none of them had previously delivered an eye health intervention. They received two days of training using a curriculum developed through the process described in Additional file 1. The content of the training and the key messages that the peer supporters delivered to participants are described in the protocol [3]. An allowance was provided to peer supporters for telephone communication with participants, but no other financial incentives were given. Weekly telephone calls between the principal investigator and peer supporters were carried out to share progress, build a sense of belonging, and address any challenges emerging during the program.

The county has a well-equipped eye clinic at the Kerugoya county referral hospital. Patients at the clinic are attended on a walk-in basis. There were four eye health workers (one ophthalmologist, three ophthalmic clinical officers) in the county during the study period. Guidelines for screening and management of diabetic retinopathy were launched at the national level 3 months before the start of the main trial, and were also being implemented in Kirinyaga county [15].

Design

This is a mixed-methods process evaluation of a cRCT. We used the CFIR to guide the process because it is comprehensive and can be used to develop the evaluation tools, guide the content analysis, and aid interpretation of findings [16].


Table 2 Measures for the level of implementation

Implementation measure	Questions related to DURE study	Quantitative indicator(s)— compared with target	
Recruitment and retention	How successful were the recruitment and retention procedures?	-Proportion of DSGs that agreed to participate (%) -Proportion of participants invited who agreed to participate (%) -Peer educators recruited, trained and retained (<i>n</i>) -Retention rate of clusters and participants in the study (%)	
Reach What proportion of the intended audience was exposed to the intervention? Dosage delivered What percentage of interventions was delivered most/least successfully by implementers?		-People who were referred (n) -People who received individual reminders (n) -People who attended group meetings (n)	
		-Group talks delivered (<i>n</i>) -Referrals made (<i>n</i>) -Individual reminders given(<i>n</i>)	
Dose received What percentage of the intervention was received most/least successfully by the target audience		-Proportion of group sessions attended by each participant (%) -Referrals given (<i>n</i>)	
Fidelity How much of the intervention was delivered as intended (adherence)? What parts were not delivered?		-Adherence to content of group talks -Adherence to frequency of group talks	
Acceptability How acceptable is the intervention for current and future implementation?		Acceptance of the intervention by peer supporters and participants Willingness of stakeholders to scale up the intervention in future	

DSG Diabetes Support Group, DURE Uptake of Retinal Examination in Diabetes study

Table 3 Eligibility criteria for participants and peer-supporters

Criterion	Participants	Peer-supporters
Age > 18 years	\checkmark	
Member of a diabetes support group	\checkmark	
Will reside in the county for the next 12 months	\checkmark	
Has a mobile phone	\checkmark	
Willing to participate in the study	\checkmark	
Had not had a screening exam in the last 12 months	\checkmark	×
Has had a screening exam in the preceding 12 months	×	
Willing to be a peer-supporter	N/A	
Willing to commit 2 days for training	N/A	
Willing to commit many hours to peer-support work	N/A	
Fluent in Kikuyu or Kiswahili	N/A	
Already attending DR screening	×	
Already receiving treatment for DR	×	×
Has a debilitating illness	×	×

" $\sqrt{}$ indicates participants or peer-supporters included in the study

"×"indicates participants or peer-supporters excluded in the study $N\!/\!A$ Not applicable

Data collection

Data on recruitment and retention, fidelity, reach, and dose were collected routinely during the trial activities and collated through document review. Peer supporter training records provided information on attendance and content of training. Trial registers provided data on recruitment, while DSG meeting attendance registers captured participants attendance and retention, as they provided the date of the meeting and the list of attendees. DSG meeting minutes, peer supporter diaries, and the research team's activity logs, and field notes contained detailed information on the intervention activities, personnel involved, duration, frequency, and resources used.

We studied the sample interview questions available on http://cfirguide.org/ and tailored our data collection tools to gather information relevant to the DURE study. These questions related to the stakeholders' perception of the intervention and how it worked/did not work.

Ten interviews were conducted with purposively selected key informants to represent recipients, implementers, administrators, and policy-makers. Key informants were recruited until data saturation was reached. Face to face interviews were conducted by the first author in English at locations convenient to the key informant, using a semi-structured interview schedule. Interviews lasted 30–60 min and were captured through field notes.

We conducted four focus group discussions in community settings with 7 participants in the intervention arm who did not take up screening; 7 participants in the intervention arm who took up screening; 8 participants from the control arm; and 7 peer supporters. Focus group discussions were conducted in the Kikuyu language by the first author and two research assistants who were considered culturally appropriate but not involved in the trial implementation. Discussions were audio-recorded, transcribed, and translated into English.

Data analysis

Quantitative data on recruitment and retention, reach, fidelity, and dose were analysed for descriptive summary statistics.

A thematic content analysis of qualitative data was undertaken based on the different constructs and subconstructs of the CFIR: (1) Intervention characteristics (e.g., acceptability of the intervention, compatibility with existing DSG programs, relative advantages, or disadvantages of the intervention, and suggested adaptations); (2) Outer setting (e.g., perceived role of the Ministry of Health (MoH) policies and guidelines in driving which services were implemented); (3) Inner setting (e.g., perceptions about organizational factors within DSGs and the eye clinic, that might have affected the implementation); (4) Individual characteristics of peer supporters who delivered the intervention (e.g., knowledge, attitudes, self-efficacy about their role); and (5) Implementation processes (e.g., planning, engagement, and execution factors that may have affected the delivery and reception of the DURE intervention). Using the CFIR as a template framework, the principal investigator and two other analysts (1 male, 1 female) read the transcripts, coded the data independently, and grouped the codes into themes. Analysis began as soon as the first interview was completed, and then proceeded concurrently with data collection until data saturation was reached. The analysts reviewed the codes iteratively to check for potential biases, and to verify the emerging themes. Discrepancies between coders were resolved through discussion and review of the original transcripts.

Data (qualitative and quantitative) from all sources were organized under the respective constructs and subconstructs to facilitate triangulation and to identify which factors affected the acceptability, recruitment, retention, reach, fidelity, and dose of implementation.

The first author had undertaken training on implementation research and clinical trials, and had expertise on the technical content of the intervention. All the analysts had skills in quantitative and qualitative research methods.

Results

How was the intervention implemented? *Recruitment and retention*

All the 16 DSGs in the county accepted to participate in the trial (2 in the pilot trial and 14 in the main trial). All clusters were retained throughout the study. Of 837 members of DSGs approached to participate the main trial, 86 did not meet eligibility criteria (Fig. 2), and 17 did not consent, thus 734 participants were recruited and participated in the trial. Of these, 31 (4.2%) were lost to follow-up during the trial (95.8% follow-up rate rate). The 14 peer supporters (age range was 29-58 years) received the training program. All had at least secondary education, with three having achieved a tertiary level qualification, although this was not a requirement. One peer supporter was absent from day 72 as another DSG project that was being implemented a different county contracted him for their team. The remaining 13 actively participated in the DURE trial until the end of the trial. The research team maintained weekly contact with peer supporters and attended some of the DSG meetings, which minimized the likelihood of loss to follow-up or missing data. Recruitment procedures had already been tested in the pilot trial [17].

Reach

Out of 369 PLWD recruited into the intervention arm, 92% attended the first group talk, while 72% attended all six talks. Seventy-eight percent attended



three or more talks (50%), and this ranged from 68% in the most rural DSG to 88% in urban DSGs. One hundred percent of the participants were referred for the screening examination and were issued with a referral card. Peer supporters also gave monthly telephone reminders to those who had not yet taken the screening exam. Seventy-four percent of participants

received at least one telephone reminder to attend screening.

Dose delivered and dose received

The 14 peer supporters attended the training. During the 6 months of the study, peer supporters in the seven DSGs in the intervention arm delivered a group talk each month (total 42 talks, 100% of planned talks were delivered). We found that 34/42 (81%) meetings had an attendance of \geq 80% of the trial participants.

Of the targeted 369 telephone reminders in the first month, 273 (74%) telephone reminders were made as some of the participants took the screening exam immediately after the first group talk and before the telephone reminders. Forty-eight percent of participants received a reminder each month (six reminders in total) over the 6 months trial period. There were frequent reports of participants being unreachable on phone, as the phones were switched off, or they had changed the number. In such cases, the peer supporter arranged a personal visit to the participant to give a face-to-face reminder (this was a local adaptation of the intervention). During the first two months of implementation, peer supporters from two different DSGs made weekly calls to the principal investigator to seek feedback (in addition to the weekly calls that all peer supporters received from the principal investigator). This was a further form of intervention adaptation; these two DSGs had the highest rates of implementation fidelity.

Fidelity

There was 100% adherence to the frequency of the group talks (one group talk every month). However there were occasional changes on the actual date of the group talk each month (due to other communal activities), such that the intervals between the group talks were not constant. Adherence to content of group talks met the required threshold in 83% of the group talks. Some 43 participants could not be reached on telephone at least once during the trial, and the peer supporters gave them a reminder through a face-to-face communication as mentioned above. We monitored adaptations, which we interpreted as evidence of ownership and adaptation to meet contextual needs.

Acceptability

The acceptance of the intervention in the study population was high, given the high retention in the study. The intervention was perceived as beneficial and acceptable to participants because it was bridging information gaps.

I am happy you people are coming here to us, because we are benefiting. Although I have not yet gone [for screening], I now know that I should not wait to have problems with my vision and I know where I should go [for screening] (FGD participant)

There was satisfaction with the study procedures among the participants. This was not surprising as we had tested these with a pilot study [17]. We had anticipated participants to be uncomfortable with temporary blurring of vision due to dilating drops, but it was not perceived to be a significant problem because they were forewarned about it.

The difficult seeing after the medicine in the eye...it was not as serious as I expected,... it was not like you could not see at all..., and by the time I was going home I could see very well (FDG participant)

All 10 key informants were willing to scale up the intervention to other counties in future, using the DURE tools such as the PS training curriculum and the community entry mechanism.

The DURE curriculum for the peer supporter training is very comprehensive and easy to follow, and we want to formally adopt it in our peer supporter manual, so that we can use it in our routine training of peer supporters (KDDA national representative)

Eye care workers accepted to conduct the screening for DR, even though at first they were concerned about a sudden increase in the screening workload.

Initially we had concern that this might increase the workload, but we found that this was only a temporary effect as many participants came at the same time for the screening, which is unlikely to happen in the repeat screening visits. It is good to see the participants asking for the screening... they already have the information about it (eye health worker)

How was the intervention experienced?

Table 4 shows the five CFIR constructs, the sub-constructs identified within them and examples of the related quotes.

DURE intervention characteristics

Stakeholders considered the peer-supporter-led intervention to have relative advantage, as it was not feasible to have health workers go to the community to give the same intervention. All PLWD also perceived the DURE intervention as a relative advantage compared with the usual practice where they were not offered screening. In particular, participants found the referral card highly valuable as it represented personalized care and was perceived to make it easier to navigate interaction with eye care providers. Peer supporters valued the training and task shifting which gave them confidence and recognition that they did not have before.

The intervention components were easy to implement along with the usual duties of peer supporters, and were adaptable to suit local needs (such as additional face-toface reminders for participants who could not be reached by phone). Although research assistants

Table 4 Quotes on CFIR constructs

CFIR construct/sub-construct	Sample quotes
Intervention characteristics	
Relative advantage	This is more effective than leaving it to health workers from the hospital to go to the community to educate the PLWD, for that is not always feasible (MoH national representative) The training we received was very good. People appreciate my work when I give them talks (Peer supporter) The referral cardhave a look (showing the card)it has my name, and the date of my next clinic at the clinic I just showed it (to the staff) and I was attended (FGD participant who attended clinic) I still have my referral card I never leave it behindso I know that the day I go to the eye clinic I will show it and get checked (FGD participant who did not attend clinic)
Complexity	Training the peer supporters was not difficult, the training slides are easy to use (trainer) We are not doing things that are very different from what is usually done[in DSGs]we were already familiar with group talks, what we have not been doing was the telephone remindersand giving referrals, but that is not burdensome (peer supporter) When I learnt the reason for the test, and was given the referral cardall I needed was to present myself at the clinic. You could go even the following day, anytimeand you only needed to go once. We went together several of us. Can anyone say that it is difficult? (FGD participant who had screening)
Adaptability and flexibility	I wanted the PLWD to go to the eye clinic as soon as possibleso if I could not reach them on phone because they had put off the phone, I took it upon myself to go to their homes (peer supporter) We carry out the DURE activities because they fit well with our other activitiesI go on with my usual work on the farm except for the DSG meetings (peer supporters) During recruitment, we realized that we needed additional personnel in the recruitment team, so we expanded the team (member of research team)
Cost	The intervention uses existing resources in the community and in the hospitalthis is the biggest advantage because it can scaled up without cost limitations(MoH county representative) People's pockets are differentwhen I went there I had only 100 shillings I paid 50 for registration. They asked me to pay another 50 for eye examination. I paid it because I had to get the screening. But someone else will just say they will come next week. (FGD participant who attended screening)
Relative disadvantage	The only problem is that at both the diabetes clinic and the eye clinic they make you wait queuing two timesthen in the eye clinic they put some medicine in your eyes and ask you to wait again you can end of wasting a lot of time waiting. Why should I que twice? (FGD participant who attended screening) But why don't you focus on preventing the complications, rather than just screening? For me

CFIR construct/sub-construct	Sample quotes
	I have begun with doing exercises, but later I will go for the screening. [names peer supporter] will take me there (FGD participant who did not attend screening)
Outer setting	
Diabetes as a health priority	We are keen on sustainable interventions for NCDs and we have to go to where the patients are foundWe should not wait for patients to come but go to them. This is sustainable because the trained peer educators will remain in the DSG to educate more people. We will work with them more. (MoH representative at national level)
Clinical guidelines for DR are used as a national governance tool that is also useful for resource mobilization	The clinical guidelines have been very helpful. Earlier on I did not routinely screen those who have good vision. Now I dilate and screen all those that come here. We also order more dilating drops (eye health worker)
Peer supporters mitigated potential implementation challenges such as political events	The presidential elections were nullified, everyone left the (DSG) meeting to go and watch the news and had I not been passionate to mobilise members there would have been very poor attendance at the next meeting (peer supporter)
Intervention fits within the norms of the health care system	The county health services, including eye care and diabetes care services supported this innovative involvement of peer supporters because we all want to improve quality of life for PLWD (MoH representative at county level)
External outreach camps	There were two external mobile outreach camps organized at a church by a private care provider some of the people preferred to go for the screening here because it was nearer (DSG county lead) With the mobile outreach clinic, you know it is only for one day, so you don't want to miss the opportunity. For the hospital, some of the people, even if they live near, do not attend They keep on postponing because the eye clinic will always be here (Peer supporter)
Inner setting	
Tension for change	I know someone who doesn't go out of his home now, because he can't seethat is why we have been told not to wait till we have eye problems (FGD participant who took screening) I have never had my eyes checked Can you check me today? Or give us the referral cards so we can go [to the eye clinic] tomorrow (FGD participant from control arm) We have to find a way of easily identifying those any diabetes patient who has not been screened. May be label their files so that they can easily identified (eye health worker) We have seen people going blindnobody will come from outside to stop it we have to do something ourselves (KDDA county representative)
Compatibility	We give the group talks as part of the monthly DSG meetings (peer supporter) We want to have all PLWD screened for DR, thus this intervention is contributing to that mandate (eye care worker)
DSG Organisational Culture	For us we are always open to new things that can help us who live with diabetes, so we are

CFIR

FIR construct/sub-construct	Sample quotes
	happy to work with you on thisit empowers us to not just to go for screening, but to engage in advocacy for diabetes eye health (KDDA local representative) In DSGs, we know about volunteering and for the good of our people, we all have to work together to ensure everyone goes for the eye check I do not mind giving my time to do this, though of course it requires extra time I am happy people got tested (peer supporter) At the DSG I tell them my experience with screeningwe don't hide things from one another(PLWD who has taken screening) In some of the DSGs, participants came togetherwe would have a large group turn up at one gothey would tell me they all agreed to come together (eye health worker) Here we like to share about ourselves openly, we don't hide things, we are not afraid to open up or keep reminding one another about attending screening)
Incentives and rewards	We do not get paid for this work, it is about volunteer work, people who do not want to volunteer their time cannot do this work (peer supporter) But since we are doing good work, and we spend a lot of time on it, if we were paid we could do even more (peer supporter)
Readiness for implementation	We are planning a peer supporter training in [names county]we want you to come and train them so that they start doing the same in [names county] (KDDA national representative) Now that you have done this with some groups, you also have to come to our groups and give us the intervention, you should not leave us out (PLWD from control group)
Adjustments in the eye health system	Sometimes, people did not screen for diabetic retinopathy if the patient's vision was good. But now we have been reminded to screen all PLWD annually and we have started doing that (eye health worker) The eye clinic has recently been renovated, we saw the governor launch it and we heard that it has all the equipment, so we are now happy to go there. (FGD participant who attended screening)
Community volunteers (CVs) reinforced key messages	Community volunteers really support us because they reinforce what we say. In our DSG, we have a member who is a community volunteer I usually call her to speak after I given the group talkit is better when the message comes from two people. (peer supporter)
Geographical barriers hinder uptake of screening	Getting to the eye clinic is a problem because the easiest way is to take a <i>boda boda</i> (motorbike taxi) to the main road and then wait for a <i>matatu</i> (public van). I avoid <i>boda boda</i> because I have a back problem, so I just wait for the outreach camp. (FGD participant who has not attended screening) From this experience, the cost of mobility must be borne by the provider, not the PLWD. We

CFIR construct/sub-construct	Sample quotes
	must find a way of going to the DSG for screening, rather than asking them to come. (member of the steering team) For me, I haven't gone to the eye clinic. I am looking for the fare. Why can't you come to do the test here? (FGD participant who has not attended screening)
Peer supporter characteristics	
Knowledge and beliefs about the intervention	The peer supporter at [DSG] informed me that they have started a WhatsApp group for peer supportersto discuss how they can do more to prevent diabetes complications in generalthey feel the work they are doing with DURE can be expanded (PI field notes) I observed that the peer supporters enjoyed giving the group talks, and the key messages were easy to explain and it was a social activity, unlike the paper work which was more of an individual task. They still did the documentation since they were trained to do it. (research assistant)
Individual identification with the role of PS	In our support group, most people have gone to the eye clinic, because [names peer supporter] is very active, and he makes us laugh when he is giving the talkyou cannot get tiredand every time he meets you he will remind you, even at church (FGD participant who attended clinic) I always see [name] here in the diabetes clinic, bringing his DSG members. Then he also takes them to the eye clinic. Sometimes they tell him they do not have the money for the hospital fee but he insists and they pay (diabetes care worker)
Individual stage of change	All the peer supporters had already taken screening so they must have been good role models (KDDA national representative) None of the peer supporters had any previous training on delivering a diabetes eye health intervention, you could tell that they liked itthe novelty of the information seems to have been a motivator (Trainer)
Personal attributes	"I did not do as much work as [name], though he and I are the supporters in our group. But he did very wellyou know he is younger, 'sharp sharp' (slang for exuberance) and men can do this work more easily " (Female peer supporter) [Name] is ever punctual so we know (DSG) meetings will run on time. She is a teacher so she explains very well. That is why many people don't miss the meetings, and most of us got tested the very first month (FGD participant who attended screening) What I have seen, is that he [peer supporter] is self-sacrificingfrom the heart he closes his business of selling clothes to bring PLWD to both the diabetes clinic and the eye clinic (diabetes clinician) We did not know whether keeping the peer supporters engaged over six months would be challengingI would say selection of peer supporters is important as they have to be highly motivated and committed (member of steering team)

CFIR construct/sub-construct	Sample quotes
Self-efficacy of peer supporters	He took me to the eye clinic, together with others he did not feel bothered about waiting in the queue with us (FGD participant who attended screening) When I saw that the first five members had gone for screening right after I gave the first talk, I knew I was doing it right, I felt motivated me to continue with the work (peer supporter) When I observed [names peer supporters giving the talk], they performed so well, they answered all the questions. I think it is because they were trained well (research assistant)
Process of implementation	
Planning	We were very happy to be involved from the beginningwe have always insisted on being involved as equal partners in things that concern us, so we participated (KDDA national representative) I remember the meeting we had at the beginningwhen our chairman of KDDA came and introduced the projectwe agreed to support (peer supporter)
Engaging	The research team was really committed they were always available and we worked so well together, it made us not to leave the work half- way (peer supporter) we even took lots of photos The community volunteers, they really embrace uswe support one another in the work (peer supporter) I looked forward to the call from the PI every week – it gave me motivation (peer supporter) Regular briefing helped us to keep involved (steering committee member)
Executing	We were of course concerned about the feasibility of the intervention since we have not used the DSG platform before. But we had success with the pilot trial, so this proved not to be a major issue (member of project steering group) The DSGs in the control arm are left out, but we have understood that they can get the intervention thereafter (peer supporter)
Evaluating	I am looking forward to the findings of the study (research nurse) You need to give us a copy of the results (MoH county representative) We will organize a forum to share the results with the stakeholders (MoH national representative)

DSG Diabetes Support Group, DURE Uptake of Retinal Examination in Diabetes study, FGD Focus Group Discussion, KDDA Kenya Defeat Diabetes Association, MoH Ministry of Health, PI Principal Investigator

observed that documentation tasks were timeconsuming for peer supporters, these peer supporters did not perceive it as difficult. Given that the pilot trial had provided an opportunity to test-drive the intervention, all stakeholders recognized the intervention as "fit for trial". Stakeholders noted that the intervention utilized existing resources, rather than requiring additional resources, which pointed to the advantage of cost-efficiency and possibility for scaling up. At the individual level, some PLWD indicated an inability or unwillingness to pay the hospital consultation fee at the eye clinic might be a financial

Page 13 of 17

barrier to screening. Another relative disadvantage was the need for queuing at the eye clinic, especially because those who attended the diabetes clinic on the same day had to queue twice. One participant expressed skepticism about prioritizing screening rather than lifestyle interventions (such as physical exercise).

Outer setting

Stakeholders highlighted the growing health priority given to diabetes and other non-communicable diseases within national health policy framework as an important outer setting construct that increased stakeholder interest in the implementation of the DURE intervention. The intervention was also perceived to be responding to patients' need for a patient-centered approach to care, in addition to being aligned with the norms of the health system such as increasing efficiency and access to services.

The recent implementation of the clinical guidelines for DR had sensitized the diabetes and eye health workers that all PLWD need annual screening for DR. Eye care providers also found the guidelines useful for mobilizing required resources such as mydriatic eye drops. The guidelines were considered an aid to implementing the intervention.

Disruptions in the sociopolitical environment presented a potential outer setting constraint. During the study period, there was a disputed election that was subsequently nullified and had to be repeated. It can be challenging to maintain participant attendance to DSG meetings or to screening during periods of political turmoil. The peer supporters mitigated this potential disruption through persuasive communication with participants. The research team similarly maintained communication with all the stakeholders to ensure continued engagement, fidelity, and availability of screening at the eye clinic.

There were two external outreach eye camps in the county during the study period. Some participants attended screening at these camps instead of the eye clinic. This is because they offered the advantage of proximity, convenience, and the perception of being a scarce but valuable opportunity for screening, which constituted an external incentive. As they used the same screening guidelines, any participants screened at the outreach site was taken to have the outcome of interest.

Inner setting

The implementation climate is a key sub-construct within the inner setting construct that was found to be associated with the DURE implementation. Among the peer supporters, there was tension for change that resulted from the training, since they perceived that PLWD are vulnerable to vision loss from DR. Similarly, eye care workers expressed the need to detect and treat DR in a timely manner, as most patients presented with advanced DR. The PLWD reported having taken up screening early in the intervention because the group talks raised awareness about their vulnerability to DR, thus raising the relative priority of taking up screening.

Peer supporters found the intervention to have high compatibility as it was seen to respond to the tension for change, and it was designed to fit within the usual support group activities. To this extent, the compatibility was a facilitator for implementation. Participants tended to turn up at the eye clinic in groups especially early in the trial, and since screening procedures take time, there was a risk of overloading the health system. However, we recognized that this potential implementation barrier was foreseeably transient, since the participants would get individualized appointments for subsequent screening. Based on this premise, the eye care workers supported the implementation, and thus the potential challenge transitioned to become an enabler. This theme also emerged in relation to acceptability (above).

The organizational culture of DSGs is another subconstruct that influenced the implementation. One of the participants referenced that DSGs are usually receptive for "new things" especially those that empower the PLWD. Participants pointed to the culture of self-disclosure, which was associated with the willingness of participants to update the peer supporter and DSG members on whether they had taken up screening. Further, the culture of collective action led participants in a DSG to team up and go together for screening. Due to the culture of volunteerism, peer supporters were willing to commit time to deliver the intervention and sometimes to accompany the participants to the eye clinic. However, a potential threat to volunteerism was also voiced by peer supporters who expressed that they could be doing other things (opportunity cost) and that since their work was effective they should receive some incentives from the government.

Readiness for implementation was epitomized by the interest of DSG leadership at national level to begin scale up of peer-supporter training and to incorporate the DURE training curriculum in the peer-supporter training manual. Participants referenced national-level stakeholder interest to scale up the intervention to other counties. On the other hand, eye care workers articulated the need to equip the eye clinic with more staff and technology for screening.

Participants in the control arm requested for the intervention to be implemented in their own DSGs as they felt left out. This was already planned to meet the ethical obligation to ensure that control groups also benefit from the intervention [18]. This implementation in control arm DSGs was implemented subsequent to the trial.

A potential barrier in the structural characteristics construct of the inner setting is the geographical terrain and distance to the eye clinic for geographically remote participants—distance, unsuitable transport options, cost, and time for travel were noted to be challenges for participants of some DSGs.

Characteristics of peer supporters

On the sub-construct of knowledge and beliefs about the intervention, all peer supporters attended the training and rated it as a very useful learning experience because they had no previous exposure to eye health. The training was articulated as a facilitator of implementation since it created self-efficacy to deliver the intervention and to answer any questions from participants. Both the training and the task-shifting inherent to the intervention were valued as sources of personal fulfilment and increased role-recognition by peers and health workers. However, supporters identified that their contribution in the intervention risked being overlooked or under-recognized as they had not received formal recognition such as certificates. Providing certificates was a challenge at this stage because we needed to have the training curriculum formally adopted before certification. Providing certificates may facilitate implementation in scale up of the intervention, by stemming potential burnout or turnover of peer supporters.

Regarding individual stage of change, peer supporters expressed a collective (rather than individual) sense of accountability to implement the intervention: "we are the people to make a contribution to reducing the number of people going blind". This may relate to the culture of collective action, a theme already highlighted under inner setting constructs. Being volunteers, their motivation came from a sense of altruism, but they were also motivated by the training and subsequent initial success of seeing participants attend the screening. The 13/14 peer supporters remained engaged with the trial until the end, and they reported that the weekly telephone contact with the principal investigator kept them engaged. Community volunteers were a further source of motivation, even though they do not have diabetes. This is because they are also volunteers in the community, are well versed with the health system and they reinforced the key messages.

Cultural adaptation was further highlighted at the peer supporter level, especially relating to gender and age norms in the performance of peer support roles. For example, a female peer supporter noted that she took on a smaller proportion of the tasks and left the rest to her younger male counterpart. However, age and gender were not sufficient to account for the effectiveness of peer supporters. Participants identified some personal attributes of peer supporters that influenced them to take up screening. These included interpersonal skills demonstrated while delivering the group talk (e.g., using humour, keeping time), persistence with follow-up, individual reminders and accompanying participants to the eye clinic.

Process

Collaborating with stakeholders emerged as an important attribute. DSG leadership at national and county level were satisfied that they had been involved in the planning of the implementation, through attending preimplementation meetings. They also noted that the community entry process (through the national, county, and support group leadership) positively influenced acceptability of the intervention.

Continuous engagement was considered critical, in the form of frequent telephone contact or face-to-face contact with all the stakeholders. Peer supporters found the training and weekly telephone calls to give them a sense of identity with the trial. Research assistants valued role modelling for their tasks, while all stakeholders valued regular feedback. In the executing sub-construct, the trial was executed as per the trial protocol. It was not feasible to mask the intervention arm to the intervention, since the intervention activities within a DSG were overt. However, we did not find any evidence of contamination between clusters, perhaps because participants were not privy to the intervention allocation of other DSGs. Regarding reflecting and evaluating, we regularly reviewed the progress and quality of implementation with peer supporter and eye care workers. Peer supporters constantly reflected on their own performance and gave updates particularly on the reach and fidelity. All key informants and participants looked forward to receiving the results of the trial.

Over-arching analysis of the factors influencing implementation

We found that the intervention was in alignment with health system priorities and stakeholder interest. All five constructs of the CFIR impacted the extent to which the intervention was implemented, and the sub-constructs within different constructs were inter-related. For example, among the participants, tension for change developed when the intervention raised awareness of the vulnerability to vision loss from DR. Training of peer supporters enabled them to deliver the intervention, as well as to develop self-efficacy to increase uptake of screening among participants. Getting feedback on the number of people who had attended screening further increased selfefficacy among peer supporters. Prioritization of diabetes and implementation of clinical guidelines for DR also created tension for change among health workers who conducted the screening. Figure 3 illustrates this example of inter-relatedness. Most of the factors identified in each of the constructs were facilitators of implementation, and were modifiable, which means interventions targeting these factors can improve implementation during scaling up.



Discussion

The DURE study process evaluation provides detailed information on the implementation outcomes and the context of implementation to better understand how the intervention and context interacted. This is the first trial to increase access to DR screening through peer support interventions that has also documented the implementation process. The strengths of this study include the use of a validated theory (CFIR). This theory focuses on broad constructs that are representative of the potential influences on the implementation process, which enhances the generalizability and replicability of our methods [12]. We collected data from different stakeholders, which provided an understanding of how different stakeholders experienced the intervention and its implementation strategies. We engaged PLWD who had gone for screening as well as those who had not to identify different perspectives on facilitators and enablers for the intervention. We collected extensive data using mixedmethods, and we triangulated findings from different sources for validity and to reduce the potential risk of bias. The qualitative data were in broad agreement and provided rich context for the quantitative findings, which enhances confidence in the findings.

The trial had success with implementation outcomes, i.e., acceptability, recruitment and retention, reach, fidelity, and dose. This success resulted from the perception of stakeholders and that the intervention was beneficial. The stakeholder interest, the tension for change and the increasing priority given to diabetes in the national health policy suggest that this was an opportune time for the intervention.

Stakeholders identified that the national clinical guidelines for DR positively influence the practice and readiness of eye clinics to provide screening. By sensitizing health workers to the need to screen all PLWD, the guidelines created tension for change. This in turn ensured that all who turned up for screening received the service, as it addressed potential supply-side barriers to screening. We therefore conclude that health system strengthening such as the development and implementation of clinical guidelines was a facilitator for the implementation of the intervention as well as adoption of screening behaviour by the participants. Several studies have found that health system strengthening gives traction to heath care interventions [18–21].

We observed an early response to the intervention activities. The highest attendance at group talks was at the first group talk. In addition, all participants were referred to the eye clinic for DR screening at the start of the trial. Participants who took up screening reported doing so as soon as they received the group talk and referral, and other DSG members accompanied them. This may be related to the influence of the culture of collective action as well as the effect of the group talk and referral card. We therefore hypothesise that the first group talk and the referral are the most essential components of the intervention. We also hypothesise that these components are well aligned to the causes of non-attendance to screening in this population, which we had found to be inadequate knowledge of diabetes eye complications and lack of referral of referral for screening [2].

Among those in the intervention arm who did not take up screening in the intervention arm, lack of knowledge was not identified as a barrier, suggesting that the intervention had good reach and had increased awareness even among those who did not take up screening. At the same time, participants who did not attend screening still indicated intention to take up screening. These points to the need to identify the post-awareness/intention barriers among PLWD. Grimshaw et al. (2014) had a similar finding among physicians who received an educational intervention to increase referrals of PLWD for DR screening [12]. There were some potential relative disadvantages of the intervention, such as having to queue and pay consultation fee twice if a PLWD was attending both diabetes and eye clinics. This was foreseeable, given the structure of the hospital. These individual, social, and organizational level barriers are consistent with those in the literature [12, 21–23]. Some hospitals are testing the effectiveness of conducting DR screening at the diabetes clinic to overcome this challenge [24]. There is also need to support geographically remote individuals with logistical and financial challenges, such as through multiple access points for DR screening, closer to home [25].

The role of peer supporters in delivering the intervention was a prominent facilitator. Peer supporters delivered the health education talks and telephone reminders, but in addition, they visited participants who were not reachable on phone, accompanied participants to the eye clinics, and waited with them. These are roles of peer supporters that we had envisaged would contribute to increasing the self-efficacy of participants who took up screening [3]. This is evidence that the intervention worked through the anticipated mechanism of the intervention, which strengthens the plausibility of the results [26]. The peer supporters also identified the training and task shifting as very valuable, and these were important enablers for the role of peer supporters. We also found that community volunteers to be an important enabler to peer supporter roles, which we had not anticipated. This is evidence that complex interventions interact with the context in multiple ways to influence the outcomes.

The intervention was compatible with processes and activities of both the DSGs and the eye clinics. We noted adaptation to the delivery of the intervention (but not to the content of the intervention), driven by contextual experience. Like other investigators of pragmatic public health interventions, we recognized the need for fidelity-adaptation balance, since they coexist and are both valuable [10, 27]. To mitigate the risk of intervention drift that may occur with adaptation, we monitored the level of fidelity, which was maintained. Rather than threatening fidelity, because it ensured that those who could not be reached by phone still got the scheduled reminders to attend screening.

Our findings highlight that all the constructs of the CFIR were relevant to the trial, and there were multiple determinants to implementation. We cannot attribute the success of implementation exclusively to the characteristics of the intervention, the peer supporters, the DSGs, eye clinics, the outer setting, or implementation processes. More likely, it is the combination and interconnectivity of these constructs working together. This highlights the importance of alignment of interventions to context.

The process evaluation has potential limitations. We did not conduct evaluation at multiple points in the trial; hence, we could not capture changes over time. As the study was conducted in multiple DSGs with different peer supporters, we cannot account for differences in peer support styles. The social and health system context of the trial may also affect the generalizability of its results. The potential for researcher bias by the project investigator undertaking the qualitative interviews is acknowledged; however, it was agreed among the project team that she was the most culturally appropriate person to probe about how the context affected implementation and the mechanisms of impact. We cannot rule out response bias by social desirability, but the risk was low, given that this was not a sensitive topic, and we relied on diverse sources of data.

Conclusion

The intervention was largely implemented as designed, and achieved high implementation outcomes for acceptability, recruitment, retention, reach, fidelity, and dose. There is high stakeholder interest to support scale up. The intervention worked through the expected mechanisms, but was also aided by unanticipated mechanisms. Health system strengthening was a necessary pre-requisite for implementation. We recommend that future process evaluations are carried out at multiple points and include a cost analysis.

Supplementary information

Supplementary information accompanies this paper at https://doi.org/10. 1186/s41182-019-0188-z.

Additional file 1. Methodology for development of training for peer supporters

Abbreviations

CFIR: Consolidated Framework for Implementation Research; cRCT: Cluster Randomized Controlled Trial; DR: Diabetic Retinopathy; DSG: Diabetes Support Group; DURE: Uptake of Retinal Examination in Diabetes study; FGD: Focus Group Discussion; KDDA: Kenya Defeat Diabetes Association; PLWD: People living with diabetes

Acknowledgements

We thank the county government of Kirinyaga for supporting the study. We thank Michael Gichangi (Ministry of health), Jefitha Karimurio (University of Nairobi), Alice Mwangi (Operation Eyesight Universal), and Reuben Magoko (Kenya Defeat Diabetes Association) for assistance with the community entry process, and for valuable discussion about the study intervention. We thank the peer supporters, participants, eye care workers, and the research teams for their participation.

Authors' contributions

NM, CB, AF, SG, CM, and LM were responsible for conception and designing the study. NM, MN, LM, and MaK were responsible for fieldwork and acquisition of data. NM, MiK, DM, LM, MaK, MN, CB, and AF were responsible for data analysis. All authors participated in interpretation of the data. NM drafted the paper. JR, AF, MaK, SG, CB, MiK, and DM were responsible for the critical review of the paper. NM was responsible for revision of the paper in view of feedback from the co-authors. All authors read and approved the final manuscript.

Funding

The study is funded by the Queen Elizabeth Diamond Jubilee Trust, through the Commonwealth Eye Health Consortium in the form of a grant to the first author. The funding body did not participate in the design of the study, data collection and analysis, interpretation of data, and writing the manuscript.

Availability of data and materials

The data that support the findings of this study are available from Ministry of Health, Kenya but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the Ministry of Health.

Ethics approval and consent to participate

The ethics review committees of the London School of Hygiene & Tropical Medicine (LSHTM) and the African Medical Research Foundation (AMREF) approved the DURE study. Informed written consent was obtained at the cluster level, while informed verbal consent was obtained at the individual level. The study adhered to the tenets of the Declaration of Helsinki and the principles of Good Clinical Practice.

Consent for publication

Not applicable

Competing interests

The authors declare that they have no competing interests.

Author details

¹London School of Hygiene and Tropical Medicine, London, England. ²Kenya Medical Training College, Nairobi, Kenya. ³University of Auckland, Auckland, New Zealand. ⁴Kerugoya County Referral Hospital, Kerugoya, Kenya. ⁵University of Nairobi, Nairobi, Kenya. ⁶Oxford University NHS Trust, Oxford, UK. ⁷Kenya Medical Research Institute, Nairobi, Kenya.

Received: 6 September 2019 Accepted: 26 November 2019 Published online: 06 January 2020

References

- 1. Ning C, Paul M, Yin WT. Diabetic retinopathy. Lancet. 2010;376:124–36.
- Mwangi N, et al. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Tropical Medicine and Health. 2017;45:41.
- Mwangi N, et al. Effectiveness of peer support to increase uptake of retinal examination for diabeticretinopathy: study protocol for the DUREpragmatic duster randomized clinical trialin Kirinyaga, Kenya. BMC Public Health. 2018;18:871.
- 4. Craig P, et al. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008;337:a1655.
- Bandura A, Self-efficacy. Encyclopedia of human behavior, ed. R.V. S. Vol. Reprinted in H. Friedman [Ed.], Encyclopedia of mental health. San Diego: Academic Press, 1998. 1994, New York: Academic Press.
- Eccles MP, et al., Explaining clinical behaviors using multiple theoretical models. Implementation Science, 2012.
- Hall CE, et al. A needs assessment of people living with diabetes and diabetic retinopathy. BMC Res Notes. 2016;9:56.
- 8. Moore GF, et al. Process evaluation of complex interventions: Medical Research Council guidance. BMJ. 2015;350:h1258.
- Nielsen K, et al. Succes or failure? Interpreting and understanding the impact of interventions in four similar worksites. Work Stress. 2006;20:272–87.
- Bonell C, et al., Realist randomised controlled trials: A new approach to evaluating complex public health interventions. Soc Sci Med (1982), 2012. 75(12): p. 2299-2306.
- Oakley A, et al. Process evaluation in randomised controlled trials of complex interventions. BMJ. 2005;332(7538):413–6.
- 12. Grimshaw JM, et al. Looking inside the black box: results of a theory-based process evaluation exploring the results of a randomized controlled trial of printed educational messages to increase primary care physicians' diabetic retinopathy referrals [Trial registration number ISRCTN72772651]. Implement Sci. 2014;9:86.
- Damschroder LJ, et al. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implement Sci. 2009;4:50.
- Ministry of Health, Kenya National Bureau of Statistics, and World Health Organization, Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. 2015, Ministry of Health, Division of Non-Communicable Diseases.

- Ministry of Health. Guidelines for the screening and management of diabetic retinopathy in Kenya. Nairobi: Ministry of Health; 2017.
- Breimaier HE, et al. The Consolidated Framework for Implementation Research (CFIR): a useful theoretical framework for guiding and evaluating a guideline implementation process in a hospital-based nursing practice. BMC Nurs. 2015;14:43.
- 17. Mwangi N, et al., Feasibility of a cluster randomized controlled trial on effectiveness of peer –led health education interventions to increase uptake of retinal examination for diabetic retinopathy in Kirinyaga, Kenya: an internal pilot trial. BMC Pilot and Feasibility Studies (In press), 2019.
- Fottrell E, et al. The effect of community groups and mobile phone messages on the prevention and control of diabetes in rural Bangladesh: study protocol for a three-arm cluster randomised controlled trial. Trials. 2016;17:600.
- 19. Yip JLY, et al. Process evaluation of a National Primary Eye Care Programme in Rwanda. BMC Health Serv Res. 2018;18:950.
- 20. Tanja AJ Houweling, et al. The effect of participatory women's groups on birth outcomes in Bangladesh: does coverage matter? Study protocol for a randomized controlled trial. Trials. 2011.
- 21. Lilian RR, et al. Strengthening primary eye care in South Africa: An assessment of services and prospective evaluation of a health systems support package. PLoS One. 2017;13(5):e0197432.
- 22. Courtright P, et al. Setting targets for human resources for eye health in sub-Saharan Africa: what evidence should be used? Hum Resour Health. 2016;14:11.
- Mumba M, Hall A, Lewallen S. Compliance with Eye Screening Examinations among Diabetic Patients at a Tanzanian Referral Hospital. Ophthalmic Epidemiol. 2007;14(5):306–10.
- Gichuhi S, et al. Evaluation of the Kenyatta National Hospital diabetic retinopathy screening program 2015-2016. J Ophthalmol East Centr Southern Afr. 2017;21(2(2017)).
- 25. Poore S, et al. Planning and developing services for diabetic retinopathy in Sub-Saharan Africa. Int J Health Policy Manag. 2015;4(1):19–28.
- Mbuya MNN, et al. Theory-Driven Process Evaluation of the SHINE Trial Using a Program Impact Pathway Approach. Clin Infect Dis. 2015;61(Suppl 7):s752– 2758.
- 27. Pérez D, et al. A modified theoretical framework to assess implementation fidelity of adaptive public health interventions. Implement Sci. 2016;11:91.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions



Additional notes on data collection

Trial documents such as registers, activity logs, minutes, diaries and field notes provided quantitative and qualitative data on field activities. Quantitative data was extracted on an Excel template by the study team and used to measure recruitment, reach, dose and fidelity. Qualitative data was extracted by two team members independently using thematic analysis and provided explanatory information on the quantitative parameters, as well as on adaptation and contextual factors.

Interviews with key informants were conducted by the primary investigator and captured through field notes written during the interviews. The data was not audio-recorded as field notes were considered sufficient for capturing the information, providing an audit trail and were easily accessible to the key informant during the interview for member-checking. Multiple coders carried out the analysis as explained in the paper. These factors helped to maintain objectivity and transparency, and minimize bias.

Additional notes on measurement of contamination in the DURE study

Before the study we identified processes that may lead to contamination. The intervention was delivered during DSG meetings. There were two main processes that could lead to contamination at DSG meetings. The first process was participants in the control arm attending meetings in the intervention arm. This might occur if PLWD in the control arm were visiting in the vicinity of an intervention DSG. The second process was peer supporters from the intervention arm delivering the intervention in the control arm, if they were responsible for more than one DSG.

Methods that were used to mitigate contamination are summarised in three categories: statistical design, trial conduct, and monitoring for contamination. In statistical design, we used cluster randomisation, where the intervention was delivered within geographically dispersed clusters. This minimized the likelihood of members of different DSGs coming into close contact. Regarding trial conduct, the recruitment of peer supporters from each DSG ensured that each peer supporter was only responsible for delivering the intervention in one DSG. Visiting PLWD were not eligible for recruitment into the study. To monitor contamination, participants in all trial activities (such as attendance to DSG meetings) were carefully documented in trial registries, to identify any movement of participants between DSGs. We did not find evidence of contamination.

We measured self-efficacy in all participants before and after the intervention, as discussed in chapter 10. We found that self-efficacy was low in all arms at the start of the study, but increased in the intervention arm after the intervention. Self-efficacy remained low in the control arm. This indicated that there was no receipt of intervention in the control arm, and therefore that there was little evidence of contamination. During focus group discussions with participants and peer supporters at the end of the trial, we also enquired about exposure of the control group to the intervention. We had promised the DSG leaders that we would offer the intervention to the control arm at the end of the follow-up period. The follow-up period was short (180 days) and this may have safeguarded against contamination.

Chapter 10

Effectiveness of peer support to increase attendance at DR screening - a pragmatic cluster randomized clinical trial Outcome evaluation of the DURE trial

10.1 Overview

Research paper 1 (chapter 4) provided a conceptual model on how interventions to strengthen knowledge, referral and self-efficacy of PLWD can improve uptake of eye examination. In settings that have diabetes support groups, innovative use of this resource to influence these factors may be effective in enhancing uptake of DRS uptake.

Research paper 5, which is the study protocol (chapter 7), provided further insights on how self-efficacy for taking a retinal examination might improve through peer support in the Uptake of Retinal Examination in Diabetes (DURE) study. The previous chapter (chapter 9) reported the process evaluation of the trial, which provided evidence that the intervention worked through the anticipated mechanism of the intervention.

This chapter consists of a research paper that describes the main trial results reporting the effectiveness of the intervention. The trial is reported in adherence to the Consolidated Standards of Reporting Trials (CONSORT) 2010 statement: extensions for cluster trials and for pragmatic trials (www.consort-statement.org).

Research paper 8 (being submitted to Lancet Global Health) includes data on baseline characteristics of participants, self-efficacy scores, follow-up, attendance to DRS screening and time to screening. The data on effectiveness of the intervention is stratified by gender, as there was evidence of variability of effect between men and women. This study demonstrated a significant early screening response following the intervention, and it offers a practical way to increase attendance at DRS screening, particularly in countries where DR is often detected late.

10.2 Research Paper 8

Lancet Global Health – Licences and Copyright

The following terms will apply to this publication once it is published in this journal:

Copyright and reuse

'All content is published under <u>Creative Commons</u> licensing, which enables authors to retain copyright while allowing others to copy, distribute, and make some uses of their work, provided full credit is given to them as originators. Authors will be offered a choice of two licences (CC BY or CC BY-NC-ND) depending on whether or not they wish to allow commercial reuse of their work and whether or not they wish to allow others to alter their work in the course of its reuse. Authors will be asked to sign a licence to permit our publisher, Elsevier, to publish the work in *The Lancet Global Health*. '

https://www.thelancet.com/langlo/about

London School of Hygiene & Iropical Medicine Keppel Street, London WC1E 7HT www.lshtm.ac.uk



Registry

T: +44(0)20 7299 4646 F: +44(0)20 7299 4656 E: registry@lshtm.ac.uk

RESEARCH PAPER COVER SHEET

PLEASE NOTE THAT A COVER SHEET MUST BE COMPLETED <u>FOR EACH</u> RESEARCH PAPER INCLUDED IN A THESIS.

SECTION A – Student Details

Student	Nyawira Mwangi	
Principal Supervisor	Professor Allen Foster	
Thesis Title	Diabetic Retinopathy in Kenya: Assessment of services and interventions to improve access	

If the Research Paper has previously been published please complete Section B, if not please move to Section C

SECTION B – Paper already published

Where was the work published?			
When was the work published?	a sure and		
If the work was published prior to registration for your research degree, give a brief rationale for its inclusion			
Have you retained the copyright for the work?*	Yes	Was the work subject to academic peer review?	Yes

*If yes, please attach evidence of retention. If no, or if the work is being included in its published format, please attach evidence of permission from the copyright holder (publisher or other author) to include this work.

SECTION C – Prepared for publication, but not yet published

Where is the work intended to be published?	Lancet Global Health		
Please list the paper's authors in the intended authorship order:	Nyawira Mwangi, Covadonga Bascaran, Jacqueline Ramke, Mathew Kipturgo, Mark Ng'ang'a, Stephen Gichuhi, Min Kim, David Macloed, Consuela Moorman, Lawrence Muthami, Michael Gichangi, Allen Foster		
Stage of publication	Not yet submitted		

SECTION D – Multi-authored work

	I designed the study, conducted the research,
For multi-authored work, give full details of your role in	analysed and interpreted the data, drafted the
the research included in the paper and in the preparation	manuscript and prepared subsequent revisions
of the paper. (Attach a further sheet if necessary)	in consideration of the comments from the
	co-athors and peer reviewers

Student Signature:	Nyawira	0	<u> </u>	Date:	17 SEPT 2019
Supervisor Signature:	1	/		Date:	SERT 17, 2019

Increasing attendance at screening for diabetic retinopathy: results of the DURE cluster-randomized trial in Kenya

Nyawira Mwangi, MSc,^{1,2} Covadonga Bascaran,MSc¹ Jacqueline Ramke, PhD^{1,3} Mathew Kipturgo, MSc² Min Kim, MPH¹ Mark Ng'ang'a, MMed⁴ Stephen Gichuhi, PhD⁵ David Macleod, PhD¹ Consuela Moorman, FRCOphth⁶ Lawrence Muthami, MSc⁷ Michael Gichangi, MSc⁸ Allen Foster, FRCOphth¹

¹London School of Hygiene and Tropical Medicine

²Kenya Medical Training College, Nairobi, Kenya

³University of Auckland, New Zealand

⁴Kerugoya County Referral Hospital, Kerugoya, Kenya

⁵University of Nairobi, Nairobi, Kenya

⁶Oxford University NHS Trust, Oxford, United Kingdom

⁷Kenya Medical Research Institute, Nairobi, Kenya

⁸Ministry of Health, Nairobi, Kenya

Corresponding author: Dr. Nyawira Mwangi, Nyawira.Mwangi@lshtm.ac.uk

Abstract

Background

Diabetic retinopathy (DR) is a common complication of diabetes; it can occur without symptoms, so regular screening is essential to avoid vision loss. Access to screening for DR remains limited, particularly in low and middle-income countries (LMICs). We assessed whether combining usual care in diabetes support groups with a peer-led health education intervention was effective in increasing attendance at DR screening.

Methods

In a cluster randomised controlled trial, we randomly assigned 14 diabetes support groups (1:1) to receive a peer-led health education intervention (group talks, individual reminders and referral for screening) in addition to usual care (intervention group) or to receive usual care only (control group). The primary outcomes were proportion of participants who attended DR screening in each arm at 6 months, analysed in the intention to treat population.

[Pan African Clinical Trials Registry number: PACTR201707002430195, registered 15 July 2017].

Findings

Among 734 participants, 54.2% (200/ 369) in the intervention arm and 11.2% (41/365) in the control arm attended screening (Odds Ratio [OR] 9.4 95%CI 6.4–13.7; p<0.0001). Gender was an effect modifier (Men: OR 4.5 95%CI 2.5-7.1; Women: OR 19.5 95%CI 10.9-34.8, p<0.0001). Among those who attended screening, the median time to screening was 11 days (range 1-180) in the intervention arm and 90 days (range 1-180) in the control arm, p<0.0001. No unanticipated adverse events occurred.

Interpretation

This intervention increased attendance at DR screening in people who had not previously attended screening.

Research in context panel

Evidence before this study

Before this trial, we searched in PubMed, the Cochrane Library and trial registries on the International Clinical Trials Registry Platform for systematic reviews and randomised controlled trials (RCTs) using peer support (emotional, educational, motivational or practical assistance provided by non-professionals) to improve diabetes outcomes, including screening for retinopathy. We searched with the terms: "peer support", "diabetes", "diabetic retinopathy", "screening" and "retinal examination" for literature published in English by June 1, 2017.

We found three systematic reviews. Dale et al in 2012 included 14 RCTs and found that peer support improved clinical and behavioural outcomes in some adults with diabetes. Fisher et al in 2017 included 30 studies and found that peer support can reduce HbA1c levels. Gatlin et al in 2017 included seven studies and found that peer led self-management education can increase diabetes knowledge and reduce HbA1c. None of the studies in these reviews included any outcome related to diabetic retinopathy screening (DRS) and none of the studies was conducted in LMICs. We searched the reference lists of included studies to identify additional relevant studies, but found none.

During our study, Lawrenson et al (2018), published a systematic review in January 2018 reporting that interventions targeting patients, health professionals, the healthcare system, or a combination of these may be effective in increasing access to DRS. None of the included studies were from LMICs, and none provided evidence on the effect of peer supporter-led interventions. Thus, before our study, there was no substantial evidence on whether peer-supporter-led interventions can increase attendance to screening in LMICs.

310

Added evidence

To our knowledge, this is the first published randomised trial to assess the effect of a peer-led intervention to increase uptake of DRS among members of diabetes support groups (DSGs). We provide evidence that attendance at DRS in similar settings can substantially and rapidly improve if peer supporters are empowered to provide an intervention that combines health education and referral. This finding is novel, and of considerable potential importance not only because the intervention is feasible and acceptable, but also because a mixed-methods process evaluation of the trial identified the active ingredients of the intervention. Our findings further add value to the existing evidence because the participants were people who had never had DRS hence they are the population who need screening most urgently.

What the whole evidence means

Increasing attendance to DRS is a key objective of DR screening programs in all countries. Our results showed that it is feasible to reach PLWD who have never had screening with a peer support intervention that increases initiation of DRS. Many countries may have untapped potential for improving health-care access using existing resources such as community groups. This highlights the importance of collaboration with DSGs in order to reach this highly vulnerable group in the real world settings.

The findings have important implications for public health programs, particularly in countries where DR is often detected late with the risk of resultant visual loss and blindness. In such resource-limited settings, DR control programs should give priority to potentially feasible and sustainable public health interventions like those suggested in our study. Future studies should investigate context-specific costs of the intervention, and whether the intervention leads to maintenance of annual DRS over the long term.

Background

Retinal changes due to diabetic retinopathy (DR) were first described by Jaeger in 1856.⁽¹⁾ It is estimated that there were 420 million people living with diabetes (PLWD) in 2017, with a third having DR.⁽²⁾ Early detection and treatment are important for preventing vision loss associated with DR, therefore regular DR screening (DRS) for all PLWD is recommended.⁽³⁾ Low uptake of DRS is well documented, particularly in low and middle-income countries (LMICs), where the majority of PLWD live. Lack of screening or low uptake contribute to increased risk of vision loss from DR.⁽⁴⁾

Given the need to increase coverage of screening, it is essential that attendance at DRS be maximized as far as resources allow. Finding low cost effective interventions that can increase uptake of screening in the real world setting is challenging for DR screening programs. Lawrenson in 2018, in a systematic review on attendance at DRS, provided evidence that various interventions targeting patients, healthcare professionals or the healthcare system are associated with meaningful improvements in DRS attendance.⁽⁵⁾ However, the authors noted that the interventions were 'black box' whose active ingredients were not easily identifiable. All the 66 studies included in this review were conducted in high income countries, hence there is need for evidence from LMICs.⁽⁵⁾ There is also need for evidence on interventions that are targeted to those who need DRS most— those who have never had screening, as they are highly vulnerable to vision loss from DR.

Improving the knowledge of PLWD about eye screening is essential if avoidable vision loss is to be prevented.⁽⁶⁾ The lack of knowledge of PLWD on diabetes eye health is known to be a modifiable predictor of DRS uptake, especially in LMICs.^(7, 8) Knowledge can raise PLWD's perception of vulnerability and severity of DR, hence building self-efficacy for adherence to screening recommendations. A recent systematic review of barriers and enablers for DRS reported higher odds of uptake of DRS when PLWD were provided health education.⁽⁷⁾ The effect of health

education interventions led by peers within community-based diabetes support groups (DSGs) on attendance at DRS remains equivocal, although peer support is beneficial for other outcomes.⁽⁹⁻¹¹⁾

The Uptake of Retinal Screening in Diabetes (DURE) trial was a two arm (1:1) pragmatic clusterrandomized clinical trial (cRCT) designed to test the hypothesis that the proportion of PLWD attending DRS would be higher in DSGs with the peer supporter-led intervention than in DSGs offering the usual standard of care in DSGs. The mechanism of effect of the intervention was predicted to be through increasing the self-efficacy of PLWD to adopt screening behaviour. The primary outcome was the proportion attending DRS in either arm during the six-month study period. We defined DRS as measurement of visual acuity and a retinal examination through a dilated pupil conducted by an eye care worker using either an ophthalmoscope, a slit lamp microscope or a retinal camera, which reflects current practice in Kenya.⁽¹²⁾

We chose a cRCT design instead of an individually randomised trial because peer support is normally provided at group level in DSGs, and to reduce the effect of intervention contamination, since patients in the same DSG often interact with one another. The trial design was guided by the United Kingdom's Medical Research Council framework for developing complex interventions.⁽¹³⁾ Patient representatives were involved in the planning and implementation of the intervention. We anticipate that the trial findings will inform future strategies for improving screening services and subsequent early detection and treatment of vision threatening DR (VTDR).

Methods

Setting

Kenya is a LMIC, as defined by the World Bank⁽¹⁴⁾, and the prevalence of diabetes is 2% in the 18-64 years age group.⁽¹⁵⁾ Kirinyaga is a rural county located in central Kenya. Our formative research had established that only 12% of PLWD who are already attending diabetes services in Kirinyaga had ever taken DRS.⁽⁸⁾ DRS services are available at the county hospital on a walk-in basis, but

313

they are under-utilized. An estimated 25%-30% of PLWD are registered members of DSGs, while an additional 20% of PLWD attend DSG meetings occasionally even though they are not members.

Study design and conduct

The study protocol was peer-reviewed and a favourable ethics opinion was received from the ethics committees of the London School of Hygiene and Tropical Medicine (LSHTM) and the African Medical Research Foundation. The trial was conducted in accordance with the principles of the International Conference on Harmonisation of Good Clinical Practice under the oversight of the study sponsor, LSHTM. Written informed consent was obtained at the cluster level, while participants gave verbal consent to receive the study interventions and follow-up.

The sample size was estimated using standard formulas for cRCTs recommended by Hayes and Bennett.⁽¹⁶⁾ Based on the data from formative research, with 50 PLWD in each DSG and 12% of PLWD attending DRS services in the control arm, and using a coefficient of variation of 0.25, 14 clusters (7 in each arm, Fig 10-1) were required in order to have 80% power to detect a twofold increase in uptake in the intervention arm (with statistical significance at the 5% level).

A research nurse recruited the participants from each group through simple random sampling prior to randomization. Table 10-1 shows the eligibility criteria while Fig 10-2 (at the end of the chapter) shows the participant flow in the trial. Computer generated random numbers were obtained remotely by a statistician on STATA 15 (StataCorp 2017) and used for allocation of clusters to study arm. The allocation sequence was concealed from trial personnel until the moment of assignment.

For each cluster in the intervention arm, two peer supporters (1 male, 1 female) were selected according to the exclusion and inclusion criteria and received a two-day training to deliver the intervention. The content of the training is described in the trial protocol⁽¹⁷⁾ and summarized in Box 1. In addition to usual care at DSGs, the intervention arm had three components: monthly

314

group talks on diabetes eye health, individual referral for DRS using a referral card and monthly individual telephone reminders to take DRS. Peer supporters received airtime to facilitate the telephone calls, as well as reimbursement for any logistical costs, but no other incentives. Research assistants monitored the implementation of the intervention. Self-efficacy to take a retinal examination (a participant's confidence in his ability to obtain a retinal examination, despite potential barriers) was assessed at baseline, after the first group talk and at the end of the study period, using a structured questionnaire. The primary outcome was attendance at DRS within six months. The records at Kerugoya County Hospital eye clinic were monitored daily and the identifier cards of participants that attended screening were deposited in a specific container by the eye care team. These cards were then collected and given to the outcome assessment nurse (not involved in recruitment or follow-up).

Criterion	Participants	Peer supporters
Age > 18 years		· ·
Member of a diabetes support group		
Will reside in the county for the next 12 months		
Has a mobile phone		
Willing to participate in the study		
Had not had a screening exam in the last 12 months		
Has had a screening exam in the preceding 12 months		
Willing to be a peer supporter		
Willing to commit two days for training		
Willing to commit many hours to peer support work		
Fluent in Kikuyu or Kiswahili		
Already attending DR screening		
Already receiving treatment for DR		
Has a debilitating illness		

Table 10-1: Eligibility criteria for participants and peer supporters

KEY	Include	Exclude	Not applicable
-----	---------	---------	----------------

The recruitment nurse, research assistants and the outcome assessor had no training in eye care, and were masked to the intervention allocation, to avoid contamination and bias. The eye care providers were also masked to the intervention allocation. It was not possible to mask the study participants or peer supporters in the intervention arm, because the peer supporters' activities within the DSG were overt. As the primary outcome was assessed from hospital records, we concluded that the lack of masking could not incentivise over-reporting or under-reporting in either arm.

Box 1: Trial Interventions		
Domain	Intervention group	Control group
Usual care	Monthly group meetings with general diabetes education talks, blood sugar and blood pressure measurements	Monthly group meetings with general diabetes education talks, blood sugar and blood pressure measurements
Intervention		
Peer-supporters training	Two days training following a structured curriculum regarding diabetes eye disease, retinal screening, role of peer-supporters, communication and other aspects specified in the protocol ⁽¹⁷⁾	
Group education	Monthly group education provided by trained peer- supporters, with structured content on diabetic eye disease and retinal screening as specified in the protocol ⁽¹⁷⁾	
Individual participant	Monthly individual telephone	
reminders	reminder by peer-supporters	
	to participants to take a	
	the protocol ⁽¹⁷⁾	

Statistical analysis and reporting

This study is reported in accordance with the Consolidated Standards of Reporting Trials

(CONSORT) 2010 statement and its extensions to cluster randomized studies⁽¹⁸⁾ and pragmatic

trials⁽¹⁹⁾. The CONSORT flow chart and the checklists are included as appendices. All data analysis was carried out using Stata 15 (StataCorp, College Station, Texas, USA 2017) and following a pre-specified analysis plan (Appendix 9).

We calculated descriptive statistics for participants' characteristics and compared the study arms. Continuous variables are summarised by mean and standard deviation, SD (if normally distributed), or median and inter-quartile range (IQR) if distribution was skewed. Categorical data are summarised as frequencies and percentages. We further analysed the intra-DSG correlation for these variables, to address a gap found in previous DRS studies.⁽⁵⁾

Our primary analysis was by intent-to-treat population, and we did not conduct interim analyses. The primary outcome (attendance at DRS) was binary, and analysed at the participant level as well as at cluster level. Tests of crude association with exposure to the intervention and with selfefficacy scores were performed. The measure of effect is presented as odds ratios (OR) with 95% confidence intervals (CI).



Fig 10-1: Location of the clusters and the eye clinic within Kirinyaga county

We generated Kaplan-Meier survival curves to illustrate the difference in time-to-attendance between the two arms. We assessed the difference in time-to-attendance with hazard ratios (HRs) estimated by Cox regression. As the assumption for proportional hazards was not met, we estimated HRs for narrower time bands, within which the proportional hazard assumption holds.

We performed logistic regression for bivariate and multivariable analysis, while accounting for clustering, to identify which of the independent variables were predictors of attendance to DRS. We also constructed nested multivariable logistic models including all potential confounders and did not find any evidence of a confounding effect.

Role of the funding source

The funder of the study had no role in study design, data collection, data analysis, data interpretation, or writing of the report. The first author had full access to all the data in the study and had final responsibility for the decision to submit for publication.

Results

Baseline characteristics

All the 14 DSGs completed the trial. Baseline data was collected for all 734 participants (369 in intervention and 365 in control arm). The two arms were balanced for baseline characteristics (Table 10-2). Appendix 8 shows the intra-cluster correlation coefficients (ICC) for the baseline anthropometric and clinical characteristics. The ICC is low for all the variables, indicating minimal clustering effect.

Characteristic	Control (N=365)	Intervention (N=369)
Age (years) mean (SD)	59.6 (11.6)	56.5 (11.5)
Gender -Female n (%)	211 (57.8%)	208 (56.4%)
Duration of diabetes years, median (IQR)	4 (1 - 9)	4 (2-8)
Duration of DSG membership median (IQR)	2 (1-4)	2 (1-3)
Education (highest level completed) N, %		
Did not complete primary	108(29.6%)	84(22.8%)
Primary	167(45.8%)	171(46.1%)
Secondary	69(18.9%)	84(22.8%)
Post-secondary	21(5.8%)	30(8.1%)
Employed n (%)	294(80.1%)	297(80.5%)
Known comorbidity with hypertension (N, %)	34 (9.3%)	61(16.5%)
Body Mass Index mean (SD)	28.3 (5.0)	27.9 (5.1)
Waist circumference cm (mean, SD)	89.9 (8.5)	87.2 (9.6)
Random Blood Sugar mmol/l (mean, SD)	8.0 (3.2)	7.9 (3.0)
Systolic blood pressure mmHg (mean, SD)	137.8 (19.4)	134.3 (18.8)
Diastolic blood pressure mmHg (mean, SD)	76.6 (12.3)	77.4 (11)
Self-Efficacy (mean, SD)	13.4/50 (1.2)	13.5/50 (1.2)

Table 10-2: Baseline characteristics of study participants (N=734)

Follow-up

Some 4.2% (n=31) participants were lost to follow-up (20/369 in intervention arm and 11/365 in control arm). They were lost to follow up relatively late in the study (median 150 days, interquartile range 96-156 days). The reasons for loss to follow-up were travel (Intervention 7, Control 2); illness, not related to the intervention (Intervention 2, Control 1); untraceable (Intervention 11, Control 8). No adverse reactions were reported among the 369 participants who received the intervention. Participants who attended for retinal examination screening experienced temporary blurring of vision due to the effect of mydriatic eye drops, but this was anticipated and explained to the participants in advance. All of them described this effect as minor.

Proportion attending screening

In the intervention arm, 54.2% (200/ 369) and in the control arm 11.2% (41/365) PLWD attended DRS. Compared with usual care, the intervention quintupled the number of participants taking screening. Overall, participants in the intervention arm had 9.4 times the odds of taking screening compared to those in the control arm (Odds Ratio 9.4, 95% Cl 6.4–13.7; p<0.0001). Given the apparent imbalance between trial arms at baseline in terms of age, education, known comorbidity with hypertension and waist circumference, adjusting for these covariates (Odds Ratio 9.4, 95% Cl 6.3 - 14; p<0.0001).

We obtained similar results in both participant level and cluster level analysis, Fig 10-3 shows the attendance by cluster. The influence of clustering on attendance to screening was reflected in an ICC of 0.3, (95% CI 0.05-0.54, p<0.0001), meaning that 30% of the variation in attendance to screening can be explained by the variation between DSGs. Attendance at DR screening was markedly high in two of the intervention DSGs, as compared to the other DSGs. Process evaluation (chapter 9) showed that this was due to the exceptional performance of the peer supporters in the task of delivering the intervention.



Figure 10-3: Proportion of PLWD that attended screening in each cluster

Although the intervention effect was high in both men and women, there was a more pronounced effect in women. The odds of taking screening in women was higher than in men (OR 19.5, 95% CI 10.9-34.9, p<0.001 compared to OR 4.2 95% CI 2.5-7.1, p<0.001), which was confirmed through logistic regression (interaction p<0.0001). Gender is thus an effect modifier. We did not find variability in the pooled effect estimates for other variables, hence there were no strong reason to perform subgroup analysis for other variables.

The scores for self-efficacy to take DRS at baseline did not differ between the two arms (median score 13/50, IQ z\d R 12-14 for both arms, p=0.2). After exposure to the first dose of the intervention there was strong evidence of a difference in self-efficacy scores, intervention arm (median 43/50, IQR 42-45) and control arm (median 14/50, IQR 13-15), p<0.0001) (Fig 10-4). This increase in self-efficacy in the intervention arm was associated with DRS (OR 9.4, 95% CI 6.4-13.7, p<0.0001).



Fig 10-4 Self-efficacy scores in intervention and control arms

Time to screening

Trends in attendance to DRS over the study period were consistently higher in the intervention arm compared with the control arm, (Fig 10-5 and Table 10-3). However DRS was only taken up by 54% (200/369) of the intervention arm over the 6 months, of which 183 (50%) had their screening within the first 30 days of the trial after the first group talk and referral for screening. This indicates that the intervention had an early benefit for DRS attendance, but only an additional 16 (4%) attended DRS over the next five months of further exposure to the intervention. In the control arm, DRS attendance was spread more evenly across the follow-up period at 1-3% of control participants/ month. The median time to screening was 11 days (range 1-180) in the intervention arm and 90 days (range 1-180) in the control arm, p<0.0001.


Fig 10-5 Kaplan-Meier analysis of time from first group talk to attendance at DRS (A) and in relation to the group talks (B)

	Proportion of participants attending screening			
Time period (days)	Intervention arm (n=369)	Control arm (n=365)	Hazard ratio [CI]	p value
0 -30	183(49.6%)	4 (1.1%)	63.0 [20.4 - 194.6]	<0.001
31-60	14 (3.8%)	7 (1.9%)	4.3 [1.6 - 11.8]	<0.001
61-90	2 (0.5%)	10 (2.7%)	0.4 [0.1 - 2.7]	<0.001
91-180	1 (0.3%)	20 (5.5%)	0.1 [0.02 - 2.6]	<0.001

Table 10-3: Time period of trial in which attendance at DR screening occurred

Discussion

We found strong evidence that the intervention increased uptake of screening. To our knowledge, this is the first trial to investigate the effect of peer-led intervention on DR screening rates within diabetes support groups in any part of the world. The trial population was homogeneous in that they had not previously taken screening, hence they are the group particularly in need of screening.⁽⁶⁾ The large effect of the intervention is likely to be because the intervention addressed known barriers to screening in this community, which are lack of referral and lack of knowledge on diabetes eye health.⁽⁸⁾ Secondly, the high effect might be because the PLWD in our trial had never had screening, and other studies have reported that populations with low screening rates at baseline show the largest improvements.

We found an early response—PLWD exposed to the intervention tended to attend screening right after the first group talk and subsequent referral, and further group talks did not tend to increase the effect. This important finding has not been reported in the literature. In addition, similar to Zhang et al, we observed a ceiling effect, whereby additional reminders did not increase attendance at DR screening.⁽⁶⁾ Given the relatively low probability of further improvement after exposure to the active ingredients, it may be appropriate to limit the intervention to a single group talk and referral, and then repeat the intervention after six months. Additional intervention components are likely required to increase uptake among those who did not respond to this intervention.

In this population of African PLWD, diabetes was of short duration on average (median duration 4 years) and glycemic control at the time of study was poor. The demographic and anthropometric characteristics of this group are comparable to the characteristics of the national population 18-64 years, as described in the Kenya STEPwise survey for risk factors for non-communicable disease.⁽¹⁵⁾ The results can be used to improve services in populations with a similar profile.

We envisaged that the intervention would cause effect by increasing PLWD's self-efficacy to initiate screening and this was confirmed. However, 46% of the PLWD in the intervention arm still did not take screening even though the overall self-efficacy in the group increased. This suggests that some participants face additional post-self-efficacy barriers that are not addressed by this intervention. Similar to other studies, our process evaluation (chapter 9) found that geographical barriers (for PLWD living in rural remote areas) as well as direct and indirect costs of screening are additional challenges for PLWD.⁽²⁰⁾ On the other hand, it is also possible that in some individuals there is a considerable time lag between improvement in self-efficacy and uptake of screening.

In the control arm, 11% attended DRS. This might be because all participants were asked if they have ever had a DR screening exam in the baseline assessment, which may have encouraged them to think about the need for DRS and motivated them to go for screening. Gender was an effect modifier with a more pronounced effect in women. This is fortunate, since women are known to have a higher burden of VI and lower use of eye care services, as compared to men.^(21, 22)

Younger age and higher education were associated with increased odds of attendance to DRS. Considering the implications of DR-related vision loss on quality of life, and economic consequences such as lost productivity, it is of interest that young PLWD attend screening. In contrast, some studies have reported younger age to be associated with poor attendance to

326

screening.⁽²⁰⁾ There is consensus in most studies that higher education improves access to eye care services.^(21, 23)

In women, known comorbidity with hypertension and poor control of blood pressure were associated with increased odds of attendance to DRS.⁽²⁾ This is fortunate, since poor blood pressure control is a known risk factor for DR, and these women at high risk are more likely to attend DRS. However the same effect was not seen in men with hypertension.

In men, being employed was associated with reduced odds of attendance, perhaps because the DSG activities and the screening service are mostly available during regular working hours, when they are at work. We did not note a similar association for women, possibly reflecting different circumstances around employment. This is an indicator that there are additional barriers for men, and it might explain why the intervention effect in men was less pronounced than in women. It might be useful to target work spaces with additional interventions for men in employment.

Two DSGs in the intervention arm had higher performance than other DSGs, due to the high performance of peer supporters in their tasks. This shows the need for DR screening programs using this intervention to pay attention to the performance of peer supporters, and the factors that influence this performance. Peer supporters are volunteers and have different motivators for performance. Incentives that are responsive to their needs may sustain high performance.⁽²⁴⁾ These incentives may include formal employment with remuneration. However it is unclear whether employment would stifle the creativity, flexibility and sense of altruism that underpins this performance.^(24, 25) Non-financial incentives including training and recognition through provision of badges, t-shirts or bicycles to help with transportation are appreciated by community volunteers in Kenya.^(26, 27) The right package of incentives that is essential for program

The strengths of our study are the cluster randomised study design and robust study procedures that followed international guidance, with oversight from a registered clinical trials unit. It is a pragmatic study, conducted in real world settings and using peer supporters to deliver the intervention, which facilitates sustainability and expansion to larger scale. The choice of intervention was based on barriers and enablers identified in literature review and formative research.⁽⁸⁾ The follow up rate was high, and the primary outcome was measured objectively with data for the outcome being available for all participants. The trial steering team included a diabetes educator; this expertise is known to be particularly important in health education interventions for PLWD. The trial was also accompanied by a robust theory-driven process evaluation, hence it is not a 'black-box' intervention.^(5, 28)

The trial provides an important example of task shifting to increase screening coverage. The task of providing health education on diabetes eye health was shifted from the clinic to the community, and from health workers to the peer supporters in DSGs. In turn the peer supporters referred the PLWD to the clinic, thus facilitating DRS. This linkage is important for reaching underreached groups with screening interventions. Our study thus contributes to ongoing discussions on the role of task shifting in eye health.

Our study has limitations that also indicate directions for future research. We used a single geographic setting so the results may not be widely generalizable. However, they are likely to be replicable in LMIC settings with diabetes support groups and poor uptake of DRS. Longer-term follow-up studies are needed to determine the sustainability of the effect of peer support on attendance to regular annual screening, as opposed to a one-time effect. We did not collect data on socioeconomic status and did not measure cost-effectiveness.

The findings have important implications for clinical and public health policy, particularly in lowincome countries where DR is often detected late. This peer supporter-led intervention may be effective in countries with similar health system structures, an opportunistic DR screening model and a low attendance to DRS. Many countries have diabetes support groups with peer supporters;

328

innovative use of this resource in enhancing uptake of DRS uptake may be implemented with minimum disruption to the existing health system.

This study may assist health workers understand the relationship between self-efficacy and DRS when attending to individual PLWD.⁽²⁹⁾ Our findings should assist policy makers in developing DRS strategies that utilise existing resources to improve coverage of screening. The next step in scaling up the intervention in Kenya would be for the Ministry of Health to collaborate with DSGs and train peer supporters to deliver the intervention in all 47 counties. We are presently planning this approach, which has stakeholder support.

Conclusion

The pragmatic DURE intervention resulted in a positive and early response for attendance at DRS offering a practical way to increase screening. The intervention was effective for both men and women, although it had a greater impact on women. Further research is needed to assess the best timing for repeating the intervention and the cost-effectiveness of this strategy.

Declarations

Consent for publication

Not applicable

Availability of data and materials

The data that support the findings of this study are available from Ministry of Health, Kenya but restrictions apply to the availability of these data, which were used under license for the current study, and so are not publicly available. Data are however available from the authors upon reasonable request and with permission of the Ministry of Health.

Competing interests

The authors declare that they have no competing interests.

Abbreviations

CI	Confidence interval
cRCT	Cluster randomized clinical trial
DR	Diabetic retinopathy
DRS	Diabetic retinopathy screening
DURE	Uptake of Retinal Examination in Diabetes study
DSG	Diabetes support group
HbA _{1c}	Glycated haemoglobin
HIC	High income country
HR	Hazard ratio
IQR	Interquartile range
LMICs	Low and middle income countries
LSHTM	London School of Hygiene & Tropical Medicine
OR	Odds ratio
PLWD	People living with diabetes
RCT	Randomized clinical trial
SD	Standard deviation
VTDR	Vision-threatening diabetic retinopathy

Acknowledgements

We thank the county government of Kirinyaga and the Kenya Defeat Diabetes Association for supporting the study. We thank the peer supporters, participants, and the research teams for their participation. We are grateful to the Queen Elizabeth Diamond Jubilee Trust through the Commonwealth Eye Health Consortium for funding the study.

REFERENCES

1. Kalantziz G, Angelou M, Poulakou-Rebelakou E. Diabetic retinopathy: An historical assessment. Hormones International Journal of Endocrinology and Metabolism. 2006;5(1):72-5.

2. Yau JWY, Rogers SL, Kawasaki R, Lamoureux EL, Kowalski JW, Bek T, et al. Global prevalence and major risk factors of diabetic retinopathy. Diabetes Care. 2012;35:556-64.

3. International Council of Ophthalmology. ICO Guidelines for Diabetic Eye Care-updated 2017. San Francisco, California: International Council of Ophthalmology, 2017.

4. Scanlon PH. The English National Screening Programme for diabetic retinopathy 2003–2016. Acta Diabetologica. 2017;54(6):515-25.

5. Lawrenson JG, Graham-Rowe E, Lorencatto F, Burr J, Bunce C, Francis JJ, et al. Interventions to increase attendance for diabetic retinopathy screening (Review). Cochrane Database of Systematic Reviews. 2018;2018(Issue 1 Art. No.: CD012054).

6. Zhang X, Norris SL, Saadine J, Chowdhury FM, Horsley T, Kanjilal S, et al. Effectiveness of interventions to promote screening for diabetic retinopathy. Am J Prev Med. 2007;33(4):318-35.

7. Mapa Mudiyanselage Prabhath Nishantha Piyasena, Gudlavalleti Venkata S. Murthy, Jennifer L. Y. Yip, Clare Gilbert, Maria Zuurmond, Tunde Peto, et al. Systematic review on barriers and enablers for access to diabetic retinopathy screening services in different income settings. PLoS ONE. 2019;14(4):e0198979.

8. Mwangi N, Macleod D, Gichuhi S, Muthami L, Moorman C, Bascaran C, et al. Predictors of uptake of eye examination in people living with diabetes mellitus in three counties of Kenya. Tropical Medicine and Health. 2017;45(41).

9. Fisher EB, Boothroyd RI, Elstad EA, Hays L, Henes A, Maslow GR, et al. Peer support of complex health behaviors in prevention and disease management with special reference to diabetes: systematic reviews. Clinical Diabetes and Endocrinology. 2017;3(1):4.

10. Dale JR, Williams SM, Bowyer V. What is the effect of peer support on diabetes outcomes in adults? A systematic review. Diabetic Medicine. 2012;29(11):1361-77.

11. Gatlin TK, Serafika R, Johnson M. Systematic review of peer education intervention programmes among individuals with type 2 diabetes. Journal of Clinical Nursing. 2017;26(23-24):4212-22.

12. Ministry of Health. Guidelines for the screening and management of diabetic retinopathy in Kenya. Nairobi: Ministry of Health, 2017.

13. Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M. Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ. 2008;337(a1655).

14. World Bank. Income levels 2019. Available from: https://data.worldbank.org/income-level/lower-middle-income.

15. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases, 2015.

16. Hayes RJ, Bennett S. Sample size calculation for cluster randomized trials. International Journal of Epidemiology. 1999;28:319-26.

17. Mwangi N, Ng'ang'a M, Gakuo E, Gichuhi S, Macleod D, Moorman C, et al. Effectiveness of peer support to increase uptake of retinal examination for diabeticretinopathy: study protocol for the DUREpragmatic cluster randomized clinical trialin Kirinyaga, Kenya BMC Public Health. 2018;18(871).

18. Campbell MK, Piaggio G, Elbourne DR, Altman DG, for the CONSORT Group. Consort 2010 statement: extension to cluster randomised trials. BMJ. 2012;345.

19. Zwarenstein M, Treweek S, Gagnier JJ, Altman DG, Tunis S, Haynes B, et al. Improving the reporting of pragmatic trials: an extension of the CONSORT statement. BMJ. 2008;337.

20. Leese GP, Boyle P, Feng Z, Emslie-Smith A, Ellis JD. Screening uptake in a well-established diabetic retinopathy screening program: the role of geographical access and deprivation. Diabetes Care. 2008;31:2131–5.

21. Ehrlich JR, Stagg BC, Andrews C, Kumagai A, Musch DC. Vision Impairment and Receipt of Eye Care Among Older Adults in Low- and Middle-Income Countries. JAMA Ophthalmology. 2019;137(2):146-58.

22. Jacqueline Ramke, Anthony B. Zwi, Anna Palagyi, Ilse Blignault, Clare E. Gilbert. Equity and Blindness: Closing Evidence Gaps to Support Universal Eye Health. Ophthalmic Epidemiology. 2015;22(5):297-307.

23. Cleland CR, Burton MJ, Hall C, Hall A, Courtright P, Makupa WU, et al. Diabetic retinopathy in Tanzania: prevalence and risk factors at entry into a regional screening programme. Trop Med Int Health. 2016;21(3):417-26.

24. Kok MC, Broerse JEW, Theobald S, Ormel H, Dieleman M, Taegtmeyer M. Performance of community health workers: situating their intermediary position within complex adaptive health systems. Human Resources for Health. 2017;15(1):59.

25. Cherrington A, Ayala GX, Elder JP, Arredondo EM, Fouad M, Scarinci I. Recognizing the diverse roles of community health workers in the elimination of health disparities: from paid staff to volunteers. Ethn Dis. 2010;20(2):189-94.

26. Aseyo RE, Mumma J, Scott K, Nelima D, Davis E, Baker KK, et al. Realities and experiences of community health volunteers as agents for behaviour change: evidence from an informal urban settlement in Kisumu, Kenya. Human Resources for Health. 2018;16(1):53.

27. Ormel H, Kok M, Kane S, Ahmed R, Chikaphupha K, Rashid SF, et al. Salaried and voluntary community health workers: exploring how incentives and expectation gaps influence motivation. Human Resources for Health. 2019;17(1):59.

28. Damschroder LJ, Aron DC, Keith RE, Kirsh SR, Alexander JA, Lowery JC. Fostering implementation of health services research findings into practice: a consolidated framework for advancing implementation science. Implementation Science 2009;4(50).

29. Hall CE, Hall AB, Kok G, Mallya J, Courtright P. A needs assessment of people living with diabetes and diabetic retinopathy. BMC Res Notes. 2016;9(56).

Fig 10-2 Trial flow chart

Enrolment		We invited 16 clusters to			
		participate			
			Two clusters participated in the pilot trial		
Cluster		14 clusters were recruited			
recruitment into		for the main trial			
pilot trial					
		Cluster members assessed			
		for eligibility (837)			
			86 did not meet eligibility criteria		
			 (48 were not regular members of the DSGS, 15 did not have a phone and 23 will not reside in the county for the period of the trial) 17 declined participation 		
		Participants recruited (734)			
Random	Intervention arm: 7		Control arm: 7 clusters, 365		
Allocation	clusters, 369		participants		
	participants, 14 peer				
	educators				
	Usual care				

Intervention	Peer educators received		All received usual care		
	a 2 day training				
	, ,				
	All participants received				
	the intervention				
	(monthly group talk)				
	Those who had not				
	attended eye exam				
	received a weekly				
	individual reminder				
		+			
			•		
Follow-up	349 participants and 13		354 participants were retained over		
	peer educators were		180-day follow-up period.		
	retained over 180-day		11 were lost to follow-up		
	follow-up period.				
	20 were lost to follow-				
	up.				
	1 peer educator got a				
	job in another county				
	and was unavailable day				
	72 - 180				
Interim estimate	369 participants		365 participants included		
of primary	included		0 participants excluded		
outcome	0 participants excluded		Intention-to-treat analysis		
	Intention to treat				
	analysis				

Figure 10-2: Flow diagram for the trial

Table 10-4: Predictors of attendance at DR screening

Variable	Attended	Did not attend	Odds	[95% Confidence	p value
	screening (241)	screening (493)	Ratio	Interval]	
			9.4	[6.4 - 13.7]	<0.001
Control	41	324	1		
Intervention	200	169	9.4	[6.4 - 13.7]	<0.001
Age (years)			0.5	[0.4 - 0.7]	<0.001
<45 years	48	45	1		
45-64 years	142	272	0.5	[0.3 - 0.8]	0.002
>65 years	51	176	0.3	[0.2 - 0.5]	<0.001
Gender			1.18	[0.9 - 1.6]	0.3
Male	97	218	1.00		
Female	144	275	1.18	[0.9 - 1.6]	0.3
Duration since diagnosis of diabetes			1.0	[0.9 - 1.2]	0.9
<6 years	158	313	1		

6-9 years	30	78	0.8	[0.5 - 1.2]	0.2
10-20 years	41	80	1	[0.7 - 1.5]	0.9
>= 20 years	12	22	1.1	[0.5 - 2.2]	0.8
Duration of membership to DSG			0.6	[0.5 - 0.9]	0.003
<3 years	165	287	1		
3-5 years	74	191	0.7	[0.5 - 0.9]	0.02
>=6 years	2	15	0.2	[0.05 - 1.02]	0.05
Education			1.5	[1.3 - 1.8[<0.001
Did not complete primary	38	154	1		
Primary	116	222	2.1	[1.4 - 3.2]	<0.001
Secondary	64	89	2.9	[1.8 - 4.7]	<0.001
Post-secondary	23	28	3.3	[1.7 - 6.4]	<0.001
Employed			0.8	[0.5 - 1.1]	0.2
Unemployed	54	89	1		
Employed	187	404	0.8	[0.5 - 1.1]	0.2
Hypertension			2.0	[1.3-3.1]	0.001
Previously diagnosed with hypertension	45	50	2.0	[1.3-3.1]	
Not previously diagnosed with hypertension	196	443	1		
Body Mass Index			0.9	[0.8 - 1.1]	0.2

Normal	74	131	1		
Increased	167	362	0.8	[0.6 - 1.1]	0.2
Waist circumference			0.9	[0.7 - 1.3]	0.7
Normal	109	215	1		
Increased	132	278	0.9	[0.7 - 1.3]	0.7
Random Blood Sugar			1.01	[0.8 - 1.4]	0.9
Hypoglycaemia	6	14	1		
Normoglycaemia	129	263	1.14	[0.4 - 3.0]	0.8
Hyperglycaemia	106	216	1.14	[0.4 - 3.1]	0.9
Systolic blood pressure at baseline mmHg			0.9	[0.6 - 1.3]	0.5
Normal	54	101	1		
Elevated	187	392	0.9	[0.6 - 1.3]	0.5
Diastolic blood pressure at baseline mmHg			1.3	[1.0 - 1.8]	0.07
Normal	143	326	1		
Elevated	98	167	1.3	[1.0 - 1.8]	0.07
Visual Acuity			1.1	[0.8-1.6]	0.5
Normal	176	372	1		
Visual Impairment	65	121	1.1	[0.8-1.6]	0.5



Section E

"I fully realize that I have not succeeded in answering all of your questions... Indeed, I feel I have not answered any of them completely. The answers I have found only serve to raise a whole new set of questions, which only lead to more problems, some of which we weren't even aware were problems. To sum it all up... In some ways I feel we are confused as ever, but I believe we are confused on a higher level, and about more important things."

> Mary Ann Raywid, 1928-2010 Scholar, author and activist

Chapter Eleven: Thesis summary

11.0 Overview

This thesis documents lessons learned regarding the assessment of diabetic retinopathy (DR) services in Kenya and interventions to improve access to these services. In this chapter, we summarize the PhD narrative, including the overall strengths and limitations, highlight the key findings, consider the implications for policy, practice, research, and dissemination, and provide recommendations and conclusions.

11.1 The PhD narrative

The thesis is organized in five sections A-E and 11 chapters.

Section A provides an overview of the research (Chapter 1) and background

information from the literature (Chapter 2).

Section B describes the health system assessment for diabetes and DR in three counties of Kenya (Chapter 3 and 4).

Section C describes the health system strengthening activities, mainly in knowledge translation of the evidence to develop clinical guidelines (Chapter 5) and a training program to address supply side gaps that hinder availability of DR services in the country (Chapter 6).

Section D describes the clinical trial, whether the DURE intervention increases attendance at screening in Kirinyaga county (Chapter 7-10).

Section E provides a summary discussion for the thesis (Chapter 11).

The strength of this research is the progression of thinking from an understanding of the global literature on DR services, assessment of the Kenyan health system, identification of a critical gap in services, and culminating in the design, implementation and evaluation of an intervention to increase access to DR screening. This progression of research, where each step informed the next step, made it possible to develop a significant knowledge base. In addition, each of those steps was based on a strong theoretical underpinning, which facilitated a deeper understanding of the findings.

The main limitation of the research concerns the generalizability of the findings. The findings may apply to the wider Kenyan population and possibly a wider African population with similar social, cultural and health system context, and a similar patient profile. However, the findings may not be generalizable to other settings.

11.2 Overview of DR services

The aim of DR services is to prevent DR-related vision loss. The growing increase in prevalence of diabetes in all parts of the world places the spotlight on these services, since a third of people living with diabetes (PLWD) at any point will have DR and one in 10 will require immediate treatment for vision-threatening DR (VTDR). The services encompass a broad approach consisting of: first, prevention of DR through metabolic control of diabetes; second, DR screening among PLWD, for early detection of DR; third, timely treatment of those with VTDR; and fourth, rehabilitation of those with significant sight loss. This comprehensive approach includes services provided in the community and in diabetes and eye care clinics, although the extent of integration between these services varies in different parts of the world.

While high-income countries have successful DR programs that provide these services, sub-Saharan Africa (SSA) has a high level of unmet need for these services. The International Diabetes Federation reports that the projected increase in the prevalence of diabetes over the next two decades will be disproportionately concentrated in this region. We can expect a commensurate increase in the need for DR services, given that all PLWD are at risk of DR. As DR is asymptomatic until some sight is lost, early detection depends on the provision of regular screening for DR for all PLWD.

The first objective was a review of the literature on the epidemiology of both diabetes and DR globally and in Africa. In chapter two, the evidence from populationbased studies, clinic-based studies and systematic reviews on the risk factors, prevention and treatment of DR was presented. It is notable that globally there is plenty of literature on these conditions, but the evidence from Africa is sparse. This was not a surprising finding, as this discrepancy is ubiquitous in global health research.

As explained in section 11.5, we did not conduct a systematic review. However, the principles of systematic review were applied to ensure robustness and breadth of the literature review. The population, intervention, comparator and outcomes (PICO) were defined, and the London School of Hygiene and Tropical Medicine provided access to a wide range of peer reviewed and grey literature. This made it possible to

342

perform a comprehensive and balanced synthesis of the evidence, as reported in chapter two. Important gaps were identified in the literature, which would be of interest to the academic and policy audience.

The literature review was useful for identifying the priorities and challenges in DR services. One such challenge is screening for DR. There is evidence that DR meets established criteria for a screening programme, and that such screening is costeffective, at least in high-income settings that have well-established systematic screening programs and a higher prevalence of diabetes. A few papers reported that DR screening (or opportunistic case detection) in SSA is fragmented. It is not clear whether screening is cost-effective in SSA, but a first step would be to investigate this in a well-documented project within an African health system.

Kenya is one of the growing economies in SSA and the geographical context of this research, therefore the general health system in Kenya is described in chapter two.

11.3 DR services in Kenya

The prevalence of diabetes in Kenya is reported at 2% of the population 18-64 years (2015).¹ However, there are no population-based data on the prevalence of DR. The number of people requiring DR services can be estimated using existing evidence from the global literature, and this has been presented in chapter two in section A. Health systems influence the implementation and success of DR programs. Health systems research to improve DR services is thus an important area, in order to close existing gaps for PLWD.

The second objective of the thesis was to conduct a health system assessment. Section B describes the health system assessment for diabetes and DR in three counties of Kenya, representing rural, urban and semi-urban areas. The assessment employed mixed methods (quantitative and qualitative), was guided by the WHO health system approach and the tracer condition approach, and obtained patients' perspectives in the health system assessment. The use of these mixed methods and approaches provided a more comprehensive understanding of the health system.

Quantitative methods facilitated the analyses of patient-level predictors of use of DR services, such as duration of diabetes. Quantitative methods were also useful for identifying groups of PLWD who may be most at risk of not receiving screening services (e.g. those without symptoms, or those from a rural county), thereby contributing evidence to inform the design of targeted interventions to improve uptake in these groups. Barriers to access of services were also identified.

With qualitative methods (semi-structured interviews with service providers and key informants), strengths and weaknesses of the health system could also be explored. The use of the tracer approach alongside the WHO health systems approach facilitated the link of various health system indicators to the specific outcomes of interest in diabetes eye health, as discussed in section B.

Major gaps were found in the area of awareness and access. There were also gaps in availability and affordability of DR services, but the main barriers to entry to DR services were lack of awareness and lack of access. Although it is recommended that 100% of PLWD have an annual screening examination, only 25.6% of PLWD attending diabetes clinics had ever had a DR screening examination in their lifetime, which demonstrates a need-demand mismatch (or a screening gap) in DR services. Only 13.3% had a screening examination in the preceding year as recommended in clinical guidelines. This 87% unmet need for screening (screening gap) emerged as the priority area for DR interventions, since it is a bottleneck in the continuum of DR services. The findings presented in this thesis support the findings of other studies that the uptake of DR screening is low in low-resource settings.

The PLWD in Kirinyaga (a rural county) were the least likely to have had an eye examination in the preceding 12 months, with PLWD in Nairobi (urban county) having 2.6 times (95% CI 1.1–7.1) and PLWD in Nakuru (semi-urban county) having approximately three times increased uptake (95% CI 1.1–8.0). The main predictors for attendance at screening were referral from diabetes services and knowledge of diabetes complications. Participants referred for an eye examination had almost eight times the odds of having attended an eye examination in the last 12 months compared to those who had not been referred (OR 7.9, 95% CI 3.7–16.4, p < 0.001). Participants who had a knowledge of diabetes eye complications had four times the odds (OR 3.9, 95% CI 1.6–9.1) of attending as compared to those who had no knowledge of eye complications.

The strong association of knowledge of diabetes complications and uptake of screening is suggestive of the influence of demand-side barriers to access. Only 24.4% of participants had ever been referred from the diabetes clinic for a retinal examination, which on the other hand reflects a supply-side barrier. Additional supply side barriers obtained from qualitative interviews were: the lack of local clinical guidelines for DR; lack of mechanisms to ensure that patients referred for screening are not lost in the referral pathway between diabetes and eye care services; lack of point of care tools, such as checklists or protocols; lack of systematic patient education; and health workforce challenges such as lack of access to continuous professional development on DR.

This study helped to identify important opportunities in the health system. These include a high stakeholder interest in strengthening services, well-established support groups, a strong non-communicable disease strategy, and strong governance structures at national and county level. These strengths and gaps have been discussed in detail in section B the thesis.

In the papers from the health system assessment (research papers 1 and 2), these findings have been presented as the basis for exploring an approach to fill the health system gaps. Interventions to increase attendance to screening need to address the barriers and take the enablers in the health system into account.

11.4 Health system strengthening

As screening is the entry to DR services, and attendance at screening is low, our aim was to increase attendance to screening. However, it would be unethical and unnecessary to increase attendance at screening where existing supply-side barriers to DR services translate to the services being unavailable or of sub-optimal quality. Therefore there was need for health system strengthening (as a third objective) to facilitate the availability of an appropriate standard of care, including screening, treatment, referral and follow-up. Stakeholders identified the priorities for health system strengthening at national level. The priorities identified were national clinical guidelines and easily accessible resources for continuous professional development for health workers. Both of these were important opportunities for knowledge translation informed by research evidence. The scientific evidence identified through our literature review and stakeholder engagement were useful for translating the evidence into clinical guidelines and a training programme that can be accessed easily.

Although several studies have pointed out the need for health system strengthening ahead of intervention research, most of the health system strengthening activities reported have not involved knowledge translation and have not been described in detail. In addition, there is limited literature on the processes and outputs from these health system strengthening activities. In order to address this gap, section C of the thesis provides a detailed description of these activities.

Clinical guidelines are an important governance tool in health systems. There is a paucity of literature on how context-relevant guidelines can be developed in Africa, despite the potential value of these guidelines. This study has provided methodological contributions on this process and contributed to expanding the literature on this subject. The process and outputs of guideline development are described in the research papers 3 and 4. The scope of the guidelines is screening, diagnosis, treatment and follow-up for DR. The guidelines have been implemented nationally, not just in the study area. An online training course on the control of DR has been developed, targeting multidisciplinary teams involved in DR care. The development of the course is described in chapter six.

347

The health system strengthening was done with significant input from stakeholders in both the diabetes and eye care services. This is important because it facilitates ownership and implementation by the stakeholders, as well as capacity building that can be applied in the health system. We have described the technical aspects of the health system strengthening that could aid replication of these interventions elsewhere.

11.5 Development of an intervention to increase uptake of screening

The fourth objective of the thesis was to develop an intervention to increase attendance at DR screening and to test it in a randomized controlled trial. This study involved a community-based intervention relevant for an LMIC setting, where the unmet need for DR screening is higher, and hence the intervention had the potential for significant health system benefits.

The intervention in this trial utilizes diabetes support groups (DSGs) and peersupporters because they are existing resources in this context, a strength of the health system. The process of intervention development has been guided by the findings of the health system assessment, stakeholder consultation, review of the evidence and by the UK Medical Research Council's guidance on the development of complex interventions. The intervention fits the criteria for a complex intervention because it had several components, required different behaviours by PLWD, peersupporters and eye care workers, and required implementation at multiple settings at community level and at eye clinics. Multiple stakeholders in diabetes and eye care services were involved in intervention development. It was important to combine these approaches, as described in Section D of the thesis, in order to develop an intervention that could improve uptake of screening. The trial protocol for the trial is presented as research paper 5.

Before the trial, the literature on peer-support interventions to improve diabetes outcomes as well as the role of peer-support in other chronic conditions was reviewed. We did not conduct a systematic review, since previous systematic review evidence on peer-support for diabetes was already available and because of time limitations. This evidence has been presented in the thesis. The studies included in the systematic reviews were mainly conducted in high-income countries, while the peer-support models were mainly facility-based and led by health workers. In addition, the studies were mostly focused on other outcomes than DR screening.

11.5.1 Conceptual framework for the intervention

In developing the conceptual framework for the intervention it was hypothesised that a multi-faceted intervention delivered by peer-supporters would create awareness in PLWD regarding screening, build self-efficacy, link the PLWD with the eye clinic and provide reminders to attend screening. The basic principle behind the Self-Efficacy Theory by Albert Bandura is that individuals are more likely to engage in activities for which they have high self-efficacy in comparison to those in which they have low self-efficacy.^{2, 3} Self-efficacy has been shown to be a strong and consistent predictor of health behaviours such as physical activity. However, self-efficacy is task specific, meaning that it is crucial to identify the mechanisms in which self-efficacy is developed for a given behaviour, like attending DR screening. Self-efficacy had not been investigated previously in relation to uptake of DR screening services.

We developed the conceptual framework deductively using the self-efficacy theory. We also considered the barriers to attendance to screening identified by PLWD during the health system assessment. Our framework is thus novel and has not been validated in other populations. However, it could serve as a starting point for replication or expansion in future interventions.

11.6 Implementation of the intervention

This research has been conducted in the pragmatic context, which means the results are likely to be applicable to everyday practice and are important to policy makers looking for scalable interventions to increase access to DR screening in Kenya. The clinical trials unit at the London School of Hygiene and Tropical Medicine provided oversight of the trial.

The population included in the research were PLWD who are attending diabetes services but have never had eye screening for DR. Although this population is large in this context, in future screening programs will need to target a more heterogeneous population (PLWD that have previously been screened and those that have never been screened) hence these differences would need to be considered.

11.6.1 Pilot trial

Two clusters (DSGs) participated in a pilot trial to investigate the feasibility of the trial. The criterion for continuation to the main study was the feasibility of

recruitment of 50 participants in the trial, 80% monthly follow-up rates for individuals and no attrition of clusters. The pilot study met these success criteria, indicating that a full trial of the intervention was feasible. The pilot trial also provided additional benefits, such as documenting high acceptability and an interim measure of the effectiveness of the intervention. The findings of the pilot trial are described in research paper 6. The interpretation of these findings was that the full trial was feasible and we did not need to change the study procedures. We did not conduct a process evaluation for the pilot study, preferring to defer it to the main trial.

11.6.2 Full trial

11.6.2.1 Self-efficacy outcomes

The measurement of self-efficacy is particularly relevant for interventional studies aiming to improve specific health behaviours. The majority of studies evaluating the effects of interventions on health behaviour rely only on the distal outcome measures. The results of research that includes a measure that is proximal to the screening behaviour are relevant in that they provide insights on the intervention pathway; in this trial, that the intervention increased self-efficacy to take screening in the intervention arm. This increase occurred early in the intervention, corresponding to the first group talk and referral. Based on the findings of the process evaluation (chapter 9), evaluation of self-efficacy and the evaluation of the primary outcome (chapter 10), an association between the increase in self-efficacy and attendance at screening was inferred. However, some of the PLWD who had high self-efficacy after receiving the intervention did not attend screening, suggesting that there are additional barriers that were not addressed by the intervention.

11.6.2.2 Primary outcome evaluation

The intervention was highly effective for increasing uptake of DR screening, and this effect occurs early in the intervention, followed by a ceiling effect. Further, there was strong evidence of effect modification by gender, with women showing a more pronounced effect than men for exposure to the intervention and attendance at screening. This unusual effect was difficult to explain. However, it suggests that gender is a significant determinant of screening behaviour. It also corroborates the finding of other researchers that behaviour change and interventions to change behaviour are complex.⁴

These findings would suggest that the level of awareness of PLWD is a major driver of attendance to DRS. The intervention effect was primarily due to the first group talk, following which 50% of participants in the intervention arm took up screening. Only a further 4% took up screening in the following 5 months (making a total of 54%), hence it is clear that further group talks and telephone reminders to take screening have limited returns. Although the intervention increased self-efficacy in the intervention arm, 46% of participants exposed to the intervention did not go for screening over the follow-up period. This suggests that raised self-efficacy is necessary but not sufficient, as there are post-self-efficacy barriers to screening.

We found that 11% of the PLWD in the control arm took screening, which may have been because at baseline they were asked whether they had ever taken screening. It is worth considering how participation in the research itself (such as responding to the baseline questionnaire) may have influenced the results obtained. It is also possible that a similar effect may account for part of the response in the intervention arm. This would suggest that the effect size might be smaller in a non-research setting. These results of the outcome evaluation are described in the research paper 8.

In this study, the primary outcome was the proportion of PLWD who attended DR screening by the end of the study period. This is an essential outcome indicator to chart progress in DR screening programs, and we have shown that it is possible to measure it easily and objectively. This measure also provides important evidence related to uptake, access and coverage of DR services. Screening programs in resource-restrained settings may benefit from identifying a single measure of intervention effect that can be used for monitoring, for which we would recommend this outcome. As gender is an effect modifier in this intervention, we would recommend that this outcome be routinely analysed by gender.

There were many additional questions on screening which were not addressed in this study. For instance, the cost-effectiveness of screening and the prevalence of DR in PLWD were not included as they were beyond the scope of the thesis. However, they are important input and impact factors that would further complete the understanding of the intervention. Given that screening programs are not uniform and they evolve over time, it is not possible to have a universal monitoring tool that provides a complete picture of a screening program. Each program needs to choose the priority indicators for monitoring and evaluation, considering the resources available.

11.6.3 Process evaluation

The Consolidated Framework for Implementation Research (CIFR) guided the mixedmethods process evaluation of the trial. This framework, linked with principles of Implementation Science, was useful in identifying how, why and for whom the intervention worked. The quantitative component enabled the evaluation of implementation outcomes, in terms of recruitment, retention, reach, dose and fidelity. These important process indicators may be used for monitoring this intervention, however their measurement is labour-intensive, requiring lots of documentation and time, especially when the intervention is delivered in its current form. The measurement of the indicators is more viable for routine settings, if the intervention is limited to just one group talk as we recommend later in this chapter.

Both quantitative and qualitative components of the process evaluation contributed to the investigation of potential influences on attendance at screening. Findings generated using different methods could therefore be synthesised, compared, and triangulated to increase the validity and facilitate a deeper understanding of how the intervention worked. The results indicated that all the elements of the CIFR had an influence on how the intervention worked. There was evidence of local adaptation of the intervention during implementation, to suit local contexts, which is described in research paper 7.

Participants who took up screening reported doing so soon after receiving the first group talk. This is attributable to the finding that the first group talk addressed the main barriers to screening attendance, which were lack of awareness and lack of referral. In addition, the influence of the culture of collective action led to the

354

majority of the PLWD attending screening together early in the intervention. These findings provide evidence that the first group talk and the referral are the most essential components of the intervention, as mentioned earlier. Research paper 7 describes these findings in detail.

11.7 Implications

11.7.1 Implications for research

Whilst the findings of this research contribute to evidence for improving DR screening services in Kenya, they also demonstrate that gaps remain and present opportunities for future research. As an example, further longitudinal research to enable greater understanding of the effect of the intervention on the future cycles of annual DR screening for the participants.

There is considerable social, cultural, and health system variation in different settings and the intervention may work differently in other settings. Further research is required to explore the extent to which the peer-led intervention is transferrable for use in other contexts and countries. We recommend that research interventions should be: preceded by a situation analysis to identify the existing forms of peersupport; robust in design; comprehensively described and process evaluations conducted to gain insights into their mechanism of action.

Priority research areas in the short-term period are:

- 1. How can the ceiling effect of the intervention be broken?
- 2. How feasible, cost-effective and sustainable is the scale-up of the intervention for the health system?

- 3. What is the long-term effect of the intervention on initiation and maintenance of screening for PLWD?
- 4. What is the effect of adding further components to the intervention (such as photography-based screening in diabetes clinics)?
- 5. Why is the intervention effect more pronounced in women?

11.7.2 Implications for DR programs

DR programs in Kenya

Given the implementation of the clinical guidelines that explicitly emphasise annual screening for all PLWD, it is anticipated that screening services will be developed and become available to all PLWD in all parts of the country.

This study has introduced a new intervention that is not already part of the existing DR program. The findings presented in this thesis suggest that the intervention components could feasibly be delivered as they were delivered in this trial, at community DSGs with DR screening being provided at the existing eye clinic with the required capacity. The intervention is acceptable to the PLWD, the peer-supporters, health-care providers and other stakeholders. This trial tested the effectiveness of the intervention for PLWD who have never had screening. However, the intervention could be offered to all PLWD, as envisaged in the DR guidelines. The intervention could be scaled up to other counties in Kenya that have DSGs in the current form and services to examine and treat patients with DR, though cost-effectiveness data is lacking.

The intervention could potentially be modified to just one group talk and referral to the eye clinic. The group talk could then be repeated after six months, and monitored to see if this repetition will increase uptake of screening. A further group talk should also be provided before the next screening cycle, at 12 months. The monthly group talks and monthly telephone reminders can be left out, as they did not seem to be critical ingredients for the effect of the intervention.

The 'bottom-up' approach to health education in this intervention is advantageous because it is delivered to the target audience that may benefit from it. It is a suitable model for use with other chronic conditions for which peer groups exist. The training curriculum for peer-supporters developed in this study is already being adopted by the Kenya Defeat Diabetes Association as part of their peer-support manual. The online DR course is already available for use by health workers, and it will continue to be available to support lifelong learning within and outside Kenya.

It is important to note that while this was a pragmatic study, the implementation phase required a lot of follow-up by the principal investigator (PI) and regular documentation by the peer-supporters. For example, the PI and peer-supporters had weekly telephone contact, which would not be considered standard 'real-world' practice in any context. When the programme is implemented in non-study conditions, it will be necessary to see if these roles are required and if so who may be suitable to fill them. In Kenya, this might be the focal person for NCDs or the focal person for eye care in each county. Thus during program scale-up it will be necessary for further health system strengthening to consider these questions. In addition, there will be a need for continuous engagement of the communities and

357

DSGs. Although the intervention is a 'low-hanging fruit', it is critical for policy makers and DR program implementers to invest adequately in the scale-up phase if the research finding is to be turned into a practical and effective health service activity.

DR programs outside Kenya

The Kenya model of national DR guidelines has already been used to develop the DR guidelines in Ghana through the DR Network. The online training program is open to participants in all parts of the world.

In principle, the DSG intervention can be implemented in other contexts that have similar DSG and health system structures, after adaptation to account for the local context. Funders of international DR programs may therefore wish to invest in such a process of adaptation before program implementation.

11.7.3 Implications for practice

Pocock and Stone⁵ in their paper 'The primary outcome is positive —Is that good enough?' recommend that researchers reflect on the following questions to interpret whether the evidence from a clinical trial with positive results is sufficient to advance clinical practice (Table 11-1). These questions have provided a useful framework for appraisal and interpretation of the trial findings, and other researchers might find them useful.

Table 11-1: Is the evidence sufficient?

1. Does a p value of <0.05 provide strong enough evidence?

There is very strong evidence of a large intervention effect, which is not just based on an interpretation of the statistical significance of the p value

2. What is the magnitude of the treatment benefit?

Compared with usual care, the intervention quintupled the number of PLWD taking screening. Given that these are PLWD who had never taken screening, this effect has important implications for increasing screening among this unreached group. The literature suggests that 10 people need to be screened to find one person who requires treatment, however further research is needed to investigate if this is true in LMIC settings with no or limited previous screening programmes.

3. Is the primary outcome clinically important (and internally consistent)?

Yes, attendance to screening is important for early detection of DR. However, we did not investigate the prevalence of DR among those screened. The literature evidence indicates that a third of PLWD have DR and one in 10 VTDR.

4. Are secondary outcomes supportive?

Yes the secondary outcomes in our study were evaluated in the process evaluation, such as implementation outcomes (high acceptability, fidelity, reach and dose), and they support the findings of the outcome evaluation.

5. Are the principal findings consistent across important subgroups?
The intervention effect was high in both men and women. It was markedly higher in women.

6. Is the trial large enough to be convincing?

Although the sample size was sufficient for our study objectives, we

recommend replication of the study with larger samples in different LMICs.

7. Was the trial stopped early?

No

8. Do concerns about safety counterbalance positive efficacy?

No

9. Is the efficacy-safety balance patient-specific?

No, this intervention is considered safe for all participants.

10. Are there flaws in trial design and conduct?

No the study design and conduct is robust, we have discussed the

limitations of the study.

11. Do the findings apply to my patients?

The findings are context-specific, hence generalizability is limited to similar contexts.

11.8 Dissemination

The findings contained in this thesis have been disseminated, and will further be disseminated to a local, national and international audience of PLWD, diabetes support groups, health workers, policy-makers, program implementers and researchers.

The findings have been shared in publications, conferences, stakeholders' feedback meetings, peer research meetings, pubic engagement activities, reports to funders, and knowledge translation activities. The specific forums where the findings have been shared are shown as an appendix. We have received positive feedback from our dissemination. This has prompted us to further reflect on the implications of our findings, and on the next steps. Some of the data on secondary outcomes is still being analysed and will be disseminated in similar forums.

11.9 Conclusion

The overall aim of the PhD study was to assess DR services and provide evidence to improve the services. The findings provide new evidence to further the discussion on health systems for diabetes and DR and targeted interventions to improve access to DR services.

We have introduced a peer-led health education intervention in community settings and showed that it is effective in increasing attendance to screening. The main findings show that the intervention had a positive effect on screening attendance. The positive results are largely due to the increased awareness resulting from one group talk given by a peer-supporter and the referral of PLWD for a screening eye clinic. As this is the first study to document the role of peer-led interventions to increase

attendance at DR screening, the results provide a useful contribution to the

knowledge base on the use of community resources for DR care in LMICs, which

often goes undocumented in practice.

REFERENCES

1. Ministry of Health, Kenya National Bureau of Statistics, World Health Organization. Kenya STEPwise survey for non-communicable diseases risk factors 2015 report. Ministry of Health, Division of Non-Communicable Diseases; 2015.

2. Bandura A. Self-Efficacy. S RV, editor. New York: Academic Press; 1994.

3. Gao J, Wang J, Zheng P, Haardörfer R, Kegler MC, Zhu Y, et al. Effects of self-care, self-efficacy, social support on glycemic control in adults with type 2 diabetes. BMC Fam Pract. 2013;14(66).

4. Kelly MP, Barker M. Why is changing health-related behaviour so difficult? Public Health. 2016;136:109-16.

5. Pocock SJ, Stone GW. The Primary Outcome Is Positive — Is That Good Enough? New England Journal of Medicine. 2016;375(10):971-9.

The Road Not Taken

Two roads diverged in a yellow wood, And sorry I could not travel both

[...]

I shall be telling this with a sigh Somewhere ages and ages hence: Two roads diverged in a wood, and I— I took the one less travelled by, And that has made all the difference·

Robert Frost (1874-1963) American poet

Appendix 1 Courses undertaken

Modules

Modules	Organizer
Statistics for Epidemiology	LSHTM module
Extended Epidemiology	LSHTM module
Principles of Social Research	LSHTM module
Ophthalmic Epidemiology	LSHTM MOOC
Clinical Trials	LSHTM short course
Implementation Research	World Health Organization TDR MOOC
Diabetic Eye Disease- Strengthening Services	LSHTM MOOC
	Modules Statistics for Epidemiology Extended Epidemiology Principles of Social Research Ophthalmic Epidemiology Clinical Trials Implementation Research Diabetic Eye Disease- Strengthening Services

Transferable Skills courses Organiser

Friday, 02 October 2015	UCL Careers Workshop: Academic Career Planning	Bloomsbury Postgraduate Network
Tuesday, 06 Oct 2015	An Introduction to Semi Structured Interviews	LSHTM
Wednesday, 07 Oct 2015	MS Excel 2013: Formulae, Functions and Formatting	LSHTM
Wednesday, 07 Oct 2015	Introduction to Qualitative Analysis	LSHTM
Friday, 09 Oct 2015	MS Excel 2013: Data Management	LSHTM
Wednesday, 14 Oct 2015	MS PowerPoint 2013: for academic presentations	LSHTM
Wednesday, 28 Oct 2015	Your PhD Part 1 - Reading for a PhD - The First Important Steps	Bloomsbury Postgraduate Network
Wednesday, 04 Nov 2015	Oral Presentations: What Makes it Good?	LSHTM
Thursday, 05 Nov 2015	Research Information Skills 1	LSHTM Library
Friday, 06 Nov 2015	UCL Careers Workshops – Academic Careers Planning	Bloomsbury Postgraduate Network
Thursday, 12 Nov 2015	MS Excel 2013: Graphs & Charts	LSHTM
Friday, 13 Nov 2015	Research Information Skills 2	LSHTM Library
Thursday, 26	Your PhD Part 2 - Management Skills for	Bloomsbury Postgraduate
Nov 2015	Researchers	Network
Friday, 04 Dec 2015	Introduction to Data Management in the Social Sciences	LSHTM
Monday, 07 Dec 2015	Oral Presentations: What Makes it Good?	LSHTM

Tuesday, 14 Jun 2016	Introduction to Working with NVIVO 10	LSHTM
Wednesday, 15 Jun 2016	Working with NVIVO 10	LSHTM
Wednesday, 15 Jun 2016	Making Your Thesis Legal - & Depositing it Online	LSHTM
Wednesday, 15 Mar 2017	Publication Workshop 2: Publication Ethics	LSHTM
Thursday, 23 Mar 2017	Journal Metrics and the Publishing Landscape – a talk from leading Scientific Publishers- Elsevier	Bloomsbury Postgraduate Network
Friday, 24 Mar 2017	Fundamentals of Giving A Poster Presentation	Bloomsbury Postgraduate Network
Tuesday, 28 Mar 2017	Introduction to Research Support and Integrity	Bloomsbury Postgraduate Network
Thursday, 30 Mar 2017	Using Social Media for research	Bloomsbury Postgraduate Network
Friday, 31 Mar 2017	Skills for Conflict Resolution	Bloomsbury Postgraduate Network
Monday, 03 Apr 2017	Travel Safety and Security	LSHTM
Wednesday, 19 Apr 2017	Writing up your Research for Publication and Social Media	LSHTM
Monday, 24 Apr 2017	Travel Procedures for Students	LSHTM
Tuesday, 25 Apr 2017	EndNote X7: managing your references and bibliographies	LSHTM
Tuesday, 02 May 2017	Sexual Aggression: Avoidance and Survival	LSHTM
Tuesday 09 May 2017	MS Excel 2016: Data Management	LSHTM
Thursday, 10 May 2016	Social Media for Research	LSHTM
Friday 02 Jun 2017	MS Excel 2016: Graphs and Charts	LSHTM
Tuesday, 13 June 2017	Conducting Systematic literature Search	LSHTM Library
Wednesday, 21 June 2017	Understanding Journal Metrics	University of London / LSHTM
Thursday, 23 June 2018	Peer Review	LSHTM
Monday, 21 January 2019	Peer Review	Elsevier Researcher Academy
Friday, 10 May 2019	Improving your Assertiveness	LSHTM
Wednesday, 22 May 2019	Preparing to Submit Your Thesis	LSHTM

Key: LSHTM-London School of Hygiene and Tropical Medicine; MOOC-Massive Open Online Courses; UCL-University College London

Appendix 2: PhD Timeline

ACTIVITY	201	5			2016	5			2017	7			2018	3			2019	Ð			2020
	Q1	Q2	Q3	Q4	Q1	Q2	Q3	Q4	Q1	Q2	Q3	Q4	Q1	Q2	Q3	Q4	Q1	Q2	Q3	Q4	Q1
Award of Research grant																					
Registration for PhD																					
Literature review			· ·																		
Health system assessment																					
Health system strengthening																					
Clinical trial																					
Writing up																					
Viva																					

*Q=Quarter

Appendix 3-DURE Standard Operating Procedures

Measuring Clinical and Anthropometric parameters

Weight

Participant should be weighed in light dress (remove all heavy items of clothing or items in pockets before being weighed) and without shoes. Ensure the scale reads 'zero' before weighing the patient. Use the same scale for all participants. Record the weight to one decimal place in the enrolment form. Inform the participant of the measurement findings

Height

Volunteer must remove his/her shoes and hat before having his/her height measured.

Record the height in cm to no decimal places in the enrolment form.

Inform the participant of the measurement findings

Waist circumference

To measure the waist:

- Participant in upright standing position
- Find the bottom of the ribs and the top of the hips
- Wrap a soft tape measure around the waist, midway between these points, usually at the level of the umbilicus
- Participant should breathe out naturally before taking the measurement
- Record the measurement to the nearest 0.5cm on the enrolment form

Inform the participant of the measurements

The cutoffs are 94cm and 80cm for men and women respectively (recommended by IDF for sub-Saharan Africa)*

*The metabolic syndrome--a new worldwide definition. Alberti KG, Zimmet P, Shaw J, IDF Epidemiology Task Force Consensus Group. Lancet. 2005 Sep 24-30; 366(9491):1059-62.

Blood pressure

Participant must rest for at least 5 minutes in the seated position before measuring blood

pressure

Use a digital device for measuring blood pressure

Measure blood pressure in the seated position, at the upper arm, using an appropriate size of cuff

Record the systolic and diastolic blood pressure in the enrolment form

Inform the participant of the measurement findings

Patients with blood pressure >140/90mmHg or >90/60mmHg should be referred to the physician

Random blood sugar

Participant should be seated Check that meter code and test strip code match Apply single use disposable non sterile gloves Using the single use lancet, obtain a blood sample from the side of the finger Avoid using thumb or index finger The finger may bleed without assistance, but may need 'milking' gently Apply a drop of blood to the strip by holding the patient's finger to the edge of the strip until the yellow window is completely filled with blood Dispose of used lancet into a sharps container Apply digital pressure to stop bleeding Remove test strip from meter and switch meter off Remove gloves and dispose appropriately Record the blood sugar on the enrolment form If blood sugar is <=4.4mg/dl, give a glucose feed immediately Inform the participant of the measurement findings

Visual acuity

Snellen chart is placed 6m away at eye level and in good lighting Test each eye separately, starting with the right eye Test visual acuity with glasses if participant uses glasses Record the visual acuity on the enrolment form

Appendix 4



CONSORT 2010 checklist of information to include when reporting a pilot or feasibility randomized trial

Item	Description	Reported on
		page number
Title	Identification of study as randomised pilot or feasibility trial	1
Authors	Contact details for the corresponding author	1
Trial design	Description of pilot trial design (e.g. parallel, cluster)	4 and 5
Methods		
Participants	Eligibility criteria for participants and the settings where the pilot trial was conducted	4
Interventions	Interventions intended for each group	5 and 6
Objective	Specific objectives of the pilot trial	3
Outcome	Prespecified assessment or measurement to address the pilot trial objectives	5
Randomization	How participants were allocated to interventions	5
Blinding (masking)	Whether or not participants, care givers, and those assessing the outcomes were blinded to group assignment	5
Results		
Numbers randomized	Number of participants screened and randomised to each group for the pilot trial objectives	8
Recruitment	Trial status	8
Numbers analysed	Number of participants analysed in each group for the pilot objectives	8
Outcome	Results for the pilot objectives, including any expressions of uncertainty	8
Harms	Important adverse events or side effects	8
Conclusions	General interpretation of the results of pilot trial and their implications for the future definitive trial	11
Trial registration	Registration number for pilot trial and name of trial register	1
Funding	Source of funding for pilot trial	11

Citation: Eldridge SM, Chan CL, Campbell MJ, Bond CM, Hopewell S, Thabane L, et al. CONSORT 2010 statement: extension to randomised pilot and feasibility trials. BMJ. 2016;355.

Appendix 5

Developing the competences and training content for peer supporter training

Context: The content of the training is incremental to the existing training manual for peer supporters.

Process

The following steps were completed:

- Identification of a local gap. The health system assessment for diabetes and diabetic retinopathy identified that uptake of eye examination in three counties is low. A peer-led educational and referral intervention targets to address the demand side barriers to uptake of eye examination.
- Literature review of peer support. This showed that peer support has been effective in contributing to education of PLWD and linkage with health care resources.
- A review of the existing peer support training manual has been conducted. We noted that it does not have content on diabetes eye health.
- 4. Literature review on development of training programs identified the factors associated with greater impact of training initiatives to be: alignment to local needs and priorities, country ownership of the training program, competency-based training, and a sustainability strategy.
- Competencies for peer supporters have been developed, based on the literature and the identified local need
- 6. Assessment of the social accountability of this training. Social accountability in medical education is defined as an education that responds to the requirements and expectations of the society. This training has been assessed for relevance, cost-effectiveness, quality and equity.

- a. Relevance: The training is relevant because it addresses a need that has been identified in this community.
- b. Cost-effectiveness: The training is cost-effective because it is low cost in terms of training resource. It requires only two days and hence it can easily be added to the existing training without causing significant disruption to the training arrangements during the scale up phase. In addition, if necessary a shorter version of the training can also be created.
- c. Quality Assurance: The identification of concise competencies provides a consistent standard of knowledge and skill. The training will also consist of structured practical sessions and role modelling to aid application of knowledge.
- Equity: The training contributes to enhancing equity because it empowers disadvantaged individuals and groups to obtain the eye examination that they require.
- 7. Stakeholder consultation: The draft competencies were reviewed by ophthalmologists, endocrinologists, physicians, diabetes educators, public eye health specialists and the umbrella body for peer supporters (Kenya Defeat Diabetes Association).
- The curriculum was presented at a local workshop and a local conference for discussion.
 This was done to foster a sense of ownership among the stakeholders.

Appendix 6: CONSORT 2010 checklist for DURE cluster randomised trial

Section/Topic	ltem No	Standard Checklist Item	Extension for Cluster Designs	Page No *
Title and abstract				
	1a	Identification as a randomised trial in the title	Identification as a cluster randomised trial in the title	1
	1b	Structured summary of trial design, methods, results, and conclusions (for specific guidance see CONSORT for abstracts) ^{1,2}	Table 2 (below)	2
Introduction				
Background and objectives	2a	Scientific background and explanation of rationale	Rationale for using a cluster design	5
	2b	Specific objectives or hypotheses	Whether objectives pertain to the cluster level, the individual participant level or both	5
Methods				
Trial design	За	Description of trial design (such as parallel, factorial) including allocation ratio	Definition of cluster and description of how the design features apply to the clusters	5,6
	3b	Important changes to methods after trial commencement (such as eligibility criteria), with reasons		None
Participants	4a	Eligibility criteria for participants	Eligibility criteria for clusters	Table 1
	4b	Settings and locations where the data were collected		6
Interventions	5	The interventions for each group with sufficient details to allow replication, including how and when	Whether interventions pertain to the cluster level, the individual participant level or both	6

		they were actually administered		
Outcomes	6a	Completely defined pre- specified primary and secondary outcome measures, including how and when they were assessed	Whether outcome measures pertain to the cluster level, the individual participant level or both	7
	6b	Any changes to trial outcomes after the trial commenced, with reasons		None
Sample size	7a	How sample size was determined	Method of calculation, number of clusters(s) (and whether equal or unequal cluster sizes are assumed), cluster size, a coefficient of intracluster correlation (ICC or <i>k</i>), and an indication of its uncertainty	6
	7b	When applicable, explanation of any interim analyses and stopping guidelines		7
Randomisation:				
Sequence generation	8a	Method used to generate the random allocation sequence		6
	8b	Type of randomisation; details of any restriction (such as blocking and block size)	Details of stratification or matching if used	6
Allocation concealment mechanism	9	Mechanism used to implement the random allocation sequence (such as sequentially numbered containers), describing any steps taken to conceal the sequence until interventions were assigned	Specification that allocation was based on clusters rather than individuals and whether allocation concealment (if any) was at the cluster level, the individual participant level or both	6
Implementation	10	Who generated the random allocation sequence, who enrolled participants, and who	Replace by 10a, 10b and 10c	

		assigned participants to interventions		
	10a		Who generated the random allocation sequence, who enrolled clusters, and who assigned clusters to interventions	6
	10b		Mechanism by which individual participants were included in clusters for the purposes of the trial (such as complete enumeration, random sampling)	6
	10c		From whom consent was sought (representatives of the cluster, or individual cluster members, or both), and whether consent was sought before or after randomisation	6
Blinding	11a	If done, who was blinded after assignment to interventions (for example, participants, care providers, those assessing outcomes) and how		7
	11b	If relevant, description of the similarity of interventions		N/A
Statistical methods	12a	Statistical methods used to compare groups for primary and secondary outcomes	How clustering was taken into account	7,8
	12b	Methods for additional analyses, such as subgroup analyses and adjusted analyses		7,8
Results				
Participant flow (a diagram is strongly recommended)	13a	For each group, the numbers of participants who were randomly assigned, received intended treatment, and	For each group, the numbers of clusters that were randomly assigned, received intended treatment, and were analysed for the primary outcome	Fig 2

		were analysed for the primary outcome		
	13b	For each group, losses and exclusions after randomisation, together with reasons	For each group, losses and exclusions for both clusters and individual cluster members	Fig 2
Recruitment	14a	Dates defining the periods of recruitment and follow- up	Period of follow-up	8
	14b	Why the trial ended or was stopped		N/A
Baseline data	15	A table showing baseline demographic and clinical characteristics for each group	Baseline characteristics for the individual and cluster levels as applicable for each group	Table 2
Numbers analysed	16	For each group, number of participants (denominator) included in each analysis and whether the analysis was by original assigned groups	For each group, number of clusters included in each analysis	8
Outcomes and estimation	17a	For each primary and secondary outcome, results for each group, and the estimated effect size and its precision (such as 95% confidence interval)	Results at the individual or cluster level as applicable and a coefficient of intracluster correlation (ICC or k) for each primary outcome	7,8
	17b	For binary outcomes, presentation of both absolute and relative effect sizes is recommended		8
Ancillary analyses	18	Results of any other analyses performed, including subgroup analyses and adjusted analyses, distinguishing pre-specified from exploratory		8
Harms	19	All important harms or unintended effects in each group (for specific guidance see CONSORT for harms ³)		8

Discussion				9-11
Limitations	20	Trial limitations, addressing sources of potential bias, imprecision, and, if relevant, multiplicity of analyses		11
Generalisability	21	Generalisability (external validity, applicability) of the trial findings	Generalisability to clusters and/or individual participants (as relevant)	11
Interpretation	22	Interpretation consistent with results, balancing benefits and harms, and considering other relevant evidence		9-11
Other information				
Registration	23	Registration number and name of trial registry		2
Protocol	24	Where the full trial protocol can be accessed, if available		6
Funding	25	Sources of funding and other support (such as supply of drugs), role of funders		7

* Note: page numbers optional depending on journal requirements

Full bibliographic	Campbell MK, Piaggio G, Elbourne DR, Altman DG; for the CONSORT Group. Consort 2010 statement: extension to
reference	cluster randomised trials. BMJ. 2012;345: e5661.

REFERENCES

- ¹ Hopewell S, Clarke M, Moher D, Wager E, Middleton P, Altman DG, et al. CONSORT for reporting randomised trials in journal and conference abstracts. *Lancet* 2008, 371:281-283
- ² Hopewell S, Clarke M, Moher D, Wager E, Middleton P, Altman DG at al (2008) CONSORT for reporting randomized controlled trials in journal and conference abstracts: explanation and elaboration. *PLoS Med* 5(1): e20
- ³ Ioannidis JP, Evans SJ, Gotzsche PC, O'Neill RT, Altman DG, Schulz K, Moher D. Better reporting of harms in randomized trials: an extension of the CONSORT statement. *Ann Intern Med* 2004; 141(10):781-788.

Appendix 7: Checklist of items for reporting pragmatic trials - DURE trial

Section	Item	Standard CONSORT description	Extension for pragmatic trials
Title and abstract	1	How participants were allocated to interventions (e.g., "random allocation," "randomised," or "randomly assigned")	
Introduction			
Background	2	Scientific background and explanation of rationale	Describe the health or health service problem that the intervention is intended to address Page 5
Methods			
Participants	3	Eligibility criteria for participants; settings and locations where the data were collected	Eligibility criteria should be explicitly framed to show the degree to which they include typical participants and/or, where applicable, typical providers (e.g., nurses), institutions (e.g., hospitals), communities or localities (e.g., towns) and settings of care Table 1
Interventions	4	Precise details of the interventions intended for each group and how and when they were actually administered	Describe extra resources added to (or resources removed from) usual settings in order to implement intervention. Indicate if efforts were made to standardise the intervention or if the intervention and its delivery were allowed to vary between participants, practitioners, or study sites Page 6 Describe the comparator in similar detail to the
			Page 6, detailed information is provided in the trial protocol
Objectives	5	Specific objectives and hypotheses	

Section	Item	Standard CONSORT description	Extension for pragmatic trials
Outcomes	6	Clearly defined primary and secondary outcome measures and, when applicable, any methods used to enhance the quality of measurements (e.g., multiple observations, training of assessors)	Explain why the chosen outcomes and, when relevant, the length of follow-up are considered important to those who will use the results of the trial Page 5
Sample size	7	How sample size was determined; explanation of any interim analyses and stopping rules when applicable	If calculated using the smallest difference considered important by the target decision maker audience (the minimally important difference) then report where this difference was obtained Page 6
Randomisation— sequence generation	8	Method used to generate the random allocation sequence, including details of any restriction (e.g., blocking, stratification)	
Randomisation— allocation concealment	9	Method used to implement the random allocation sequence (e.g., numbered containers or central telephone), clarifying whether the sequence was concealed until interventions were assigned	
Randomisation— implementation	10	Who generated the allocation sequence, who enrolled participants, and who assigned participants to their groups	
Blinding (masking)	11	Whether participants, those administering the interventions, and those assessing the outcomes were blinded to group assignment	If blinding was not done, or was not possible, explain why Page 7
Statistical methods	12	Statistical methods used to compare groups for primary outcomes; methods for additional analyses, such as subgroup analyses and adjusted analyses	

Section	Item	Standard CONSORT description	Extension for pragmatic trials
Results			
Participant flow	13	Flow of participants through each stage (a diagram is strongly recommended)—specifically, for each group, report the numbers of participants randomly assigned, receiving intended treatment, completing the study protocol, and analysed for the primary outcome; describe deviations from planned study protocol, together with reasons	The number of participants or units approached to take part in the trial, the number which were eligible, and reasons for non-participation should be reported Figure 2
Recruitment	14	Dates defining the periods of recruitment and follow-up	
Baseline data	15	Baseline demographic and clinical characteristics of each group	
Numbers analysed	16	Number of participants (denominator) in each group included in each analysis and whether analysis was by "intention-to- treat"; state the results in absolute numbers when feasible (e.g., 10/20, not 50%)	
Outcomes and estimation	17	For each primary and secondary outcome, a summary of results for each group and the estimated effect size and its precision (e.g., 95% CI)	
Ancillary analyses	18	Address multiplicity by reporting any other analyses performed, including subgroup analyses and adjusted analyses, indicating which are pre-specified and which are exploratory	
Adverse events	19	All important adverse events or side effects in each intervention group	
Discussion			
Interpretation	20	Interpretation of the results, taking into account study hypotheses, sources of potential bias or imprecision, and the dangers associated with multiplicity of analyses and outcomes	

Section	Item	Standard CONSORT description	Extension for pragmatic trials
Generalisability	21	Generalisability (external validity) of the trial findings	Describe key aspects of the setting which determined the trial results. Discuss possible differences in other settings where clinical traditions, health service organisation, staffing, or resources may vary from those of the trial. This information is published in the process evalu ation paper.
Overall evidence	22	General interpretation of the results in the context of current evidence	

1. Campbell MK, Piaggio G, Elbourne DR, Altman DG, for the CONSORT Group. Consort 2010 statement: extension to cluster randomised trials

BMJ. 2012;345

2. Zwarenstein M, Treweek S, Gagnier JJ, Altman DG, Tunis S, Haynes B, Oxman AD, Moher D, for the CONSORT and Pragmatic Trials in Healthcare (Practihc) group, ^[1]Improving the reporting of pragmatic trials: an extension of the CONSORT statement. BMJ 2008; 337; a2390

Appendix 8: Intra-cluster correlation coefficients (ICC) for demographic, anthropometric and clinical variables

Variable	ICC	95% CI
Education	0.026	0.000 - 0.060
Occupation	0.021	0.000 - 0.081
Duration of diabetes	0.004	0.000 - 0.022
Duration of membership to	0.072	0.007 - 0.136
diabetes support group		
Weight	0.000	0.000 - 0.015
Height	0.003	0.000 - 0.020
Waist circumference	0.112	0.023 - 0.200
Random blood sugar	0.037	0.012 - 0.158
Systolic blood pressure	0.008	0.000 - 0.029
Diastolic blood pressure	0.035	0.000 - 0.075

Interpretation

The intra-cluster correlation coefficients for all the variables are very small, thus there is very little effect of clustering. This means that clustering did not reduce the power of the study to detect true differences between the study arms.

Appendix 9

DURE Statistical Analysis Plan (SAP)

Study title	Effectiveness of peer support to increase uptake of retinal examination
	for diabetic retinopathy: pragmatic cluster randomized clinical trial in
	Kirinyaga, Kenya
Acronym	DURE
Pan African Clinical	PACTR201707002430195
Trials registration	
Study Protocol	https://bmcpublichealth.biomedcentral.com/track/pdf/10.1186/s12889-
	<u>018-5761-6</u>
SAP version	1
Author	Nyawira Mwangi, supervisors and statisticians listed below
Principal investigator	Nyawira Mwangi
Supervisors	
	Allen Foster
	Covadonga Bascaran
	Lawrence Muthami
Statisticians	
	David Macleod
	Lawrence Muthami
	Min Kim
Advisory committee	
	Stephen Gichuhi
	Consuela Moorman

David Macleod

Min Kim

AimTo evaluate, by means of a pragmatic cluster randomized controlled
trial, the effectiveness of a peer supporter- led community education
programme in Kirinyaga county, Kenya.

HypothesisThe proportion of people living with diabetes (PLWD) having a retinal
examination for DR is higher in diabetes support groups (DSGs) allocated
to the peer supporter-led educational package than in DSGs randomized
to the usual standard of care.

 Research questions
 1. To what extent can health education delivered by peer

 supporters increase the demand for annual retinal examination

 among PLWD?

2. What are the contextual factors that determine the effectiveness of the intervention?

Study design Two arm (1:1) pragmatic cluster randomized clinical trial

Level of statistical P<0.05, two-sided

significance

 Sample size
 Based on standard formula for cluster randomized trials by Hayes and

 Bennett
 Bennett

*There were no previous studies that assessed attendance to diabetic retinopathy screening (DRS) in DSGs, and hence there was no published literature on intracluster correlation for DRS.

The health system assessment had shown that only 12.2% of the PLWD in Kirinyaga had ever taken a DRS. We based our sample size determination on this evidence, as well as on a discussion with experts

	about what magnitude of effect would be clinically relevant (two-fold
	increase in attendance to DRS)
	Power: 80%, alpha 5%, 50 participants in each group (including 15% loss)
	Pilot study: 2 clusters (DSGs), 50*2=100 participants
	Main trial: 14 clusters (DSGs), 50*14=700 participants
Reliability assessment	Cronbach alpha of 0.70
for self-efficacy	
questionnaire	
Baseline data	Demographic, clinical and self-efficacy data
Randomization	- Done after recruitment
	- Use computer generated random numbers generated on Stata
	15 away from study site by statistician (DM)
	- Block randomization in groups of 4 clusters
Follow-up trial period	Six months
Stopping rules	None - no stopping rules for safety, futility, efficacy or lack of power. We
	do not expect to stop early, based on our findings at the pilot trial.
Population to be	All those who were randomized, by intention to treat analysis
analysed	
Interim analysis	Not planned
Primary Outcome	Attendance to DRS, measured at the end of 180 days
Secondary outcomes	These will be assessed as a part of the process evaluation at 6 months:
	1. Contextual factors that affect the effectiveness of
	the intervention
	2. Characteristics of peer supporters associated with
	uptake of eye examination

3. Barriers to uptake of eye examination among

PLWD.

ImplementationAcceptability, recruitment and retention, reach and doseoutcomesdelivered/received will be assessed during the process evaluation and
reported as proportions and percentages

Timing of data Data cleansing will be performed upon completion of the 180-day

analysis follow-up of the last cluster/participant included in the study. The final

Presentation ofData will be presented with their values, confidence intervals and presultsvalues.

analysis will be conducted thereafter.

Comparison of-We will list general patient characteristics in a baseline characteristicsbaselinetable.

characteristics -Data will be presented as mean with standard deviation (SD) when normally distributed or as median with interquartile range in case of skewed data.

-Dichotomous and categorical data will be presented in proportions.

-Normality of the data will be assessed using histograms.

-Linearity will be assessed using scatter plots.

-Differences between continuous variables will be assessed using Student's t-tests or Mann-Whitney-U test, depending on normality, whereas the Chi-squared test or Fisher's exact will be used for categorical values. Reference for-Visual acuity- WHO classification of Visual Impairmentinterpretation of the-Blood sugar – reference ranges from the World Health Organizationanthropometric,-Blood pressure – reference ranges used by the Ministry of Health,clinical and metabolicKenya: 60-90mmHg diastolic and 90-40mmHg systolicparameters-Body Mass Index: standard international categories-Waist circumference cut offs used for sub-saharan Africa (80cm in
women and 94cm in men)

Reporting primaryFrequencies and percentages of those that attend DRS in each arm willoutcomebe reported

Relationship between We will conduct a linear regression analysis for continuous variables and**primary outcome and** a logistic regression for dichotomous variables.

explanatory variables A univariable regression analysis will be conducted on demographic and clinical variables against the outcome variable.

A univariable regression analysis will be conducted on all variables and all variables will be used for inclusion in the multivariable model. As there are no previous studies that include these variables into one model estimating effect on DRS, we will not include any variable on a theory driven basis. We will construct the multivariable model using backward stepwise regression by starting with all variables in the model, testing the removal of each variable in turn, and deleting variables whose loss gives the most statistically insignificant deterioration in the fit of the model. This process will be repeated until no further variables can be deleted without a statistically significant loss of fit. Adjustment forWe will control for Type 1 errors by adjusting our primary outcomes formultiplicitydifferent confounders.

Subgroup analysisWe will conduct analysis for these subgroups as they reflect the
perspective from which a patient approaches DRS: age, gender,
education, occupation, current diabetes treatment, comorbidity with
hypertension, and current challenges with diabetes services

Survival analysisAnalysis of time-to-DRS data will be done (participants will be censored
at 180-days of follow-up).

Categorical variables will be analysed using the log-rank test and continuous variables will be assessed using a univariable Cox proportional hazard regression analysis.

Analysis of survival data will be presented by Kaplan-Meier survival curves when independent variables are dichotomous or categorical. A univariable Cox regression analysis will be conducted on clinical and demographic variables with 180-day DRS attendance as the dependent variable.

Covariates with a of p < 0.25 in the univariable analysis will be included in the multivariable model

Reporting guidelinesCONSORT 2010 statement and its extensions to abstracts, clusterrandomized clinical trials and pragmatic trials

Missing data In our pilot trial we collected baseline data and primary outcome data, and obtained some insight into the likelihood of missing data in our main trial. We expect to have no missing data for the baseline variables. For survival data, any participant lost to follow up before DRS is considered censored from the last day of follow-up.

- Statistical softwareStatistical analyses will be performed using Stata version 15(StataCorp, 4905 Lakeway Drive, College Station, Texas 77845 USA
http://www.stata.com)
- Publication of results We aim to publish three results papers
 - 1. The pilot trial results
 - 2. The primary outcome
 - 3. Process evaluation

Appendix 10: Research Dissemination

Category	Audience	Venue	Date	
Research meetings				
	CEHC/QEDJT meeting	London	October 2019	
	ICEH research meeting	ICEH	May 2019	
	CEHC/QEDJT research meeting	London	March 2019	
	Publications (see papers in thesis)	-	-	
	Kenya DR working group	Nairobi	Diverse dates	
Conferences				
	Ophthalmological Society of Kenya Conference	Nairobi, Kenya	November 2018	
	COECSA 2018	Addis Ababa, Ethiopia	August 2018	
	Nordic Ophthalmology Congress	Oslo, Norway	August, 2018	
	UNESCO 'Open Education for a Better World' Workshop	Vipava, Slovenia	July, 2018	
	World Ophthalmology Congress	Barcelona, Spain	June 2018	
	Fourth Global Forum on Human Resource for Health	Dublin, Ireland	November 2017	
	COECSA 2017	Kampala, Uganda	August 2017	
	COECSA 2016	Naivasha, Kenya	August 2016	
Public Engagemen	t			
	World Diabetes Day	Nairobi, Kenya	Nov, 2017	
	World Health Day	Nairobi, Kenya	April 2017	
	Three Minute thesis competition	London	May 2017	
	Three Minute thesis competition	London	May 2018	

	Lay persons visiting LSHTM for public engagement	London	May 2017
	World Diabetes Day	Nairobi	Nov 2016
Implementation partners			
	Kenya DR Stakeholders meeting	Nairobi	August 2017
	Kenya DR Research Group meeting	Nairobi	February 2018
	KDDA strategic planning meeting	Embu	Sept 2018
	KDDA strategic planning meeting	Nairobi	Sept 2017
	Kenya DR Stakeholders meeting	Nairobi	May 2016
CEHC - Commonwealth Eye Health Consortium			

COECSA - College of Ophthalmology of Eastern, Central and Southern Africa

DR - Diabetic Retinopathy, ICEH-International Centre for Eye Health

KDDA - Kenya Defeat Diabetes Association

LSHTM - London School of Hygiene and Tropical Medicine

QEDJT -Queen Elizabeth Diamond Jubilee Trust

UNESCO - United Nations Educational, Scientific and Cultural Organization

Amref Health Africa in Kenya



REF: AMREF – ESRC P227/2016

April 20, 2016

Nyawira Mwangi London School Hygiene and Tropical Medicine/KMTC Email: nyawiramwangi@yahoo.com Phone: +254722727839

Dear Dr. Nyawira,

RESEARCH PROTOCOL: SITUATION ANALYSIS OF THE HEALTH SYSTEM FOR DIABETES AND DIABETIC RETINOPATHY IN KENYA

Thank you for submitting your research protocol to the AMREF Ethics and Scientific Review Committee (ESRC).

This is to inform you that the ESRC has approved your protocol. The approval period is from April 20, 2016 to April 19, 2017 and is subject to compliance with the following requirements:

- a) Only approved documents (informed consents, study instruments, advertising materials etc) will be used.
- b) All changes (amendments, deviations, violations etc) are submitted for review and approval by AMREF ESRC before implementation.
- c) Death and life threatening problems and severe adverse events (SAEs) or unexpected adverse events whether related or unrelated to the study must be reported to the ESRC immediately.
- d) Any changes, anticipated or otherwise that may increase the risks or affect safety or welfare of study participants and others or affect the integrity of the research must be reported to AMREF ESRC immediately.
- e) Submission of a request for renewal of approval at least 60 days prior to expiry of the approval period (attach a comprehensive progress report to support the renewal).
- f) Clearance for export of biological specimen or any form of data must be obtained from AMREF ESRC, NACOSTI and Ministry of Health for each batch of shipment/export
- g) Submission of an executive summary report within 90 days upon completion of the study. This information will form part of the data base that will be consulted in future when processing related research studies so as to minimize chances of study duplication and/or plagiarism.

Please do not hesitate to contact the ESRC Secretariat (esrc.kenya@amref.org) for any clarification or query.

Yours sincerely,

Prof. Mohamed Karama Chair, AMREF ESRC

CC: Dr. George Kimathi, WASH Programme Manager, AMREF Kenya and Vice Chair AMREF ESRC Samuel Muhula, Monitoring & Evaluation and Research Manager, AMREF Kenya

Amref Health Africa in Kenya

REF: AMREF – ESRC P301/2016

14th February 2017

Mwangi Nyawira Kenya Medical Training College P.O. Box 2 – 00202 Kenyatta National Hospital, Nairobi Tel: +254722727839 Email: <u>nyawiramwangi@yahoo.com</u>

Dear Dr Nyawira,

health africa

RESEARCH PROTOCOL: DIABETIC RETINOPATHY IN KIRINYAGA COUNTY, KENYA: UPTAKE OF RETINAL EXAMINATION IN PATIENTS WITH DIABETIS MELLITUS STUDY

Thank you for submitting your research protocol to the Amref Ethics and Scientific Review Committee (ESRC).

This is to inform you that the ESRC has approved your protocol. The approval period is from 14th February 2017 to 14th February 2018 and is subject to compliance with the following requirements:

- a) Only approved documents (informed consents, study instruments, advertising materials etc) will be used.
- b) All changes (amendments, deviations, violations etc) are submitted for review and approval by Amref ESRC before implementation.
- c) Death and life threatening problems and severe adverse events (SAEs) or unexpected adverse events whether related or unrelated to the study must be reported to the ESRC immediately.
- d) Any changes, anticipated or otherwise that may increase the risks or affect safety or welfare of study participants and others or affect the integrity of the research must be reported to Amref ESRC immediately.
- e) Submission of a request for renewal of approval at least 60 days prior to expiry of the approval period (attach a comprehensive progress report to support the renewal).
- f) Clearance for export of biological specimen or any form of data must be obtained from Amref ESRC, NACOSTI and Ministry of Health for each batch of shipment/export.
- g) Submission of an executive summary report within 90 days upon completion of the study. This information will form part of the data base that will be consulted in future when processing related research studies so as to minimize chances of study duplication and/or plagiarism.

Please do not hesitate to contact the ESRC Secretariat (esrc.kenya@amref.org) for any clarification or query.



CC: Dr. George Kimathi, Director Capacity Development Institute, Amref Health Africa and Vice Chair Amref ESRC

Samuel Muhula, Monitoring & Evaluation and Research Manager, Amref Health Africa in Kenya

Amref Health Africa in Kenya

REF: AMREF - ESRC P301/2016

January 25, 2018

Nyawira Mwangi P.O Box 2, 00202, Nairobi, Kenya Email: nyawiramwangi@yahoo.com Tel: +254722727839

Dear Dr. Mwangi,

amre

health africa

RESEARCH PROTOCOL: DIABETIC RETINOPATHY IN KIRINYAGA COUNTY, KENYA: UPTAKE OF RETINAL EXAMINATION IN PATIENTS WITH DIABETES MELLITUS

Thank you for submitting your protocol to the Amref Health Africa Ethics and Scientific Review Committee (ESRC). Your protocol was reviewed during a meeting held on January 25, 2018.

This is to inform you that the ESRC has approved the annual renewal of your protocol. The approval period is from January 25, 2018 to January 24, 2019 and is subject to compliance with the following requirements:

- a) Only approved documents (informed consents, study instruments, advertising materials etc.) will be used.
- b) All changes (amendments, deviations, violations etc.) are submitted for review and approval by Amref ESRC before implementation.
- c) Death and life threatening problems and severe adverse events (SAEs) or unexpected adverse events whether related or unrelated to the study must be reported to the ESRC immediately.
- d) Any changes, anticipated or otherwise that may increase the risks or affect safety or welfare of study participants and others or affect the integrity of the research must be reported to Amref ESRC immediately.
- e) Submission of a request for renewal of approval at least 60 days prior to expiry of the approval period (attach a comprehensive progress report to support the renewal).
- f) Clearance for export of biological specimen or any form of data must be obtained from Amref ESRC, NACOSTI and Ministry of Health for each batch of shipment/export.
- g) Submission of an executive summary report within 90 days upon completion of the study. This information will form part of the data base that will be consulted in future when processing related research studies so as to minimize chances of study duplication and/or plagiarism.

Please do not hesitate to contact the ESRC Secretariat (esrc.kenya@amref.org) for any clarification or query. AMREF

Yours sincerely,

Prof. Mohamed Karama

Chair, Amref Health Africa ESRC

- 30125 00100, NA CC: Dr. George Kimathi, Director Institute of Capacity Development, Amref Health Africa and Vice Chair Amref Health Africa ESRC
 - Samuel Muhula, Monitoring & Evaluation and Research Manager, Amref Health Africa in Kenya

P O Box 30125-00100 Nairobi, Tel: +254 (0)20 699 4000, Fax: +254 (0)20 699 2531. www.amref.org
London School of Hygiene & Tropical Medicine

Keppel Street, London WC1E 7HT United Kingdom Switchboard: +44 (0)20 7636 8636

www.lshtm.ac.uk



Observational / Interventions Research Ethics Committee

Dr Nyawira Mwangi LSHTM

25 May 2016

Dear Nyawira

Study Title: Assessment of health system performance for Diabetes and Diabetic Retinopathy in Kenya in order to recommend and implement a package of interventions to strengthen DR services'

LSHTM Ethics Ref: 10904

Thank you for responding to the Observational Committee's request for further information on the above research and submitting revised documentation.

The further information has been considered on behalf of the Committee by the Chair.

Confirmation of ethical opinion

On behalf of the Committee, I am pleased to confirm a favourable ethical opinion for the above research on the basis described in the application form, protocol and supporting documentation as revised, subject to the conditions specified below.

Conditions of the favourable opinion

Approval is dependent on local ethical approval having been received, where relevant.

Approved documents

The final list of documents reviewed and approved by the Committee is as follows:

Document Type	File Name	Date	Version
Protocol / Proposal	Nyawira_Project_Proposal_Feb2016	21/02/2016	final
Protocol / Proposal	Patient interview (Diabetes)	21/02/2016	final
Protocol / Proposal	Patient interview (DR)	21/02/2016	final
Protocol / Proposal	Service provider interview (Diabetes)	21/02/2016	final
Protocol / Proposal	Service provider interview (DR)	21/02/2016	final
Investigator CV	CURRICULUM VITAE	21/02/2016	final
Information Sheet	Consent to Participate in the Study	21/02/2016	final
Information Sheet	Information sheet for service providers	21/02/2016	final
Investigator CV	CV Foster 2015	27/02/2016	final
Investigator CV	Covadonga Bascaran CV 2016	01/03/2016	final
Covering Letter	Cover letter for clarifications	23/05/2016	1
Information Sheet	Information sheet Kiswahili	23/05/2016	update
Information Sheet	Information sheet for participants	23/05/2016	update

Information Sheet	Consent to Participate in the Study	23/05/2016	update
Information Sheet	Consent Form- Kiswahili	23/05/2016	update
Local Approval	AMREF Approval letter	23/05/2016	final
Protocol / Proposal	Semi-structured interview guide for key informants for diabetes services_240516	24/05/2016	updated
Protocol / Proposal	Semi-structured interview guide for key informants for diabetic retinopathy services_240516	24/05/2016	update

After ethical review

The Chief Investigator (CI) or delegate is responsible for informing the ethics committee of any subsequent changes to the application. These must be submitted to the Committee for review using an Amendment form. Amendments must not be initiated before receipt of written favourable opinion from the committee.

The CI or delegate is also required to notify the ethics committee of any protocol violations and/or Suspected Unexpected Serious Adverse Reactions (SUSARs) which occur during the project by submitting a Serious Adverse Event form.

At the end of the study, the CI or delegate must notify the committee using an End of Study form.

All aforementioned forms are available on the ethics online applications website and can only be submitted to the committee via the website at: http://leo.lshtm.ac.uk

Additional information is available at: www.lshtm.ac.uk/ethics

Yours sincerely,



Professor John DH Porter Chair

<u>ethics@lshtm.ac.uk</u> <u>http://www.lshtm.ac.uk/ethics/</u>

Improving health worldwide

London School of Hygiene & Tropical Medicine

Keppel Street, London WC1E 7HT United Kingdom Switchboard: +44 (0)20 7636 8636

www.lshtm.ac.uk

LONDON SCHOOL of HYGIENE &TROPICAL MEDICINE

Observational / Interventions Research Ethics Committee

Dr. Nyawira Mwangi LSHTM

5 June 2017

Dear Dr. Nyawira Mwangi,

Study Title: Diabetic Retinopathy in Kenya: Uptake of Retinal Examination in patients with Diabetes Mellitus (DURE) study

LSHTM ethics ref: 11853

Thank you for your application for the above research, which has now been considered by the Interventions Committee.

Confirmation of ethical opinion

On behalf of the Committee, I am pleased to confirm a favourable ethical opinion for the above research on the basis described in the application form, protocol and supporting documentation, subject to the conditions specified below.

Conditions of the favourable opinion

Approval is dependent on local ethical approval having been received, where relevant.

Approved documents

The final list of documents reviewed and approved by the Committee is as follows:

Document Type	File Name	Date	Version
Investigator CV	Covadonga Bascaran CV 2016	21/08/2016	1
Investigator CV	CV Foster 2015	21/08/2016	1
Investigator CV	Nyawira_LSHTM_CV	21/08/2016	1
Protocol / Proposal	Information sheet for participants	01/11/2016	1
Protocol / Proposal	Information sheet Kiswahili	01/11/2016	1
Protocol / Proposal	Consent to Participate in the Study	01/11/2016	1
Protocol / Proposal	Kiswahili_Consent_Form	01/11/2016	1
Protocol / Proposal	Recruitment Form	01/11/2016	1
Protocol / Proposal	OUTCOME EVALUATION FORM	01/11/2016	1
Protocol / Proposal	TOPIC GUIDE FOR PARTICIPANTS	01/11/2016	1
Protocol / Proposal	TOPIC GUIDE FOR FOCUS GROUP DISCUSSION WITH PEER SUPPORTERS	01/11/2016	1
Protocol / Proposal	TOPIC GUIDE FOR INDEPTH INTERVIEWS WITH EYE CARE PROVIDERS	01/11/2016	1
Protocol / Proposal	Evaluation of peer supporters training	01/11/2016	1
Advertisements	RECRUITMENT PROCEDURE	01/11/2016	1
Safety Information	SAFETY INFORMATION	01/11/2016	1

Protocol / Proposal	Nyawira_CRT_Proposal_FINAL	17/11/2016	final
Information Sheet	Information sheet for team leaders of patient support groups	20/02/2017	final
Information Sheet	Consent to Participate in the Study_team lead	20/02/2017	final
Information Sheet	Information sheet Kiswahili	20/02/2017	final
Information Sheet	Information sheet for participants	20/02/2017	final
Information Sheet	Consent to Participate in the Study	20/02/2017	final
Information Sheet	Kiswahili_Consent_Form	20/02/2017	final
Local Approval	Approval Letter P301-2016	20/02/2017	final
Sponsor Letter	QA973_Sponsor confirmation_22.02.17	22/02/2017	final
Covering Letter	Cover letter-Nyawira Mwangi	24/05/2017	May 2017
Protocol / Proposal	Nyawira_CRT_Proposal_May2017	24/05/2017	May 2017
Protocol / Proposal	Information sheet for team leaders of patient support groups	24/05/2017	May 2017
Protocol / Proposal	Information sheet for participants	24/05/2017	May 2017
Protocol / Proposal	Information sheet for participants (Kiswahili)	24/05/2017	May 2017
Protocol / Proposal	Consent to Participate in the Study	24/05/2017	May 2017
Protocol / Proposal	Consent form for participants (Kiswahili)	24/05/2017	May 2017
Protocol / Proposal	Participant Recruitment Form	24/05/2017	May 2017
Protocol / Proposal	Topic guide for FGD with participants	24/05/2017	May 2017
Protocol / Proposal	Topic guide for focus group discussion with peer supporters	24/05/2017	May 2017
Protocol / Proposal	Topic guide for indepth interviews with eye care providers	24/05/2017	May 2017
Protocol / Proposal	Peer Support Training Programme evaluation	24/05/2017	May 2017
Information Sheet	Consent to Participate in the Study	24/05/2017	May 2017
Information Sheet	Consent form for participants (Kiswahili)	24/05/2017	May 2017
Information Sheet	Information sheet for participants	24/05/2017	May 2017
Information Sheet	Information sheet for participants (Kiswahili)	24/05/2017	May 2017
Information Sheet	Information sheet for team leaders of patient support groups	24/05/2017	May 2017
Advertisements	Participant Recruitment Procedure	24/05/2017	May 2017

After ethical review

The Chief Investigator (CI) or delegate is responsible for informing the ethics committee of any subsequent changes to the application. These must be submitted to the Committee for review using an Amendment form. Amendments must not be initiated before receipt of written favourable opinion from the committee.

The CI or delegate is also required to notify the ethics committee of any protocol violations and/or Suspected Unexpected Serious Adverse Reactions (SUSARs) which occur during the project by submitting a Serious Adverse Event form.

An annual report should be submitted to the committee using an Annual Report form on the anniversary of the approval of the study during the lifetime of the study.

At the end of the study, the CI or delegate must notify the committee using an End of Study form.

All aforementioned forms are available on the ethics online applications website and can only be submitted to the committee via the website at: http://leo.lshtm.ac.uk

Additional information is available at: www.lshtm.ac.uk/ethics



Chair

<u>ethics@lshtm.ac.uk</u> <u>http://www.lshtm.ac.uk/ethics/</u>

Improving health worldwide



No: No:	Uptake of Retinal Examination in Diabetes (DURE) Name:	Uptake of Retinal Examination in Diabetes (DURE) Name
--	--	---

To the glory of God and for the good of our people.